# Scaling Limits of Critical Systems IN RANDOM GEOMETRY

Ellen Grace Powell Queens' College

Supervised by: Prof. Nathanaël Berestycki.

This dissertation is submitted for the degree of Doctor of Philosophy

April 2017









**Preface** This dissertation is the result of my own work and includes nothing which is the outcome of work done in collaboration except as declared in the Preface and specified in the text. The work in Chapter 5 was completed in collaboration with H.Wu, and the work in Chapter 4 with J.Aru and A.Sepúlveda. In all these collaborations, the contribution of each collaborator was equal.

This dissertation is not substantially the same as any that I have submitted, or, is being concurrently submitted for a degree or diploma or other qualification at the University of Cambridge or any other University or similar institution except as declared in the Preface and specified in the text.

I further state that no substantial part of my dissertation has already been submitted, or, is being concurrently submitted for any such degree, diploma or other qualification at the University of Cambridge or any other University or similar institution except as declared in the Preface and specified in the text. It does not exceed the prescribed word limit for the relevant Degree Committee.

## ACKNOWLEDGEMENTS

Firstly, to my supervisor, Nathanaël. Thank you so much for all the time you have given me: for the great problems, the discussions, and for encouraging me when I needed it the most. Your confidence in me has meant a great deal, and I have learnt such a lot from you. Thanks also to everyone at Queens', especially to Julia and Richard, for my mathematical education, and for all your advice and support over the years. Thank you to the CCA, EPSRC and my fantastic cohort of fellow PhD students for making the last few years so interesting and diverse, and to my coauthors, Juhan, Avelio and Hao, for sharing your time and ideas with me so generously.

Of course, special thanks go to my family. To my wonderful fiancé Tom: thank you for all your love and support, and for keeping me sane (almost)! Thank you to my sister, Eve, for always being there for a laugh, or a cry, and for reminding me what is important in

life. And finally, to my parents. You have given me so much encouragement, and supported me in so many ways; I can never thank you enough. This is dedicated to you.

### Contents

1	Intr	Introduction 7				
	1.1	Backg	round	7		
	1.2	Results and Outline				
	1.3	Prelim	inaries	12		
		1.3.1	Brownian Motion	12		
		1.3.2	Gaussian free field	14		
		1.3.3	Gaussian multiplicative chaos	20		
		1.3.4	Liouville quantum gravity	22		
		1.3.5	Branching Brownian motion and connections with Gaussian multi-			
			plicative chaos.	23		
		1.3.6	Schramm–Loewner evolutions	26		
		1.3.7	Connection between SLE and the GFF $\hdots$	33		
		1.3.8	Branching Processes and the Brownian CRT	39		
	1.4	Questi	ions	42		
<b>2</b>	Inva	ariance	e principles for branching diffusions in bounded domains	45		
	2.1	Introd	uction	45		
		2.1.1	Related Work	48		
		2.1.2	Organisation of the Paper and Main Ideas	49		
	2.2	Prelim	inaries	51		
		2.2.1	Diffusions as Trees	51		
		2.2.2	Spectral Theory and Martingales	51		
	2.3	The P	hase Transition	54		
		2.3.1	The Supercritical Case	54		
		2.3.2	The Subcritical Case	55		
		2.3.3	The Critical Case	57		
	2.4	Survival at Criticality: Proof of Theorem 2.4		60		
		2.4.1	Spine Decomposition	60		
		2.4.2	Asymptotics for the Survival Probability	62		
		2.4.3	Asymptotics for $a(t)$	66		
	2.5	The Conditioned System				
	2.6	Convergence to the Brownian CRT $\ldots$		80		
		2.6.1	Martingale Convergence	82		
		2.6.2	Connection with the height function	91		

		2.6.3	Convergence to the CRT					
		2.6.4	Preliminaries on convergence of metric measure spaces					
		2.6.5	Proof of the main theorem					
3	Critical Gaussian chaos: convergence and uniqueness in the derivative							
	nor	malisa	tion 105					
3.1 Introduction $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$		Introd	uction $\ldots \ldots \ldots$					
	3.2 Preliminaries		ninaries					
		3.2.1	Log-correlated fields, 2d Gaussian free field and $\star\text{-scale}$ invariant					
			kernels					
		3.2.2	Convolution with mollifiers					
		3.2.3	Maxima of the mollified fields					
		3.2.4	Previous works on subcritical and critical Gaussian multiplicative					
			chaos					
		3.2.5	Critical measures					
		3.2.6	Bessel processes					
	3.3	Proof	of Theorem 3.1					
		3.3.1	A uniformly integrable family					
		3.3.2	Convergence					
	3.4	*-scale	e invariant kernels $\ldots \ldots 136$					
4	Approximating the Liouville measure using local sets of the Gaussian							
	free	field	139					
	4.1	Introd	uction $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $139$					
	4.2	Prelin	inaries on the Gaussian free field and its local sets					
		4.2.1	Two useful families of local sets					
	4.3	Overv	iew of the Liouville measure and loop constructions of $[{\rm Aid}15]$ 145					
		4.3.1	Subcritical regime					
		4.3.2	Critical regime					
		4.3.3	Measures constructed using nested loops					
	4.4	Local	set approximations of the subcritical Liouville measure					
	4.5 Generalizations		alizations $\ldots \ldots 150$					
		4.5.1	Non-simply connected domains and general boundary conditions 150					
		4.5.2	Dirichlet-Neumann GFF					
	4.6	Critic	al and supercritical regimes					
		4.6.1	(Super)critical regime					

		4.6.2	The derivative martingale in the critical regime	5
		4.6.3	Seneta-Heyde rescaling	1
5	Lev	el Line	es of the Gaussian free field with general boundary data 16'	7
	5.1	Introd	luction $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $16$	7
	5.2	Prelin	inaries $\ldots \ldots 176$	3
		5.2.1	Regulated functions and functions of bounded variation 175	3
		5.2.2	A result on the convergence of curves	3
		5.2.3	The zero boundary Gaussian free field	6
		5.2.4	Local sets for the GFF	8
		5.2.5	$SLE_{\kappa}(\rho)$ processes	0
		5.2.6	Level lines of the GFF with piecewise constant boundary data 18	2
		5.2.7	First generalizations to the GFF with general boundary data 18	3
	5.3	Non-b	oundary intersecting regime	7
	5.4	Monotonicity		0
	5.5	Estimates on crossing probabilities		
	5.6	Existence of the coupling—proof of Theorem 5.2		1
	5.7	Proof	of Theorems 5.3 to 5.5	7
	5.8	Proof	of Theorem 5.6 and concluding remarks	9

#### Abstract

This thesis focusses on the properties of, and relationships between, several fundamental objects arising from critical physical models. In particular, we consider Schramm–Loewner evolutions, the Gaussian free field, Liouville quantum gravity and the Brownian continuum random tree.

We begin by considering branching diffusions in a bounded domain  $D \subset \mathbb{R}^d$ , in which particles are killed upon hitting the boundary  $\partial D$ . It is known that such a system displays a phase transition in the branching rate: if it exceeds a critical value, the population will no longer become extinct almost surely. We prove that at criticality, under mild assumptions on the branching mechanism and diffusion, the genealogical tree associated with the process will converge to the Brownian CRT.

Next, we move on to study Gaussian multiplicative chaos. This is the rigorous framework that allows one to make sense of random measures built from rough Gaussian fields, and again there is a parameter associated with the model in which a phase transition occurs. We prove a uniqueness and convergence result for approximations to these measures at criticality.

From this point onwards we restrict our attention to two-dimensional models. First, we give an alternative, "non-Gaussian" construction of Liouville quantum gravity (a special case of Gaussian multiplicative chaos associated with the 2-dimensional Gaussian free field), that is motivated by the theory of multiplicative cascades. We prove that the Liouville (GMC) measures associated with the Gaussian free field can be approximated using certain sequences of "local sets" of the field. This is a particularly natural construction as it is both local and conformally invariant. It includes the case of nested  $CLE_4$ , when it is coupled with the GFF as its set of "level lines".

Finally, we consider this level line coupling more closely, now when it is between  $SLE_4$  and the GFF. We prove that level lines can be defined for the GFF with a wide range of boundary conditions, and are given by  $SLE_4$ -type curves. As a consequence, we extend the definition of  $SLE_4(\rho)$  to the case of a continuum of force points.

### 1 Introduction

#### **1.1** Background

This thesis will aim to explore, and hopefully gain a little insight into, the behaviour of a few canonical models in random geometry. These models, that we will describe shortly, are all linked in that they arise naturally from physical models at their "critical" point. That is, at a point (usually a special temperature), where a phase transition occurs.

We will be interested in the macroscopic behaviour of such systems. Take a discrete lattice model (for example, a random model for the spins of particles in a magnet) and observe what happens in the scaling limit, as the number of particles tends to infinity. Perhaps surprisingly, it turns out that in many cases, what one gets in the end doesn't actually depend very much on the initial model at all. This is the notion of universality.

The main objects studied here will all be universal in this sense. Let us start, as an example, with Brownian motion. Take any random walk in space with mean zero increments, and then speed up the walk and rescale the size of the steps accordingly. The following result tells us that, if we do this in the correct way, the walk will converge to a Brownian motion.

**Theorem 1.1** (Donsker's invariance principle (1-dimension)). Let  $(X_k)_{k=1}^{\infty}$  be a sequence of i.i.d. random variables with mean 0 and variance 1. Set  $S_n = \sum_{k=1}^n X_k$  and define a process  $S = (S_t : t \in [0,1])$  by linear interpolation:  $S_t = S_{\lfloor t \rfloor} + (t - \lfloor t \rfloor)(S_{\lfloor t \rfloor + 1} - S_{\lfloor t \rfloor})$ . Then the laws of the process

$$S^N := \left(\frac{1}{\sqrt{N}}S_{Nt} : 0 \le t \le 1\right)$$

converge weakly to that of a Brownian motion on  $(C[0,1], \mathcal{B}(C[0,1]))$ .

In particular, the scaling limit in the above theorem does not actually depend on the details of the random walk we chose in the first place. For this reason, Brownian motion can be thought of as a canonical notion of random curve.

However, there are some things that Brownian motion lacks the complexity to describe. For example, the natural interfaces that appear in many planar lattice models (say, between positive and negative spins in a magnet) become increasingly fractal as the number of particles tends to infinity, but do not look like the trace of a Brownian motion. A great breakthrough was made in our understanding of these interfaces when Oded Schramm, [Sch00], introduced a family of random fractal curves known as Schramm–Loewner evolutions (SLE) as candidates for their scaling limits. Indeed, it has since been proved that in several cases, [CS12, LSW04, SS09, Smi01], such interfaces do in fact converge to SLE.

Another natural question, which has been of great interest to the probability community in the last few years, is whether the notion of a canonical random path (i.e. Brownian motion) can be extended to higher dimensions. For example, in twodimensions, can we say what a random sphere should look like? This problem, which can be phrased more concretely in terms of putting a "random metric" on the 2sphere, has attracted much recent discussion and attention. Although seemingly a very difficult question, astounding progress on this has been made in several directions. We will discuss two different approaches to this now.

Just as Brownian motion is a limit of random walks, it seems natural to search for a "random sphere" by considering sensible discrete approximations. In this case, the analogous things to look at are random planar maps; essentially, large collections of polygons glued together to give a metric space with the topology of a sphere. A major breakthrough in this direction came when Le Gall [LG13] and Miermont [Mie13] proved, independently, that for certain natural choices of random planar maps these metric spaces have a scaling limit, known as the Brownian map.

**Theorem 1.2** (Convergence to the Brownian map). Suppose that p = 3 or an even integer greater than or equal to 4. Let  $\mathbf{m}_n$  be a uniformly chosen rooted p-angulation of the sphere (that is, a finite connected graph embedded in  $\mathbb{S}^2$ , viewed up to orientation preserving homeomorphisms of  $\mathbb{S}^2$ , with a distinguished oriented root edge and with n faces of degree p.) Let  $d_{gr}$  be the graph distance on  $\mathbf{m}_n$ . Then there exists a positive constant  $\lambda_p$  such that

$$(\mathbf{m}_n, \frac{\lambda_p}{n^{1/4}} d_{gr}) \xrightarrow{(d)} (\mathbf{m}_\infty, d^*)$$

as  $n \to \infty$ , with respect to the Gromov-Hausdorff topology on compact metric spaces. The limit  $(\mathbf{m}_{\infty}, d^*)$ , known as **the Brownian map**, is a random compact metric space that is almost surely homeomorphic to  $\mathbb{S}^2$ .

This limit is clearly a good candidate for our "uniform" random surface. The only drawback is that it does not come with a canonical conformal embedding into the sphere. Thus, it does not give us immediately the canonical conformal structure that is desirable from the perspective of Liouville quantum gravity.

Another approach to this question has been to try and define the limiting metric space directly in the continuum, using a further canonical object known as the Gaussian free field (GFF). This is the natural analogue of Brownian motion in two dimensional time: it is a planar, conformally invariant Gaussian field, that also arises from numerous physical models. The GFF describes the fluctuations of an electric potential, the scaling limit of various discrete "height-function" models, such as the dimer model, and is also a centrally important object in quantum field theory.

Motivated by non-rigorous constructions in the physics literature, one would like to define a random "Liouville quantum gravity" surface from a Gaussian free field h, roughly, by taking it to have Riemannian metric tensor equal to  $e^{\gamma h} dz$ . In fact, the Gaussian free field is too rough to be defined pointwise, so to define these things rigorously is a non-trivial task. Nevertheless, remarkable progress has been made recently by Miller and Sheffield, who are able to define a metric on the sphere using the GFF, and also connect it to the Brownian map [MS15a, MS15b, MS16e].

The final object that will be considered in this thesis is Aldous' continuum random tree [Ald91]. This is a random metric space that again arises as the scaling limit of various critical models. As an example, take the Galton–Watson tree: a simple and classical model associated with many natural biological processes. It has been proved [Ald93, LGD02, Mie08], that if one takes a critical (multitype) Galton–Watson tree and conditions it to be large in some sense, then it will converge after rescaling to the Brownian CRT.

**Theorem 1.3** (Convergence of critical (single-type) Galton–Watson trees). Suppose that  $\mathcal{L}$  is an offspring distribution with  $\mathbb{E}[\mathcal{L}] = 1$  and  $\mathbb{E}[\mathcal{L}^2] = \sigma^2$ . Let  $T_n$  be a sample of the associated Galton–Watson tree, viewed as a random metric space with distance  $d_n$ , and conditioned to reach height n. Then

$$(T_n, \frac{1}{n}d_n) \xrightarrow{(d)} (\mathcal{T}_{\mathbf{e}}, d_{\mathbf{e}})$$

with respect to the Gromov-Hausdorff topology as  $n \to \infty$ , where  $(\mathcal{T}_{\mathbf{e}}, d_{\mathbf{e}})$  is the Brownian continuum random tree: the real tree encoded by a Brownian excursion  $\mathbf{e}$  conditioned to reach height 1.

In this thesis we will see another example, where the CRT appears as a scaling limit for the genealogy of critical branching diffusions.

To conclude, we remark that one of the most fascinating things about this collection of canonical objects, and indeed many others that have not been mentioned here, is the web of connections between them. For example, it turns out that SLE curves actually appear as level lines and flow lines of the Gaussian free field [Dub09, SS09, MS16a], and in another direction, [DMS12] proves that Liouville quantum gravity can be constructed as a "mating" of two continuum random trees. We will be particularly interested in exploring such connections here. Specifically, we will look more closely at the relationship between SLE and the GFF in Chapter 5, and at the relationship between branching processes and Liouville quantum gravity in Chapter 4.

#### **1.2** Results and Outline

We begin, in Section 1.3, by giving a brief overview of the main objects appearing in this thesis. This will include relevant definitions, results, and a little historical context. We then conclude the introduction by mentioning a few open problems, and discussing possible directions for future research.

In Chapter 2, we move on to our first main topic: invariance principles for branching diffusions in bounded domains. We consider branching diffusions in a bounded domain D of  $\mathbb{R}^d$  in which particles are killed upon hitting the boundary  $\partial D$ . It is known [Sev58, Wat65] that such a process undergoes a phase transition when the branching rate exceeds a critical value. We investigate the system at criticality, and prove an asymptotic for the probability of survival up to large times. We show further that the genealogical tree associated with such a critical process converges to Aldous' continuum random tree, under appropriate rescaling. This result holds under only a mild assumption on the domain, and is valid for all branching mechanisms with finite variance, and a general class of diffusions.

Next, in Chapter 3, we turn our attention to the theory of Gaussian multiplicative chaos. This is the framework, developed by Kahane in the 1980s, that allows one to rigorously define random measures of the form " $e^{\gamma h(z) - \frac{\gamma^2}{2} \operatorname{var}(h(z))} dz$ ", when  $\gamma > 0$  is a parameter and h is a rough Gaussian field. It turns out that at a critical value  $\gamma = \sqrt{2d}$ , where d is the dimension of the underlying space, the usual construction of these measures yields something degenerate. Our main result in this chapter is that, if we approximate the field h by convolving with a mollifier, and use a "derivative" renormalisation at criticality (formally, taking the derivative of the subcritical measures in  $\gamma$  and evaluating at  $\gamma = 2$ ), we can obtain a non-trivial limit measure.

One of the fields covered by this result is the 2-dimensional Gaussian free field. As discussed in the Section 1.1, the chaos measure in this case (known as the Liouville measure) is of particular interest. In Chapter 4, we give an alternative construction of this measure, that is inspired by the theory of branching random walks and multiplicative cascades. We prove that the Liouville measure can be approximated, including at criticality, using certain sequence of "local sets" for the Gaussian free field. This is in an extremely simple and natural construction, as it is defined entirely using the local geometry of the field (in particular, not using any Gaussian process machinery). It also resolves a conjecture of Aïdekon, [Aid15].

Finally, in Chapter 5 we study in more detail the "level line" coupling [SS09] between  $SLE_4$  and the Gaussian free field. We generalise the existing theory by considering a free field with general boundary data F, rather than the specific

examples of boundary data that had been considered before. We show that the level lines of this field exist as continuous curves, under only the assumption that F is regulated (i.e., admits finite left and right limits at every point), and satisfies certain inequalities. Moreover, we show that these level lines are almost surely determined by the field. This allows us to define and study a generalization of the  $SLE_4(\underline{\rho})$ process, now with a continuum of force points. A crucial ingredient is a monotonicity property in terms of the boundary data which strengthens a result of Miller and Sheffield [MS16a] and is also of independent interest.

Chapter 5 is joint work with Hao Wu from the University of Geneva and Chapter 4 was completed in collaboration with Juhan Aru and Avelio Sepúlveda from ETH Zürich. The work in Chapter 5 is based on the accepted paper [PW17], and the work in Chapters 2, 3 and 4 are based on the preprints [Pow17b], [Pow17a] and [APS17]. All of this work was supported by the UK Engineering and Physical Sciences Research Council (EPSRC) grant EP/H023348/1 for the University of Cambridge Centre for Doctoral Training, the Cambridge Centre for Analysis.

#### **1.3** Preliminaries

We will now discuss in a little more detail some of the main objects that we will be concerned with.

#### 1.3.1 Brownian Motion

As discussed in the introduction, Brownian motion plays a central role in our understanding of physical systems. It is a canonical object to study for probabilists, since it arises from the most simple model of motion: the random walk. Indeed, Donsker's invariance principle (see for example [MP10]), tells us that any unbiased random walk will converge after rescaling to a Brownian motion.

We will briefly recall here some basic properties of Brownian motion, but this is by no means a complete survey. For a more detailed introduction see, for example, [MP10].

Brownian motion can be characterised in many different ways, but we will use the following definition:

**Definition 1.4** (Brownian Motion). Let  $(B_t)_{t\geq 0}$  be a continuous random process in  $\mathbb{R}^d$  starting from 0. If  $(B_t)_{t\geq 0}$  is a zero-mean Gaussian process with

$$\mathbb{E}[B_s B_t] = s \wedge t$$

for all  $s, t \ge 0$ , then we say that  $(B_t)_{t\ge 0}$  is a Brownian motion.



Figure 2: A one-dimensional Brownian motion.

**Remark 1.5.** To start a Brownian motion at a position x other than 0, as we will often want to do, we simply consider the process  $x + B_t$ , where  $B_t$  is as in Definition 1.4.

One of the most important properties of Brownian motion is the following Markov property. It says that if one stops a Brownian motion at any given time t, its evolution from time t onwards will simply be that of another Brownian motion, starting from  $B_t$ , but otherwise completely independent of what happened before. We write  $(\mathcal{F}_s^B)_{s\geq 0}$  for the filtration generated by  $(B_s)_{s\geq 0}$ .

**Lemma 1.6** (Weak Markov Property). For any  $s \ge 0$ , the process  $(B_{t+s} - B_s)_{t\ge 0}$ is a standard Brownian motion, independent of  $\mathcal{F}_s^B$ .

In fact, this property does not only hold at deterministic times. To describe this phenomena we need the notion of *stopping times*.

**Definition 1.7** (Stopping times). A random variable T defined on the same probability space  $(\Omega, \mathcal{F}, \mathbb{P})$  as a Brownian motion  $(B_t)_{t\geq 0}$  is a stopping time for the Brownian motion if, for all  $t \geq 0$ ,

$$\{T \le t\} \in \mathcal{F}_t^B.$$

These times are precisely those at which we have a stronger version of Lemma 1.6. For T and B as above, we define the *stopped*  $\sigma$ -algebra by setting

$$\mathcal{F}_T^B = \{ A \in \mathcal{F} : A \cap \{ T \le t \} \in \mathcal{F}_t^B \ \forall t \ge 0. \}$$

**Lemma 1.8** (Strong Markov Property). Let B be a standard Brownian motion and T be a stopping time for B which is almost surely finite. Then the process

$$(B_{t+T} - B_T)_{t \ge 0}$$

is a standard Brownian motion, independent of  $F_T^B$ .

A variant of Brownian motion that will be relevant to what comes next is the *Brownian Bridge*. This is a Brownian motion on some interval of time, let's say [0, 1], that is conditioned to start and end at 0. Although this is an event of probability zero, one can make sense of this conditioning, and obtain a process which is characterised as follows.

**Definition 1.9** (Brownian Bridge). Let  $(W_t)_{t \in [0,1]}$  be a continuous Gaussian process with

- $W_0 = W_1 = 0$  almost surely.
- $\mathbb{E}[W_t] = 0 \ \forall t \ and$
- $\operatorname{cov}(W_s, W_t) = s \wedge t st \text{ for all } s, t \ge 0.$

Then we say that W is a Brownian bridge on [0, 1].

We conclude this section by mentioning a very special property of Brownian motion in two dimensions. This is the property of *conformal invariance*. We will see it appearing again and again in the following sections, and it will play a central role in this thesis. To define it, we need the following fundamental theorem:

**Theorem 1.10** (Riemann Mapping Theorem). Let  $\mathbb{D}$  be the open unit disc and  $D \subset \mathbb{C}$  be any proper, simply connected domain. Then there exists a conformal isomorphism  $\Phi: D \to \mathbb{D}$ . That is a bijection  $\Phi$  such that  $\Phi$  and  $\Phi^{-1}$  are holomorphic with non-vanishing derivative.

**Lemma 1.11** (Conformal Invariance of Brownian Motion). Let D and D' be simply connected complex domains and let B, B' be Brownian motions in D and D' respectively, starting at z, z' and stopped upon leaving D, D'. Let  $\Phi : D \to D'$  be a conformal isomorphism. Then  $\Phi(B)$  and B' have the same law, up to a time change.

#### **1.3.2** Gaussian free field

The planar Gaussian free field (GFF) is another canonical, conformally invariant object in two dimensions. In fact, we will see that it is the natural generalisation of Brownian motion, or more specifically, the Brownian bridge, to two-dimensional time. This section will mainly follow the approach taken in [Ber15b]. We refer the reader to these excellent notes for further details and proofs, as well as to [She07] and [Wer].

We will begin by briefly discussing the discrete counterpart of the GFF, the *Discrete Gaussian free field*, in order to get some intuition. Take a finite graph G = (V, E) with a special set of boundary edges  $\partial$ , and let  $X_n$  be a random walk on G. Write  $\mathbb{P}_x$  for the law of  $X_n$ , when it is started from position  $x \in V$ .

**Definition 1.12** (Discrete Green Function). The discrete Green function G(x, y) is defined, for  $x, y \in V$  by

$$G(x,y) := \frac{1}{d(y)} \mathbb{E}_x \left( \sum_{n=0}^{\infty} \mathbb{1}_{\{X_n = y; \tau > n\}} \right)$$

where  $\tau$  is the first time that  $X_n$  hits  $\partial$ .

We then define the Discrete GFF on G in a very simple way. Let

$$V = \{x_1, \cdots, x_n\}.$$

**Definition 1.13** (Discrete GFF). The Discrete GFF is the centered Gaussian vector  $(h(x_i))_{1 \le i \le n}$  on V, with covariances given by the discrete Green function.





From this definition, one can in fact prove that the density of h with respect to  $\prod dx_i$  takes a very special form. It is proportional to

$$\exp\left(-\frac{1}{4}\sum_{x,y\in V:x\sim y}(h(x)-h(y))^2\right)$$

where the term  $\sum_{x,y\in V:x\sim y} (h(x) - h(y))^2$  is what is known as the *Dirichlet energy* of h: an energy that is minimised by harmonic functions. Thus, the DGFF can be considered a natural perturbation of harmonic functions, just as the random walk, or Brownian motion, can be considered a perturbation of linear ones.

We would like the continuum Gaussian free field to also be such a perturbation. Unfortunately, it turns out that in this case, as we will discuss shortly, it cannot even be defined as a function! With this in mind, perhaps the simplest way to think of the GFF is as a limit of discrete Gaussian fields, in the same way that Brownian motion is often thought of as the limit of random walks. Indeed, one can define it in this way, and this is useful to bear in mind. However, we will approach things more directly. To do this, we first need an analogue of the discrete Green function. Let  $D \subset \mathbb{C}$  be a bounded domain, and let  $p^{D}(t, x, y)$  be the transition density <sup>1</sup> for a Brownian motion killed when leaving D.

**Definition 1.14** (Continuum Green Function). We define the Green function on D by

$$G_D(x,y) := \int_0^\infty p^D(t,x,y) \, dt$$

Similarly to Definition 1.13, we would like to define the continuum GFF to be the centered Gaussian function with covariances given by the Green function. However, there is now a problem, as the Green function is not actually well defined. In fact, it can be characterised as the only function in D that blows up logarithmically on the diagonal and is harmonic away from it.<sup>2</sup>

Although this singularity prevents us from defining the GFF pointwise, we can still define it in other ways. One approach, using the Kolmogorov extension theorem, is to define it as functional acting on the set

$$\mathcal{M} := \{ \rho = \rho^+ - \rho^- \text{ s.t. } \int_{D^2} \rho^{\pm}(x) \rho^{\pm}(y) G_D(x, y) \, dx \, dy < \infty \}$$

of signed measures with finite Green energy. We can think of this action as "integrating" the GFF against these measures (which will make more sense in light of the following discussion, see Lemma 1.17.)

**Definition 1.15** (Gaussian free field zero boundary conditions [Ber15a]). There exists a unique stochastic process  $(h_{\rho})_{\rho \in \mathcal{M}}$  indexed by  $\mathcal{M}$  such that for every  $\rho_1, \dots, \rho_n \in \mathcal{M}$ 

$$(h_{\rho_1},\cdots h_{\rho_n})$$

is a centered Gaussian vector with

$$\operatorname{cov}(h_{\rho_i}, h_{\rho_j}) = \int_{D^2} G_D(x, y) \rho_i(x) \rho_j(y) \, dx \, dy$$

**Remark 1.16.** Recall Definition 1.9 of the Brownian bridge: it is the unique centered Gaussian process on [0,1] with covariances given by the function  $G(s,t) = s \wedge t - st$ . In fact, this function G is exactly the Green function on [0,1].

<sup>&</sup>lt;sup>1</sup>That is,  $p^{D}(t, x, y) = p(t, x, y) \mathbb{P}_{x,y}^{t}(W_{s} \in D \ \forall s \in [0, t])$ , where  $p(t, x, y) = (2\pi t)^{-1} \exp(-|x - y|^{2}/2t)$  is the transition density for Brownian motion in  $\mathbb{R}^{d}$  and  $\mathbb{P}_{x,y}^{t}$  is the law of a Brownian bridge from x to y on [0, t].

<sup>&</sup>lt;sup>2</sup>More precisely, it is the only function such that  $G_D(x, y) = -\log(|x - y|) + O(1)$  as  $y \to x, \forall x \in D$ and  $\Delta G_D(x, \cdot) = -2\pi \delta_x(\cdot)$  as a distribution.

We can intepret the above as saying that the GFF has "covariances given by the Greens' function". But how does the definition fit in with our intuition in terms of Dirichlet energy? This can be made clearer with an alternative approach.

It turns out that one can also view the GFF as a random distribution. That is, as a continuous linear map  $\mathcal{D}(D) \to \mathbb{R}$ , where  $\mathcal{D}(D)$  is the set of smooth, compactly supported functions in D.<sup>3</sup> We denote the set of distributions by  $\mathcal{D}'(D)$ , and endow it with the weak-\* topology.

Define the Sobolev space  $H_0^1(D)$  to be the Hilbert space completion of  $\mathcal{D}(D)$ with respect to the *Dirichlet inner product*:

$$(f,g)_{\nabla} := \int_D \nabla f \cdot \nabla g.$$

Let  $(f_n)_{n\geq 1}$  be an orthonormal basis of  $H_0^1(D)$  and let  $(\alpha_n)_{n\geq 1}$  be a collection of i.i.d N(0,1) random variables. Then we can also construct the GFF using the following approximation:

Lemma 1.17. Set

$$h_n := \sum_{i=1}^n \alpha_i f_i.$$

This sum converges almost surely in the space of distributions (in fact, in the Sobolev space  $H_0^{-\varepsilon}(D)$  for any  $\varepsilon > 0$ ). Moreover, the limit agrees in distribution with the process h from Definition 1.15, when it is restricted to  $\mathcal{D}(D)$ .



Figure 4: A simulation of the discrete GFF using a lot of terms.

**Remark 1.18.** In fact, a density argument can be used to show that the above definition extends directly to give a process satisfying Definition 1.15 (see [Ber15b, Theorem 1.16]). With this in mind, we write  $(h, \rho)$  for  $h_{\rho}$  in Definition 1.15, as is standard for distributions.

<sup>&</sup>lt;sup>3</sup>We use the topology on  $\mathcal{D}(D)$  which is defined by saying that a sequence  $f_n \to 0$  if there exists a compact  $K \subset D$  supporting all the  $f_n$ , and such that  $f_n$  and  $f'_n$  converge uniformly to 0 on D.

Note that for any distribution f in  $\mathcal{D}'(D)$ , and function  $g \in \mathcal{D}(D)$ , one can define the Dirichlet inner product  $(f,g)_{\nabla}$  using the definition of the distributional derivative. This definition is just the natural extension of the Gauss–Green formula (integration by parts), and so we have that  $(f,g)_{\nabla} = -(f,\Delta g)$ . Thus, for the Gaussian free field, whenever g is such that  $-\Delta g = \rho \in \mathcal{M}$  we set

$$(h,g)_{\nabla} := (h,\rho).$$

This is then a Gaussian random variable with mean 0, and variance  $(g, g)_{\nabla}$ .

We should also point out that, in all of the above, we have been considering a Gaussian free field with *zero boundary conditions*. In fact, we will often want to work in a more general setting, where we allow it to have *non-zero mean*. The way we do this is simply to add on a deterministic harmonic function to the field. This harmonic function will be referred to as the *mean* of the distribution, and its restriction to the boundary will be referred to as its *boundary conditions*.

Now we will review some of the key properties of the Gaussian free field. The first is conformal invariance, and follows directly from the corresponding conformal invariance of the Green function (which in turn follows from the conformal invariance of Brownian motion, Lemma 1.11.)

**Lemma 1.19** (Conformal Invariance of the GFF). Let h be a Gaussian free field in D, and  $\phi: D \to D'$  be a conformal map. Then  $h \cdot \phi^{-1}$  has the law of a GFF on D'.

The second is a spatial Markov property. See [Ber15b, Theorem 1.17] for a proof.

**Lemma 1.20** (Weak Markov Property for the GFF). Take a deterministic closed set  $A \subset D$  and let h be a GFF with zero boundary conditions on D. Then we can write  $h = h_A + h^A$  where

- *h<sub>A</sub>* is a random distribution that is a harmonic function when restricted to *D \ A*;
- $h^A$  is a zero-boundary GFF in  $D \setminus A$ ; and
- $h_A$ ,  $h^A$  are independent.

Note the analogy between this Lemma, and Lemma 1.6. The latter says that a Brownian motion can be written, from any fixed time t onwards, as a sum of two parts:  $B_t$ , plus another independent Brownian motion. The above Lemma says that a Gaussian free field can be written, in any fixed open set, as a sum of two parts: a harmonic function, plus another independent Gaussian free field. It is therefore perhaps not particularly surprising that there is also a strong Markov property for the Gaussian free field. This is described by the notion of a local set.



Figure 5: Weak Markov property for the GFF.

**Definition 1.21** (Local sets, Strong Markov Property). Take a (now possibly random) relatively closed set  $A \subset D$  and let h be a GFF with zero boundary conditions on D. We say that A is a local set for h if there exists a distribution  $h_A$ , that is harmonic when restricted to  $D \setminus A$ , and is such that, conditionally on A,  $h^A = h - h_A$ is a zero-boundary Gaussian free field in  $D \setminus A$ .



Figure 6: Strong Markov Property for the GFF.

These sets are exactly analogous to stopping times for Brownian motion - they are random "stopping" sets at which a strong Markov property holds. One can even draw stronger analogies between particular types of local sets for the GFF and stopping times for Brownian motion. For examples of this, see [ASW15]. Local sets will play an important role in two of the papers included in this thesis, and we will see many natural, non-deterministic examples later on.

One final thing that we will discuss here is the so-called *circle average* process of the free field. This will be important in the following sections, as it is particularly useful way of approximating the field. Essentially we approximate the value of the Gaussian free field h at a point z by taking its average on smaller and smaller circles around z. That is, we define  $h_{\varepsilon}(z) := (h, \rho_{\varepsilon}^z)$ . where  $\rho_{\varepsilon}^z$  is the uniform measure on the circle of radius  $\varepsilon$  around z, and let  $\varepsilon \to 0$ . One can prove (see [DS11, Propn 3.1]) that a modification of h exists for which this process is jointly Hölder in z and  $\varepsilon$ . In fact (for fixed z) it has the same law as a Brownian motion. **Lemma 1.22.** Fix  $z \in D$  let  $h_{\varepsilon}(z)$  be defined as above, for  $\varepsilon \leq \varepsilon_0 < dist(z, \partial D)$ . Then

$$(h_{e^{-t}}(z))_{t \ge \log(1/\varepsilon_0)}$$

is a Brownian motion started from  $h_{e^{-\log(1/\varepsilon_0)}}(z)$ .

#### **1.3.3** Gaussian multiplicative chaos

Gaussian multiplicative chaos theory was introduced by Kahane [Kah85] in the 1980s in order to rigorously define measures of the form

$$\mu^{\gamma}(dz) = \mathrm{e}^{\gamma h(z)} \, dz$$

where h is a rough Gaussian field and  $\gamma > 0$  is a parameter. Although Kahane's original construction of these measures was restricted to a special class of fields h (having so called  $\sigma$ -positive kernels) it has since been extended (at least for some values of  $\gamma$ ) to general log-correlated fields [Ber15a, RV10]. By a log-correlated field we mean a Gaussian field h defined as in Definition 1.15, but with the Green function  $G_D$  replaced by another non-negative definite kernel K satisfying

$$K(x,y) = -\log|x-y| + O(1) \quad \text{as } x \to y.$$

The basic idea to construct the measures  $\mu^{\gamma}$ , is to approximate the field *h* using a sequence of smooth fields. A natural way to do this is to convolve *h* with a mollifying measure  $\theta$ , supported in a compact set and with total mass 1. This yields the approximations

$$h_{\varepsilon}(z) := h \star \theta_{\varepsilon}(z) = (\theta_{z,\varepsilon}, h)$$

for  $\varepsilon > 0$ , where  $\theta_{\varepsilon}$  is the image of the measure  $\theta$  under the map  $x \mapsto \varepsilon x$ , and  $\theta_{\varepsilon,z}$  is the image of  $\theta_{\varepsilon}$  under the map  $x \mapsto x + z$ . Then one can define the approximate measures

$$\mu_{\varepsilon}^{\gamma} := \mathrm{e}^{\gamma h_{\varepsilon}(z)} \, \mathbb{E}[\mathrm{e}^{\gamma h_{\varepsilon}(z)}]^{-1}$$

and try to take a limit as  $\varepsilon \to 0$ . Note that for each z, the process  $e^{\gamma h_{\varepsilon}(z)} \mathbb{E}[e^{\gamma h_{\varepsilon}(z)}]^{-1}$  is a martingale. We have the following result.

**Theorem 1.23** ([Ber15a, RV10]). Let h be a log-correlated field and  $\theta$  a measure supported in B(0, 1), with total mass 1 and satisfying  $\int \log(1/|w-v|)\theta(dw) = O(1)$ uniformly in  $v \in B(0, 5)$ . Then for  $\gamma < \sqrt{2d}$ ,  $\mu_{\varepsilon}^{\gamma}$  converges weakly in probability to a random measure  $\mu^{\gamma}$  as  $\varepsilon \to 0$ . Moreover,  $\mu^{\gamma}$  a.s. has no atoms,  $\mu^{\gamma}(A) > 0$  for any  $A \subset D$  open, and  $\mu^{\gamma}$  does not depend on the choice of mollifier  $\theta$ . **Remark 1.24.** This includes the case when h is the 2d-GFF in a domain D and  $\theta$  is uniform measure on the unit circle. In Chapter 4, based on [APS17], we will give an alternative construction of this measure using only the local geometry of the field and its local sets.

Note that this theorem only holds for  $\gamma < \sqrt{2d}$ . Indeed, it is known [RV10], that the measures  $\mu_{\varepsilon}^{\gamma}$  converge to the 0 measure almost surely if  $\gamma \ge \sqrt{2d}$ . Therefore, to define something non-trivial in the critical ( $\gamma = \sqrt{2d}$ ) and supercritical ( $\gamma > \sqrt{2d}$ ) cases, it is necessary to give an extra *push* as to the measures  $\varepsilon \to 0$ . That is, to make an additional renormalisation. These cases turn out to be much more tricky to deal with than the subcritical case. One of the main reasons for this is that the  $\mu_{\varepsilon}^{\gamma}$  mass of a fixed set is no longer uniformly integrable, meaning that existence of a limit is much harder to show.

We first discuss the critical case. In analogy to the corresponding results for Gaussian multiplicative cascades (a discrete counterpart of Gaussian multiplicative chaos related to the branching random walk, see Section 1.3.5) we expect to be able to renormalise at criticality and obtain a non-trivial limit measure in two different ways:

- (1) by instead considering  $\sqrt{\log(1/\varepsilon)}\mu_{\varepsilon}^{\sqrt{2d}}$  or
- (2) by taking the derivative measures  $D_{\varepsilon}(dz) := \frac{d}{d\gamma} \mu_{\varepsilon}^{\gamma}(dz)|_{\gamma = \sqrt{2d}}$

The first is a deterministic renormalisation, known as the Seneta–Heyde rescaling, and has the advantage of yielding a sequence of positive approximating measures. The downside, however, is that  $\sqrt{\log(1/\varepsilon)} e^{\gamma h_{\varepsilon}(z)} \mathbb{E}[e^{\gamma h_{\varepsilon}(z)}]^{-1}$  is no longer a martingale for each z. On the other hand, the derivative, random, renormalisation preserves the martingale property. The drawback here is that it gives only a sequence of signed measures.

Both problems are reasonably difficult to get around. However, in [DRSV14a, DRSV14b] the authors considered a special class of fields h, with so-called  $\star$ -scale invariant kernels, and used a different sequence of "cut-off" approximations to the field to define the critical chaos measures. In this specific set-up, they were able to show that both sequences (1) and (2) yield a non-trivial limiting measure, and in fact, they are the same up to a constant.

**Theorem 1.25.** Suppose h has a  $\star$ -scale invariant kernel  $K(x, y) = \int_1^\infty k(u|x - y|)/u \, dx$ , and the approximate fields  $h_{\varepsilon}$  have kernels given by

$$K_{\varepsilon}(x,y) := \int_{1}^{1/\varepsilon} \frac{k(u|x-y|)}{u} \, du.$$

Then the two sequences of approximating measures (1) and (2) converge to the same limiting measure, up to a constant  $\sqrt{2/\pi}$  - the limit of (1) is  $\sqrt{2/\pi}$  times the limit of (2).

They also extended the above result to cut-off approximations for the 2d-GFF, which does not quite fit into the \*-scale invariant framework.

However, these approximations do not correspond to convolving the field with a mollifier. To complete the picture at criticality therefore, it remains to show that (1) and (2) yield (the same) limiting measure when h is a general log-correlated field and  $h_{\varepsilon}$  is a convolution approximation.

This was done for (1) when h has  $\star$ -scale invariant kernel by Junnila and Saksman [JS17], and also when h is the 2d-GFF [JS17, HRV15]. The corresponding result (2) is the main result of [Pow17a] (and Chapter 3.)

In the supercritical case  $(\gamma > \sqrt{2d})$  less is known, but there are still alternative constructions that should lead to non-trivial chaos measures. We refer the reader to the survey [RV14] for a discussion of what is expected from the physics literature, and of recent mathematical progress (see in particular Section 6.2.)

#### 1.3.4 Liouville quantum gravity

Liouville quantum gravity is a canonical model of a random Riemannian surface, first studied by physicists in the 1980's [Pol81a, Pol81b]. It was introduced to extend the notion of Feynman path integrals (closely related to Brownian motion) to Feynman integrals over surfaces. Naïvely, one would like to define a Liouville quantum gravity surface as the Riemannian surface with metric tensor given by

$$e^{\gamma h(z)} dz$$

where  $\gamma > 0$  is a real parameter, and h is an instance of the Gaussian free field. This would result in a random surface, parameterised by say, the sphere, in which areas where h is large correspond to those with large area, and areas where h is small correspond to those with small area.

Note the connection with the chaos measures  $\mu^{\gamma}$  described in the previous section. The difference is that here we want to define a random surface (as a metric space, together with a conformal structure) rather than just a volume measure. It turns out that this is a much more difficult problem.

Connection to the Brownian map. As discussed in Section 1.1, another approach to LQG is to consider the scaling limit of certain sequences of random planar maps. This yields a random metric space, known as the Brownian map, by the work of

Le Gall [LG13] and Miermont [Mie13]: see Theorem 1.2. This metric space has the topology of the sphere, but comes without any canonical conformal embedding. So, on one hand we have a metric space with no conformal structure, and on the other, we have a conformal structure with no metric. It has long been conjectured that these two objects should be closely related in the case when  $\gamma = \sqrt{8/3}$ , known as *pure gravity*. Moving forward, however, has proved extremely difficult.



Figure 7: A Random Planar Map (Simulation by Nicolas Curien).

Despite the difficulties presented by this problem, Miller and Sheffield [MS15a, MS15b, MS16e] have recently found a way to connect the two. Using a random growth process called QLE they are able to put a metric on  $\sqrt{8/3}$ -Liouville quantum gravity which agrees in law with the Brownian map. Conversely, they can show that the Brownian map comes almost surely with a canonical conformal structure, and that the resulting conformal sphere (equipped with a measure) agrees in law with  $\sqrt{8/3}$ -Liouville quantum gravity. This represents an enormous breakthrough in our understanding of these random surfaces.

## **1.3.5** Branching Brownian motion and connections with Gaussian multiplicative chaos.

In this section, we will provide a (very) brief introduction to branching Brownian motion. This will not only be the central object of study in Chapter 2, but is also linked to the theory of Gaussian multiplicative chaos. The process in the simplest case (binary branching) is defined as follows:

**Definition 1.26** (Branching Brownian Motion). We start, at time 0, with a single particle at position  $x \in \mathbb{R}^d$ .

- This particle moves according the law of a Brownian motion in  $\mathbb{R}^d$ .
- After an Exp(β)-distributed waiting time, independent of the motion, the particle splits into two.
- Each offspring particle repeats stochastically the behaviour of the initial particle, starting from the point of fission, and independently of one another.



Figure 8: A Branching Brownian Motion (simulation by Matt Roberts).

This definition can be extended, as will be done in Chapter 2, to more general branching mechanisms and individual particle motions. Here we will study the behaviour of the process when we introduce a bounded domain to the model, and kill particles upon hitting the boundary. It turns out that doing this induces a phase transition in the system, and we will study what happens at criticality. However, we will not discuss this further until Section 1.3.8, and leave the precise definitions to Chapter 2.

One crucial tool in Chapter 2 will be to make us of an certain martingale associated with the process.

**Lemma 1.27** (The Exponential Martingale). Let  $(X_t^1, \dots, X_t^{N_t})$  denote the positions of particles in a branching Brownian motion (in the whole of  $\mathbb{R}^d$ ) at time t. Then for  $\lambda \in \mathbb{R}$  the process

$$W_{\lambda}(t) := \sum_{i=1}^{N_t} e^{\lambda X_t^i + (\lambda^2/2 - \beta)t}$$

is a martingale.

Now, the discrete time counterpart of BBM, the branching random walk, turns out to be closely related to Gaussian multiplicative chaos. Indeed, the branching



Figure 9: Branching Brownian Motion in a bounded domain (simulation by Henry Jackson).

random walk gives rise to a very similar construction of measures - known as *multiplicative cascades*.

To illustrate this, let us describe multiplicative cascade measures on the interval [0,1]. Denote the dyadic subintervals of [0,1] by  $\mathcal{I}_n = \{[k/2^n, (k+1)/2^n] : 0 \le k \le 2^n - 1\}$  and let  $\mathcal{I} = \bigcup_n \mathcal{I}_n$ . Take W > 0 any random variable with mean one and let  $(W_I)_{I \in \mathcal{I}}$  be independent copies of W. Then we can define a sequences of measures on  $([0,1], \mathcal{B})$  inductively, by setting  $\mu_0$  to be Lebesgue measure and defining

$$\mu_{n+1} = \sum_{I \in \mathcal{I}_n} W_I \cdot \mu_n |_I$$

for each  $n \ge 0$ . The following result is originally due to Kahane and Peyrière [KP76].

**Theorem 1.28.** Define  $\mu_n$  as above. Then

- μ<sub>n</sub> → μ a.s. for some limiting measure μ (in the sense of weak convergence of measures.)
- If in addition E[W log<sub>2</sub> W] < 1, then μ([0,1]) > 0 almost surely and μ almost surely has no atoms.

Note the similarity to Theorem 1.23.

We can alternatively use the random variable W to define a branching random walk in the following way. Begin at time 0 with one particle (at 0) and after the first discrete time step, let this particle branch into two. Let each of the children of the initial particle be displaced from 0 (the position of their parent) by an independent random variable with law  $\log(W)$ . Repeat the procedure. It is not hard to see that the exponential martingale associated with this process (as in Lemma 1.27, when  $\lambda = 1$ ) has exactly the same law as  $\mu_n([0, 1])$ . Thus, the theory of multiplicative cascades boils down to the theory of exponential martingales for the branching random walk (see for example, [BK77, Kyp00, Lyo97]).

In a similar fashion, many classical techniques from the theory of branching processes can be adapted to prove results about Gaussian multiplicative chaos. This is due to the fact that the circle average process of a GFF is in fact closely linked with branching Brownian motion. We saw in Lemma 3.12 that the circle average process around a single point  $z \in D$  is a Brownian motion. However, there is also an underlying branching structure. Roughly speaking, if one takes the circle averages around two different points z and w, they will be the same until the circles become disjoint, and then will move independently. For a rigorous statement along these lines see [DRSV14b, Appendix A].

We will see more connections between branching processes and LQG measures in Chapter 4. In fact, our construction of the Liouville measure using local sets is simply an analogue of the fact that branching random walk martingales also converge along certain random time sets known as *stopping lines* [BK04].

#### **1.3.6** Schramm–Loewner evolutions

The aim of this section will be to give a brief introduction to the theory of Schramm– Loewner evolutions (SLE). These are a family of random fractal curves in the plane, that were introduced by Oded Schramm [Sch00] as the only possible candidates for the scaling limits of certain interfaces in discrete lattice models. Schramm realised that any such scaling limits would have to satisfy two defining properties (conformal invariance and a certain spatial Markov property) and was able to classify the range of possibilities as a one-parameter family.

As an example, let us consider percolation in the triangular lattice. Place a triangular lattice of mesh size  $\delta$  on the upper half plane  $\mathbb{H} = \{z \in \mathbb{C} : \Im \mathfrak{m}(z) > 0\}$  and colour the sites of the lattice black or white with probability p, independently of one another. This model displays a phase transition when p = 1/2; it is the largest value of p for which all the monochromatic clusters have finite size almost surely. As with many such models, the system displays interesting behaviour at criticality.

One natural thing to consider is the collection of interfaces between monochromatic clusters, which results in a collection of discrete loops. In fact, this is rather complicated to study, and so in order to simplify things we introduce some boundary conditions. We insist that all vertices in the lattice lying the negative real line are coloured black (we assume that the lattice has been placed such that  $\pm \delta/2$  are lattice points and 0 is not), and that all lattice points on the positive real line are coloured white. This induces a single interface, from the origin to  $\infty$ , that keeps black vertices on its left, and white vertices on its right. It is known as the *exploration process* for percolation.

The problem is to describe what happens to this exploration interface as the mesh size of the lattice goes to 0. It was proved by Smirnov [Smi01] that the interface converges to a continuous curve known as  $SLE_6$ . The fact that the scaling limit must be given by this curve is a consequence of properties of the discrete interfaces (conformal invariance, and a "domain Markov" property) that should be preserved in the continuum.

To define the SLE curves, we will need a few tools from complex analysis. For more details we refer the reader to the classical textbook [Law05] or to the lecture notes [BN], where one can also find proofs of all the results.

**Complex**  $\mathbb{H}$ -hulls A complex domain  $D \subset \mathbb{C}$  is a non-empty connected open subset of the complex plane. We say that D is simply connected if  $\mathbb{C} \setminus D$  is connected in  $\mathbb{C} \cup \infty$ . As we saw in Section 1.3.1, such domains are connected by a powerful theorem, the Riemann Mapping Theorem. This says that given any two simply connected domains we can choose a conformal isomorphism mapping one to the other.

We say that  $K \subset \mathbb{H}$  is a *complex*  $\mathbb{H}$ -*hull* if K is bounded and  $H := \mathbb{H} \setminus K$  is a simply connected domain. This means that K is connected to the real line, and H is a neighbourhood of  $\infty$  in  $\mathbb{H}$ . For any such hull, we know by the Riemann Mapping Theorem that there must exist a conformal isomorphism  $g_K : H \to \mathbb{H}$ . Moreover, we can fix this isomorphism by requiring it to have a certain "hydrodynamic normalisation" at  $\infty$ . That is, we define  $g_K$  to be the unique conformal isomorphism from  $H \to \mathbb{H}$ , such that  $g_K(z) - z \to 0$  as  $z \to \infty$ . One can prove that, given this choice of  $g_K$ , we have the expansion

$$g_K(z) = z + \frac{a_K}{z} + O(|z|^{-2})$$

as  $z \to \infty$ , where  $a_K \ge 0$  is a real constant.

**Definition 1.29** (Half-plane capacity). This constant  $a_K$  is known as the half plane capacity of K and denoted by hcap(K).

In some sense, the half-plane capacity measures the size of the hull K, when "viewed from infinity". Indeed, one can prove that

$$\operatorname{hcap}(K) = \lim_{y \to \infty} y \mathbb{E}_{iy}[Im(B_T)]$$

where B is a Brownian motion started at iy and T is the time that it leaves the domain H. In particular, the half plane capacity increases as a hull increases: if  $K \subset K'$  are two complex  $\mathbb{H}$ -hulls, then  $hcap(K) \leq hcap(K')$ .

#### Loewner Chains

**Definition 1.30** (Loewner Chain). A Loewner chain is a family  $(K_t)_{t\geq 0}$  of complex  $\mathbb{H}$ -hulls satisfying the following two properties:

- $K_t$  is increasing:  $K_s \leq K_t$  for  $s \leq t$ , and
- $K_t$  satisfies the following local growth property: for any  $T \ge 0$

$$\sup_{s,t\in[0,T],|s-t|\leq h} \operatorname{rad}\left(g_{K_s}(K_t\setminus K_s)\right) \to 0 \ as \ h\to 0$$

where rad(K) is the radius of the smallest semicircle that can be placed on the real line to completely encapsulate K.



For any such family one can prove that, for all  $t \ge 0$ , there exists a unique  $\xi_t \in \mathbb{C}$  with  $\xi_t \in \overline{g_{K_t}(K_{t+h} \setminus K_t)}$ . This process  $(\xi_t)_{t\ge 0}$  turns out to be a continuous, real-valued function of time, and is known as the *Loewner transform* of the chain, or the *driving function*.

Another thing to point out is here (which of course requires some complex analysis to prove) is that the map  $t \mapsto hcap(K_t)$  induces a homeomorphism of  $[0, \infty)$ . Thus, we may assume (by reparameterisation) that

$$hcap(t) = 2t \ \forall t \ge 0$$

whenever we are considering a Loewner chain.

The observation of Loewner was that continuous driving functions  $\xi_t$  and Loewner chains  $K_t$  are in fact in one-to-one correspondence. Moreover, they are connected, together with the sequence of maps  $g_t : \mathbb{H} \setminus K_t \to \mathbb{H}$ , by a certain differential equation. **Theorem 1.31** (Loewner's Theorem). Let  $(\xi_t)_{t\geq 0}$  be a continuous real valued function and for each  $z \in \mathbb{H}$  let  $g_t(z)$  be the maximal solution to the Loewner equation

$$\frac{\partial g_t(z)}{\partial t} = \frac{2}{g_t(z) - \xi_t}, \ g_0(z) = z$$

which exists on some time interval  $[0, \zeta(z)]$  by classical ODE theory. Let  $K_t = \{z \in \mathbb{H} : \zeta(z) \leq t\}.$ 

Then  $(K_t)_{t\geq 0}$  is a Loewner chain, parameterised so that  $hcap(K_t) = 2t$ , and with driving function  $\xi_t$ . Moreover,  $g_t$  is the unique sequence of maps  $\mathbb{H} \setminus K_t \to \mathbb{H}$  with the hydrodynamic normalisation at  $\infty$ .

- **Remark 1.32.** The time,  $\zeta(z)$ , when the solution to the Loewner equation for  $g_t(z)$  fails to exist, is the time that the growing hull  $K_t$  "swallows" the point z.
  - One can prove that continuous curves  $(\gamma_t)_{t\geq 0}$  in  $\mathbb{H}$  which do not cross themselves and have  $|\gamma_t| \to \infty$  as  $t \to \infty$  are examples of Loewner chains. In this case the map  $g_t$  sends the tip of the curve,  $\gamma_t$ , to the point  $\xi_t$  (where  $g_t$  is extended by continuity.)

Chordal SLE We are now ready to define Schramm–Loewner evolutions.

**Definition 1.33** (Chordal SLE in  $\mathbb{H}$  from  $0 \to \infty$ ). For  $\kappa > 0$ , SLE<sub> $\kappa$ </sub> in  $\mathbb{H}$  from 0 to  $\infty$  is defined to be the Loewner chain driven by  $\xi_t = \sqrt{\kappa}B_t$  where  $B_t$  is a standard Brownian motion.

Not all Loewner chains are generated by curves in the sense of Remark 1.32. Hence the obvious question: is SLE? This was answered by Rohde and Schramm for  $\kappa \neq 8$  in their seminal work [RS05] on SLE. It was later completed for the case  $\kappa = 8$  by Lawler, Schramm and Werner, [LSW04].

**Theorem 1.34** (Rohde–Schramm, Lawler–Schramm–Werner). For every  $\kappa > 0$ , SLE<sub> $\kappa$ </sub> is generated by a a random continuous curve  $\gamma$ .

Thus, SLE are random curves *driven* by Brownian motion. Intuitively, one can think of SLE as random curves travelling from 0 to  $\infty$  which change direction whenever an underlying Brownian motion increases or decreases.

One of the first things to note about SLE is that, due to the scaling property of Brownian motion ( $B_t$  has the same law as  $\sqrt{t}B_1$  for any t), SLE is itself scale invariant. That is, for any  $r \ge 0$  if  $(\gamma_t)_{t\ge 0}$  is an SLE<sub> $\kappa$ </sub> process, then the rescaled process  $(r^{-1/2}\gamma_{rt})_{t\ge 0}$  also has the law of an SLE<sub> $\kappa$ </sub>.



Figure 10:  $SLE_4$  (Simulation by Tom Kennedy).

This says that SLE is invariant under conformal maps of  $\mathbb{H}$  that fix 0 and  $\infty$  (indeed scaling maps represent all such maps.) This allows us to define SLE, by conformal invariance, in any simply connected domain and between any two marked boundary points.

**Definition 1.35** (Chordal SLE).  $SLE_{\kappa}$  is a collection  $(\mu_{D,a,b})$  of laws on continuous curves, indexed by triples (D, a, b) where D is a simply connected domain and a and b are two marked boundary points. The law  $\mu_{\mathbb{H},0,\infty}$  is that given by Definition 1.33. For any other triple (D, a, b),  $\mu_{D,a,b}$  is defined to be the image of  $\mu_{\mathbb{H},0,\infty}$  under the (unique) conformal isomorphism sending  $\mathbb{H}$  to D, 0 to a and  $\infty$  to b.

Chordal SLE defined in this way is therefore a conformally invariant process, in the following sense. Take an  $SLE_{\kappa}$  curve  $\gamma$ , in a domain D from a to b, and apply a conformal map  $\varphi: D \to D'$  that sends a to a' and b to b'. Then  $\varphi(\gamma)$  has the law of an  $SLE_{\kappa}$  in D' from a' to b'.

The other characteristic property of SLE is the following:

**Definition 1.36** (Domain Markov Property). Suppose that  $\gamma$  is an  $\text{SLE}_{\kappa}$  in Dfrom a to b (that is,  $\gamma$  has the law  $\mu_{D,a,b}$ ) and that  $\tau$  is a stopping time for  $\gamma$ . Then, conditionally, on  $\gamma[0, \tau]$ , the remainder of the curve  $\gamma[\tau, \infty)$  has law given by  $\mu_{D\setminus\gamma[0,\tau],\gamma(\tau),b}$ . This is the domain Markov property of  $(\mu_{D,a,b})$ .

In fact, it was proved by Schramm [Sch00] that these two properties completely characterise SLE. This is what makes them the only possible candidates for certain scaling limits of discrete models.

**Proposition 1.37** (Characterisation of SLE, [Sch00]). Suppose that  $(\mu_{D,a,b})$  is a family of laws on continuous curves (indexed by triples as in Definition 1.35) that is conformally invariant and satisfies the domain Markov property. Then there exists  $\kappa > 0$  such that  $(\mu_{D,a,b})$  is chordal SLE<sub> $\kappa$ </sub>.



Figure 11: Domain Markov Property of SLE.

The behaviour of  $SLE_{\kappa}$  paths, of course, depends on the value of  $\kappa$ . Indeed, the paths should arise from physical models lying in different, so-called, universality classes. It turns out that the behaviour can be classified in to three distinct phases, [RS05].

**Theorem 1.38** (Rohde–Schramm). For different values of  $\kappa$  the behaviour of  $SLE_{\kappa}$  can be described as follows:

- For κ ≤ 4, γ is almost surely a simple curve, and γ<sub>t</sub> ∈ H for all t > 0 (i.e. it does not hit the real line);
- For  $\kappa \in (4, 8)$   $\gamma$  has double points, and  $\gamma \cap \mathbb{R} \neq \emptyset$ ;
- For  $\kappa \geq 8$ ,  $\gamma$  is almost surely a space filling curve. That is,  $\gamma[0,\infty) = \overline{\mathbb{H}}$ .



Figure 12:  $SLE_2$ ,  $SLE_4$  and  $SLE_6$  from left to right (Simulations by Tom Kennedy).

Thus,  $\kappa = 4$  and  $\kappa = 4$  can be thought of as critical values for SLE. In Chapter 5 we will focus on a special connection with the Gaussian free field in the  $\kappa = 4$  case.

One final thing to note about SLE, as is perhaps already visible from the pictures, is that the curves generating them have Hausdorff dimension strictly bigger than one. That is, they are fractal. The dimension of the  $SLE_{\kappa}$  curves was computed by Beffara [Bef08] and shown to be equal to  $\min(2, 1 + \frac{\kappa}{8})$ .

Scaling Limits As we have already mentioned several times, the defining properties of SLE mean that they are widely believed to be the scaling limits of a whole class of interfaces in discrete lattice models. The continuous parameter  $\kappa$  should correspond

to what physicists call the "universality class" of the discrete model, and is related to the "central charge" c of conformal field theory via the relationship

$$c = \frac{(8-3\kappa)(\kappa-6)}{2\kappa}.$$

However, proving that these scaling limits exist is often extremely difficult, and results that we have obtained so far have been much celebrated:

- SLE<sub>2</sub> was proved by Lawler, Schramm and Werner [LSW04] to be the scaling limit of the loop erased random walk (or equivalently, by Wilson's algorithm, the branches of the uniform spanning tree);
- SLE<sub>3</sub> has been proved to be the limit of the interface between positive and negative spins in the critical Ising model [CS12, CDCH<sup>+</sup>14];
- SLE<sub>4</sub> corresponds to the "level lines" of the Gaussian free field [SS09, SS13];
- SLE<sub>16/3</sub> has been proven to arise from the exploration interface in the critical FK model [CS12, CDCH<sup>+</sup>14];
- SLE<sub>6</sub> is the limit of the exploration path in critical percolation [CN07, Smi01]; and finally
- SLE<sub>8</sub> can be shown to be the scaling limit of the path separating the uniform spanning tree from its dual, [LSW04].

But there is much left to do. For example, when  $\kappa = 8/3$ , SLE<sub> $\kappa$ </sub> satisfies a special *restriction* property ([LSW03]) which leads us to believe that it should the scaling limit of a model known as the self-avoiding random walk. This is one of many open problems in this area.

 $SLE_{\kappa}(\rho)$  processes.  $SLE_{\kappa}(\rho)$  processes are a variant of  $SLE_{\kappa}$  where the curves have an additional attraction, or repulsion, from certain "force points" in the domain or on its boundary. The vector  $\rho$  tells us how strong this attraction or repulsion is, and in which direction. This is rigorously described by the addition of an extra drift term in the differential equation defining their driving functions (they are still Loewner chains, but now the driving function is the solution of an SDE related to Brownian motion rather than Brownian motion itself.)

However, the new differential equations are somewhat more tricky to deal with. It has only been shown recently, through a remarkable connection with the Gaussian free field [MS16a, WW16], that the associated Loewner chains are actually generated by continuous curves. In [PW17], which will make up Chapter 5, we generalise the notion of  $SLE_4(\rho)$  to a continuum of force points and show that this property still holds.

**Conformal Loop Ensembles (CLE)** Recall that in our earlier discussion of critical percolation, the most natural interfaces to look at were those separating *all* the monochromatic clusters in the model. Instead of a single interface from 0 to  $\infty$ , this produces a collection of loops in the upper half plane. It is believed that these should converge, as the mesh size of the lattice goes to 0, to CLE.

 $\text{CLE}_{\kappa}$  are collections of loops in complex domains (indexed by a real parameter  $\kappa$ ), that again satisfy conformal invariance and a specific form of spatial Markov property. They were introduced by Sheffield and Werner [SW12], and can be defined either using something known as a *Brownian loop-soup*, or by using a branching variant of  $\text{SLE}_{\kappa}(\rho)$  [She09]. This is why, locally,  $\text{CLE}_{\kappa}$  loops look like  $\text{SLE}_{\kappa}$  curves (a fact that should not be surprising given the interpretation in terms of interfaces.)

Again, it is known that conformal loop ensembles arise as scaling limits in a number of cases:  $\kappa = 3$  corresponds to interfaces in the critical Ising model [CS12, CDCH<sup>+</sup>14];  $\kappa = 4$  to contour lines of the Gaussian free field [MS11, ASW15];  $\kappa = 16/3$  to cluster interfaces in critical FK percolation [CS12, CDCH<sup>+</sup>14]; and  $\kappa = 6$  to cluster interfaces for critical percolation on the triangular lattice [CN07, Smi01]. Once more, however, the theory is far from complete.

#### 1.3.7 Connection between SLE and the GFF

Many intriguing connections between Schramm–Loewner evolutions and the Gaussian free field have been uncovered in recent years. Although both are canonical, conformally invariant objects in the plane, the link between them is subtle, and not at all obvious.

The main connection we will focus on in this thesis, see Chapter 5 in particular, is the interpretation of SLE curves as "level lines" or "flow lines" of the GFF [Dub09, MS16a, SS09, SS13, WW16]. This was first studied by Schramm and Sheffield [SS09], who proved the following. Take a discrete GFF on a lattice approximation to a domain  $D \subset \mathbb{C}$  (with certain specific boundary conditions) and extend it to a continuous function on D. Then the zero level-line of this field (that is, the line along which the field is equal to 0) converges in distribution, as the mesh size of the lattice goes to 0, to an SLE<sub>4</sub>. Hence SLE<sub>4</sub> is the scaling limit of level lines of the discrete GFF.

The question follows: is it possible to make sense of  $SLE_4$  as a level line of the continuum Gaussian free field? This was again answered by Schramm and Sheffield [SS13]. The construction they came up with is this:

**Theorem 1.39** (SLE<sub>4</sub>-GFF coupling). *SLE*<sub>4</sub> and the GFF can be coupled as follows.

• Take an SLE<sub>4</sub>  $\gamma$  in the upper half plane from 0 to  $\infty$ .



Figure 13: A level line of the discrete GFF (from [SS09]). Different colours represent varying height of the field.

- On either side of the curve γ sample an independent GFF. On the left with boundary conditions -λ (recall, this means that we add on the constant function -λ to the field on the left) and on the right with boundary conditions +λ. Here, λ = π/2.<sup>4</sup>
- The law of the resulting field (we can interpret it as a field on the whole of H, since the SLE curve has Lebesgue measure 0) is just another GFF. Now it has special boundary conditions given by the function F<sub>0</sub> = −λ1<sub>(-∞,0)</sub> + λ1<sub>[0,∞)</sub> on the real line (recall, this means that it has the law of a zero-boundary GFF plus the harmonic extension of F<sub>0</sub>.)

The proof of this is by a relatively simple martingale argument, that we will sketch below.



Figure 14: Construction of the coupling.

Proof of Theorem 1.39. Let  $(\gamma_t)_{t\geq 0}$  be a SLE<sub>4</sub> curve from 0 to  $\infty$  in  $\mathbb{H}$ , parameterised by half-plane capacity. By Theorem 1.38, we know that this is a continuous curve which does not touch the boundary of  $\mathbb{H}$ . Let  $\eta_t$  be the harmonic extension in  $\mathbb{H} \setminus \gamma[0, t]$  of the function on the boundary given by

•  $-\lambda$  on the left-hand side of  $\gamma[0, t]$  and  $\mathbb{R}_{-}$ , and

 $<sup>^{4}</sup>$ However, in general its specific value depends on the choice of normalisation for the Green function, and so it is usually better to leave it vague.

•  $+\lambda$  on the right-hand side of  $\gamma[0, t]$  and  $\mathbb{R}_+$ ,

as in Figure 15. Recall that  $g_t$ , the unique conformal map  $\mathbb{H} \setminus \gamma[0,t] \to \mathbb{H}$  with



Figure 15: The boundary values of the harmonic function  $\eta_t$  in  $\mathbb{H} \setminus \gamma[0, t]$ .

hydrodynamic normalisation at  $\infty$  satisfies, for any fixed  $z \in \mathbb{H}$ ,

$$\frac{dg_t(z)}{dt} = \frac{2}{g_t(z) - W_t} \quad \forall t \text{ a.s.}$$

$$(1.1)$$

where  $W_t = 2B_t$  and  $B_t$  is a standard Brownian motion. We also set  $f_t(z) := g_t(z) - W_t$  so that  $f_t(\gamma(t)) = 0$ .

Then we can calculate explicitly that

$$\eta_t(z) = -\arg f_t(z) + \pi/2$$

for any  $z \in \mathbb{H}$ . A simple application of Itô's formula, using (1.1), then gives us that  $\eta_t(z)$  is a continuous local martingale. In fact, by boundedness, it is a true martingale and so has an almost sure limit  $\eta_{\infty}(z)$  as  $t \to \infty$ . We also define

$$G_t(z,w) = G_{\mathbb{H}}(f_t(z), f_t(w)) \quad \text{and} \quad E_t(p) = \iint_{\mathbb{H}^2} p(z)G_t(z,w)p(w)dzdw \; ; \; \; p \in C_c^{\infty}(\mathbb{H}).$$

Note that  $G_t(z, w)$  and  $E_t(p)$  are non-decreasing in t for any  $z, w \in \mathbb{H}$  and positive test function p. This means that the limits  $G_{\infty}(z, w)$  and  $E_{\infty}(p)$  also exist in this case.

Recall, we would like to prove that if we take a field  $\tilde{h}$  which is given by the function  $\eta_{\infty}$  plus an independent Gaussian free field on either side of  $\gamma([0,\infty))$ , then the law of  $\tilde{h}$  is that of a zero-boundary Gaussian free field plus the harmonic function  $\eta_0$ . To this end, it is enough to show that for any positive test function p in  $C_c^{\infty}(\mathbb{H})$ , the random variable  $(\tilde{h}, p)$  is Gaussian with mean  $(\eta_0, p)$  and variance  $E_0(p)$ .

Fix such a p. We will show that for each  $z \in \mathbb{H}$ ,  $(\eta_t, p)$  is a continuous martingale with quadratic variation  $-E_t(p)$ . This will allow us to conclude because then we have, for any  $\mu > 0$ 

$$\mathbb{E}[\exp(-\mu(\tilde{h}, p))] = \mathbb{E}[\mathbb{E}[\exp(-\mu(\tilde{h}, p))] | \gamma([0, \infty)]$$
$$= \mathbb{E}[\exp(-\mu(\eta_{\infty}, p) + \frac{\mu^2}{2}E_{\infty}(p))]$$
$$= \mathbb{E}[\exp(-\mu(\eta_0, p) + \frac{\mu^2}{2}E_0(p))]$$

by the optional stopping theorem.

To verify the claim, observe that since  $\eta_t$  is a continuous local martingale, so is  $(\eta_t, p)$  by Fubini's theorem. Thus we only need to show that  $(\eta_t, p)^2 + E_t(p)$  is a continuous local martingale. However, this can be seen by noting that it is a double integral of the function  $\eta_t(z)\eta_t(w) + G_t(z,w)$  against p(z)p(w) on  $\mathbb{H}$ , and again applying Fubini. The latter process is easily verified to be a martingale for each  $z, w \in \mathbb{H}$ , using Itô's formula and the explicit expression for  $G_{\mathbb{H}}$ .

So, this provides a coupling between  $SLE_4$  and the continuum Gaussian free field in which the curve can in some sense be interpreted as a level line of the field. In fact, we can see that it is actually more like a "level cliff"; there is a constant height gap of  $2\lambda$  between either side of the curve. This, in essence, is a consequence of the roughness of the field.

One important and highly non-trivial property of this coupling, is that the Gaussian free field we get in the end actually determines the  $SLE_4$  curve uniquely. Of course we would expect this from a continuous field and its level line, but in the case of the GFF it is not so clear. Despite the deep nature of this result, the proof in [SS13] again follows a simple and elegant argument.

So now, what about different boundary conditions? The choice of  $F_0$  is clearly very specific. In order to make sense of this question we need a more general definition (cf. Definition 5.1.)

**Definition 1.40** ([MS16a, WW16]). Suppose that F is  $L^1$  with respect to harmonic measure on  $\mathbb{R}$  viewed from some point in  $\mathbb{H}$  and that h is a zero boundary GFF in  $\mathbb{H}$ . If  $(K_t, t \ge 0)$  is a Loewner chain and  $(g_t, t \ge 0)$  is the corresponding sequence of conformal maps, set  $f_t = g_t - W_t$ , and let  $V_t^R(x)$  (resp.  $V_t^L(x)$ ) be the image of  $x \ge 0$  (resp.  $x \le 0$ ) under  $g_t$ . Let  $\eta_t^0$  be the bounded harmonic function on  $\mathbb{H}$  with
boundary values (see Figure 16)

$$\begin{cases} F(f_t^{-1}(x)), & \text{if } x \ge V_t^R(0^+) - W_t, \\ \lambda, & \text{if } 0 \le x < V_t^R(0^+) - W_t, \\ -\lambda, & \text{if } V_t^L(0^-) - W_t \le x < 0, \\ F(f_t^{-1}(x)), & \text{if } x < V_t^L(0^-) - W_t, \end{cases}$$

and define, for  $z \in \mathbb{H} \setminus K_t$ ,

$$\eta_t(z) = \eta_t^0(f_t(z)).$$

We say that K is a level line of h + F if there exists a coupling (h, K) such that the following domain Markov property holds: for any finite K-stopping time  $\tau$ , given  $K_{\tau}$ , the conditional law of  $(h + F)|_{\mathbb{H} \setminus K_{\tau}}$  is equal to the law of  $h \circ f_{\tau} + \eta_{\tau}$ .

More plainly, suppose we have a curve  $\gamma$  that is coupled with a Gaussian free field with boundary data F. Then we say that  $\gamma$  is a level line of the field if, for any stopping time  $\tau$ , the curve  $\gamma[0, \tau]$  is a local set for the field, and the associated harmonic function has boundary values as shown in the lefthand side of Figure 16.



Figure 16: The lefthand side shows the boundary values of the harmonic function  $\eta_{\tau}$  in  $\mathbb{H} \setminus K_{\tau}$ . This is the image under  $f_{\tau}^{-1}$  of the harmonic function  $\eta_{\tau}^{0}$  in  $\mathbb{H}$ , whose boundary values are shown on the right hand side.

This definition is satisfied by the coupling of Schramm and Sheffield described above (by an easy adaptation of the proof given here). Moreover, the argument can be extended to show that level lines do in fact exist for a very wide class of boundary data F. Wu and Wang [WW16] proved the existence for piecewise constant functions F, and later Wu and Powell ([PW17], see Chapter 5) extended this to any function that is *regulated* (has left and right limits at every point). These couplings again satisfy the desirable property that the curve is determined by the field, and the curves themselves are given by  $SLE_4(\rho)$  processes. See Chapter 5 for the specific relationship between  $\rho$  and the boundary data F.

Another point of interest is that this relationship provides a way to couple together many different  $SLE_4(\rho)$  processes. That is, as level lines of a single Gaussian free field, to which lots of different boundary data is added. This results in a nice interaction between the curves; they are in fact monotonic in the corresponding boundary data [WW16, PW17]. Note that this makes sense in terms of their interpretation as level lines.

A final consequence we will mention, is that these couplings allow previously less-tractable properties of the SLE curves to be attacked from the point of view of the GFF. For example, it is not immediately obvious from their definitions in terms of Loewner chains that  $SLE_4(\rho)$  processes are actually generated by curves at all (although it was proved for  $SLE_4$  in [RS05]). In general, this result comes as a consequence of their coupling with the GFF [MS16a, WW16, PW17].

Flow line coupling. When  $\kappa \neq 4$  the coupling between  $\text{SLE}_{\kappa}$  and the GFF becomes even more exotic. This is the subject of the Imaginary Geometry series by Jason Miller and Scott Sheffield [MS16a, MS16b, MS16c, MS16d]. It turns out that  $\text{SLE}_{\kappa}$ curves with  $\kappa \neq 4$  can be interpreted as so-called "flow lines" of the Gaussian free field. Heuristically, let *h* be a GFF and consider the "vector field" described by  $e^{ih/\chi}$ for  $\chi > 0$  (although of course this is not really well defined). Then one can try to trace the flow lines of this field, that is, curves  $\eta$  that are solutions of the differential equation

$$\eta'(t) = e^{ih(\eta(t))/\chi}$$

The heuristic is that what one should get is an  $SLE_{\kappa}$   $(\chi = 2/\sqrt{\kappa} - \sqrt{\kappa}/2)$ .

Rigorously, one can define flow lines as curves coupled with the GFF in a similar way to Definition 1.40, but with an additional "winding" term in the definition of the harmonic function. Again, different boundary conditions of the field correspond to  $\text{SLE}_{\kappa}(\rho)$  for different values of  $\rho$ .

**Coupling with CLE**. We should also mention here the coupling between the Gaussian free field and  $CLE_4$ . This somehow represents a fuller description of the GFF's level lines; indeed,  $CLE_4$  arises when one considers the scaling limit of the entire collection of contour lines of a discrete, zero-boundary condition GFF. Of course this results in a collection of loops rather than a single curve. In the continuum, the coupling can be described as follows:

- Sample a  $CLE_4$  in a domain D.
- Define a function  $h_1$ , by taking its value inside each  $\text{CLE}_4$  loop to be given by an independent random variable, taking values  $\pm 2\lambda$  with equal probability.
- Sample independent  $CLE_4$ 's inside all of these loops and repeat the procedure above in the second generation loops to obtain a function  $h'_2$ .
- Let  $h_2 = h_1 + h'_2$  and iterate.

• It is known [MS11] that the limit of  $h_n$  as  $n \to \infty$  is a zero-boundary Gaussian free field.

This coupling was first discovered by Miller and Sheffield [MS11], but see also [ASW15] for a proof.

Note that in all these couplings, the loops or curves define non-trivial examples of local sets for the corresponding GFF. In [ASW15] the authors construct even more general local set couplings, in which the harmonic functions can take 2 arbitrary constant values (so the coupling above with first generation  $\text{CLE}_4$  loops corresponds to  $\pm 2\lambda$ .) These can be thought of as being analogous to uniformly integrable stopping times for Brownian motion.

These local sets, and the coupling between the GFF and  $CLE_4$  will also play an important role in Chapter 4. Here we will construct the Liouville measure as a *multiplicative cascade measure* using  $CLE_4$ , in the natural way suggested by the above construction of the GFF.

Quantum Zipper. To conclude this section, we briefly mention another coupling between SLE and the GFF, due to Scott Sheffield [She16]. Roughly speaking, one takes two special Liouville quantum gravity surfaces (known as quantum wedges) and conformally welds them to obtain a new surface equipped with a curve. It turns out that what results is simply another Liouville quantum gravity surface, equipped with an  $SLE_{\kappa}$  ( $\kappa = \gamma^2$ .) This coupling can be thought of as the coupling emerging from the real counterpart of the martingale that produces the imaginary geometry coupling described above. For more details, see [She16] or [Ber15b].

#### **1.3.8** Branching Processes and the Brownian CRT

This brings us to our final object of consideration. To begin, suppose we have a finite tree, that is, a finite connected graph with no cycles. Then we can view this as a metric space in a natural way, by giving each branch of the tree length one, and letting the distance between any two points on the tree be given by the length of the unique path that joins them without backtracking. Generalising this *unique path* property gives us the notion of a continuum, or real, tree.

**Definition 1.41** (Real Tree). A metric space  $(\mathcal{T}, d)$  is said to be a real tree if, for all  $v, w \in \mathcal{T}$  the following two conditions hold:

- There exists a unique isometric map  $\phi_{v,w} : [0, d(v, w)] \to \mathcal{T}$  with  $\phi_{v,w}(0) = v$ and  $\phi_{v,w}(d(v, w)) = w$ .
- Any continuous injective map [0,1] → T that joins v and w has the same image as φ<sub>v,w</sub>

One way to define is a real tree is the following: take a continuous function  $f: [0, \infty) \to [0, \infty)$  with f(0) = 0 and define a "distance" function on  $[0, \infty) \times [0, \infty)$  by

$$d_f(s,t) = f(s) + f(t) - 2\min_{r \in [s,t]} f(r)$$

whenever  $s \leq t$ . It is easy to verify that this defines a pseudometric on  $[0, \infty)$ . Thus, quotienting by the equivalence relation ~ that identifies points with  $d_f(\cdot, \cdot) = 0$  we obtain a metric space  $(\mathcal{T}_f, d_f) := ([0, \infty)/ \sim, d_f)$ . One can prove, see for example [LG06], that this metric space is a real tree.

To connect this with our initial intuition, consider a finite tree, and turn it into a continuum object by joining the vertices with branches of length one. The metric described at the beginning of this section then produces a real tree. Actually, it is a real tree of the form  $(\mathcal{T}_f, d_f)$ , where the function f is given by the *contour function*  $C_t$  of the tree (see Figure 17) below.



Figure 17:  $C_t$  (on the right) is the contour function of the tree on the left-hand side. This is the function that we get if we traverse the tree at speed one in a *depth-first order* with backtracking (see figure), and measure how our height is changing with time.

The Brownian continuum random tree of David Aldous [Ald91], is an example of a particularly natural probability law on real trees.

**Definition 1.42** (Brownian CRT). The Brownian CRT is the metric space

$$(\mathcal{T}_{\mathbf{e}}, d_{\mathbf{e}})$$

where **e** is a normalised Brownian excursion.

By a normalised Brownian excursion we mean, intuitively, a Brownian path that is conditioned to begin at 0, stay positive on the interval [0, 1] and end at 0 at time 1 (although this takes a bit of work to define rigorously). We could also consider a Brownian excursion that is conditioned to reach a fixed positive height, rather than



Figure 18: A Brownian CRT (Simulation by Igor Kortchemski).

to last a fixed length of time. For more rigorous definitions of these objects see, for example, [IM65].

This real tree turns out to be universal in that it arises from many different constructions. We will be mainly concerned with the view of it as a limit of conditioned critical Galton Watson trees (to be discussed shortly). However, it can also be defined as a certain law on *spanning subtrees*, or using a so-called *line breaking construction*, see [Ald91].

Let us now turn to Galton Watson trees. Let L be a random variable, with mean m and variance  $\sigma^2$ ; this will be our *offspring distribution*. The GW tree associated with L is the tree generated by the random population growth model defined as follows:

- Initially, at time 0, there is one particle.
- At time 1 this particle dies, and gives birth to a random number of offspring, with distribution L.
- Each offspring particle then stochastically repeats the behaviour of its parent, completely independently of everything else.

It is a classical result of Kolmogorov [Kol38] that there is a phase transition in this model when the mean m of the offspring distribution is equal to 1. For  $m \leq 1$  the population becomes extinct almost surely, and for m > 1 it survives with positive probability. We will focus here on the critical case m = 1.

We consider the natural tree associated with this process. It turns out [Ald93, LGD02], that if this tree is conditioned to reach a large height n (which is the same as conditioning the process to survive to generation n) then it will converge after rescaling to the Brownian CRT (in this case we mean the tree generated by a Brownian excursion conditioned to reach a fixed height). Similar statements also exist in terms of conditioning the process to have a large total progeny.

The Brownian CRT is in fact the scaling limit of a much wider class of models. For example, one generalisation of the Galton–Watson process is to take a multitype version, where the particles also have types that influence their offspring distribution. In this case, there is again a notion of criticality, and it was proved by Miermont [Mie08] that the associated critical conditioned trees converge to the CRT. In Chapter 2 we will extend this to a result concerning genealogical trees of critical branching diffusions. Take branching Brownian motion say, (see Definition 1.26), in a bounded domain. There is again a phase transition for extinction here, and criticality corresponds to a certain value of the branching rate. In [Pow17b] we prove that the genealogical tree of such a critical process, when it is conditioned to be large, also converges after rescaling to the Brownian CRT. See Chapter 2 for a proof of this fact.

We will conclude this preliminary section in what seems like a suitable manner: by mentioning one last connection. This time, between the CRT and Liouville quantum gravity. Recall the discrete approach to LQG, where one considers the scaling limits of certain random planar maps ([LG13, Mie13]). One of the key steps in the proof that this scaling limit exists is to encode the maps using a correspondence known as the Cori–Vauquelin–Schaeffer bijection. This is a correspondence between random planar maps and *labelled trees*. It turns out, in a similar vein to the above discussion, that the labelled trees corresponding to a suitable sequence of random planar maps actually converge to a Brownian CRT, together with a labelling function. The interpretation of this limit in terms of the original random planar maps is what is used to prove the existence of the limiting Brownian map.

In the continuum, the connection between LQG and the Brownian CRT was established in the paper [DMS12] of Duplantier, Miller and Sheffield. Here they find a way to "mate" a pair of correlated CRTs, that in the end produces a continuum LQG surface.

## 1.4 Questions

There were of course many more interesting questions that arose during this PhD, that we have so far not been able to address. Here we present a selection, that we hope will be answered at some point in the future:

• We know from Chapter 2 that the genealogical structure of critical branching diffusions in *nice, bounded domains* converge to the Brownian CRT. However, if one removes the assumption of boundedness, or regularity, then more exotic things may occur. For example, in the half line with absorption at the

origin, we know by [BBS14] that at least the survival probability decays completely differently (although this is also for a slight variant of the model, where particles are given a drift towards 0.).

So, there may be *some* rough or unbounded domains where the critical genealogy *is* given by a Brownian CRT, but we expect that there should also be some where it is not (although explicit examples are not known to my knowledge). It is therefore an interesting and seemingly open problem to try and classify *all* possible scaling limits, depending on the geometry of the domain.

- Another natural extension would be to consider what happens if you allow the offspring distribution to be more general. For example, if you remove the assumption of finite variance, or allow it to depend on the spatial motion, will the behaviour remain the same?
- Some of the motivation for Chapter 4 comes from the following question: how does the Liouville measure,  $\mu^{\gamma}$ , of a fixed Gaussian free field vary with  $\gamma$ ? Current work with Juhan Aru, Avelio Sepúlveda and Xin Sun will prove that it varies continuously in the space of measures, and converges to 0 as  $\gamma \rightarrow 2$ . The approach will make use of the new construction of  $\mu^{\gamma}$  from Chapter 4. We also hope to use this to prove that the renormalised limit  $\mu^{\gamma}/(2-\gamma)$  converges to the critical derivative measure as  $\gamma \uparrow 2$ .
- On that note, can we say what happens to the Liouville measure (for some fixed γ) if we put a dynamic on the underlying GFF? For example, take the Ornstein–Uhlenbeck dynamic (defined by putting an independent OU process on each of the coefficients α<sub>i</sub> from Lemma 1.17). In the subcritical case, the dynamical Liouville measure should vary continuously. But what happens at criticality?
- This leads to an analogous question regarding the level, or flow, lines of the Gaussian free field. We know from [SS13, MS16a, WW16, PW17] that the GFF determines it level lines. Furthermore, we will see in Chapter 5 that the level lines are in some sense continuous in the boundary data. But are they continuous in general (i.e. with respect to some topology on the space of fields)? What happens to the level lines or flow lines when we put an Ornstein–Uhlenbeck dynamic on the field? We believe that, at least for level lines, there should be *exceptional times* for the process. For example, there should be times when the SLE<sub>4</sub> curves hit the boundary of the domain. This, however, is proving to be a challenging problem, as the relationship between the GFF and its level lines remains rather mysterious.
- We can also ask what happens if we change the parameter  $\kappa$  in the coupling of

 $SLE_{\kappa}$  and the GFF. In this case we may also see special values of  $\kappa$  at which the associated curve *jumps*. As above, this seems to be a little tricky to make rigorous.

Finally, the theory of level lines of the GFF with general boundary data, Chapter 5, is a little way from being complete. We prove the existence of level lines in the case when a certain inequality is satisfied, that corresponds to the non-existence of a *continuation threshold* for the associated SLE<sub>4</sub>(ρ) process. However, we should be able to construct level lines without this assumption, at least up to the continuation threshold (i.e. until the level line hits a "bad" part of the boundary). Construction up to this time is done in [MS16a, WW16] for the case of piecewise constant boundary data, and also beyond this time in [MSW16] for the case of one force point. Extending our results to these types of situation is a work in progress with Hao Wu.

# 2 Invariance principles for branching diffusions in bounded domains

## 2.1 Introduction

This article concerns branching diffusions in a bounded domain D of  $\mathbb{R}^d$ . These are processes in which individual particles move according to the law of some diffusion, are killed upon exiting the domain, and branch into a random number of particles (with distribution A, independent of position) at rate  $\beta > 0$ . Whenever such a branching event occurs, each of the offspring then independently repeats the behaviour of its parent, starting from the point of fission. Throughout, the configuration of particles will be denoted by

$$(X_t^1, \cdots, X_t^{N_t})$$

where  $N_t$  is the number of particles alive at time t, and we will write  $\mathbb{P}_x$  for the law of the process initiated from a point  $x \in D$ . We will always assume that the offspring distribution has mean m > 1 and finite variance, and that the generator  $L = -\frac{1}{2} \sum_{i,j} a^{ij} \partial x_i \partial x_j + \sum_i b^i \partial x_i$  of the diffusion is uniformly elliptic and self-adjoint with smooth coefficients.

It is known [Sev58, Wat65] that such a system exhibits a phase transition in the branching rate: for large enough  $\beta$  there is a positive probability of survival, but for small  $\beta$ , including at criticality, there is almost sure extinction. The critical value of  $\beta$  is equal to  $\frac{\lambda}{m-1}$ , where  $\lambda$  is the first eigenvalue of L on D with Dirichlet boundary conditions. The main goal of this paper will be to study the system at criticality and find a scaling limit for the resulting genealogical tree. This is the continuous planar tree that is generated purely by the birth and death times of particles in the system, and encodes no information about the spatial movement. More precisely, for given y > 0, we condition the diffusion to survive until time ny > 0 and look at the associated genealogical tree  $\mathcal{T}_n^y$ , equipped with its natural distance  $d_n^y$ . Rescaling the distances by a factor n produces a sequence of random compact metric spaces  $(\mathcal{T}_n^y, \frac{1}{n}d_n^y)_{n\in\mathbb{N}}$ . We will prove that this sequence converges in distribution to a conditioned Brownian Continuum Random Tree as  $n \to \infty$ , with respect to the Gromov-Hausdorff topology. Indeed, if we write  $(\mathcal{T}_{e^y}, d_{e^y})$  for the real tree whose contour function is given by  $e^y$ , a Brownian excursion conditioned to reach level y, then we obtain the following result.

**Theorem 2.1.** Suppose that  $D \subset \mathbb{R}^d$  is a bounded  $C^1$  domain and that L is uniformly elliptic and self-adjoint with smooth coefficients. Further suppose that A has

mean m > 1 and finite variance, and  $\varphi \in C^1(\overline{D})$  where  $\varphi$  is the first eigenfunction of L on D. Then for any y > 0, and any starting point  $x \in D$ ,

$$(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}) \xrightarrow[n \to \infty]{} (\mathcal{T}_{e^y}, d_{e^y})$$

in distribution, with respect to the Gromov-Hausdorff distance, where

$$\alpha = \sqrt{\frac{4(m-1)}{\lambda \langle 1, \varphi \rangle \mathbb{E}[A^2 - A] \int_D \varphi(y)^3 \, dy}}$$

**Remark 2.2.** One sufficient condition to ensure that the hypotheses of Theorem 2.1 are satisfied is to assume that the boundary of D is  $C^{2+\lfloor d/2 \rfloor}$ . Standard regularity theory of elliptic partial differential equations, see for example [Eva98, §6.3], then implies that  $\varphi \in C^1(\overline{D})$ . However, this is also satisfied in many other cases.

On the way to proving Theorem 2.1 we obtain several other results on critical branching diffusions, which are interesting in their own right as well as being essential to our method. We start with the phase transition. This was first proved by Sevast'yanov [Sev58] and Watanabe [Wat65], but has also been reworked and generalised in recent years, for example in [EK04], which studies local versus global extinction in unbounded domains. The precise description is as follows:

**Theorem 2.3** ([Sev58], [Wat65]). Let  $D \subset \mathbb{R}^d$  be a bounded domain, satisfying a minimal regularity assumption (see Condition 2.9). Suppose that L is a uniformly elliptic self-adjoint operator with smooth coefficients and that A is a distribution with finite mean m > 1. Then there are two possibilities for the long term behaviour of the branching diffusion determined by L and A in D, according to the value of the branching rate  $\beta$ . Namely, for any starting position  $x \in D$ , if  $\lambda$  is the principal eigenvalue of L on D with Dirichlet boundary conditions then,

- (1) for  $\beta > \frac{\lambda}{m-1}$  the process survives for all time with positive probability.
- (2) for  $\beta \leq \frac{\lambda}{m-1}$  the process becomes extinct almost surely.

Moreover, if  $\beta \leq \frac{\lambda}{m-1}$  then  $\mathbb{P}_x(N_t > 0) \to 0$  uniformly in D.

In the statement above we have taken some care to specify the regularity required on the domain, which is not detailed in the earlier works. Essentially we require that the eigenfunctions of the Laplacian converge to 0 pointwise on the boundary, and that Brownian motion started from a point on the boundary leaves the domain immediately with probability one. We will provide alternative proof of this Theorem, which in contrast to the earlier more analytic proofs in [Sev58], [Wat65], uses arguments centred around martingales (also appearing in [EK04]) arising naturally from the definition of the process. This proof will show that the stated regularity assumptions are sufficient.

The rest of the paper will focus on the behaviour of the system at criticality, starting with an asymptotic for the survival probability.

**Theorem 2.4.** Suppose that the domain D is  $C^1$ , that L is as in Theorem 2.3, and that A has finite variance. Then, in the critical case  $\beta = \frac{\lambda}{m-1}$ , for all  $x \in D$  we have

$$\mathbb{P}_x\left(N_t > 0\right) \sim \frac{1}{t} \times \frac{2(m-1)\varphi(x)}{\lambda\left(\mathbb{E}[A^2] - \mathbb{E}[A]\right)\int_D \varphi(y)^3 \, dy}$$
(2.1)

as  $t \to \infty$ . Here  $\varphi$  is the first eigenfunction of L on D, normalised to have unit  $L^2$  norm.

This asymptotic then allows us to study the behaviour of the system when it is conditioned to survive for a long time, which is important for the proof of Theorem 2.1. One tool that we will use is a classical *spine* change of measure, under which the process has a distinguished particle, the spine, which is conditioned to remain in D forever (as in [Pin85]). Along this spine, families of ordinary critical branching diffusions immigrate at rate  $\frac{m}{m-1}\lambda$  according to a biased offspring distribution. Note that there is no extinction under this new measure, which we denote by  $\mathbb{Q}_x$ . We will prove that changing measure in this way is in fact somewhat close to conditioning on survival for all time, in the sense of the following Proposition.

**Proposition 2.5.** Assume the hypotheses of Theorem 2.4. Then for any  $T \ge 0$ ,  $x \in D$  and  $B \in \mathcal{F}_T$ , where  $\mathcal{F}$  is the natural filtration of the process, we have

$$\lim_{t \to \infty} \mathbb{P}_x \left( B | N_t > 0 \right) = \mathbb{Q}_x(B).$$
(2.2)

Furthermore, we are able to prove a Yaglom type limit theorem for the positions of the particles in the system at time t, given survival.

**Theorem 2.6.** For any measurable function f on D such that  $\int_D f(x)^2 \varphi(x) dx < \infty$ , we have

$$\left( t^{-1} \sum_{i=1}^{N_t} f(X_t^i) \middle| N_t > 0 \right) \to Z$$

in distribution as  $t \to \infty$ , where Z is an exponential random variable with mean

$$\frac{\lambda\left(\mathbb{E}[A^2] - \mathbb{E}[A]\right)\langle\varphi, f\rangle_{L^2(D)}\int_D\varphi^3}{2(m-1)}.$$

One consequence of Theorem 2.6 (or rather its proof) is that it allows us to describe the limiting distribution of the particles in the system at time t, given sur-

vival. It turns out that this is the law with density  $\varphi$ , normalised to be a probability distribution.

Corollary 2.7. Let

$$\mu_t := \frac{1}{N_t} \sum_{i=1}^{N_t} \delta_{X_t^i}$$

be the uniform distribution on all particles alive at time t, given survival. Then, for each f as in Theorem 2.6, we have that

$$\mu_t(f) \to \mu(f)$$

in distribution, and hence in probability, as  $t \to \infty$ , where

$$\mu(f) = \frac{\int_D \varphi(x) f(x) \, dx}{\int_D \varphi(x) \, dx}.$$

As this paper was being completed, the author learnt that similar results to Theorem 2.4 and Theorem 2.6 have also been shown by Asmussen and Hering in [AH83]. However their proofs are completely different from those in the current paper and, more importantly, our method provides several new and crucial ingredients for the proof of Theorem 2.1.

#### 2.1.1 Related Work

It is interesting to note the analogy between Theorems 2.3-2.6, and the classical results from the theory of Galton-Watson processes. Indeed, for critical Galton-Watson processes, Kolmogorov [Kol38] proved an asymptotic for the probability of survival up to time n;

$$\mathbb{P}(Z_n > 0) \sim \frac{c}{n}$$

where  $Z_n$  is the population size at time n, and the constant depends on the variance of the offspring distribution. Moreover, Aldous [Ald91],[Ald93] and Duquesne and Le Gall [LGD02] showed that if you condition a critical Galton-Watson process to reach a large generation or have a large total progeny, then you have a scaling limit for the resulting tree. This limit is in the Gromov-Hausdorff topology, after rescaling distances in the tree appropriately, and the limiting object is the Continuum Random Tree, [Ald91]. In fact, this result can be extended to multitype Galton-Watson processes, as in [Mie08], where the same scaling limit exists. Since a branching diffusion can be thought of as a limit of multitype Galton-Watson processes, considering the types to be positions and discretising the domain appropriately, this is the first indication we should be able to obtain a similar scaling limit. **Remark 2.8.** The constant  $\alpha$  in Theorem 2.1 is exactly what one obtains formally from the convergence in [Mie08], considering the branching diffusion to be a scaling limit of appropriate multitype Galton-Watson processes. However, Miermont's proof strategy is to make an induction on the number of types, and the lack of uniformity in the estimates as the number of types grows means that it does not extend to the set up considered here. Instead we use a combination of probabilistic and analytic ideas; see Section 2.1.2 for a sketch of the argument.

An asymptotic for the survival probability has also been considered previously, see [BBS14] and [Kes78], in the case of branching Brownian motion with absorption at the origin, where the branching rate is kept constant and each particle moves with a drift  $-\mu$ , which is varied. In this set up, there is a critical value of  $\mu = \mu_c$ above which extinction occurs with probability one. The *near-critical* system, as  $\mu$ approaches its critical value from below, has also been studied, and in [BBS11] a limit, as  $\mu \uparrow \mu_c$ , is found for the probability of survival for all time as a function of the initial position. However, these results are quite different from ours as we do not allow our domain to be unbounded. The proofs of Theorems 2.1 and 2.4 do not extend to this situation, and in fact, we would expect to see a variety of behaviours for the critical system if we remove the assumption of boundedness. It would be an interesting problem to explore the possible cases here, and classify which domains fall into the regime of Theorem 2.1 and Theorem 2.4.

## 2.1.2 Organisation of the Paper and Main Ideas

We begin, for completeness and in order to introduce key concepts for the latter part of the paper, by providing a full proof of Theorem 2.3 in the case of Brownian motion with binary branching. This also allows us to make precise the regularity that is required on the domain for this statement to be true, see Condition 2.9. The main idea behind the proof we will give is to exploit the existence of a certain martingale

$$M_t = e^{(\lambda - \beta)t} \sum_{i=1}^{N_t} \varphi(X_t^i),$$

where  $\varphi$  is the first eigenfunction of  $-\frac{1}{2}\Delta$  on D, with unit  $L^2$  norm, and  $\lambda$  is the first eigenvalue. We show that its properties (which depend on  $\beta$ ) change critically at the point  $\beta = \lambda$ . These critical features turn out to determine the long term behaviour of the entire process, and thus provide the result of the theorem.

We will then turn to the proofs of the remaining theorems. Again we will give these for binary branching Brownian motion, and wherever adaptation for general diffusions and branching mechanisms is required, we will indicate the necessary changes. Any extra arguments are in fact minor, which is why we prefer to highlight the simplest case. This allows us to keep the arguments clear and avoid introducing extra notation.

The proof of Theorem 2.4 proceeds by a combination of probabilistic arguments, and analysis of the Fisher-Kolmogorov-Petrovskii-Piscounov equation. This is a partial differential equation which is known to be satisfied by the survival probability for branching Brownian motion; first noted by McKean [McK75] in the one-dimensional case, and used as the main tool by Sevast'yanov in the original proof of Theorem 2.3. Naively, we can write the survival probability in an  $L^2$  expansion with respect to the orthonormal basis of  $L^2(D)$  given by eigenfunctions of the Laplacian. Since the survival probability satisfies the FKPP equation we get a family of coupled ODEs from the coefficients. However, this is tricky to analyse directly. Instead, we apply a probabilistic line of reasoning, changing measure by  $M_t/M_0$  to get a spine characterisation of the system as discussed in the introduction. This allows us to deduce that the survival probability decays like  $a(t)\varphi(x)$  as  $t \to \infty$ , where a(t) is the first coefficient in our expansion. Thus, our problem is reduced to the study of a single ODE. From here elementary analysis, combined with some extra information obtained from the probabilistic arguments, yields the result. We then prove Theorem 2.6 and Corollary 2.7, using the method of moments and a Many-to-Few Lemma.

The remainder of the paper is devoted to the proof of Theorem 2.1. Again working primarily in the case of Brownian motion with binary branching, we take an i.i.d. sequence of critical processes and concatenate the height functions of their associated trees. We would like to find a process which approximates this, and will converge after rescaling to a reflected Brownian motion: an analogue of the Lukasiewicz path for Galton-Watson trees. Just as the martingale  $M_t$  roughly measures the size of our system as we increase time, exploring it in a different, depth-first, order provides another martingale that is a proxy for the height function. After strengthening our result, Corollary 2.7, for the conditioned system, we can prove that the quadratic variation of this martingale is essentially linear, and so we obtain an invariance principle.

We then have to prove that this martingale is indeed a good approximation to the height process. This is one of the main difficulties, as the reversibility tools that are key to proving this for the Lukasiewicz path in the Galton-Watson case are lost. Instead, we must use precise estimates, and a delicate ergodicity argument related to our spine change of measure. This is one of the reasons that our machinery from the proof of Theorem 2.4 is so essential. Tightness arguments then allow us to conclude.

Acknowledgements I would like to thank Nathanaël Berestycki, for suggesting this problem and for many helpful discussions and suggestions.

# 2.2 Preliminaries

#### 2.2.1 Diffusions as Trees

As stated in the introduction,  $\mathbb{P}_x$  will denote the law of our branching diffusion, initiated from the point  $x \in D$ . We will consider this as a law on continuous planar trees, where every vertex is also marked with a position in D. In this representation, the vertices at a given height t in the tree will correspond to the particles alive in the system at time t, and their marks will correspond to their positions. To complete the definition of the tree we also need to decide, at every branch point, how the branching subtrees are ordered from left to right. To do this, we assume that given the number of offspring at a branching event, this ordering is chosen uniformly at random. This gives us a law on planar, or equivalently labelled, rooted trees. We emphasise here that these trees, unlike Galton-Watson trees, are in *continuous time*. Note that, in Theorem 2.1, we are considering them without their marks (in fact, the marks are irrelevant to the tree considered as a metric space.) The final point to make, is that when we use our notation  $(X_t^1, \dots, X_t^{N_t})$  for the system at time t, the indices correspond to the ordering of the vertices from left to right in the tree, and the  $X_t^i$  are their positions, or marks.



Figure 19: An example of the continuous tree generated by a branching diffusion. The two vertices at height t (marked with dots) have positions, or marks, given by  $X_t^1$  and  $X_t^2$ .

## 2.2.2 Spectral Theory and Martingales

From now on we will work in the case of Brownian motion with binary branching, unless stated otherwise.

As discussed in the introduction, the behaviour of branching Brownian motion killed when leaving a given domain D will be closely related to the spectral properties of the Laplacian on that domain. Throughout, we will let  $\{\lambda_i\}_{i\geq 1}$  denote the eigenvalues of  $-\frac{1}{2}\Delta$  on D with Dirichlet boundary conditions, with corresponding eigenfunctions  $\{\varphi_i\}_{i\geq 1}$ , normalised to have unit  $L^2$  norm. Recall that under this normalisation the eigenfunctions form an orthonormal basis for  $L^2(D)$ , the eigenvalues are real with

$$\lambda := \lambda_1 < \lambda_2 \le \lambda_3 \le \dots \to \infty$$

and the first eigenfunction

 $\varphi := \varphi_1$ 

is strictly positive in the domain. Also note that the eigenfunctions are in  $C^{\infty}(D)$ , and assuming the domain is sufficiently regular, converge pointwise to 0 on  $\partial D$ . In particular, they are all bounded.

**Condition 2.9.** For the proof of Theorem 2.3 we will assume that the domain D is regular enough that:

- (1) The eigenfunctions converge to 0 pointwise on the boundary, and
- (2) Every point  $x_0$  on the boundary satisfies, for every  $\varepsilon > 0$ ,  $\lim_{x \to x_0} \mathbf{P}_x (\tau_D > \varepsilon) = 0$ ,

where  $\mathbf{P}_x$  is the law of Brownian motion started from the point x and  $\tau_D$  is the first time it hits the boundary  $\partial D$ .

These are very weak regularity conditions on the domain D. For example, see [GT83, Theorem 8.29], both conditions are satisfied by any domain satisfying a uniform exterior cone condition.

To set up some notation, let  $p^{\mathbb{R}^d}(t, x, y)$  be the transition density for Brownian motion on  $\mathbb{R}^d$  and set, for  $x, y \in D$ ,

$$p^{D}(t, x, y) = p^{\mathbb{R}^{d}}(t, x, y) - \mathbf{E}_{x} \left[ p^{\mathbb{R}^{d}}(t - \tau_{D}, B_{\tau_{D}}, y) \, \mathbb{1}_{\{\tau_{D} \leq t\}} \right].$$

Then we have

$$0 \le p^{D}(t, x, y) \le p^{\mathbb{R}^{d}}(t, x, y) \le \frac{1}{(2\pi t)^{d/2}}$$
(2.3)

and by the strong Markov property  $p^D$  is the transition density of Brownian motion *killed* when leaving the domain D. This means, in particular, that for all integrable functions f we have

$$\int_{D} f(y) p^{D}(t, x, y) \, dy = \mathbf{E}_{x} \left[ f(B_{t}) \mathbb{1}_{\{\tau_{D} > t\}} \right]$$
(2.4)

for t > 0 and  $x \in D$ . It can be shown (see for example [MNV09, Remark 2.1]) that if D satisfies Condition 2.9, then for any bounded continuous function f on D which vanishes on  $\partial D$ ,

$$\int_{D} p^{D}(t, x, y) f(y) dy = \mathbf{E}_{x} \left[ f(B_{t}) \mathbb{1}_{\{\tau_{D} > t\}} \right]$$
(2.5)

is the unique solution of the heat equation in the domain, with initial data f and Dirichlet boundary data. One consequence of this is that for any of the eigenfunctions  $\varphi_i$ , we have

$$\mathbf{E}_{x}\left[\varphi_{i}(B_{t\wedge\tau_{D}})\right] = \mathrm{e}^{-\lambda_{i}t}\,\varphi_{i}(x) \tag{2.6}$$

for all  $x \in D$ . This leads to the following decomposition for functions  $f \in L^2(D)$ . Here and throughout the rest of the paper,  $\langle \cdot, \cdot \rangle$  will represent the usual inner product on  $L^2(D)$ .

**Lemma 2.10.** If  $f \in L^2(D)$  then for all  $t \ge 0$  and  $x \in D$ 

$$\mathbf{E}_{x}\left[f(B_{t})\mathbb{1}_{\{\tau_{D}>t\}}\right] = \sum_{1}^{\infty} e^{-\lambda_{i}t} \varphi_{i}(x) \left\langle\varphi_{i}, f\right\rangle.$$

*Proof.* First note that  $p^{D}(t, x, y)$  is bounded for all  $x, y \in D$  by (2.3), and that  $(p^{D}(t, x, \cdot), \varphi_{i}) = e^{-\lambda_{i}t} \varphi_{i}(x)$  by (2.5) and (2.6). Therefore, since the  $\varphi_{i}$ 's form an orthonormal basis of  $L^{2}(D)$ , we have that

$$\sum_{i=1}^{n} e^{-\lambda_i t} \varphi_i(x) \varphi_i(\cdot) \to p^D(t, x, \cdot)$$

in  $L^2$ . Thus for any  $f \in L^2(D)$ , we may conclude that

$$\mathbf{E}_x\left[f(B_t)\mathbb{1}_{\{\tau_D>t\}}\right] = \left(p^D(t,x,\cdot),f\right) = \lim_{n\to\infty} \left(\sum_{i=1}^n e^{-\lambda_i t} \varphi_i(x)\varphi_i(\cdot),f\right)$$

which yields the result.

Another consequence of (2.6) is the existence of an additive martingale for the system. Indeed, a straightforward application of the branching Markov property and the Many-to-One Lemma (see for example [HH07] or [HR15]) which tells us that

$$\mathbb{E}_x\left[\sum_{i=1}^{N_t} f(X_t^i)\right] = e^{\beta t} \mathbf{E}_x\left[f(B_t)\mathbb{1}_{\{\tau_D > t\}}\right]$$

for measurable f, provides the following:

Lemma 2.11. The process

$$M_t = e^{(\lambda - \beta)t} \sum_{i=1}^{N_t} \varphi(X_t^i)$$

is a martingale under  $\mathbb{E}_x$ , for each  $x \in D$ .

Moreover, by positivity of  $\varphi$  in D,  $M_t$  is a positive martingale, and thus by the martingale convergence theorem converges almost surely to an almost surely finite limit. This is a strong indicator of the existence of a phase transition, since the properties of the limit  $M_{\infty}$ , which depend on  $\beta > 0$  and change critically at  $\beta = \lambda$ , determine the long term behaviour of the system.

**Remark 2.12.** The above theory directly extends to more general diffusions, with generators and offspring distributions as in Theorem 2.3. In particular, Condition 2.9 will provide the required degree of regularity. Here we have that

$$M_t = e^{(\lambda - \beta(m-1))t} \sum_{i=1}^{N_t} \varphi(X_t^i)$$

defines a positive martingale, using a generalisation of the Many-to-One Lemma.

## 2.3 The Phase Transition

In this section we will provide a proof of Theorem 2.3.

#### 2.3.1 The Supercritical Case

We begin by supposing that  $\beta$  is strictly greater than  $\lambda$ , in which case the proof can be summarised as follows. We already know that  $M_t(\beta) \to M_\infty$  as  $t \to \infty$ . We will show that for this range of  $\beta$ , the martingale  $M_t$  is square integrable and so the limit  $M_\infty$  cannot be degenerate. Thus, since  $e^{(\lambda-\beta)t} \to 0$  as  $t \to \infty$ , it must be the case that

$$\sum_{i=1}^{N_t} \varphi(X_t^i) \to \infty$$

with positive probability. Boundedness of  $\varphi$  then implies that  $N_t \to \infty$  on this event.

In order to prove square integrability, we need a generalisation of the Many-to-One Lemma to a Many-to-Two form. This is well know, see for example [HR15, Lemma 1] or [Wat67, Equation (2.23)]. Lemma 2.13 (Many-to-Two). If f, g are measurable functions on D then

$$\mathbb{E}_{x}\left[\sum_{i=1}^{N_{t}} f(X_{t}^{i}) \sum_{i=1}^{N_{t}} g(X_{t}^{i})\right] = \mathbf{E}_{x}\left[e^{\beta t} f(B_{t})g(B_{t})\mathbb{1}_{\{\tau_{D}>t\}}\right] \\ + \mathbf{E}_{x}\left[\int_{0}^{t\wedge\tau_{D}} 2\beta e^{\beta s} \mathbf{E}_{B_{s}}[\sum_{1}^{N_{t-s}} f(X_{t-s}^{i})]\mathbf{E}_{B_{s}}[\sum_{1}^{N_{t-s}} g(X_{t-s}^{i})]\right]$$

where B is standard Brownian motion started at  $x \in D$  and  $(X_t^1, \dots, X_t^{N_t})$  represents branching Brownian motion as usual.

Applying the Lemma with  $f = g = \varphi$ , and using (2.6) to rewrite the expectation terms in the integrand, along with Fubini, we see that

$$\mathbb{E}_x\left[M_t^2\right] = e^{2(\lambda-\beta)t} \mathbf{E}_x\left[e^{\beta t} \varphi(B_{t\wedge\tau_D})^2\right] + 2\beta \int_0^t e^{2(\lambda-\beta)s} \mathbf{E}_x\left[e^{\beta s} \varphi(B_{s\wedge\tau_D})^2\right] ds.$$

Since we also have the bound,

$$\mathbf{E}_{x}\left[\mathrm{e}^{\beta t}\,\varphi(B_{t\wedge\tau_{D}})^{2}\right] \leq \|\varphi\|_{\infty}\mathbf{E}_{x}\left[\mathrm{e}^{\beta t}\,\varphi(B_{t\wedge\tau_{D}})\right] \leq \varphi(x)\|\varphi\|_{\infty}\,\mathrm{e}^{(\beta-\lambda)t}$$

for all t and x, we can substitute this in and integrate to see that

$$\mathbb{E}_{x}\left[M_{t}^{2}\right] \leq \varphi(x) \|\varphi\|_{\infty} \left(1 + \frac{4\beta}{\lambda - \beta}\right)$$

for all t. Thus we obtain square integrability.

**Remark 2.14.** For the more general set up, with generator L and offspring distribution A as in Theorem 2.3, a generalisation of the Many-to-Two lemma, see [HR15], proves uniform integrability of the martingales. This again shows that there is a positive probability of survival in the supercritical case.

## 2.3.2 The Subcritical Case

Now let us suppose that  $\beta < \lambda$ . The convergence of the martingale in this case, along with the fact that  $e^{(\lambda - \beta)t} \to \infty$  as  $t \to \infty$ , means that in fact

$$\sum_{i=1}^{N_t} \varphi(X_t^i) \to 0$$

almost surely as  $t \to \infty$ . Although this does not show that  $N_t \to 0$  a.s. for all such  $\beta$  immediately, Lemma 2.10 ensures that  $\sum_{i=1}^{N_t} \varphi(X_t^i)$  being small must mean that

 $N_t$  is small with high probability. This is made explicit by the following Lemma, which is essentially a consequence of Lemma 2.10.

**Lemma 2.15.** Let  $\varepsilon > 0$  be given. Then there exists  $T(\varepsilon) \ge 0$  and an absolute constant K, such that for all  $t \ge T(\varepsilon)$  and all  $x \in D$  we have

$$\mathbf{P}_x\left(\{\varphi(B_t) < \varepsilon\} \cap \{t < \tau_D\}\right) \le \mathrm{e}^{-\lambda t} K\varepsilon.$$

*Proof.* Let  $f_{\varepsilon}(x) = \mathbb{1}_{[0,\varepsilon)}(\varphi(x)) = \mathbb{1}_{[0,\varepsilon)} \cdot \varphi$ . This is clearly in  $L^2$  (it is bounded by 1), so we may apply Lemma 2.10 to obtain that

$$\mathbf{P}_{x}\left(\{\varphi(B_{t})<\varepsilon\}\cap\{t<\tau_{D}\}\right)=\mathbf{E}_{x}\left[f_{\varepsilon}(B_{t})\mathbb{1}_{\{\tau_{D}>t\}}\right]=\sum_{i=1}^{\infty}\mathrm{e}^{-\lambda_{i}t}\left(\varphi_{i},f_{\varepsilon}\right)\varphi_{i}(x).$$

Intuitively, since  $\lambda = \lambda_1 < \lambda_i$  for  $i \ge 2$ , we expect that the sum should behave roughly like  $e^{-\lambda t}(\varphi, f_{\varepsilon})\varphi(x)$  as t becomes large. Indeed we can bound it above, using that  $(\varphi, f_{\varepsilon}) \le \operatorname{vol}(D)\varepsilon$ , by

$$e^{-\lambda t} \left( \|\varphi\|_{\infty} \operatorname{vol}(D)\varepsilon + e^{-\gamma t} \left| \sum_{i=2}^{\infty} e^{-(\lambda_i - \lambda_2)t} \left\langle \varphi_i, f_{\varepsilon} \right\rangle \varphi_i(x) \right| \right)$$
(2.7)

where  $\gamma > 0$  is the spectral gap for D. Therefore, it is enough to show that the expression in the modulus above is bounded by some absolute constant, for all t large enough. To do this, observe that since  $\lambda_i \to \infty$  as  $i \to \infty$ , there exists an N such that  $(\lambda_i - \lambda_2) > \frac{\lambda_i}{2}$  for all  $i \ge N$ . This means that

$$\begin{split} \left| \sum_{i=2}^{\infty} \mathrm{e}^{-(\lambda_i - \lambda_2)t} \left\langle \varphi_i, f \right\rangle \varphi_i(x) \right| &\leq \sum_{i=2}^{\infty} \mathrm{e}^{-(\lambda_i - \lambda_2)t} \left| \left\langle \varphi_i, f \right\rangle ||\varphi_i(x)| \right. \\ &\leq N \operatorname{vol}(D) \sup_{1 \leq i \leq N} \|\varphi_i\|^2 + \sum_{N+1}^{\infty} \mathrm{e}^{-\frac{\lambda_i}{2}t} \left| \left\langle \varphi_i, f \right\rangle ||\varphi_i(x)| \right. \end{split}$$

for all t, where the first term in the final line is a constant depending only on D. Furthermore, by Cauchy-Schwarz we have

$$\left|\sum_{N+1}^{\infty} e^{-\frac{\lambda_i}{2}t} \left| \left\langle \varphi_i, f \right\rangle \left| \left| \varphi_i(x) \right| \right| \le \sqrt{\sum_{i=N+1}^{\infty} e^{-\lambda_i t} \varphi_i(x)^2} \sqrt{\sum_{i=N+1}^{\infty} \left\langle \varphi_i, f \right\rangle^2} \right|$$

where the first term is less than  $\|p^D(t/2, x, \cdot)\|_{L^2} \leq (\pi t)^{-d/4}$  and the second is less than  $\|f\|_{\infty}$ . This can clearly be bounded uniformly for all t large enough.  $\Box$ 

**Corollary 2.16.** Let  $\tau_D$  be the hitting time of  $\partial D$ , for a Brownian motion started

at  $x \in D$ . Then

$$\mathbf{P}_x\left(\tau_D > t\right) \le A \,\mathrm{e}^{-\lambda t}$$

for all  $t \ge T = T(\|\varphi\|_{\infty})$ , where A is a constant independent of x and T.

*Proof.* This follows immediately from the above taking  $\varepsilon = \|\varphi\|_{\infty}$ .

**Lemma 2.17** (Expectation of the population size). Suppose  $\beta \leq \lambda$ . Then for all  $x \in D$  and all  $t \geq T = T(\|\varphi\|_{\infty})$ , we have

$$\mathbb{E}_x[N_t] \le A \,\mathrm{e}^{(\beta - \lambda)t}$$

where A is a constant, independent of t and x.

*Proof.* This is a straightforward consequence of the Many-to-One Lemma and the above Corollary.  $\hfill \Box$ 

To prove almost sure extinction in the subcritical case, it is enough to show that  $\mathbb{P}_x(N_t > 0) \to 0$  as  $t \to \infty$ . However, this is immediate from Lemma 2.17, since

$$\mathbb{P}_x \left( N_t > 0 \right) \le \mathbb{E}_x \left[ N_t \right] \le A e^{(\beta - \lambda)t}$$

which indeed tends to 0 in the case  $\beta < \lambda$ .

## 2.3.3 The Critical Case

When  $\beta$  is equal to  $\lambda$ , it is still the case that branching Brownian motion with parameter  $\beta$  dies out almost surely. However, since we can no longer rely on the fact that  $e^{(\beta-\lambda)t} \to 0$  as  $t \to \infty$ , the decay of  $\mathbb{E}_x[N_t]$  from Lemma 2.17 is lost, and we must apply a slightly more delicate argument. To improve the situation, we make use of the following Lemma, which can be found in [Wat65, Lemma 2.1]. We provide a proof for completeness.

Lemma 2.18 ([Wat65]). For all  $x \in D$ 

$$\mathbb{P}_x (N_t \to 0 \text{ or } N_t \to \infty \text{ as } t \to \infty) = 1.$$

Proof. Since  $N_t$  is integer-valued it is sufficient to prove that  $\mathbb{P}_x (N_t = k \text{ i.o.}) = 0$  for every  $k \in \mathbb{N}$ . Fix k and define a sequence of hitting and leaving times  $(L_n, H_n)_{n\geq 1}$ , by letting  $L_1$  be the first time t that  $N_t \neq N_0$ , and  $H_1$  be the first time that  $N_t = k$ . Then inductively, let  $L_n$  be the first time after  $H_{n-1}$  that  $N_t \neq k$ , and  $H_n$  the first time after this that  $N_t = k$ . We have to show that  $\mathbb{P}_x (H_n < \infty) \to 0$  as  $n \to \infty$ . Set  $\gamma$  to be the infinum over all  $x \in D$  of the probability that a Brownian motion started at x leaves the domain before an independent exponential waiting time. It is easily verified, using for example Corollary 2.16, that  $\gamma > 0$ . Then we have that  $\mathbb{P}_x(H_1 < \infty) \leq \mathbb{P}(N_{L_1} \neq 0) \leq (1 - \gamma)$  and inductively, using the Markov property for the k particles alive at each time  $H_j$ , that  $\mathbb{P}_x(H_n < \infty) \leq (1 - \gamma)(1 - \gamma^k)^{n-1}$ . This completes the proof.  $\Box$ 

This, together with Lemma 2.15, provides the basis for the proof of Theorem 2.3 in the critical case. By Lemma 2.18, it will be sufficient to prove that  $\mathbb{P}_x(N_t \to \infty \text{ as } t \to \infty) = 0$  for all  $x \in D$ . Since we know that  $\sum_{1}^{N_t} \varphi(X_t^i) \to M_\infty < \infty$  almost surely, letting  $A_k$  be the event that  $\{N_t \to \infty\} \cap \{\sum_{1}^{N_t} \varphi(X_t^i) \le k \text{ eventually}\}$ , we can write this probability as

$$\mathbb{P}_{x}\left(N_{t}\to\infty\right)=\sup_{k}\mathbb{P}_{x}\left(A_{k}\right)$$

since the  $A_k$ 's are increasing. Thus, it will be enough to show that  $\mathbb{P}_x(A_k) = 0$  for all k > 0. To do this, fix k, and observe that  $\mathbb{P}_x(A_k) = \lim_{m \to \infty} \mathbb{P}_x(A_k^m)$  where  $A_k^m$ is the event that  $\{\{N_t \ge m\} \cap \{\sum_{1}^{N_t} \varphi(X_t^i) \le k\}$  eventually}. However, for

$$\{N_t \geq m\} \cap \{\sum_1^{N_t} \varphi(X_t^i) \leq k\}$$

to occur, it must be the case that one of the particles in the system at time t has  $\varphi(X_t^i) \leq \frac{k}{m}$  (since  $\varphi$  is positive.) Hence,

$$\mathbb{P}_x\left(\{N_t \ge m\} \cap \{\sum_{1}^{N_t} \varphi(X_t^i) \le k\}\right) \le \mathbb{P}_x\left((\sum_{i=1}^{N_t} \mathbb{1}_{\{\varphi(X_t^i) \le \frac{k}{m}\}}) \ge 1\right) \le \mathbb{E}_x\left[\sum_{i=1}^{N_t} \mathbb{1}_{\{\varphi(X_t^i) \le \frac{k}{m}\}}\right]$$

which is less than  $K_{\overline{m}}^{\underline{k}}$  for  $t \ge T(k/m)$ , as a consequence of Lemma 2.15. Since the probability of this holding for *all* large times is certainly smaller than the probability of it holding at time T(k/m) say, we see that

$$\mathbb{P}_{x}(A_{k}) = \lim_{m \to \infty} \mathbb{P}_{x}(A_{k}^{m}) \le \lim_{m \to \infty} K \frac{k}{m} = 0$$

as required.

**Remark 2.19.** The proofs given above for the subcritical and critical cases rely purely on spectral properties of the Laplacian. These still hold for our more general diffusions, so no adaptation of the arguments is required.

To complete the proof of Theorem 2.3 we must show that the decay of the survival probability in the critical case (the subcritical case having been dealt with by Lemma 2.17) is uniform in D. However, this will follow from elementary analysis, once we have noted that the survival probability is a solution of the FKPP equation in D. This was first observed by McKean, [McK75], in one dimension, but the proof extends directly to our situation. In fact, the relationship with the FKPP equation is the key tool employed in [Wat65] and [Sev58] to prove Theorem 2.3.

**Lemma 2.20.** Assume that D satisfies Condition 2.9 and let  $u(t, x) := \mathbb{P}_x (N_t > 0)$ for the critical system. Then  $u \in C^{2,1}(D \times (0, \infty)) \cap C(\overline{D})$  is a solution of

$$\begin{aligned} \partial u/\partial t &= \frac{1}{2} \Delta u + \lambda (u - u^2) & on \ D \times (0, \infty) \\ u(x, 0) &= \mathbbm{1}_{\{x \in D\}} & on \ \overline{D} \times \{0\} \\ u(x, t) &= 0 & on \ \partial D \times (0, \infty). \end{aligned}$$
 (2.8)

Proof. Conditioning on the first branching time of the process, we can write

$$u(x,t) = e^{-\lambda t} \mathbf{P}_x(\tau_D > t) + \int_0^t \lambda e^{-\lambda s} \mathbf{E}_x \left[ (2u - u^2)(B_s, t - s) \mathbb{1}_{\{\tau_D > s\}} \right]$$

An easy differentiation after making the change of variables  $s \leftrightarrow t - s$  in the integral then provides (2.8). Note that  $u(t, x) \to 0$  as  $x \to \partial D$  since

 $\mathbb{P}_x \left( N_t > 0 \right) \leq 1 - \mathbb{P}_x \left( \text{the process becomes extinct before the first branching time} \right)$  $\leq 1 - e^{-\lambda s_1} \mathbb{P}_x \left( \tau_D \leq s_1 \right)$ 

for any  $s_1$ , where the last line can be made arbitrarily small by first taking  $s_1$  to 0, and then using the property (2) of Condition 2.9. Thus  $u \in C(\overline{D})$ .

To conclude the proof of 2.3, we note the continuous functions u(x,t) are clearly decreasing in t and converge to the continuous function 0 for each  $x \in \overline{D}$ . This is a compact set and so by Dini's theorem, an elementary result from real analysis (see for example [Rud76, Theorem 7.13]), the decay must indeed be uniform.

**Remark 2.21.** In the case of a more general diffusion with generator L and branching mechanism determined by offspring distribution A, the partial differential equation (2.8) becomes

$$\frac{\partial u}{\partial t} = -Lu + \frac{\lambda}{m-1} \left( (1-u) - G(1-u) \right)$$

where G is the probability generating function of A. This results in the same regularity for u.

## 2.4 Survival at Criticality: Proof of Theorem 2.4

Throughout this section, we will work in the critical case  $\beta = \lambda$  (for binary branching Brownian motion) and also from now on assume the domain D to have  $C^1$  boundary. We will prove the asymptotic (2.1) for the survival probability, using a combination of spine techniques, and analysis of the FKPP equation.

## 2.4.1 Spine Decomposition

It turns out that a helpful approach in the proof of (2.1) will be to change measure via the  $\mathbb{P}_x$ -martingale  $M_t$  from the previous section, see Lemma 2.11, and use a spine construction to describe the behaviour of  $\lambda$ -branching Brownian motion under this new measure. This is a so called *spine change of measure*, in that it changes the law of the initial particle, but then all subprocesses branching off this *spine* still behave as ordinary  $\lambda$ -branching Brownian motions.

To make sense of this, we extend our probability measure  $\mathbb{P}_x$  to a probability measure  $\overline{\mathbb{P}}_x$  on a bigger space, by choosing one distinguished line of descent which we call the *spine*. We let the initial particle be part of the spine and then, whenever a spine particle splits, the new spine particle is chosen uniformly from its children. We denote the natural filtration of this new process by  $(\overline{\mathcal{F}}_t)_{t\geq 0}$ , and the position of the spine by  $(\xi_t)_{t\geq 0}$ . Then it follows from the fact that  $e^{\lambda t} \varphi(B_t)/\varphi(x)$  is a martingale for our individual particle motion under  $\mathbf{P}_x$  (a consequence of (2.6)) that the process

$$\overline{M}_t = \frac{\varphi(\xi_t)}{\varphi(x)} 2^{S_t}$$

is a martingale with respect to  $\overline{\mathcal{F}}_t$ , where  $S_t$  is the number of branch points along the spine before time t. For a proof of this, see [Rob10, Theorem 2.4], and see also [Rob10] and [HH09] for details of the above construction. We can therefore define a new measure  $\overline{\mathbb{Q}}_x$  on the same probability space as  $\overline{\mathbb{P}}_x$  via

$$\frac{d\overline{\mathbb{Q}}_x}{d\overline{\mathbb{P}}_x}\Big|_{\overline{\mathcal{F}}_t} = \overline{M}_t.$$

Then there is a nice characterisation of the process under  $\overline{\mathbb{Q}}_x$ , which follows from the classical spine theory for such changes of measure developed in [HH09]; but see also [HH07], [CR88] and [HHK12] among others. Using the fact that under the change of measure for Brownian motion defined by

$$\left. \frac{d\mathbf{Q}_x}{d\mathbf{P}_x} \right|_{\mathcal{F}_t} = \frac{\mathrm{e}^{\lambda t} \,\varphi(B_t)}{\varphi(x)} \tag{2.9}$$

we have that the new particle evolves as a Brownian motion conditioned to remain in D, see [Doo57], [Pin85], we obtain a description of the whole process under  $\overline{\mathbb{Q}}_x$ , as summarised in the Lemma below. We refer the reader to the papers cited above for more details, and proof of the characterisation.

**Lemma 2.22.** Under the measure  $\overline{\mathbb{Q}}_x$ , the law of  $\lambda$ -branching Brownian motion with a distinguished spine can be constructed as follows:

- The initial particle evolves as a Brownian motion started at x and conditioned to remain in D for all time.
- At an accelerated rate of  $2\lambda$  it splits into two particles.
- One of these particles, the spine particle, is chosen uniformly at random and goes on to repeat stochastically the behaviour of the initial ancestor.
- The other particle goes on to perform an independent λ-branching Brownian motion, starting from the point of fission.

Alternatively, we can think of the process as being formed by a single spine particle, which evolves as a Brownian motion conditioned to remain in D, and along which ordinary  $\lambda$ -branching Brownian motions immigrate (branch off) at rate  $2\lambda$ .

We also let  $\mathbb{Q}_x := \overline{\mathbb{Q}}_x|_{\mathcal{F}_t}$  be the corresponding measure on the original filtration  $\mathcal{F}_t$  of the branching process. Then we have that

$$\left. \frac{d\mathbb{Q}_x}{d\mathbb{P}_x} \right|_{\mathcal{F}_t} = \frac{\sum_{i=1}^{N_t} \varphi(X_t^i)}{\varphi(x)} = \frac{M_t}{\varphi(x)}$$
(2.10)

(again see [Rob10, Theorem 2.4]). Furthermore, the probability under  $\overline{\mathbb{Q}}_x$  that a certain particle  $X_t^j$  at time t is the spine particle, given  $\mathcal{F}_t$ , is equal to

$$\varphi(X_t^j) / \sum_{i=1}^{N_t} \varphi(X_t^i), \qquad (2.11)$$

see [Rob10] or [HR14, Remark 1.2].

**Remark 2.23.** For more general diffusions and branching mechanisms, we obtain a similar characterisation of the system after changing measure by the corresponding martingale. In this case, the spine particle will move under the law of the original diffusion conditioned to remain in the domain (defined by the same change of measure as in (2.9)), and at rate  $\frac{m}{m-1}\lambda$ , a certain number of the original branching diffusions will immigrate. The number of these immigrants has the size-biased

offspring distribution A, where

$$\mathbb{P}\left(\tilde{A}=k\right)=\frac{k+1}{m}\mathbb{P}\left(A=k+1\right)$$

for  $k \geq 0$ . Note the spine particle itself is not included in  $\tilde{A}$ .

One important fact we will use is that the position of the spine particle in the decomposition described by Lemma 2.22 converges quickly to an equilibrium distribution, with density  $\varphi^2$ . In fact, if we assume that  $\frac{1}{2}\Delta$  is *intrinsically ultracontractive* for the domain (for which a Lipschitz assumption is enough), it is well known that this convergence is uniform in the starting position.

Lemma 2.24. Suppose that D is a bounded Lipschitz domain. If

$$K^{D}(t, x, y) = \frac{e^{\lambda t} p^{D}(t, x, y)\varphi(y)}{\varphi(x)}$$

is the transition density for Brownian motion conditioned to remain in D, then for any  $\varepsilon > 0$  there exists a constant  $C_{\varepsilon}$  depending only on the domain such that

$$\left|\frac{K^D(t, x, y)}{\varphi(y)^2} - 1\right| \le C_{\varepsilon} e^{-\gamma t}$$

for all  $t > \varepsilon$  and  $x, y \in D$  where  $\gamma := \lambda_2 - \lambda_1 > 0$  is the spectral gap for -1/2 the Laplacian on D.

*Proof.* See for example [DS84] or [Bañ99, Equation (1.8)].

**Remark 2.25.** The corresponding convergence to equilibrium when the single particle motion is governed by a generator L as in Theorem 2.3 still holds whenever the domain is Lipschitz (see [Bañ99].)

#### 2.4.2 Asymptotics for the Survival Probability

Using this spine decomposition, and the fact that the law of the spine particle converges to an equilibrium distribution as  $t \to \infty$ , we may first deduce that we have an asymptotic for the survival probability which is of the correct form.

**Proposition 2.26.** Uniformly in  $x \in D$ 

$$\mathbb{P}_x \left( N_t > 0 \right) \sim a(t) \varphi(x)$$

as  $t \to \infty$ , where

$$a(t) := \int_D \mathbb{P}_z \left( N_t > 0 \right) \varphi(z) \, dz$$

converges to 0 as  $t \to \infty$ , and is independent of x.

*Proof.* The key idea for the proof of this is to write, recalling (4.6),

$$\frac{\mathbb{P}_x\left(N_t > 0\right)}{\varphi(x)} = \mathbb{Q}_x\left[\frac{1}{\sum_{i=1}^{N_t}\varphi(X_t^i)}\right]$$

and then show that the right hand side essentially does not depend on x for large t. The intuition behind this is that under the new measure, the position of the spine particle will converge very quickly to equilibrium. Then, contributions to the sum in the denominator from subprocesses branching off the spine before its position has become well mixed are unlikely to occur, as these have the law of standard  $\lambda$ -branching Brownian motions, which we know are unlikely to survive for a long time.

To begin, for  $t_0 \leq t$ , write

$$\sum_{i=1}^{N_t} \varphi(X_t^i) = M_t := M_{t_0,t} + M_{0,t_0}$$

where  $M_{0,t_0}$  is the sum of all contributions to  $M_t$  from subprocesses branching off the spine before time  $t_0$ . Also define  $f(r) := \mathbb{Q}_{\varphi^2}[1/\sum_{i=1}^{N_r} \varphi(X_r^i)]$ , where  $\varphi^2$  in the subscript indicates that the initial position is distributed according to the probability measure with density  $\varphi^2$ . Note that this is a function of r only. Then for any  $t_0 \leq t$ 

$$\frac{1}{M_t} = \frac{1}{M_{t_0,t}} + \left(\frac{1}{M_t} - \frac{1}{M_{t_0,t}}\right) = \frac{1}{M_{t_0,t}} - \frac{M_{0,t_0}}{M_t M_{t_0,t}}$$

and so

$$\mathbb{Q}_{x}\left[\frac{1}{\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})}\right] = f(t-t_{0}) + \left(\mathbb{Q}_{x}\left[\frac{1}{M_{t_{0},t}}\right] - f(t-t_{0})\right) - \mathbb{Q}_{x}\left[\frac{M_{0,t_{0}}}{M_{t}M_{t_{0},t}}\right]$$

where we label the error (second and third) terms above by  $\epsilon_x^1(t, t_0)$  and  $\epsilon_x^2(t, t_0)$ respectively. The plan is to show that we can choose  $t_0(t) < t$  such that both the error terms become small as  $t \to \infty$ . The reason we expect these terms to decay is as in the heuristic discussion above, since the equilibrium distribution for Brownian motion conditioned to remain in D is precisely  $\varphi^2$ .

Now observe that since the quantities whose expectations we are evaluating are  $\mathcal{F}_t$  measurable, their  $\mathbb{Q}_x$  and  $\overline{\mathbb{Q}}_x$  expectations are the same, and we may work with either. Considering the definition of  $M_{t_0,t}$ , and conditioning on  $\mathcal{G}_{t_0}$ , for  $(\mathcal{G}_s)_{s\geq 0}$  the filtration generated by the position of the spine up to time s (a subfiltration of

 $(\overline{\mathcal{F}}_s)_{s\geq 0})$  we see that

$$\epsilon_x^1(t,t_0) = \overline{\mathbb{Q}}_x[\overline{\mathbb{Q}}_x[1/M_{t_0,t}|\mathcal{G}_{t_0}]] - f(t-t_0)$$

is simply the difference in expectation of the function of y,

$$\mathbb{Q}_{y}\left[\frac{1}{\sum_{i=1}^{N_{t-t_{0}}}\varphi(X_{t-t_{0}}^{i})}\right] = \mathbb{P}_{y}\left(N_{t-t_{0}} > 0\right)/\varphi(y)$$

under the  $\overline{\mathbb{Q}}_x$  law of the spine particle at time  $t_0$ , and the law with density  $\varphi^2$ . As  $t_0$  increases, we know that the law of the spine particle approaches the law with this density, and by Lemma 2.24 we can quantify this with the bound

$$\sup_{x,y\in D} \left| \frac{K^{D}(t_{0}, x, y)}{\varphi(y)^{2}} - 1 \right| \le C e^{-\gamma t_{0}}$$
(2.12)

for all  $t_0 > 1$  say, where  $K^D(s, x, y)$  is the transition density of Brownian motion conditioned to remain in D. Hence we have the estimate, for all  $t_0 > 1$ :

$$|\epsilon_x^1(t,t_0)| \le \sup_{x,y \in D} \left| \frac{K^D(t_0,x,y)}{\varphi(y)^2} - 1 \right| \int_D \varphi(y) \mathbb{P}_y \left( N_{t-t_0} > 0 \right) \, dy \le C \, \mathrm{e}^{-\gamma t_0} \, f(t-t_0).$$
(2.13)

To bound the second term, write  $A_{t_0}^t$  for the event that some subprocess branching from the spine before time  $t_0$  survives until (total) time t. Since  $M_{0,t_0}/M_t M_{t_0,t}$ is positive and less than or equal to  $1/M_{t_0,t}$  we see that

$$\left|\epsilon_x^2(t,t_0)\right| \le \overline{\mathbb{Q}}_x \left[\frac{1}{M_{t_0,t}} \mathbb{1}_{A_{t_0}^t}\right].$$

Again, to estimate this we condition; but now on  $\tilde{\mathcal{G}}_{t_0}$ , where  $\tilde{\mathcal{G}} \supset \mathcal{G}$  is the filtration which also contains information about the branching points along the spine. The reason for doing this is that we know, given the position of the spine  $(\xi_s)$  for  $0 \leq s \leq t_0$  and all its branching points, that the subprocesses branching off the spine before  $t_0$  and the process continuing on from  $\xi_{t_0}$  are independent. Thus, the term on the right above is equal to

$$\overline{\mathbb{Q}}_{x}\left[\overline{\mathbb{Q}}_{x}\left[\mathbbm{1}_{A_{t_{0}}^{t}}|\tilde{\mathcal{G}}_{t_{0}}\right]\overline{\mathbb{Q}}_{x}\left[\frac{1}{M_{t_{0},t}}\middle|\tilde{\mathcal{G}}_{t_{0}}\right]\right] = \overline{\mathbb{Q}}_{x}\left[\overline{\mathbb{Q}}_{x}\left[\mathbbm{1}_{A_{t_{0}}^{t}}|\tilde{\mathcal{G}}_{t_{0}}\right]\frac{\mathbb{P}_{\xi_{t_{0}}}\left(N_{t-t_{0}}>0\right)}{\varphi(\xi_{t_{0}})}\right],$$

where we can now show that the conditional probability  $\overline{\mathbb{Q}}_x[\mathbb{1}_{A_{t_0}^t}|\tilde{\mathcal{G}}_{t_0}]$  is small. Indeed, since the probability that any subprocess branching off the spine before  $t_0$  survives until total time t is less than  $\sup_{w \in D} |\mathbb{P}_w(N_{t-t_0} > 0)|$ , we see that

$$\overline{\mathbb{Q}}_{x}\left[\mathbb{1}_{A_{t_{0}}^{t}}|\tilde{\mathcal{G}}_{t_{0}}\right] \leq S_{t_{0}} \sup_{w \in D} \left|\mathbb{P}_{w}\left(N_{t-t_{0}}>0\right)\right|,$$

where  $S_{t_0}$  is the number of such subprocesses. Moreover, again for  $t_0 > 1$  say by (2.12), we have

$$\overline{\mathbb{Q}}_x\left[\frac{\mathbb{P}_{\xi_{t_0}}\left(N_{t-t_0}>0\right)}{\varphi(\xi_{t_0})}\right] \lesssim f(t-t_0)$$

where the implied constant depends only on D. Combining all of the above, and noting that  $S_{t_0}$  is independent of the motion under  $\overline{\mathbb{Q}}_x$  with  $\overline{\mathbb{Q}}_x[S_{t_0}] = 2\lambda t_0$  provides the final estimate

$$|\epsilon_x^2(t,t_0)| \le \tilde{C}t_0 f(t-t_0) \sup_{w \in D} |\mathbb{P}_w \left( N_{t-t_0} > 0 \right)|$$
(2.14)

for all  $t_0 > 1$ , where  $\tilde{C}$  is another constant. With both these error bounds in hand, we may deduce that

$$\left|\frac{\mathbb{P}_{x}\left(N_{t}>0\right)/\varphi(x)}{f(t-t_{0})}-1\right| \leq \left|\frac{\epsilon_{x}^{1}(t,t_{0})}{f(t-t_{0})}\right| + \left|\frac{\epsilon_{x}^{2}(t,t_{0})}{f(t-t_{0})}\right| \leq C e^{-\gamma t_{0}} + \tilde{C}t_{0} \sup_{w\in D}\left|\mathbb{P}_{w}\left(N_{t-t_{0}}>0\right)\right|$$

for any  $x \in D$ , and  $1 < t_0 < t$ .

Now, since we know that  $\sup_{w \in D} |\mathbb{P}_w (N_{t-t_0} > 0)| \to 0$  as  $s \to \infty$ , it is possible to choose  $t_0(t)$  such that both  $t_0(t) \to \infty$  and  $t_0(t) \sup_{w \in D} |\mathbb{P}_w (N_{t-t_0} > 0)| \to 0$  as  $t \to \infty$ . Then we have, letting  $c(t) = f(t - t_0(t))$ , that

$$\left|\frac{\mathbb{P}_x\left(N_t > 0\right)/\varphi(x)}{c(t)} - 1\right| \to 0$$
(2.15)

as  $t \to \infty$ , uniformly for  $x \in D$ . To complete the proof, we need only show that c(t) must be asymptotically equivalent to  $a(t) := \int_D \mathbb{P}_x (N_t > 0) \varphi(x) dx$  as  $t \to \infty$ . Note that a(t) is less than  $\sup_{w \in D} |\mathbb{P}_w (N_{t-t_0} > 0)|$ , and so clearly decays with t. To see the equivalence, observe that

$$\left|\frac{a(t)}{c(t)} - 1\right| = \left|\int_D \left(\frac{\mathbb{P}_x\left(N_t > 0\right)}{c(t)} - \varphi(x)\right)\varphi(x)\,dx\right| \le \left\|\frac{\mathbb{P}_x\left(N_t > 0\right)}{c(t)} - \varphi(x)\right\|_{L^2(D)}$$
(2.16)

where the inequality holds by Cauchy-Schwarz and the fact that  $\int_D \varphi^2 = 1$ . Multiplying (2.15) by  $\varphi(x)$  and integrating tells us that the final expression converges to 0 as  $t \to \infty$ .

Remark 2.27. The proof of this Lemma remains essentially the same in our more

general framework, using Remark 2.23 in place of Lemma 2.24. Some more care is required to bound

$$\overline{\mathbb{Q}}_x[S_{t_0}],$$

as multiple processes may now immigrate at each branching point on the spine, but since the size-biased distribution has finite mean (we are assuming that A has finite variance) this is again less than some constant times  $t_0$ .

#### **2.4.3** Asymptotics for a(t)

Now, since by Lemma 2.20 we know that

$$\mathbb{P}_x \left( N_t > 0 \right) := u(x, t)$$

is a solution of the FKPP equation in D, we can find an ODE which is satisfied by a(t). This will allow us to deduce the desired asymptotic for a(t) as  $t \to \infty$ . More precisely, we have the following:

**Lemma 2.28.** Assume that D is C<sup>1</sup>. Letting  $a(t) = \int_D u(x,t)\varphi(x) dx$  we have that a(t) is differentiable for all t > 0 and

$$\frac{da}{dt}(t) = -\lambda \int_D u^2(x,t)\varphi(x)\,dx.$$
(2.17)

*Proof.* First suppose that  $u(\cdot, t) \in H_0^1(D)$  and  $\Delta u(\cdot, t) \in L^2(D)$  for all t > 0. Then, since

$$\frac{\partial u}{\partial t}(x,t) = \frac{1}{2}\Delta u(x,t) + \lambda(u(x,t) - u^2(x,t))$$

we see that  $\frac{\partial u}{\partial t}(\cdot,s) \in L^2(D)$  for all strictly positive s, and

$$\int_{D} \frac{\partial u}{\partial t}(x,s)\varphi(x)\,dx = \int_{D} \left(\frac{1}{2}\Delta u(x,s) + \lambda(u(x,s) - u^{2}(x,s))\right)\varphi(x)\,dx$$

is well defined. Furthermore, we have that  $\varphi \in H_0^1(D)$  (since  $\partial D$  is assumed  $C^1$  and  $\varphi$  vanishes on the boundary - see [Eva98, §5.5]) and that  $u \in H_0^1(D)$  by assumption. This means that we can integrate by parts against  $\varphi \in H_0^1(D)$  and use that  $\varphi$  is an eigenfunction of the Laplacian to obtain the equality

$$\int_{D} \frac{du}{dt}(x,s)\varphi(x)\,ds = -\lambda \int_{D} u^{2}(x,s)\varphi(x)dx.$$
(2.18)

Observe that the left hand side is continuous in s (for s > 0) since the right hand side must be by continuity of  $u^2$  and dominated convergence. Hence, for any t > 0, letting 0 < a < t we see that

$$-\lambda \int_D u^2(x,t)\varphi(x)\,dx = \frac{d}{dt} \int_a^t \int_D -\lambda u^2(x,s)\varphi(x)\,dx = \frac{d}{dt} \int_a^t \int_D \frac{\partial u}{\partial t}(x,s)\varphi(x)\,dxds$$

by the continuity discussed above, and (2.18). Then, applying Fubini and using the continuity of  $\frac{du}{dt}(x,s)$  in s for fixed x, we can write this as

$$\frac{d}{dt} \int_D \int_a^t \frac{\partial u}{\partial t}(x,s) ds \,\varphi(x) dx = \frac{d}{dt} \int_D u(x,t)\varphi(x) - u(x,a)\varphi(x) \, dx = \frac{da}{dt}(t).$$

Therefore, we need only show that  $u \in H_0^1(D)$  and  $\Delta u \in L^2(D)$ . However, this is simply the regularity that u obtains by virtue of being a solution of (2.8). This is a straightforward consequence of the standard regularity theory for parabolic PDEs. Since the domain is  $C^1$  and u vanishes on the boundary, it is enough to show that

$$\sum_{k=1}^{\infty} \lambda_k \langle f, \varphi_k \rangle^2 < \infty \quad \text{and} \quad \sum_{k=1}^{\infty} \lambda_k^2 \langle f, \varphi_k \rangle^2 < \infty$$
(2.19)

(see, for example, [Tho06, Lemma 3.1]). This can be proved using the Duhamel representation for u as a solution of (2.8) and a standard bootstrapping argument. We omit the straightforward calculations.

**Remark 2.29.** For the more general branching diffusion we can apply the same arguments to show that

$$\frac{da}{dt}(t) = -\frac{\lambda}{m-1} \int_D \left( G(1-u) + mu - 1 \right) \varphi(x) \, dx \tag{2.20}$$

where G is the probability generating function of A. Continuity of the right-hand side requires an extra application of the dominated convergence theorem to first see that  $G(1-u) = \mathbb{E}[(1-u)^A]$  is continuous. To bound the Sobolev norms in (2.19) one must observe that

$$\nabla(G(1-u)) = -\nabla u \mathbb{E}[A(1-u)^A]$$

is bounded in modulus by a constant times  $|\nabla u|$ .

This allows us to deduce an asymptotic for a(t), which completes the proof of Theorem 2.4.

Proposition 2.30.

$$a(t) \underset{t \to \infty}{\sim} \frac{1}{\left(\lambda \int_D \varphi(y)^3 \, dy\right) t}.$$

*Proof.* The desired asymptotic for a(t) follows fairly easily from Lemma 2.28 since we have, writing  $u(t, x) = a(t)\varphi(x) + v(t, x)$  and substituting this into (2.17), that

$$\frac{da}{dt}(t) = -\lambda a^2(t) \int_D \varphi(y)^3 \, dy - 2\lambda a(t) \int_D v(t,x)\varphi^2(x) \, dx - \lambda \int_D v^2(t,x)\varphi(x) \, dx.$$
(2.21)

Then, Proposition 2.26 tells us that the second two terms are negligible compared with the first for large t. Indeed, since  $|u(t,x)/a(t)\varphi(x)-1| = |v(t,x)/a(t)\varphi(x)| \to 0$  uniformly in x, we have that

$$\frac{da}{dt}(t) = \left(-\lambda \int_D \varphi(y)^3 \, dy - g(t)\right) a^2(t)$$
  
for  $g(t) := 2\lambda \int_D \frac{v(t,x)}{a(t)} \varphi^2(x) \, dx + \lambda \int_D \frac{v(t,x)^2}{a(t)^2} \varphi(x) \, dx$ 

where  $g(t) \to 0$  as  $t \to \infty$ . Thus we obtain, denoting differentiation with respect to t by a dot, that

$$-\lambda \int_D \varphi(y)^3 \, dy - |g(t)| \le \frac{\dot{a}(t)}{a^2(t)} \le -\lambda \int_D \varphi(y)^3 \, dy + |g(t)|.$$

Moreover, since  $\frac{da^{-1}}{dt}(t) = -\frac{\dot{a}(t)}{a^2(t)}$ , integration yields that

$$\frac{1}{a(t)} \ge \left(\lambda \int_D \varphi(y)^3 \, dy\right) (t-1) - \int_1^t |g(s)| \, ds + \frac{1}{a(1)}$$
$$\frac{1}{a(t)} \le \left(\lambda \int_D \varphi(y)^3 \, dy\right) (t-1) + \int_1^t |g(s)| \, ds + \frac{1}{a(1)}$$

where we have started from 1 to avoid any differentiability issues at 0. Note that  $\dot{a}(s)/a^2(s)$  is clearly integrable over [1, t] for any t, and so we are justified in applying the fundamental theorem of calculus here. Upon dividing by  $\left(\lambda \int_D \varphi(y)^3 dy\right) t$  we see that

$$\left|\frac{\frac{1}{(\lambda\int_D\varphi(y)^3\,dy)t}}{a(t)} - 1\right| \le \frac{1}{t} + \frac{1}{\left(\lambda\int_D\varphi(y)^3\,dy\right)}\left(\frac{1}{a(1)t} + \frac{\int_0^t|g(s)|\,ds}{t}\right).$$

The first term in the brackets on the right hand side of this expression clearly converges to 0 as  $t \to \infty$ . Furthermore, since |g| is bounded and  $|g(s)| \to 0$  as  $s \to \infty$ , so does the second. This yields the result.

Remark 2.31. Note that this Proposition, combined with the proof Theorem 2.3,

in particular (2.15), shows that

$$\left|\frac{t\lambda\int_D\varphi(y)^3\,dy\times\mathbb{P}_x(N_t>0)}{\varphi(x)}-1\right|\longrightarrow 0$$

as  $t \to \infty$ , uniformly in x.

Remark 2.32. For the more general set up, observe that

$$G(1-u) + mu - 1 = \frac{\mathbb{E}[A^2] - \mathbb{E}[A]}{2}u^2 + o(u^2)$$

by Taylor's theorem and our moment assumption on the offspring distribution A (see [DN80, Theorem A] for a general statement concerning Taylor expansions of probability generating functions). Replacing the expression for  $\frac{da}{dt}$  by (2.20) and proceeding as above, we may incorporate the error term from the integral into g(t), and reach the desired conclusion.

# 2.5 The Conditioned System

Theorem 2.4 allows us to study the law of branching Brownian motion conditioned to survive for a long time in much greater depth. One aspect of the limiting behaviour is captured by what happens to the law of the process run up to some fixed time T, if it is then conditioned to survive until a much larger time t. It turns out that this limiting description is given precisely by the evolution of the process under  $\mathbb{Q}_x$ , as described in Lemma 2.22.

Proof of Proposition 2.5. Recall, we would like to prove that for any  $T \ge 0$ ,  $x \in D$  and  $B \in \mathcal{F}_T$ , we have that

$$\lim_{t \to \infty} \mathbb{P}_x \left( B | N_t > 0 \right) = \mathbb{Q}_x(B).$$

Conditioning on  $\mathcal{F}_T$ , we see that

$$\mathbb{P}_{x}\left(B|N_{t}>0\right) = \frac{\mathbb{E}_{x}\left[\mathbbm{1}_{B}\mathbb{P}_{x}\left(N_{t}>0|\mathcal{F}_{T}\right)\right]}{\mathbb{P}_{x}\left(N_{t}>0\right)} := \frac{\mathbb{E}_{x}\left[\mathbbm{1}_{B}Y\right]}{\mathbb{P}_{x}\left(N_{t}>0\right)}$$

where we have defined

$$Y := \mathbb{P}_x \left( N_t > 0 | \mathcal{F}_T \right) = \sum_{i=1}^{N_T} \mathbb{P}_{X_T^i} \left( N_{t-T} > 0 \right) \left( \prod_{j < i} \mathbb{P}_{X_T^j} \left( N_{t-T} = 0 \right) \right).$$

Then, from our asymptotic for the survival probability and the fact that  $\frac{t}{t-T} \to 1$  as  $t \to \infty$ , it follows that

$$\frac{\mathbbm{1}_B Y}{\mathbbm{P}_x \left(N_t > 0\right)} \xrightarrow[t \to \infty]{} \frac{\sum_{i=1}^{N_T} \varphi(X_T^i)}{\varphi(x)} \mathbbm{1}_B = \frac{M_T}{M_0} \mathbbm{1}_B$$

almost surely, as  $t \to \infty$ . Moreover, we have that  $Y \leq \sum_{i=1}^{N_T} \mathbb{P}_{X_T^i} (N_{t-T} > 0) \lesssim \frac{M_T}{t-T}$  for all large enough t (recalling that our asymptotic estimates were uniform in D), and so we can dominate  $\mathbb{1}_B Y/\mathbb{P}_x (N_t > 0)$  by an integrable random variable, namely a constant multiple of  $M_T$ . The dominated convergence theorem then provides the result.

Given the asymptotic for the survival probability, it is also not too much work to prove Theorem 2.6, which gives some limiting information on the positions of particles at time t, given survival. Recall (since we are working in the branching Brownian motion case) we would like to show that for any f with  $\langle f^2, \varphi \rangle < \infty$  that we have

$$(t^{-1}\sum_{i=1}^{N_t} f(X_t^i)|N_t > 0) \to Z$$

in distribution as  $t \to \infty$ , where  $Z \sim \text{Exp}(1/\lambda \langle \varphi, f \rangle \int_D \varphi(y)^3 dy)$ . To prove Theorem 2.6, we will use the method of moments, relying on Theorem 2.4 and the following Lemma.

**Lemma 2.33.** For all  $n \in \mathbb{N}$  and  $x \in D$ ,

$$\frac{\mathbb{E}_x \left[ (\sum_{i=1}^{N_t} \varphi(X_t^i))^n \right]}{n! \varphi(x) \lambda^{n-1} (\int_D \varphi^3)^{n-1} t^{n-1}} \to 1$$

as  $t \to \infty$ , uniformly in x.

Proof of Lemma 2.33. The proof of this relies on the expression for  $\mathbb{E}_x[(\sum_{i=1}^{N_t} \varphi(X_t^i))^n]$  that one obtains from a Many-To-Few generalisation of the Many-to-Two Lemma, see below, and then proceeds by induction. We start with n = 1. This case is simple, however, because we know that

$$\left| \mathbb{E}_x \left[ \frac{\sum_{i=1}^{N_t} \varphi(X_t^i)}{\varphi(x)} \right] - 1 \right| = 0$$

for all t and x, since the expectation is that of our mean 1 martingale  $M_t/M_0$ .

We also record here that

$$\left|\frac{\mathrm{e}^{\lambda t} \mathbf{E}_{x}\left[\varphi^{2}(B_{t})\mathbb{1}_{\{t>\tau_{D}\}}\right]}{\varphi(x)\int_{D}\varphi^{3}} - 1\right| = \frac{1}{\int_{D}\varphi(y)^{3}\,dy}\int_{D}\frac{e^{\lambda t}p^{D}(t,x,y)\varphi^{2}(y)}{\varphi(x)} - \varphi(y)^{3}\,dy$$

$$(2.22)$$

$$\leq \sup_{y\in D}\left|\frac{K^{D}(t,x,y)}{\varphi(y)^{2}} - 1\right|$$

which we know is less than  $C e^{-\gamma t}$  for all x and  $t \ge 1$  say, where C is some universal constant. This fact will be crucial to the induction.

For the inductive step, we need a Many-to-Few Lemma, which is a generalisation of Lemma 2.13. This tells us, as a special case, that

$$\mathbb{E}_{x}\left[\left(\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})\right)^{n}\right] = \mathrm{e}^{\lambda t} \mathbf{E}_{x}\left[\varphi(B_{t})^{n}\mathbb{1}_{\{\tau_{D}>t\}}\right] + \sum_{j=1}^{n-1} \binom{n}{j} \int_{0}^{t} \lambda \,\mathrm{e}^{\lambda s} \mathbf{E}_{x}\left[\mathbb{1}_{\{\tau_{D}>s\}}E^{j}(t-s,B_{s})\right] \,ds$$

$$(2.23)$$

where  $E^{j}(s,x) := \mathbb{E}_{x}[(\sum_{i=1}^{N_{s}} \varphi(X_{s}^{i}))^{n-j}]\mathbb{E}_{x}[(\sum_{i=1}^{N_{s}} \varphi(X_{s}^{i}))^{j}]$ , and is proved in a very general setting in [HR15, Lemma 1]. Thus, for n > 1 we can break up

$$\left| \frac{\mathbb{E}_x \left[ (\sum_{i=1}^{N_t} \varphi(X_t^i))^n \right]}{n! \varphi(x) \lambda^{n-1} (\int_D \varphi^3)^{n-1} t^{n-1}} - 1 \right|$$

into *n* parts, each corresponding to one of the terms in (2.23). Since we know that  $e^{\lambda t} \mathbf{E}_x \left[ \varphi(B_t)^n \mathbb{1}_{\{\tau_D > t\}} \right] \leq \|\varphi\|_{\infty}^{n-1} \varphi(x)$ , the first of these terms will tend to 0 uniformly in *x* as  $t \to \infty$ . Thus we need only show that for each  $1 \leq j \leq n-1$ , we have

$$\left| \frac{\int_0^t e^{\lambda s} \mathbf{E}_x \left[ \mathbbm{1}_{\{\tau_D > s\}} E^j(t - s, B_s) \right] ds}{j!(n - j)! \varphi(x) \lambda^{n - 2} (\int_D \varphi^3)^{n - 1} t^{n - 1}} - \frac{1}{n - 1} \right| \to 0$$
(2.24)

uniformly in x as  $t \to \infty$ . In the following we set

$$Q_{y}^{k}(r) := \frac{\mathbb{E}_{y}\left[ (\sum_{i=1}^{N_{r}} \varphi(X_{r}^{i}))^{k} \right]}{k! \lambda^{k-1} (\int_{D} \varphi^{3})^{k-1} r^{k-1}}$$
(2.25)

for  $y \in D$ ,  $k \in \mathbb{N}$  and  $r \ge 0$  so that (2.24) is less than

$$\left| \frac{\int_{0}^{t} e^{\lambda s} (t-s)^{n-2} \mathbf{E}_{x} \left[ \mathbbm{1}_{\{\tau_{D}>s\}} \left( Q_{B_{s}}^{n-j}(t-s) Q_{B_{s}}^{j}(t-s) - \varphi^{2}(B_{s}) \right) \right] ds}{t^{n-1} \varphi(x) \int_{D} \varphi^{3}} + \left| \frac{\int_{0}^{t} e^{\lambda s} \mathbf{E}_{x} \left[ \varphi^{2}(B_{s}) \mathbbm{1}_{\{\tau_{D}>s\}} \right] (t-s)^{n-2} ds}{t^{n-1} \varphi(x) \int_{D} \varphi^{3}} - \frac{1}{n-1} \right|. \quad (2.26)$$

We will show that both the terms here converge to 0, uniformly in x.

The second expression is relatively easy to deal with, since  $\int_0^t (t-s)^{n-2}/t^{n-1} = 1/(n-1)$ . This means that we can pull the 1/(n-1) term, and then the modulus, inside of the integral. After doing this, and noting that  $(t-s)^{n-2}/t^{n-1} \leq 1/t$  for all  $s \in [0, t]$ , we see that the term is bounded above by

$$\frac{1}{t} \int_0^t \left| \frac{\mathrm{e}^{\lambda s} \mathbf{E}_x \left[ \varphi^2(B_s) \mathbb{1}_{\{\tau_D > s\}} \right]}{\varphi(x) \int_D \varphi^3} - 1 \right| ds.$$

From here the result follows by (2.22). Note that the bound  $\mathbf{E}_x[\varphi^2(B_s)\mathbb{1}_{\{\tau_D>s\}}] \leq e^{-\lambda s} \varphi(x) \|\varphi\|_{\infty}$  (which holds for all s and x) tells us that the integrand is uniformly bounded in x.

For the first expression, we use our induction hypothesis. This tells us that  $Q_z^{n-j}(r)Q_z^j(r)/\varphi(z)^2 \to 1$  uniformly in z as  $r \to \infty$ . It is also clear from the definition (2.25) that for fixed n and j,  $r^{n-2}Q_z^{n-j}(r)Q_z^j(r)$  is uniformly bounded on any compact interval [0, T] (for example by bounding the expectation in (2.25) by the corresponding expectation for a Yule process and using that  $\varphi$  is bounded). With this in mind we bound (2.26) above by

$$\frac{\sup_{z} \sup_{r \ge T} |Q_{z}^{n-j}(r)Q_{z}^{j}(r)/\varphi(z)^{2} - 1|}{t\int_{D} \varphi(y)^{3} dy} \int_{0}^{t-T} e^{\lambda s} \frac{\mathbf{E}_{x} \left[\mathbbm{1}_{\{\tau_{s} > 0\}} \varphi^{2}(B_{s})\right]}{\varphi(x)} ds + \frac{\sup_{z} \sup_{r \le T} (|r^{n-2}Q_{z}^{n-j}(r)Q_{z}^{j}(r)| + |r^{n-2}\varphi(z)^{2}|)}{t^{n-1}\int_{D} \varphi(y)^{3} dy} \int_{t-T}^{t} e^{\lambda s} \frac{\mathbf{E}_{x} \left[\mathbbm{1}_{\{\tau_{D} > s\}}\right]}{\varphi(x)} ds$$

where we have used that  $(t-s)^{n-2}/t^{n-1} \leq 1/t$  for the first part. Since

$$e^{\lambda s} \frac{\mathbf{E}_x[\mathbf{1}_{\{\tau_D > s\}}]}{\varphi(x)} = \int_D \frac{e^{\lambda s} p^D(s, x, y)}{\varphi(x)\varphi(y)} \varphi(y) \, dy$$

is the expectation of  $\varphi(\xi_s)$  for  $\xi$  a Brownian motion conditioned to remain in D, and therefore bounded uniformly in s and x, we obtain the convergence to 0 by letting first  $T \to \infty$  and then  $t \to \infty$  in the above.

Proof of Theorem 2.6. Lemma 2.33 combined with Theorem 2.4 tells us that for
each n > 0,

$$\mathbb{E}_x[(t^{-1}\sum_{i=1}^{N_t}\varphi(X_t^i))^n|N_t>0] \to n! \left(\lambda \int_D \varphi^3\right)^n$$

as  $t \to \infty$ , where the right hand side is the *n*th moment of an  $\text{Exp}(1/\lambda \int_D \varphi^3)$ random variable. Since convergence of the moments is enough to ensure convergence in distribution when the limiting distribution is nice enough, see for example [Bil95, Theorem 30.2], Theorem 2.6 is proved in the case  $f = \varphi$ . To deal with general f we write  $\tilde{f} = f - \langle \varphi, f \rangle \varphi$ . We will show that for any  $\varepsilon > 0$ 

$$\mathbb{P}_x(|t^{-1}\sum_{i=1}^{N_t}\tilde{f}(X_t^i)| > \varepsilon |N_t > 0) \to 0$$

as  $t \to \infty$  (uniformly in x), which implies the result by the above decomposition and the proof when  $f = \varphi$ . To do this we use Markov's inequality and the Many-to-Two Lemma. This Lemma tells us that  $\mathbb{E}_x[(t^{-1}\sum_{i=1}^{N_t} \tilde{f}(X_t^i))^2 | N_t > 0]$  is equal to

$$\frac{\varphi(x)}{t\mathbb{P}_x(N_t>0)} \left( \frac{\mathrm{e}^{\lambda t} \mathbf{E}_x[\tilde{f}(B_t)^2 \mathbb{1}_{\{\tau_D>t\}}] + 2\int_0^t \lambda \, \mathrm{e}^{\lambda s} \mathbf{E}_x[\mathbb{1}_{\{\tau_D>s\}} \mathbb{E}_{B_s}[\sum_{i=1}^{N_{t-s}} \tilde{f}(X_{t-s}^i)]^2] \, ds}{t\varphi(x)} \right)$$

where we know that the expression outside the brackets is uniformly bounded in t and x (for t large enough). We also know that

$$\frac{\mathrm{e}^{\lambda t} \mathbf{E}_x[\tilde{f}(B_t)^2 \mathbb{1}_{\{\tau_D > t\}}]}{t\varphi(x)} \le t^{-1} \left(1 + \sup_y \left|\frac{K^D(t, x, y)}{\varphi(y)^2} - 1\right|\right) (\varphi, \tilde{f}^2)$$
(2.27)

and so this term converges uniformly to 0 by the assumption on f. To conclude we use the fact that  $(\varphi, \tilde{f}) = 0$ , which was the reason for choosing  $\tilde{f}$  as we did. This means that  $\sup_{z} \varphi(z)^{-1} \mathbb{E}_{z}[\sum_{i=1}^{N_{r}} \tilde{f}(X_{r}^{i})] \to 0$  as  $r \to \infty$ , by the Many-to-One Lemma and the same argument used for the bound in (2.27). Thus we have that

$$\begin{split} (t\varphi(x))^{-1} &\int_0^t \lambda \,\mathrm{e}^{\lambda s} \,\mathbf{E}_x[\mathbbm{1}_{\{\tau_D > s\}} \mathbb{E}_{B_s}[\sum_{i=1}^{N_{t-s}} \tilde{f}(X_{t-s}^i)]^2] \,ds \\ &\leq \frac{\int_{t-T}^t \lambda \,\mathrm{e}^{\lambda s} \,\mathbf{E}_x[\mathbbm{1}_{\{\tau_D > s\}} \mathbb{E}_{B_s}[\sum_{i=1}^{N_{t-s}} \tilde{f}(X_{t-s}^i)]^2] \,ds}{t\varphi(x)} \\ &+ \sup_{r \geq T} \sup_z \left(\varphi(z)^{-1} \mathbb{E}_z[\sum_{i=1}^{N_r} \tilde{f}(X_r^i)]\right)^2 \frac{\int_0^{t-T} \lambda \,\mathrm{e}^{\lambda s} \,\mathbf{E}_x[\varphi(B_s)^2 \mathbbm{1}_{\{\tau_D > t\}}] \,ds}{t\varphi(x)} \end{split}$$

where the second term on the right can be made arbitrarily small for all  $t \ge T$  as long as T is large enough (using the bound  $e^{\lambda s} \mathbf{E}_x[\varphi(B_s)^2 \mathbb{1}_{\{\tau_D > t\}}] \le \varphi(x) \|\varphi\|_{\infty}$  on the integrand). Therefore, we are left with having to show that, for fixed T, the first term converges to 0 (uniformly in x) as  $t \to \infty$ . However, we can apply the Many-to-one Lemma and Cauchy-Schwarz to the  $\mathbb{E}_{B_s}[\cdot]$  term in the integral, and then the Markov property, to see that

$$\mathbf{E}_{x}[\mathbb{1}_{\{\tau_{D}>s\}}\mathbb{E}_{B_{s}}[\sum_{i=1}^{N_{t-s}}\tilde{f}(X_{t-s}^{i})]^{2}] \leq \mathbf{E}_{x}[\mathbb{1}_{\{\tau_{D}>s\}}e^{2\lambda(t-s)}\mathbf{E}_{B_{s}}[\tilde{f}^{2}(\tilde{B}_{t-s})] \\
\leq e^{2\lambda(t-s)}\mathbf{E}_{x}[\mathbb{1}_{\{\tau_{D}>t\}}\tilde{f}(B_{t})^{2}]$$

where the process  $\tilde{B}$  in the middle expression is just an independent Brownian motion. Using the assumption that  $(\varphi, \tilde{f}^2) < \infty$  and (2.27) once more, the result follows from computing the integral.

**Remark 2.34.** We note here that Lemma 2.33 would also hold if we started the initial particle in a random position, with density  $\varphi^2$ . This follows directly from the proof, since everything is uniform in x. In fact, since the asymptotic for the survival probability is also uniform in the starting point by Remark 2.31, we have that Theorem 2.6 holds when  $f = \varphi$  and we start the system in this random initial position. Of course this also holds for other initial configurations, but we will only use this one in the proof.

We conclude by explaining how one can obtain Corollary 2.7 from here, which describes the asymptotic distribution of a particle picked at random from the population, given survival.

Proof of Corollary 2.7 . To prove the Corollary we first show that for any f with  $(\varphi,f^2)<\infty$ 

$$\mathbb{P}_{x}\left(\left|\frac{\sum_{i=1}^{N_{t}} f(X_{t}^{i})}{\sum_{i=1}^{N_{t}} \varphi(X_{t}^{i})} - (\varphi, f)\right| > \varepsilon \left|N_{t} > 0\right) \to 0$$
(2.28)

as  $t \to \infty$ . Defining  $\tilde{f}$  as in the proof of Theorem 2.6, this is equal to

$$\mathbb{P}_{x}\left(\left|\frac{t\sum_{i=1}^{N_{t}}\tilde{f}(X_{t}^{i})}{t\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})}\right| > \varepsilon \left|N_{t} > 0\right) \le \mathbb{P}_{x}\left(t^{-1}\sum_{i=1}^{N_{t}}\tilde{f}(X_{t}^{i}) > \delta|N_{t} > 0\right) + \mathbb{P}_{x}\left(t^{-1}\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i}) < \delta/\varepsilon|N_{t} > 0\right) + \mathbb{P}_{x}\left(t^{-1}\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i}) < \delta/\varepsilon|N_{t} > 0\right)$$
(2.29)

for any  $\delta > 0$ . From the proof of Theorem 2.6, if we take a limit as  $t \to \infty$  on the right hand side, we are left with simply the probability that an exponential random variable is less than  $\delta/\varepsilon$ . Taking  $\delta \to 0$  proves (2.28). The Corollary then follows by applying the above with both f and the constant function 1, and writing

$$\sum f(X_t^i)/N_t = \sum f(X_t^i)/\sum \varphi(X_t^i) \times \sum \varphi(X_t^i)/N_t.$$

**Corollary 2.35.** For f as in Corollary 2.7, and any  $\varepsilon > 0$ , we have that

$$\mathbb{P}_{x}\left(\left|\frac{\sum_{i=1}^{N_{t}}f(X_{t}^{i})}{\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})}-(\varphi,f)\right|>\varepsilon\right|N_{t}>0\right)\to0$$
(2.30)

as  $t \to \infty$ , uniformly in the starting position x.

*Proof.* Note that the proof of Theorem 2.6 immediately gives us convergence of the first term in (2.29) to 0 as  $t \to \infty$ , for any  $\delta > 0$ , uniformly in x. Therefore, it is sufficient for us to prove that there exists a function  $g(\delta)$  converging to 0 as  $\delta \downarrow 0$ , such that for every  $\eta > 0$  there exists a T with

$$\mathbb{P}_x\left(t^{-1}\sum_{i=1}^{N_t}\varphi(X_t^i) < \delta/\varepsilon | N_t > 0\right) \le g(\delta) + \eta$$

for all  $t \ge T$  and all  $x \in D$ . To do this, we change measure to  $\mathbb{Q}_x$ , and use the fact that the spine particle under  $\overline{\mathbb{Q}}_x$  converges uniformly to an equilibrium distribution with density  $\varphi^2$ . By definition of the change of measure, and Cauchy-Schwarz, we have that

$$\mathbb{P}_{x}\left(t^{-1}\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i}) < \delta/\varepsilon | N_{t} > 0\right)$$

$$\leq \frac{\varphi(x)}{t\mathbb{P}_{x}(N_{t} > 0)}\mathbb{Q}_{x}\left[\left(\frac{1}{\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})}\right)^{2}\right]^{1/2}\mathbb{Q}_{x}\left(\frac{\sum_{i=1}^{N_{t}}\varphi(X_{t}^{i})}{t} \le \delta/\varepsilon\right)^{1/2}$$

$$(2.31)$$

which is in turn less than

$$\frac{\varphi(x)}{t\mathbb{P}_x(N_t>0)}\overline{\mathbb{Q}}_x\left[1/\varphi(\xi_t)^2\right]^{1/2}\overline{\mathbb{Q}}_x\left(\mathbb{Q}_{\xi_u}\left(\frac{\sum_{i=1}^{N_{t-u}}\varphi(X_{t-u}^i)}{t} \le \delta/\varepsilon\right)\right)^{1/2}$$

where  $\xi_s$  is the position of the spine particle at time s, under  $\overline{\mathbb{Q}}_x$ . The second inequality holds simply by positivity of  $\varphi$ , and the fact that  $\mathbb{Q}_x = \overline{\mathbb{Q}}_x$  on  $\mathcal{F}_t$ , the filtration generated by the process, but not the position of the spine, up to time t. Now, the first two terms in the product (2.31) do not depend on  $\delta$ , and are uniformly bounded in t and x, for t > 1 say. This follows from Remark 2.31 (uniform asymptotic for the survival probability), and Lemma 2.24 (convergence of the spine.) The final term of the product is also, by Lemma 2.24 again, less than or equal to

$$\left(Ce^{-\gamma u} + \int_D \varphi(y)^2 \mathbb{Q}_y \left(\frac{\sum_{i=1}^{N_{t-u}} \varphi(X_{t-u}^i)}{t} \le \delta/\varepsilon\right) dy\right)^{1/2}$$
(2.32)

for all u > 1 say, where C does not depend on x, u or t. By changing measure back to  $\mathbb{P}_y$  in the integrand, and again applying Cauchy-Schwarz, we see that the second term in the square root in (2.32) above is bounded by

$$\int_{D} \varphi(y)^{2} \left( \frac{\mathbb{P}_{y}(N_{t-u>0}) \mathbb{E}_{y}[(\sum_{i=1}^{N_{t-u}} \varphi(X_{t-u}^{i}))^{2} | N_{t-u} > 0]^{1/2}}{\varphi(y)} \times \mathbb{P}_{y} \left( \frac{\sum_{i=1}^{N_{t-u}} \varphi(X_{t-u}^{i})}{t-u} \leq \frac{t}{t-u} \delta/\varepsilon \left| N_{t-u} > 0 \right)^{1/2} \right) dy.$$

However we know that  $\varphi(y)^{-1}\mathbb{P}_y(N_{t-u>0})\mathbb{E}_y[(\sum_{i=1}^{N_{t-u}}\varphi(X_{t-u}^i))^2|N_{t-u}>0]^{1/2}$  is uniformly bounded in y as long as t-u>1 say (using the moments we calculated in Lemma 2.33 and the asymptotic for the survival probability). Since everything is positive we can take this bound outside of the integral. Furthermore, by Remark 2.34 we know that if we let  $g'(\delta)$  be the square root of the probability that an exponential random variable with mean  $\lambda \int_D \varphi(y)^3 dy$  is less than  $\delta/\varepsilon$ , then  $g'(\delta) \downarrow 0$  as  $\delta \downarrow 0$ , and

$$\int_D \varphi(y)^2 \mathbb{P}_y \left( \frac{\sum_{i=1}^{N_{t-u}} \varphi(X_{t-u}^i)}{t-u} \le \frac{t}{t-u} \delta/\varepsilon \, \middle| \, N_{t-u} > 0 \right)^{1/2} \, dy \longrightarrow g'(\delta)$$

as  $t \to \infty$ , for any fixed u. Putting all of this together proves the result.

We will use this to prove a stronger version of Corollary 2.7. We know by the Corollary that the average value of f(v) among all vertices v at large height in one tree, given survival, converges to  $\langle f, \varphi \rangle / \langle 1, \varphi \rangle$ . The next Lemma will tell us that in fact we need only look at the average over a large enough subset of these vertices. This will be helpful to us for the proof of Theorem 2.1.

**Lemma 2.36.** Let f be as in Corollary 2.7. Then for any  $\varepsilon, \rho > 0$  and  $x \in D$ 

$$\mathbb{P}_x\left(B_t(\rho)|N_t>0\right) := \mathbb{P}_x\left(\bigcup_{\substack{\rho t \le M \le N_t}} \left\{ \left|\frac{\sum_{i=1}^M f(X_t^i)}{M} - \frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle}\right| > \varepsilon \right\} \middle| N_t > 0 \right) \to 0$$

as  $t \to \infty$ .

Recall here from Section 2.2.1 how we have chosen to enumerate the particles in the system.

*Proof.* We will prove the Lemma by looking at the tree conditioned to survive until time t, and dividing the set of vertices at time t into families, depending on whether or not they have the same ancestor at some earlier time. This earlier time will be chosen such that with high probability, the average value of f over the positions of any one of these families is close to  $\langle f, \varphi \rangle / \langle 1, \varphi \rangle$ . This will show that at many places along the tth generation (vertices at height t) the average of f, taken over the positions of all previous vertices at this height, is close to what we want. To extend this to all vertices far enough along the tth generation, we will show that the size of each of these families is very small compared to t.



Figure 20: Sketch of the argument. There are  $O(g(t, \varepsilon))$  particles at time  $t - t/g(t, \varepsilon)$  with descendants at time t (marked with dots). Each of these families is likely to be good and has size  $O(t/g(t, \varepsilon))$ .

To do this, fix  $\varepsilon > 0$  and write

$$p(t,\varepsilon) := \sup_{x \in D} \mathbb{P}_x \left( \left| \frac{\sum_{i=1}^{N_t} f(X_t^i)}{N_t} - \frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle} \right| > \varepsilon/2 \left| N_t > 0 \right),$$

which by Corollary 2.35, converges to 0 as  $t \to \infty$ . This means that we can choose a function  $g(t, \varepsilon)$  such that  $g(t, \varepsilon) \to \infty$  as  $t \to \infty$ , but

$$g(t,\varepsilon) p\left(\frac{t}{g(t,\varepsilon)},\varepsilon\right) \to 0$$
 (2.33)

as  $t \to \infty$ . Indeed, since  $\sup\{p(u,\varepsilon); u \ge t\}$  converges monotonically to 0 as  $t \to \infty$ , you can choose  $g(t,\varepsilon)$  less than  $\sqrt{t}$  but still converging to  $\infty$ , such that  $g(t,\varepsilon) \sup\{p(u,\varepsilon); u \ge \sqrt{t}\} \to 0$  as  $t \to \infty$ . Then since  $p(t/g(t,\varepsilon),\varepsilon) \le \sup\{p(u,\varepsilon); u \ge t/g(t,\varepsilon)\} \le \sup\{p(u,\varepsilon); u \ge \sqrt{t}\}$ , the function g will satisfy (2.33).

As mentioned above, we will break up the particles at time t into families.

Two vertices will be in the same family if they have a common ancestor at time  $t - t/g(t, \varepsilon)$ . We let the number of these families be  $\hat{N}_{t-t/g(t,\varepsilon)}$  and set  $m_i$  to be the average value of f among the *i*th family. Here the order of the families corresponds to the order of the ancestors at time  $t - t/g(t, \varepsilon)$ . The key to the proof of this Lemma will be to show that

$$\mathbb{P}_{x}\left(\bigcup_{i=1}^{\hat{N}_{t-t/g(t,\varepsilon)}}\left\{\left|m_{i}-\frac{\langle f,\varphi\rangle}{\langle 1,\varphi\rangle}\right|>\varepsilon/2\right\}\right|N_{t}>0\right)\to0$$
(2.34)

as  $t \to \infty$ . The reason for this is that there are order  $g(t, \varepsilon)$  particles that have descendants at time t, and the probability that they are bad in the sense of (2.34) is less than  $p(t/g(t, \varepsilon))$  by definition. Then (2.33) provides the result.

To prove this rigorously however, it is more convenient to consider the unconditioned version of the probability in (2.34), noting that the event clearly does not occur if  $N_t = 0$ . To analyse this probability, we condition on the total collection of particles at time  $t - t/g(t, \varepsilon)$ . There are order 1 of these, uniformly in t by Markov's inequality (i.e. for any  $\delta > 0$  there exists a K such that  $\mathbb{P}_x(N_{t-t/g(t,\varepsilon)} > K) \leq \delta$  for all t.) Moreover, for any one of them, the probability that it

- (a) has a descendant at time t, and
- (b) the average value of f over all its descendants at time t is more than  $\varepsilon/2$  away from  $(f, \varphi)/(1, \varphi)$

is less than some constant times  $\frac{g(t,\varepsilon)}{t}p(t/g(t,\varepsilon),\varepsilon)$ . This follows from the definition of p and the asymptotic for the survival probability. Multiplying by t to account for the conditioning in (2.34) and applying (2.33) gives (2.34).

The upshot of (2.34) is that we now know, letting  $\sigma_i$  be the number of particles in the *i*th family at time *t*, that

$$\mathbb{P}_{x}\left(A_{t}|N_{t}>0\right) := \mathbb{P}_{x}\left(\bigcup_{i=1}^{\hat{N}_{t-t/g(t,\varepsilon)}} \left\{ \left|\frac{\sum_{j=1}^{\sigma_{1}+\dots+\sigma_{i}}f(X_{t}^{j})}{\sigma_{1}+\dots+\sigma_{i}} - \frac{\langle f,\varphi\rangle}{\langle 1,\varphi\rangle}\right| > \varepsilon/2 \right\} \middle| N_{t}>0\right) \to 0$$

$$(2.35)$$

as  $t \to \infty$  (observe that the families are clearly grouped together in the ordering of generation t.) This tells us that we have many "good" vertices along generation t, where the average value of f (considered over previously visited vertices) is close to what we want. To complete the proof, we must show that gaps between these good vertices, i.e. the lengths  $\sigma_i$ , are not too long.

To do this, we will prove that

$$\mathbb{P}_x\left(A_t'|N_t>0\right) := \mathbb{P}_x\left(\bigcup_{i=1}^{\hat{N}_{t-t/g(t,\varepsilon)}} \left\{\sigma_i > \frac{t}{(g(t,\varepsilon))^{1/3}}\right\} \middle| N_t>0\right) \to 0 \qquad (2.36)$$

as  $t \to \infty$ . For this we apply a similar argument to above. We consider the unconditioned probability and condition on the system at time  $t - t/g(t, \varepsilon)$ . By Markov's inequality there are order 1 particles at this time, uniformly in t, and for any one of them the probability that it

- (a) has a descendant at time t, and
- (b) the total number of descendants at time t is greater than  $t/(g(t,\varepsilon))^{1/3}$

is less than some constant times

$$\frac{g(t,\varepsilon)}{t} \times g(t,\varepsilon)^{-4/3}.$$

The first term in the product comes from the asymptotic for the survival probability. The second comes from the fact that, given survival of a process to time  $t/g(t,\varepsilon)$ , the total number of particles is roughly  $t/g(t,\varepsilon)$  times an exponential random variable by Theorem 2.6. We use Markov's inequality to get the explicit bound (uniformly in the starting point). Again multiplying by t to account for the conditioning gives (2.36).

Let us now show that  $\mathbb{P}_x(B_t(\rho)|N_t > 0) \to 0$  as  $t \to \infty$ . By the above work, and a union bound it is enough to show that

$$B_t(\rho) \subset A_t \cup A'_t$$

for all t large enough. Suppose we are on the event  $\{A_t \cup A'_t\}^c$ , and for every  $\rho t \leq M \leq N_t$ , set  $k(M) = \sigma_1 + \cdots + \sigma_i$ , where i is such that  $\sigma_1 + \cdots + \sigma_i \leq M \leq \sigma_1 + \cdots + \sigma_{i+1}$ . Then

$$\left|\frac{\sum_{i=1}^{k(M)} f(X_t^i)}{k(M)} - \frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle}\right| \le \varepsilon/2$$

for all  $\rho t \leq M \leq N_t$  simultaneously However, since we are on the event  $\{A'_t\}^c$  we also have

$$\left|\frac{\sum_{i=1}^{M} f(X_t^i)}{M} - \frac{\sum_{i=1}^{k(M)} f(X_t^i)}{k(M)}\right| \le 4\frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle} \frac{1}{\rho(g(t,\varepsilon))^{1/3}}.$$
(2.37)

Thus, for all t large enough, we must also be on the event  $\{B_t(\rho)\}^c$ .

**Remark 2.37.** The proof of Theorem 2.6 directly extends to the case of a general branching diffusion, by applying a generalised version of the Many-to-Few Lemma, see [HR15, Lemma 1]. Then Corollary 2.7 and Lemma 2.36 follow as above.

# 2.6 Convergence to the Brownian CRT

From this point onwards, we will assume that  $\varphi \in C^1(\overline{D})$  as in the statement of Theorem 2.1. In particular this means that  $|\nabla \varphi|$  is bounded on D.

Recall from Section 2.2.1 that we can consider our branching diffusions as continuous planar trees, where all the vertices are marked with a position. We write  $\mathcal{P}_x$  for the law of a sequence of i.i.d critical branching Brownian motion trees, each starting at  $x \in D$ . We will explore these trees in a *depth-first* order. This exploration is defined as follows:

- We start at the root of the first tree and move upwards (i.e. increasing height) at speed one. Whenever we reach a branching point we take the left branch.
- When we can no longer continue, we jump instantaneously to the most recent branching point that we have visited.
- We then repeat the process, starting to explore along the right branch emanating from this point.
- Whenever we can no longer continue, we jump instantaneously to the most recent branch point that we have visited, but not already jumped up to.
- When we reach the end of the first tree, we jump instantaneously to the root of the next tree and repeat.



Figure 21: A sketch of the depth-first exploration of a continuous tree. Full arrowed lines represent motion at speed one in the vertical direction. Dotted arrowed lines represent instantaneous jumps.

This is the analogue of the *lexicographical* ordering for discrete trees. Recording the height of vertices as we traverse the trees in this way gives us the *height process* 

 $H_t$  associated with the sequence. To show the convergence in Theorem 2.1 it will be important to show that this height process, when rescaled appropriately, looks like a reflected Brownian motion. To do this, we introduce a further process,  $S_t$ , which will turn out to be a martingale.

In the following, we will say that a vertex in the sequence of trees has been *visited* by time t if the exploration has passed through that point before time t. We will say that a vertex, that is also a branch point, has been *explored* by time t if it has been visited *and* jumped back to before time t. Recall that the mark associated with a vertex v corresponds to the spatial position of the particle it represents: we denote this by  $v^*$ . Finally, we write Y(v) for the set of branch points that have been visited but not explored before the time that v is visited.



Figure 22: The exploration up to time t. Branch points with filled circles have been visited and explored before time t. Branch points with empty circles have been visited but not explored. Note that the branch point furthest to the right has neither been visited nor explored before time t.  $Y(V_t)$  is the set of branch points with empty circles.

**Definition 2.38.** Let  $V_t$  be the vertex that is visited at time t in the depth-first exploration, writing  $V_t^*$  as usual for its position in D. We define for  $t \ge 0$ 

$$S_t = \varphi(V_t^*) + \sum_{v \in Y(V_t)} \varphi(v^*) - \Lambda_t \varphi(x).$$

Here,  $\Lambda_t$  is the index of the tree being visited at time t.

In fact,  $S_t$  is very closely related to the martingale  $M_t$ . Essentially they are the same process, but explored in different orders, and we will see that this is enough to preserve the martingale property. We would like to approximate the height process by  $S_t$ , and then apply an invariance principle for the martingale. This is an analogous idea to that used to prove convergence of Galton-Watson processes to the CRT in [LGD02], where  $S_t$  here plays the role of the Lukasiewicz path. We first record a

property of this process, which will be essential to showing a relationship with the height function:

**Lemma 2.39.** Let  $S'_t = S_t - \varphi(V_t^*)$  and  $I'_t = \inf_{0 \le s \le t} S'_s$ . Then

$$S'_t - I'_t = \sum_{v \in Y(V_t)} \varphi(v^*)$$

*Proof.* Since  $\varphi$  is positive, it is clear that  $I'_t = -\Lambda_t \varphi(x)$ . This implies the result.  $\Box$ 

**Definition 2.40.** The process on the right hand side in Lemma 2.39 also makes sense for a depth-first exploration of a single tree, and we denote such a process by  $\hat{S}$ . Note that this is a process that starts at 0 and is positive until the exploration of the tree is finished, i.e. an excursion.

We also write  $\hat{S}^y$  for  $\hat{S}$  conditioned to reach level y. Then  $\hat{S}^y$  is equal in law to  $S'_t - I'_t$ , restricted to the first excursion in which it exceeds y.

Observe that we can decompose S into continuous and discontinuous parts,  $S^c$ and  $S^d$ . Indeed, if we let  $S^d$  be the pure jump process that jumps up by  $\varphi(v^*)$  whenever the exploration reaches a branch point vertex v, then  $S^c = S - S^d$  is continuous. In fact,  $S^c$  is simply  $\varphi(V_t^*)$  minus a compensating sum that makes it continuous. If the exploration reaches the end of a branch at time t, then  $\lim_{s\uparrow t} \varphi(V_s^*) = 0$ , but  $\lim_{s\downarrow t} \varphi(V_s^*) > 0$ . It is easy to verify that  $S^c$  is just  $\varphi(V_t^*)$  with the jumps subtracted whenever they occur.

**Remark 2.41.** For a general diffusion and branching mechanism, as in the statement of Theorem 2.1, we define the process S in almost the same way, setting  $\varphi$ to be the first eigenfunction of the generator as usual. In this case we will only say that a branch point has been explored when all of the subtrees branching from the point have been partially explored. If a branch point has been visited but not explored before time t we let  $k_t(v)$  be the number of these subtrees that have not been explored at all before time t. We replace the sum  $\sum_{v \in Y(V_t)} \varphi(v^*)$  in the definition of S by  $\sum_{v \in Y(v_t)} k_t(v)\varphi(v^*)$ . Lemma 2.39 then holds with this sum on the right hand side. When we decompose S into its continuous and discontinuous parts, the jumps are given by  $(k-1)\varphi(v^*)$  whenever a branch point v with k branches is reached. The continuous part is again just  $\varphi(V_t^*)$  with the jumps removed.

We will return to this later, but let us now prove an invariance principle for  $S_t$ .

## 2.6.1 Martingale Convergence

**Lemma 2.42.** Under  $\mathcal{P}_x$ , for any  $x \in D$ ,  $(S_t)_{t\geq 0}$  is a locally square-integrable martingale with respect to the natural filtration generated by the depth-first exploration. Its predictable quadratic variation is given by

$$\langle S \rangle_t = \int_0^t \lambda \varphi(V_s^*)^2 + |\nabla \varphi(V_s^*)|^2 \, ds.$$

*Proof.* First observe that breaking up  $S = S^d + S^c$  gives us the expression for the predictable quadratic variation. The first term comes from the discontinuous part, which we know jumps, at rate  $\lambda$ , by  $\varphi$  applied to the position of the vertex being visited. The second comes from the fact that we can write  $dS_t^c = \varphi(V_t^*)dB_t$  for B a standard Brownian motion. This follows from the description of  $S_t^c$ , and the fact that the increments of  $(V_t^*)_{t\geq 0}$  are equal in law to those of a Brownian motion.

To see that S is a martingale, we condition on the depth first exploration up to time s and notice that we can write

$$\mathcal{E}_x\left[S_t - S_s|\mathcal{F}_s\right] = \mathbb{E}_{V_s^*}[\hat{S}_\tau - \varphi(V_s^*)] + \sum_{v \in Y(V_s)} \mathbb{E}_{v^*}[\hat{S}_{v,\tau_v} - \varphi(v^*)] + \sum_{i=1}^{\infty} \mathbb{E}_x[\hat{S}_{i,\tau_i} - \varphi(x)]$$

by the Markov property, where  $\hat{S}_{\tau}$ ,  $(\hat{S}_{v,\tau_v})_{v\in Y(V_s)}$  and  $(\hat{S}_{i,\tau_i})_{i\geq 1}$  are the processes  $\hat{S}$  for the subtrees rooted at  $v_s$ ,  $\{v\}_{v\in Y(V_s)}$  and  $\{v_i\}_{i\geq 1}$  (the roots of the remaining sequence of trees) respectively, each run up to a stopping time that *does not depend* on that subtree by conditional independence. Thus, to prove the martingale property it is enough to show that

$$\mathbb{E}_x[\hat{S}_t] = \varphi(x)$$

for any  $x \in D$  and  $t \ge 0$ .

To do this, we will approximate  $\hat{S}$  by a discrete version  $(\hat{S}_n^{\delta}, n \in \mathbb{N})$ , with  $\delta \downarrow 0$ . This process will be defined by discretising the tree using steps of size  $\delta$  in the natural way. This results in a discrete tree where every vertex is marked with a position - corresponding to its spatial position in the original tree. To define  $\hat{S}^{\delta}$ , set  $\hat{S}_0^{\delta} = \varphi(x)$  (where x is the starting position) and traverse the discrete tree in a depth-first, or lexicographical, order. If you are visiting a vertex with position y at step n, and it has children with positions  $(z_j)_{1\leq j\leq J}$ , then set  $\hat{S}_{n+1}^{\delta} - \hat{S}_n^{\delta} = \sum_{j=1}^J \varphi(z_j) - \varphi(y)$ . By considering the martingale M, it is clear that  $\mathbb{E}_x[\hat{S}_n^{\delta}] = \varphi(x)$  for every n and  $\delta$ . Moreover, for fixed  $t \geq 0$  we have that  $\hat{S}_{\lfloor t/\delta \rfloor}^{\delta} \to \hat{S}_t$  almost surely as  $\delta \to 0$ . This is clear after noting that:

- Almost surely the discrete tree only captures single branching events in any step, for all  $\delta$  small enough.
- If vertex being visited at time  $\lfloor t/\delta \rfloor$  in the discrete process corresponds to the vertex at time u in the continuous process, then  $|u-t| \leq \delta(1+W_t)$ , where  $W_t$  is

the total number of deaths plus branch points before time t in the exploration.

• The particle motion is almost-surely continuous.

Since the  $\hat{S}^{\delta}_{\lfloor t/\delta \rfloor}$  are dominated, for example by  $\|\varphi\|_{\infty}$  times the number of branch points in the continuous tree up to level t, the result follows by dominated convergence.

**Remark 2.43.** If we have a generator  $L = -\frac{1}{2} \sum_{i,j} a^{ij} \partial_{x_i x_j} + \sum_i b^i \partial_{x_i}$  as in Theorem 2.1 and an offspring distribution A with finite variance, then the same argument implies that S is a martingale. Letting  $\lambda, \varphi$  be the first eigenvalue/eigenfunction pair for -L we can calculate that the predictable quadratic variation of S is given by

$$\langle S \rangle_t = \int_0^t \frac{\lambda}{m-1} \mathbb{E}[A^2 - A] \varphi(V_s^*)^2 - \lambda \varphi(V_s^*)^2 + \sum_{i,j} \frac{\partial \varphi}{\partial x_i} \frac{\partial \varphi}{\partial x_j} a^{i,j}(V_s^*) \, ds$$

The second term here comes from the single particle motion under L and the first comes from the jumps (recall the critical branching rate in this case is  $\lambda/(m-1)$ ).

Proposition 2.44. Let

$$\sigma^2 = \frac{2\lambda \int_D \varphi(y)^3 \, dy}{\langle 1, \varphi \rangle}.$$

Then

$$\left(\frac{S_{nt}}{\sqrt{n}}\right)_{t\geq 0} \to (\sigma B_t)_{t\geq 0}$$

in distribution as  $n \to \infty$ , with respect to the Skorohod topology.

*Proof.* Since the jumps of S are bounded, this follows from the functional central limit theorem for martingales [JS87, Theorem 3.22, Chapter VIII] once we can show that for all  $t \ge 0$ 

$$\langle S^n \rangle_t \to \sigma t$$

in probability as  $n \to \infty$ . Here  $(S_t^n)_{t>0} = (S_{nt}/\sqrt{n})_{t>0}$ . However, we can write

$$\frac{\langle S^n \rangle_t}{t} = \frac{1}{nt} \int_0^{nt} \lambda \varphi(V_s^*)^2 + |\nabla \varphi(V_s^*)|^2 \, ds.$$

Then, since  $\varphi$  and  $|\nabla \varphi|$  are bounded, this follows immediately from Proposition 2.45 below. We recover  $\sigma^2$  by integrating by parts to see that  $\langle \lambda \varphi(\cdot)^2 + |\nabla \varphi(\cdot)|^2, \varphi \rangle = 2\lambda \int_D \varphi(y)^3 dy.$ 

**Proposition 2.45.** Suppose that f is a bounded, measurable function and let  $(V_s^*)_{0 \le s \le t}$  be the positions of vertices visited in the depth-first exploration before

time t as usual. Then

$$Q_t := \frac{1}{t} \int_0^t f(V_s^*) \, ds \to \frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle}$$

in  $\mathcal{P}_x$ -probability as  $t \to \infty$ .

**Remark 2.46.** In the general case (assuming Proposition 2.45) we can integrate by parts, using that  $\varphi$  is an eigenvector of L and L is self-adjoint, to see that  $\langle S^n \rangle_t \to \sigma t$  as  $n \to \infty$ , where

$$\sigma^2 = \frac{\lambda \mathbb{E}[A^2 - A] \int_D \varphi(y)^3 \, dy}{(m-1) \langle 1, \varphi \rangle}.$$

Since the jumps are not necessarily bounded in this case, to apply the FCLT for martingales we need to verify a Lindenberg-Feller type condition (for example, [JS87, (3.23), Theorem 3.22, Chapter VIII].) However, this follows immediately from the fact that we are assuming the offspring distribution to have finite variance. Then Proposition 2.44 holds.

Before we prove Proposition 2.45, let us record some of the consequences of Proposition 2.44.

**Proposition 2.47.** Let  $\Lambda_t$  be the index of the tree being visited in the exploration at time t (so  $\Lambda_0 = 1$ ). Then we have the joint convergence

$$\left(\frac{S'_{nt} - I'_{nt}}{\sqrt{n}}, \frac{\Lambda_{nt}}{\sqrt{n}}\right)_{t \ge 0} \to \left(\sigma |\beta_t|, \frac{\sigma}{\varphi(x)} L^0_t(\beta)\right)_{t \ge 0}$$
(2.38)

as  $n \to \infty$ , in distribution with respect to the Skorohod topology. Here,  $\beta$  is a standard Brownian motion started at 0 and  $L_t^0(\beta)$  is the local time of  $\beta$  at 0. Furthermore, for any y > 0, we have

$$\left(\frac{\hat{S}_{nt}^{y\sqrt{n}}}{\sqrt{n}}\right)_{t\geq 0} \xrightarrow[n\to\infty]{(d)} \left(\sigma e_t^{y/\sigma}\right)_{t\geq 0}$$
(2.39)

where  $e^{y/\sigma}$  is a Brownian excursion conditioned to reach height  $y/\sigma$ .

*Proof.* From the definition of S', we have that  $|S'_{nt}/\sqrt{n} - S_{nt}/\sqrt{n}| \le ||\varphi||_{\infty}/\sqrt{n}$  for all  $t \ge 0$  and so Proposition 2.44 implies that

$$\left(\frac{S'_{nt}}{\sqrt{n}}\right)_{t\geq 0} \to (\sigma B_t)_{t\geq 0}$$

as  $n \to \infty$  as well. Writing  $\underline{B}_t = \inf_{0 \le s \le t} B_s$  this implies the joint convergence

$$\left(\frac{S'_{nt} - I'_{nt}}{\sqrt{n}}, -\frac{I'_{nt}}{\sqrt{n}}\right)_{t \ge 0} \xrightarrow[n \to \infty]{(d)} (\sigma(B_t - \underline{B}_t), -\sigma\underline{B}_t)_{t \ge 0}$$
(2.40)

where the right-hand side by Lévy's theorem, see for example [RY91, Chapter VI, Theorem VI.2.3], is equal in distribution to

$$\left(\sigma|\beta_t|, \sigma L^0_t(\beta)\right)_{t>0}$$

However, we know that  $\Lambda_t = -I'_t/\varphi(x)$ , and so (2.38) follows. For the second claim of the Proposition, we follow [LGD02, Proposition 2.5.2]. It is well known that you can construct the process  $e^{y/\sigma}$  from a standard Brownian motion  $\beta$  by taking

$$e_t^{y/\sigma} = |\beta_{(G+t)\wedge D}|$$

where  $T = \inf\{t \ge 0 : |\beta_t| \ge y/\sigma\}$ ,  $G = \sup\{t \le T : \beta_t = 0\}$  and  $D = \inf\{t \ge T : \beta_t = 0\}$ . By the Skorohod representation theorem and (2.38) we also know that there exists a process

$$\left(Z_t^{(n)}, \Lambda_t^{(n)}\right)_{t \ge 0} \stackrel{(d)}{=} \left(\frac{S_{nt}' - I_{nt}'}{\sqrt{n}}, \frac{\Lambda_{nt}}{\sqrt{n}}\right)_{t \ge 0}$$

such that

$$\left(Z_t^{(n)}, \Lambda_t^{(n)}\right)_{t \ge 0} \xrightarrow[n \to \infty]{} \left(\sigma |\beta_t|, \frac{\sigma}{\varphi(x)} L_t^0(\beta)\right)_{t \ge 0}$$

uniformly on every compact set almost surely. This is because Skorohod convergence is equivalent to local uniform convergence when the limit is continuous. Define  $T^{(n)} = \inf\{t \ge 0 : Z^{(n)} \ge y\}$  for this sequence of processes, and  $G^{(n)}$ ,  $D^{(n)}$  in the same way as G and D above. By the remark in Definition 2.40 and (2.38), if we can prove that  $G^{(n)} \to G$  and  $D^{(n)} \to D$  almost surely, we will be done. First note that since  $\beta$  must exceed  $x/\sigma$  immediately after time T, we have that  $T^{(n)} \to T$  almost surely. This implies straight away that for all t < D we have  $t \le D^{(n)}$  for all n large enough almost surely. Now we must show that for all t > D we have  $t \ge D^{(n)}$  for all n large enough almost surely. These facts together (along with the corresponding results for G) are enough to prove the convergence. To see the final claim, we use the convergence of the local time. For any t > D, we have using basic properties of Brownian local time that  $L_t^0 > L_D^0 = L_T^0$ . The convergence of the local time therefore tells us that  $\Lambda_t^{(n)} > \frac{\sigma}{\varphi(x)} L_T^0$  for all n large enough almost surely. Since

$$\Lambda_{T^{(n)}}^{(n)} \to \frac{\sigma}{\varphi(x)} L_T^0$$

almost surely, this implies that we have also have  $\Lambda_t^{(n)} > \Lambda_{T^{(n)}}^{(n)}$  for all *n* large enough almost surely. Using the fact that  $\Lambda^{(n)}$  stays constant on  $[T^{(n)}, D^{(n)})$ , we see that  $t \ge D^{(n)}$ .

**Remark 2.48.** The above proof is not affected by changing the generator or offspring distribution, since it relies only on the convergence from Proposition 2.44.

The rest of this subsection will be devoted to the proof of Proposition 2.45, but we will first need some preliminary estimates. We write h(v) for the height of a vertex v in the the exploration.

**Lemma 2.49.** Let L be the total length of a branching Brownian motion process (i.e. how long it takes to traverse the tree in depth-first order). Then for all  $x \in D$ 

$$\mathbb{P}_x(L > t) \gtrsim \frac{\varphi(x)}{\sqrt{t}}.$$

Let  $U_t$  be a vertex picked uniformly at random from those visited before time t in the depth-first exploration under  $\mathcal{P}_x$ . Write  $U_t^*$  as usual for the position in D corresponding to its mark. Finally, denote by  $\overline{\mathcal{P}}_x$  the law on the sequence of trees plus  $U_t$ . Then we have the following control on the height of the uniform vertex.

Lemma 2.50. For all  $x \in D$ 

$$\overline{\mathcal{P}}_x\left(h(U_t) \ge C\sqrt{t}\right) \to 0$$

as  $C \to \infty$ , uniformly in t.

We will now show how we may deduce Proposition 2.45, and then go on to prove the Lemmas. Let  $N_s^i$  be the number of particles at level s in the *i*th tree of our exploration. Also write  $(X_s^{i,j})_{1 \le j \le N_s^i}$  for the positions of these particles. Finally, let I(t) be the index of the tree in which the uniform vertex lies.

Proof of Proposition 2.45. We first show that  $\mathcal{E}_x[Q_t] \to \frac{\langle f, \varphi \rangle}{\langle 1, \varphi \rangle}$  as  $t \to \infty$ . Let  $m_t$  be the average value of f among the vertices at height  $h(u_t)$  of the I(t)th tree, that have been visited before time t. Observe that by conditioning on  $m_t$ , we have

$$\mathcal{E}_x[Q_t] = \overline{\mathcal{E}}_x[f(U_t^*)] = \overline{\mathcal{E}}_x[m_t]$$

as  $t \to \infty$ . This is because, given the positions of these particles, we know that  $u_t$  is chosen uniformly from them. We will aim to show that for fixed  $\varepsilon > 0$ ,

$$\overline{\mathcal{P}}_{x}\left(A_{t}\right) := \overline{\mathcal{P}}_{x}\left(\left|m_{t} - \frac{\langle \varphi, f \rangle}{\langle \varphi, 1 \rangle}\right| > \varepsilon\right) \to 0$$
(2.41)

as  $t \to \infty$ . Then since  $m_t$  is bounded, the convergence in expectation will follow.

For the *i*th tree in our exploration and s > 0, let  $m_{s,t}^i$  be the average value of f among the vertices of tree *i* at height *s* that are visited before time *t*, and  $\tilde{N}_{s,t}^i$  be the number of such vertices. Also write  $A_{s,t}^i$  for the event that  $|m_{s,t}^i - \langle \varphi, f \rangle / \langle \varphi, 1 \rangle| > \varepsilon$ . Then, conditioning on the entire sequence of trees, we can write our probability as

$$\frac{1}{t}\mathcal{E}_x\left[\int_0^\infty \sum_{i=1}^\infty \mathbbm{1}_{\{i \le \Lambda_t\}} \mathbbm{1}_{A^i_{s,t}} \tilde{N^i_{s,t}} \, ds, \right]$$

since the probability of picking a vertex in tree i at height ds is  $\tilde{N}_s^i/t \, ds$ . Note that we can interchange the integral, expectation and sum as we like here by Fubini (the expression being bounded by 1.) Now, given  $\delta > 0$ , by Lemma 2.50 we can choose C such that  $\mathbb{P}(h(U_t) \ge C\sqrt{t}) < \frac{\delta}{3}$  for all t. Similarly we can define R(t) > 0 such that  $\overline{\mathcal{P}}_x(h(U_t) \le R(t)) = \frac{\delta}{3}$ . Note that  $R(t) \to \infty$  as  $t \to \infty$  by the law of large numbers. Indeed, for any K > 0 the proportion of time spent below height K in the exploration is less than or equal to  $\sum_{i=1}^{\Lambda_t} p_K^i/t$  where  $p_K^i$  is the time spent below level K in the *i*th tree. We also have that t is greater than or equal to  $\sum_{i=1}^{\Lambda_t-1} L^i$ where  $L^i$  is the length of the *i*th tree. However,  $p_K^i$  has finite variance and  $L^i$  does not, and so the strong law of large numbers allows us to conclude.

Since  $\mathbb{1}_{A^{s,t}} \leq 1$  this tells us that

$$\overline{\mathcal{P}}_{x}\left(A_{t}\right) \leq \frac{2\delta}{3} + \frac{1}{t}\mathcal{E}_{x}\left[\int_{R(t)}^{C\sqrt{t}}\sum_{i=1}^{\infty}\mathbb{1}_{\left\{i\leq\Lambda_{t}\right\}}\mathbb{1}_{A_{s,t}^{i}}\tilde{N}_{s}^{i}\,ds\right]$$
(2.42)

where by Fubini we can rewrite the expectation term as

$$\frac{1}{t} \int_{R(t)}^{C\sqrt{t}} \sum_{i=1}^{\infty} \mathcal{E}_x \left[ \mathbbm{1}_{\{i \le \Lambda_t\}} \mathbbm{1}_{A^i_{s,t}} \tilde{N}^i_{s,t} \right] ds.$$
(2.43)

Now for each *i*, we condition on  $\mathcal{G}_{i-1}$ : the  $\sigma$ -algebra generated by the first i-1 trees. Note that  $\mathcal{G}_{i-1}$  is independent of the *i*th tree. Moreover, the event  $\{i \leq \Lambda_t\}$  and the start time  $\tau_i$  of the *i*th tree, are measurable with respect to  $\mathcal{G}_{i-1}$ . Thus we can write

$$\mathcal{E}_{x}\left[\mathbbm{1}_{\{i\leq\Lambda_{t}\}}\mathbbm{1}_{A_{s,t}^{i}}\tilde{N}_{s,t}^{i}\middle|\mathcal{G}_{i-1}\right]=\mathbbm{1}_{\{i\leq\Lambda_{t}\}}\mathbbm{E}_{x}\left[\mathbbm{1}_{A_{s}(t-\tau_{i})}N_{s}(t-\tau_{i})\middle|N_{s}>0\right]\mathbbm{P}_{x}(N_{s}>0)$$

where the expectation is now with respect to a single branching Brownian motion tree,  $N_s(r)$  is the number of particles at level *s* before time *r* in a depth-first exploration of the tree, and  $A_s(r)$  is the event that the average of *f* among these particles is more than  $\varepsilon$  away from  $\langle \varphi, f \rangle / \langle \varphi, 1 \rangle$ .

We note here, by Theorems 2.4 and 2.6, that there exists a K such that

$$\mathbb{E}_x \left[ \left| N_s^2 \right| N_s > 0 \right]^{1/2} \le Ks, \quad \mathbb{P}_x(N_s > 0) \le K/s \tag{2.44}$$

for all s. We can also, by Lemma 2.49, choose this K such that

$$\mathcal{E}_x\left[\Lambda_t\right] \le K\sqrt{t} \tag{2.45}$$

for all t. Indeed,  $\mathcal{E}_x[\Lambda_t]$  is less than the expectation of geometric random variable, whose success rate is  $\mathbb{P}_x(L > t) \gtrsim 1/\sqrt{t}$ . Decomposing on whether or not  $N_s(t - \tau_i)$ is bigger than  $s\delta/6CK^2$  we have

$$\mathbb{E}_{x} \left[ \mathbb{1}_{A_{s}(t-\tau_{i})} N_{s}(t-\tau_{i}) \middle| N_{s} > 0 \right] \leq \frac{\delta s}{6CK^{2}} + \mathbb{E}_{x} \left[ \mathbb{1}_{B_{s}(\delta/6CK^{2})} N_{s}(t-\tau_{i}) \middle| N_{s} > 0 \right] \\ \leq \frac{\delta s}{6CK^{2}} + \mathbb{E}_{x} \left[ N_{s}^{2} \middle| N_{s} > 0 \right]^{1/2} \mathbb{P}_{x} \left( B_{s}(\delta/6CK^{2}) \middle| N_{s} > 0 \right)^{1/2}$$

where  $B_s(\cdot)$  is the event from Lemma 2.36. Using (2.44), and the conditioning above, we have that

$$\mathcal{E}_{x}\left[\mathbb{1}_{\{i\leq\Lambda_{t}\}}\mathbb{1}_{A_{s,t}^{i}}\tilde{N}_{s,t}^{i}\right]\leq\left(\frac{\delta}{6CK}+K^{2}\mathbb{P}_{x}\left(B_{s}(\delta/6CK^{2})\big|N_{s}>0\right)^{1/2}\right)\mathcal{E}_{x}[\{i\leq\Lambda_{t}\}].$$
(2.46)

This means that the expression in (2.43) is less than

$$\left(\frac{\delta}{6CK} + K^2 \sup_{s \ge R(t)} \left\{ \mathbb{P}_x \left( B_s(\delta/6K^2) \middle| N_s > 0 \right)^{1/2} \right\} \right) \frac{1}{t} \int_{R(t)}^{C\sqrt{t}} \sum_{i=1}^{\infty} \mathcal{E}_x \left[ \mathbbm{1}_{\{i \le \Lambda_t\}} \right].$$

However, by Fubini and (2.45), this is less than

$$\frac{\delta}{6} + CK^3 \sup_{s \ge R(t)} \left\{ \mathbb{P}_x \left( \left. B_s(\delta/6K^2) \right| N_s > 0 \right)^{1/2} \right\}.$$

Using Lemma 2.36, and the fact that  $R(t) \to \infty$ , we see than this is less than  $\frac{\delta}{3}$  for all t large enough. Substituting in to (2.42) proves (3.29).

To complete the proof of the Proposition, we must also show that

$$\mathcal{E}_x[Q_t^2] \to \frac{\langle f, \varphi \rangle^2}{\langle 1, \varphi \rangle^2}$$

as  $t \to \infty$ . However, letting  $U_t^1$  and  $U_t^2$  be two vertices picked independently uniformly from those visited before time t (again denoting the extended law by  $\overline{\mathcal{E}}_x$ ) we see that

$$\mathcal{E}_x\left[Q_t^2\right] = \overline{\mathcal{E}}_x\left[f(U_t^{1*})f(U_t^{2*})\right].$$

Let  $m_t^i$  and  $A_t^i$  correspond to  $m_t$  and  $A_t$  for  $U_t^i$ , i = 1, 2. Then the same reasoning as above tells us that

$$\overline{\mathcal{E}}_x\left[f(U_t^{1*})f(U_t^{2*})\right] = \overline{\mathcal{E}}_x\left[m_t^1m_t^2\right]$$

and we also have, by a union bound, that

$$\overline{\mathcal{P}}_x\left[A_t^1 \cup A_t^2\right] \to 0$$

as  $t \to \infty$ . The result follows in exactly the same way.

*Proof of Lemma 2.49.* We will prove this result using our asymptotic for the survival probability. Since we have

$$\mathbb{P}_x(L > t) \ge \mathbb{P}_x(L > t \mid N_{\sqrt{at}} > 0) \mathbb{P}_x(N_{\sqrt{at}} > 0)$$
(2.47)

and we know that the probability of survival until time  $\sqrt{at}$  decays like  $\varphi(x)/\sqrt{at}$ , it is enough to show there exists an *a* such that  $\mathbb{P}_x(L > t|N_{\sqrt{at}} > 0)$  is bounded below, uniformly in *t*. In fact, we will prove a slightly stronger statement, as it will also be of use later on. We will show that

$$\mathbb{P}_x(L \le cs^2 | N_s > 0) \to 0 \tag{2.48}$$

as  $c \to 0$ , uniformly in s. Clearly this is enough to prove the Lemma.

To do this, we first fix some compact subdomain  $D' \subset D$ . Then, we condition on the positions  $X_s^i$  of the particles at time *s* that lie within D', and consider the subprocesses continuing from these points. In particular, we consider the contributions that each subprocess makes to *L* from its first  $\mu s$  generations. Note that, given the positions  $X_s^i$ , these are independent random variables, with distribution equal to that of  $\int_0^{\mu s} N_u du$  for a branching Brownian motion started at  $X_s^i$ . This means that they all have mean  $\geq C\mu s$  and variance  $\leq C'\mu^3 s^3$  for some fixed C, C' > 0. The statement concerning the mean follows from the expression for  $\mathbb{E}_y[N_u]$ , which is bounded below by Lemma 2.33 for all *y* in the subdomain and all u > 0 by some *C*. For the variance, note that for r < u

$$\mathbb{E}_{y}[N_{r}N_{u}] = \mathbb{E}_{y}\left[N_{r}\mathbb{E}_{x}\left[N_{u}|\mathcal{F}_{r}\right]\right] \leq C''\mathbb{E}_{y}[N_{r}^{2}]$$

for some C'' > 0 (not depending on r, u or y). Since  $[\mathbb{E}_y[N_r^2] \lesssim r$  uniformly in y, the claim follows by integration.

By Theorem 2.6 we also know that for any  $\delta > 0$  there exist m and S, such that the probability of having less than ms vertices at time s in D' (conditioned on survival) is less than  $\delta$  for all  $s \geq S$ . On the complementary event, conditioned on the tree up to time s, we have ms independent random variables with mean  $\geq C\mu s$  and variance  $\leq C'\mu^3 s^3$ . By the standard application of Markov's inequality to sums of independent random variables, the probability that their sum is more than  $m\mu Cs^2/2$  away from its mean is less than  $4C'\mu/C^2m$  for all  $s \geq S$ . Since the mean is greater than  $m\mu Cs^2$ , this has to occur for the sum to be less than  $m\mu Cs^2/2$ . Taking  $\mu$  to 0, and noting that L is greater than this sum, shows that the left hand side of (2.48) can be made less than  $\delta$  as long as c is small enough and s is large enough. In fact, we can choose c small enough that this will hold for all s, due to the continuity in s of the probability in (2.48) (which you can prove by dominated convergence.) Thus the claim is proved.

Proof of Lemma 2.50. As explained in the proof of Proposition 2.45, Lemma 2.49 immediately implies that  $\mathcal{E}_x[\Lambda_t] \leq \sqrt{t}$ . Then using our asymptotic for the survival probability, we have

$$\overline{\mathcal{P}}_x\left(h(U_t) \ge C\sqrt{t}\right) \le \mathcal{E}_x\left[\sum_{i=1}^\infty \mathbb{1}_{\{i \le \Lambda_t\}} \mathbb{1}_{\{N_C^i > 0\}}\right] \lesssim \frac{1}{C\sqrt{t}} \mathcal{E}_x[\Lambda_t]$$

where the implied constant does not depend on C, x, t.

**Remark 2.51.** The proofs of Proposition 2.45 and Lemmas 2.49 and 2.50 do not need any adaptation for the more general set up.

#### 2.6.2 Connection with the height function

In order to make use of the above invariance principle, we must connect the martingale with the height function of our trees. In this section we will look at the height function, and the process  $\hat{S}$  from Definition 2.40, for the depth-first exploration of a single tree conditioned to be large. We know by Lemma 2.39 that the value of the  $\hat{S}$  when it visits a vertex v in the tree is equal to

$$\hat{S}(v) := \sum_{u \in Y(v)} \varphi(u^*).$$

We will show that for vertices with large heights, this sum is close to a constant times the height. Our approach will use an ergodicity property for the spine particle in the system under  $\overline{\mathbb{Q}}_x$ , and is inspired from [HR14].

In the following, given  $\eta > 0$ , we will say that a vertex v in a branching Brownian motion tree is  $\eta$ -bad if

$$\left|\frac{\hat{S}(v)}{h(v)} - \lambda \int_{D} \varphi(y)^{3} \, dy\right| > \eta.$$
(2.49)

We also say, for given  $T \ge 0$ , that a vertex v is  $\eta_T$ -bad if some ancestor of the vertex v at height greater than T is  $\eta$ -bad. Then we have the following estimate for the proportion of  $\eta_T$ -bad vertices:

**Proposition 2.52.** Fix  $\varepsilon, \eta > 0$  and write  $N_t^{\eta_T}$  for the collection of  $\eta_T$ -bad vertices at time t. Then we have

$$\mathbb{P}_x\left(\left.\frac{N_t^{\eta_T}}{N_t} > \varepsilon \right| N_t > 0\right) \to 0 \tag{2.50}$$

as  $T \to \infty$ , uniformly in  $t \ge T$ , for any  $x \in D$ .

By this we mean that for any  $x \in D$ , given any  $\delta > 0$ , there exists T' large enough that  $\mathbb{P}_x(N_t^{\eta_T}/N_t > \varepsilon | N_t > 0) \le \delta$  for all  $t \ge T \ge T'$ .

*Proof.* We will first show that for any  $\varepsilon > 0$ 

$$\mathbb{P}_x\left(E_{T,t}^{\varepsilon}|N_t>0\right) := \mathbb{P}_x\left(\frac{\sum_{i=1}^{N_t}\varphi(X_t^i)\mathbb{1}_{\{X_t^i|\eta_T-\text{bad}\}}}{\sum_{i=1}^{N_t}\varphi(X_t^i)} > \varepsilon \,\middle|\, N_t>0\right) \to 0$$

as  $T \to \infty$ , uniformly in t. To do this, we will use the spine decomposition given by Lemma 2.22. Recalling the definition of  $\mathbb{Q}_x$  from this section we see that the above probability is equal to

$$\mathbb{Q}_x\left[\frac{\varphi(x)/\mathbb{P}_x(N_t>0)}{\sum_{i=1}^{N_t}\varphi(X_t^i)}\mathbb{1}_{E_{T,t}^\varepsilon}\right] := \mathbb{Q}_x\left[Y_t\mathbb{1}_{E_{T,t}^\varepsilon}\right].$$

To see that this converges to 0 it is enough to prove that:

- $\mathbb{Q}_x(E_{T,t}^{\varepsilon}) \to 0$  as  $T \to \infty$ , uniformly in  $t \ge T$  and
- For every  $\delta > 0$ , there exists T' and K positive, such that  $\mathbb{Q}\left(Y_t \mathbb{1}_{|Y_t|>K}\right) \leq \delta$  for all  $t \geq T'$ .

The first point comes from the fact that the *spine particle* under this new law is unlikely to be  $\eta_T$ -bad for large T. More precisely, recall that  $\overline{\mathbb{Q}}_x$  is a law on branching processes equipped with a distinguished path, the *spine*, such that

$$\mathbb{Q}_x = \overline{\mathbb{Q}}_x \big|_{\mathcal{F}_t}$$

for  $\mathcal{F}_t$  the filtration generated by the process but not the distinguished path. Under  $\overline{\mathbb{Q}}_x$ , we know that the spine particles evolves as a Brownian motion conditioned to remain in the domain for all time, and branches at constant rate  $2\lambda$ , where each branch is either to the left or right of the spine (i.e. comes before or after in the depth-first ordering) with equal probability. Due to the mixing of this Brownian motion to a stationary distribution with density  $\varphi^2$ , see Lemma 2.24, ergodicity tells us that the probability of the spine vertex being  $\eta_T$ -bad at time t converges to 0 as  $T \to \infty$ , uniformly in  $t \geq T$ .

The connection between the motion of the spine and the event  $E_{T,t}^{\varepsilon}$  comes from the fact that under  $\overline{\mathbb{Q}}_x$ , conditioned on  $\mathcal{F}_t$ , the spine particle is chosen proportionally to  $\varphi$ , see (2.11). Indeed, since

$$\sum_{i=1}^{N_t} \varphi(X_t^i) \mathbb{1}_{\{X_t^i \eta_T - \mathrm{bad}\}} / \sum_{i=1}^{N_t} \varphi(X_t^i)$$

is positive, it is enough, for the first point, to show that its  $\mathbb{Q}_x$  expectation converges to 0 (uniformly in  $t \ge T$  as  $T \to \infty$ ). However, this follows directly from the above and (2.11) since

$$\frac{\sum_{i=1}^{N_t} \varphi(X_t^i) \mathbb{1}_{\{X_t^i \eta_T - \text{bad}\}}}{\sum_{i=1}^{N_t} \varphi(X_t^i)} = \overline{\mathbb{Q}}_x \text{ (spine } \eta_T \text{-bad at time } t | \mathcal{F}_t)$$

The second point essentially says that  $(Y_t)_{t\geq 0}$  is  $\mathbb{Q}_x$  uniformly integrable. To prove it, one can use the change of measure between  $\mathbb{Q}_x$  and  $\mathbb{P}_x$  again to write

$$\mathbb{Q}_x\left[Y_t \mathbb{1}_{\{|Y_t|>K\}}\right] = \frac{\mathbb{P}_x\left(\{|Y_t|>K\} \cap \{N_t>0\}\right)}{\mathbb{P}_x(N_t>0)} = \mathbb{P}_x\left(|Y_t|>K|N_t>0\right).$$

Since  $\varphi(x)/t\mathbb{P}_x(N_t > 0)$  is uniformly bounded above for  $t \ge 1$  say, by Theorem 2.4, we just need to show that for any  $\delta > 0$  there exists K and T' such that

$$\sup_{t \ge T'} \mathbb{P}_x \left( \frac{\sum_{i=1}^{N_t} \varphi(X_t^i)}{t} < 1/K \middle| N_t > 0 \right) \le \delta.$$
(2.51)

However, this is a direct consequence of the convergence given by Theorem 2.5, since we know that for fixed K the probability above converges, as  $t \to \infty$ , to the probability that an exponential random variable is less than 1/K.

We must now deduce that

$$\mathbb{P}_x\left(\left.\frac{N_t^{\eta_T}}{N_t} > \varepsilon \right| N_t > 0\right) \to 0$$

uniformly in  $t \geq T$  as  $T \to \infty$  from the fact that

$$\mathbb{P}_x(E_{T,t}^{\varepsilon}|N_t > 0) \to 0.$$

The idea behind this is that  $\sum_{i=1}^{N_t} \varphi(X_t^i) \mathbb{1}_{\{X_t^i \eta_T - \text{bad}\}} / \sum_{i=1}^{N_t} \varphi(X_t^i)$  is a reasonable approximation to  $N_t^{\eta_T} / N_t$  on survival at large times. By Corollary 2.7, we know that for any  $\delta > 0$  there exist r and T' positive such that

$$\sup_{t \ge T'} \mathbb{P}_x \left( \left. \frac{N_t^{D_r}}{N_t} > \frac{\varepsilon}{2} \right| N_t > 0 \right) \le \delta/2$$
(2.52)

where  $D_r = \{y \in D : \varphi(y) < 1/r\}$  and  $N_t^{D_r}$  is the number of particles in  $D_r$  at time t. Also, write  $N_t^{D_r^c,\eta_T}$  for the number of particles that are  $\eta_T$ -bad and lie in  $D_r^c$  at time t. Then

$$\frac{\sum_{i=1}^{N_t} \varphi(X_t^i)}{N_t} \le \|\varphi\|_{\infty} \quad \text{and} \quad \frac{N_t^{D_r^c,\eta_T}}{\sum_{i=1}^{N_t} \varphi(X_t^i) \mathbbm{1}_{\{X_t^i|\eta_T-\text{bad}\}}} \le r.$$

Bounding  $N_t^{\eta_T}$  above by  $N_t^{D_r} + N_t^{D_r^c,\eta_T}$ , and choosing  $T \geq T'$  large enough that  $\mathbb{P}_x(E_{T,t}^{(2||\varphi||_{\infty}r)^{-1}\varepsilon}) \leq \delta/2$ , we have that

$$\mathbb{P}_x\left(\left.\frac{N_t^\eta}{N_t} > \varepsilon \right| N_t > 0\right) \le \delta$$

for all  $t \geq T$ .

**Remark 2.53.** We can extend the proof to show that for any c > 0 and  $x \in D$ ,

$$\mathbb{P}_x\left(\left.\left\{\frac{N_{ct}^{\eta_T}}{N_{ct}} > \varepsilon\right\} \cap \left.\left\{N_{ct} > 0\right\}\right| N_t > 0\right) \to 0 \tag{2.53}$$

uniformly in  $ct \wedge t \geq T$  as  $T \to \infty$ . This follows immediately from (2.50), since

$$\begin{split} \mathbb{P}_x \left( \left\{ \frac{N_{ct}^{\eta_T}}{N_{ct}} > \varepsilon \right\} \,\cap\, \left\{ N_{ct} > 0 \right\} \middle| N_t > 0 \right) &= \mathbb{P}_x \left( \left\{ \frac{N_{ct}^{\eta_T}}{N_{ct}} > \varepsilon \right\} \,\cap\, \left\{ N_{ct} > 0 \right\} \middle| N_{ct} > 0 \right) \\ &\times \frac{\mathbb{P}_x (N_{ct} > 0)}{\mathbb{P}_x (N_t > 0)} \end{split}$$

where  $\mathbb{P}_x(N_{ct} > 0)/\mathbb{P}_x(N_t > 0)$  is uniformly bounded in  $ct \wedge t \geq T$ .

**Remark 2.54.** For the more general set up, the above arguments do not need to be changed, except to replace  $\lambda \int_D \varphi(y)^3 dy$  in (2.49) by

$$\frac{\lambda \mathbb{E}[A^2 - A]}{2(m-1)} \int_D \varphi(y)^3 \, dy$$

where A is the offspring distribution and  $\varphi$  as usual becomes the first eigenfunction of the generator. This follows from Remark 2.23, since branching occurs along the spine at rate  $(m/m-1)\lambda$  and the number of younger siblings at each such point has expectation  $\mathbb{E}[(A^2 - A)/2]$ .

The next Lemma provides the key connection between  $\hat{S}$  and the height function, for a branching Brownian motion tree that is conditioned to survive for a long time.

**Lemma 2.55.** Write  $\overline{\mathbb{P}}_x$  for the law of a branching Brownian motion started at x, plus a vertex u chosen uniformly from it. Then for any  $\eta > 0$  we have

$$\overline{\mathbb{P}}_x(u \ is \eta_T \text{-} bad \mid N_t > 0) \to 0,$$

uniformly in  $t \geq T$  as  $T \to \infty$ .

*Proof.* By conditioning on  $N_s$  and  $N_s^{\eta_T}$  for all  $0 \le s < \infty$  and L, which is the total length of the tree as usual, we see that

$$\overline{\mathbb{P}}_x\left(u \text{ is } \eta_T \text{-bad} \mid N_t > 0\right) = \mathbb{E}_x\left[\int_0^\infty \frac{N_s^{\eta_T}}{N_s} \frac{N_s}{L} \, ds \,\middle|\, N_t > 0\right].$$

Thus we need to show that, given  $\varepsilon > 0$ , we have

$$\mathbb{E}_x \left[ \int_0^\infty \mathbb{1}_{\{|N_s^{\eta_T}/N_s| > \varepsilon/2\}} \frac{N_s}{L} \, ds \, \middle| \, N_t > 0 \right] \le \varepsilon/2$$

for all  $t \ge T$ , whenever T is large enough. First note that by (2.48), we can pick a c such that  $\mathbb{P}_x \left( L \le ct^2 | N_t > 0 \right) \le \varepsilon/8$  for all t, and also using our asymptotic for the survival probability, can choose an R such that  $\overline{\mathbb{P}}_x \left( N_{Rt} > 0 | N_t > 0 \right) \le \varepsilon/8$  for all t. Moreover, since

$$\mathbb{E}_x\left[\int_0^s N_s \, ds \,\middle|\, N_t > 0\right] \le Mst$$

for some M and all s, t, we have that  $\mathbb{E}_x[\int_0^{bt} N_s ds | N_t > 0] \leq Mbt^2$  and can therefore choose b > 0 small enough that this is less than  $\varepsilon ct^2/8$ . Combining this with the condition on L, our problem is reduced to showing that

$$\lim_{T \to \infty} \frac{1}{ct^2} \mathbb{E}_x \left[ \int_{bt}^{Rt} \mathbb{1}_{\{|N_s^{\eta_T}/N_s| > \varepsilon/2\}} N_s \, ds \, \middle| \, N_t > 0 \right] \le \varepsilon/8$$

uniformly in  $t \geq T$ . However, by Remark 2.53 we know that

$$\sup_{bt \ge T} \sup_{s \in [bt,Rt]} \mathbb{P}_x\left(\left\{ \left|N_s^{\eta_T}/N_s\right| > \varepsilon/2 \right\} \cap \left\{N_s > 0\right\} \right| N_t > 0\right) \to 0$$

as  $T \to \infty$  (the uniformity comes straight from the proof.) Hence, applying Fubini,

Cauchy-Schwarz and integrating, similarly to in the proof of Proposition 2.45, we obtain the result.  $\hfill \Box$ 

We will use the above to show that, if we choose k particles  $(u_1, \dots, u_k)$  uniformly from a branching Brownian motion tree conditioned to survive up to time t and define the matrices

$$(d_t^S(u_i, u_j))_{1 \le i < j \le k} = \frac{\hat{S}(u_i) + \hat{S}(u_j) - 2\hat{S}(v_{i,j})}{t}$$

and

$$(d_t^H(u_i, u_j))_{1 \le i < j \le k} = \frac{h(u_i) + h(u_j) - 2h(v_{i,j})}{t}$$

where  $v_{i,j}$  is the most recent common ancestor of  $u_i$  and  $u_j$ , then the two are essentially the same up to the multiplicative constant  $\lambda \int_D \varphi(y)^3 dy$ . Note that  $d_t^H$ is actually how we define distances in the genealogical tree (after rescaling by t) and

$$\lambda \int_D \varphi(y)^3 \, dy = \frac{\sigma}{\alpha}$$

where  $\alpha$  is the constant from the statement of Theorem 2.1.

**Proposition 2.56.** Let  $d_t^S$ ,  $d_t^H$  and  $(u_1, \dots, u_k)$  be defined as above. Then for any  $\varepsilon > 0$ 

$$\mathbb{P}\left(\left\|\left(\frac{\sigma}{\alpha}d_t^H(u_i, u_j) - d_t^S(u_i, u_j)\right)\right\| > \varepsilon\right| N_t > 0\right) \to 0$$
(2.54)

as  $t \to \infty$ , where the distance is the Euclidean distance between  $k \times k$  matrices.

Proof. We prove this in the case k = 2, the general result following by a union bound. Lemma 2.55 tells us that if we pick a vertex u uniformly at random from a tree conditioned to survive to time t, then with high probability the value of  $\hat{S}(u)/h(u)$  is close to  $\sigma/\alpha$ . In fact, the probability that this holds at *all* points along u's ancestry (except for at very small times) is very large. Hence, if we pick any pair of vertices uniformly, since  $d_t^H$  and  $d_t^S$  depend only on their joint ancestry and the values of  $\hat{S}$  along it, they will be close (when  $d_t^H$  is multiplied  $\sigma/\alpha$ ) with high probability.

To make this more precise, pick an  $\eta > 0$ . Then we can pick T large enough that the probability of two uniformly chosen vertices  $u_1$ ,  $u_2$  from the tree under  $B_n^H$  being  $\eta_T$ -bad is arbitrarily small, uniformly in  $t \ge T$ . We can also pick R large enough the probability that  $[h(z_1) \land h(z_2), h(z_1) \lor h(z_2)] \subset [t/R, Rt]$  is arbitrarily close to 1, uniformly in t (see the proof of Lemma 2.55.)

Then, on the event that  $u_1$  and  $u_2$  are both  $\eta_T$ -good, and  $[h(u_1) \wedge h(u_2), h(u_1) \vee$ 

 $h(u_2) \subset [t/R, Rt]$ , we have, as long as  $t \geq RT$ 

$$\left|\frac{\hat{S}(u_i)}{h(u_i)} - \sigma/\alpha\right| \le \eta$$

for i = 1, 2. The other contributions to  $d_t^H$  and  $d_t^S$  come from the height (resp. the value of  $\hat{S}$ ) at the most recent common ancestor u of the two vertices. There are two possibilities: either  $h(u) \leq T$ , or not. In the second case we have that

$$\left|\frac{\hat{S}_n(u)}{h(u)} - \sigma/\alpha\right| \le \eta$$

and so

$$\left| d_t^S(u_1, u_2) - \frac{\sigma}{\alpha} d_t^H(u_1, u_2) \right| \le 4R\eta.$$

In the first we have

$$\left| d_t^S(u_1, u_2) - \frac{\sigma}{\alpha} d_t^H(u_1, u_2) \right| \le 2R\eta + \left( \frac{2\sigma}{\alpha} + \sup_{s \le T} \frac{N_s}{s} \right) \frac{T}{t}.$$

However for fixed T, we have that  $\mathbb{P}_x(\sup_{s \leq T} |N_s/s| \geq K | N_t > 0)$  converges to 0 as  $K \to \infty$ , uniformly in t (see Proposition 2.5). This proves the convergence in probability.

Immediately from the proof, we also get the following Corollary, which we will need later on.

**Corollary 2.57.** We also have for any  $\varepsilon > 0$  and c > 0

$$\mathbb{P}\left(\left\|\left(\frac{\sigma}{\alpha}d_{ct}^{H}(u_{i}, u_{j}) - d_{ct}^{S}(u_{i}, u_{j})\right)\right\| > \varepsilon \right| N_{t} > 0\right) \to 0$$
(2.55)

as  $t \to \infty$ , where the distance is the Euclidean distance between  $k \times k$  matrices.

**Remark 2.58.** The proofs of Lemma 2.55 and Proposition 2.56, given all the previous work, are exactly the same for the more general set up.

## 2.6.3 Convergence to the CRT

## 2.6.4 Preliminaries on convergence of metric measure spaces

Before we can prove Theorem 2.1, we must introduce various notions of convergence for metric spaces, and more generally, for metric measure spaces. Although our aim is to prove convergence of conditioned genealogical trees in the sense of Gromov-Hausdorff distance between metric spaces, it turns out to be helpful to go through the framework of metric measure spaces. We first recall the definition of the Gromov-Hausdorff metric on  $X_c$ : the space of (isometry classes of) compact metric spaces.

**Definition 2.59.** The Gromov-Hausdorff distance between  $(X, r_X)$  and  $(Y, r_Y)$  in  $\mathbb{X}_c$  is given by

$$d_{GH}((X, r_X), (Y, r_Y)) = \inf_{g_X, g_Y, Z} d_H^{(Z, r_Z)}(g_X(X), g_Y(Y)),$$

where the infinum is taken over all isometric embeddings  $g_X, g_Y$  from X and Y to a common metric space  $(Z, r_Z)$ , and  $d_H^{(Z, r_Z)}$  is the usual Hausdorff distance on  $(Z, r_Z)$ .

Now we will briefly discuss some modes of convergence for metric measure spaces, and how they are related, both with each other and the above. For us, a *metric measure space*  $(X, r, \mu)$  will be a compact metric space, equipped with a finite Borel measure. These will be considered up to isometry, where  $(X, r, \mu) \sim (X', r', \mu')$  if there exists a measure preserving isometry between X and X'. We denote the set of (isometry classes) of these spaces by X. We will be interested in the *Gromov*-*Prohorov* metric and the *Gromov-Hausdorff-Prohorov* metric on X. We will begin by defining the so-called *Gromov-Weak* topology.

**Definition 2.60.** [GPW09, Definition 2.3] We will call a function  $\Phi : \mathbb{X} \to \mathbb{R}$ a polynomial if there exists an  $k \in \mathbb{N}$  and a bounded continuous function  $\phi : [0,\infty)^{\binom{k}{2}} \to \mathbb{R}$  such that

$$\Phi((X,r,\mu)) = \int \mu^{\otimes k} (d(x_1,\cdots,x_n)) \phi((r(x_i,x_j))_{1 \le i < j \le k}),$$

where  $\mu^{\otimes k}$  is the product measure of  $\mu$ . Write  $\Pi$  for the set of all polynomials.

**Definition 2.61.** [GPW09, Definition 2.8] A sequence  $\mathcal{X}_n \in \mathbb{X}$  is said to converge to  $\mathcal{X} \in \mathbb{X}$  with respect to the Gromov-weak topology if and only if  $\Phi(\mathcal{X}_n)$  converges to  $\Phi(\mathcal{X})$  in  $\mathbb{R}$ , for all polynomials  $\Phi \in \Pi$ .

It was proved in [GPW09, Theorem 5] that this topology is metrised by the *Gromov-Prohorov metric* defined below.

**Definition 2.62.** The Gromov-Prohorov distance between  $\mathcal{X} = (X, r_X, \mu_X)$  and  $\mathcal{Y} = (Y, r_Y, \mu_Y)$  in  $\mathbb{X}$  is given by

$$d_{GP}(\mathcal{X}, \mathcal{Y}) = \inf_{g_X, g_Y, Z} d_{Pr}^{(Z, r_Z)}((g_X)_*(\mu_X), (g_Y)_*(\mu_Y)),$$

where the infinum is as in Definition 2.59 and  $d_P^{(Z,r_Z)}$  is the Prohorov distance between probability measures on  $(Z, r_Z)$ . Finally, we define the Gromov-Hausdorff-Prohorov metric [ADH13], [Mie09] on  $\mathbb X.$ 

**Definition 2.63.** Let  $\mathcal{X}, \mathcal{Y}$  be as in Definition 2.62. The Gromov-Hausdorff-Prohorov distance between  $\mathcal{X}$  and  $\mathcal{Y}$  is defined by

$$d_{GHP}(\mathcal{X}, \mathcal{Y}) = \inf_{g_X, g_Y, Z} \left( d_{Pr}^{(Z, r_Z)}((g_X)_*(\mu_X), (g_Y)_*(\mu_Y)) + d_H^{(Z, r_Z)}(g_X(X), g_Y(Y)) \right).$$

**Remark 2.64.** It is clear from the above definitions that convergence in the Gromov-Hausdorff-Prohorov metric implies convergence in both the Gromov-Hausdorff metric and the Gromov-Prohorov metric.

We will need a couple of facts for our proof:

**Lemma 2.65.** [GPW09, Corollary 3.1] A sequence  $\{\mathbb{P}_n\}_{n\in\mathbb{N}}$  of probability measures on  $\mathbb{X}$  converges weakly to a probability measure  $\mathbb{P}$  with respect to the Gromov-weak topology, if and only if

- (i) The family  $\{\mathbb{P}_n\}_{n\in\mathbb{N}}$  is relatively compact in the space of probability measures on  $\mathbb{X}$ .
- (ii) For all polynomials  $\Phi \in \Pi$ ,  $\mathbb{P}_n[\Phi] \to \mathbb{P}[\Phi]$  in  $\mathbb{R}$  as  $n \to \infty$ .

and

**Lemma 2.66.** [ADH13, Theorem 2.4], [BBI01, Theorem 7.4.15] A set  $\mathbb{K} \subset \mathbb{X}$  is relatively compact with respect to the Gromov-Hausdorff-Prohorov metric if and only if

- (i) There is a constant D such that  $\operatorname{diam}(\mathcal{X}) < D$  for all  $\mathcal{X} \in \mathbb{K}$ .
- (ii) For every  $\delta > 0$  there exists  $N = N_{\delta}$  such that for all  $\mathcal{X} \in \mathbb{K}$ , X can be covered by  $N_{\delta}$  balls of radius  $\delta$ .
- (*iii*)  $\sup_{\mathcal{X}\in\mathbb{K}}\mu_X(X) < +\infty$

#### 2.6.5 Proof of the main theorem

In this section we will assume that our critical branching Brownian motion is always started from position  $x \in D$ . Recall that for any y > 0 we let  $\mathcal{T}_n^y$  be the genealogical tree generated by this process, when it is conditioned to survive until time ny. This is just our usual tree, but with the marks forgotten. Denote by  $d_n^y$  the natural metric on this tree (given by the length of the path connecting two vertices by their most recent common ancestor). Finally, write  $(\mathcal{T}_{e^y}, d_{e^y})$  for the real tree whose contour function is given by  $e^y$ , a Brownian excursion conditioned to reach height y. We will often want to put a uniform measure on these trees, to make them into metric measure spaces. In the case of  $\mathcal{T}_n^y$  the measure will be denoted by  $\mu_n^y$  and will be defined by the following procedure. Let  $\phi : \mathcal{T}_n^y \to \mathbb{R}$  send the vertex visited at time t in a depth-first exploration of the tree to  $t \in \mathbb{R}$ . Then we can put a uniform measure on the image of  $\phi$  (which will almost surely be a finite interval) and pull this back to the tree. Note that choosing a vertex from  $\mathcal{T}_n^y$  according to  $\mu_n^y$  is the same as choosing a vertex uniformly from the tree, as in Section 2.6.2. To define the uniform measure  $\mu_{e^y}$  for  $(\mathcal{T}_{e^y}, d_{e^y})$ , recall that  $\mathcal{T}_{e^y}$  is the real tree encoded by the excursion  $e^y$ . This means that the metric space is (isometric to) the interval on which the excursion is supported, quotiented by the equivalence relation that  $s \sim r$ if  $\inf\{e_u^y; u \in [r, s]\} = e_r^y = e_s^y$ , with metric  $d_{e^y}(r, s) = e_r^y + e_s^y - 2\inf\{e_u^y; u \in [r, s]\}$ . The uniform measure  $\mu_{e^y}$  is then just the quotient measure of uniform measure on the interval.

Recall that, since we are working in the case of binary branching Brownian motion, we would like to show that for any y > 0

$$(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}) \xrightarrow[n \to \infty]{} (\mathcal{T}_{e^y}, d_{e^y})$$

in distribution, with respect to the Gromov-Hausdorff distance, where

$$\alpha = \sqrt{\frac{2}{\lambda \langle 1, \varphi \rangle \int_D \varphi(y)^3 \, dy}}.$$

*Proof.* An outline of the proof is as follows:

(1) We will show that if we make  $(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y})$  into a metric measure space for each n, by equipping it with the uniform measure  $\mu_n^{\alpha y}$ , then the family

$$\left(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}, \mu_n^{\alpha y}\right)_{n \ge 0}$$

is tight with respect to the Gromov-Hausdorff-Prohorov metric.

- (2) Using Propositions 2.47 and 2.56, we will show convergence of the above family, with respect to the Gromov-Prohorov metric, to  $(\mathcal{T}_{e^y}, d_{e^y}, \mu_{e^y})$ , where  $\mu_{e^y}$  is uniform measure on the real tree  $\mathcal{T}_{e^y}$ , as defined above.
- (3) This also characterises the subsequential limits with respect to the Gromov-Hausdorff-Prohorov metric, since Gromov-Hausdorff-Prohorov convergence implies Gromov-Prohorov convergence. Thus we have the convergence in distribution

$$\left(\mathcal{T}_{n}^{\alpha y}, \frac{1}{\alpha n} d_{n}^{\alpha y}, \mu_{n}^{\alpha y}\right) \xrightarrow[n \to \infty]{} (\mathcal{T}_{e^{y}}, d_{e^{y}}, \mu_{y})$$

with respect to the Gromov-Hausdorff-Prohorov metric. Consequently, by Remark 2.64, we have that

$$\left(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}\right) \xrightarrow[n \to \infty]{} (\mathcal{T}_{e^y}, d_{e^y})$$

with respect to the Gromov-Hausdorff metric, as desired.

All that remains is to verify the statements in (1) and (2). For Part (1), we need to show that for any  $\varepsilon > 0$  there exists a relatively compact  $\mathbb{K} \subset \mathbb{X}$  (wrt the Gromov-Hausdorff metric) such that

$$\inf_{n} \mathbb{P}((\mathcal{T}_{n}^{\alpha y}, \frac{1}{\alpha n} d_{n}^{\alpha y}) \in \mathbb{K}) \geq 1 - \varepsilon.$$

We will use the characterisation given by Lemma 2.66. Since all of our measures are probability measures, condition (iii) of this characterisation is trivial. We begin by proving the existence of K > 0 such that

$$\sup_{n} \mathbb{P}\left(\operatorname{diam}\left(\mathcal{T}_{n}^{\alpha y}, \frac{1}{\alpha n} d_{n}^{\alpha y}\right)\right) > 2K\right) < \varepsilon/2, \tag{2.56}$$

which gives condition (ii). However, diam  $(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y})$  is less than  $2/\alpha n$  times the maximum height of a branching Brownian motion process conditioned to survive until time  $\alpha ny$ , so this follows immediately from our asymptotic for the survival probability. To complete the proof of tightness we will consider the tree  $(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y})$  cut off at height Kn, and show that there exists an M > 0 such that the probability of this cut off tree having a  $\delta$ -net with less than  $M/\delta^4$  balls is greater than  $1 - \delta \varepsilon/2$  for all  $n \in \mathbb{N}$  and  $\delta > 0$ . By summing probabilities over the sequence  $\delta_k = 2^{-k}$ , and combining with (2.56) this provides the set  $\mathbb{K}$  that we need. This estimate is of course very crude, but will suffice for our purposes here.

To prove the claim, we will take K as above, and for any  $\delta > 0, n \in \mathbb{N}$  divide [0, Kn] into intervals of length  $n\delta/2 := b_{n,\delta}$  (so there are  $2K/\delta$  of them.) Then we will choose M such that in any one of these intervals  $[jb_{n,\delta}, (j+1)b_{n,\delta}]$  the probability of having more than  $M/2K\delta^3$  particles at time  $jb_{n,\delta}$  with descendants at time  $(j+1)b_{n,\delta}$  is less than  $\delta^2 \varepsilon/4K$ , independently of  $j, n, \delta$ . Assuming we can do this, summing up gives that the probability of this holding for any of the subintervals is less than  $\delta \varepsilon/2$ . Moreover, on the event that it doesn't happen for any subinterval, we can put a  $\delta$  ball, for each j, at every vertex in level  $jb_{n,\delta}$  that has a descendant at level  $(j+1)b_{n,\delta}$ . This will cover the tree up to level Kn, since a  $\delta$ -ball is effectively a  $\delta n$  ball (remember that lengths are rescaled), and the choice of positions means that every vertex with height in  $[jb_{n,\delta}, (j+1)b_{n,\delta})$  for  $j \ge 1$  is covered by one of the

 $\delta$  balls centred at level  $(j-1)b_{n,\delta}$ . For j = 0, vertices in this interval are covered by the ball placed at the root. In this covering, by definition of the event, there are less than  $M/\delta^4$  balls. Thus it remains to show that we can choose such an M, independently of  $\delta$ , n. This is a result of the following observations:

- By Theorem 2.4, there exists an R > 0 such that the probability of a critical branching Brownian motion starting from z surviving until time  $n\delta/2$  is less than  $R/n\delta$  for all  $n, \delta$  and z.
- Let  $\tilde{N}_{jb_{n,\delta}}$  be the number of vertices at time  $jb_{n,\delta}$  that have descendants at time  $(j+1)b_{n,\delta}$ . Then, by Wald's identity, we have for all j, n and  $\delta$

$$\mathbb{P}_{x}\left(\tilde{N}_{jb_{n,\delta}} \geq \frac{M}{2K\delta^{3}}\right) \leq \frac{2RK\delta^{2}\mathbb{E}_{x}[N_{jb_{n,\delta}}]}{Mn} \leq \frac{2CRK\delta^{2}}{Mn}$$

where  $C := \sup_{z,u} \mathbb{E}_z[N_u]$  is finite by the proof of Theorem 2.6. Note that we are not conditioning on survival here.

Therefore,

$$\mathbb{P}_x\left(\left.\tilde{N}_{jb_{n,\delta}} \ge \frac{M}{2K\delta^3}\right| N_{\alpha yn} > 0\right) \le \frac{2CRK\delta^2}{Mn} \times \mathbb{P}_x(N_{\alpha yn} > 0)^{-1}.$$

This is less than  $\delta^2 \varepsilon / 4K$  if we choose

$$M = \frac{8CRK^2}{\varepsilon} \sup_{n \ge 1} \left( n \mathbb{P}_x(N_{\alpha yn} > 0) \right)^{-1}$$

where the supremum is finite by Theorem 2.4.

Now we move on to (2). We would like to show that  $(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}, \mu_n^{\alpha y})$  converges to  $(\mathcal{T}_{e^y}, d_{e^y}, \mu_y)$  in the Gromov-Prohorov metric, or equivalently, with respect to the Gromov-weak topology. We will consider the latter formulation, in order to use the characterisation given by Lemma 2.65. Since convergence in the Gromov-Hausdorff-Prohorov sense implies convergence in the Gromov-Prohorov/Gromov-Weak sense, the argument for (1) also shows that part (i) of the characterisation (relative compactness of the laws) is satisfied. Therefore, we need only show for any polynomial  $\Phi \in \Pi$ , writing  $\mathbb{P}_n^H$  for the law of  $(\mathcal{T}_n^{\alpha y}, \frac{1}{\alpha n} d_n^{\alpha y}, \mu_n^{\alpha y})$  and  $\mathbb{P}$  for the law of  $(\mathcal{T}_{e^y}, d_{e^y}, \mu_y)$ , that we have  $\mathbb{P}_n^H[\Phi] \to \mathbb{P}[\Phi]$  as  $n \to \infty$ . To do this, we use the convergence of the martingale, Proposition 2.47, and Proposition 2.56.

Fix a polynomial  $\Phi$  from  $[0, \infty)^{\binom{k}{2}}$  to  $\mathbb{R}$ . We will first introduce some notation. We let  $B_n^H$  be the event that a branching Brownian motion tree reaches height  $\alpha yn$ , and  $B_n^S$  be the event that its associated process  $\hat{S}$ , recall the definition from Definition 2.40, reaches level  $\sigma yn$ . Note that  $\sigma/\alpha = \lambda \int_D \varphi(y)^3 dy$ , which is the constant from

Propositions 2.52 and 2.56 that relates the height function and  $\hat{S}$ . We denote by  $\mathbb{P}_n^S$  the law (on metric measure spaces) which is the same as  $\mathbb{P}_n^H$ , except that the tree is conditioned on  $B_n^S$  rather than  $B_n^H$ . To prove that  $\mathbb{P}_n^H[\Phi] \to \mathbb{P}[\Phi]$  we show that:

(i) 
$$\mathbb{P}_x(B_n^S|B_n^H) \to 1$$
 and  $\mathbb{P}_x(B_n^H|B_n^S) \to 1$  as  $n \to \infty$ .

(ii) 
$$\mathbb{P}_n^S[\Phi] \to \mathbb{P}[\Phi]$$
 as  $n \to \infty$ .

Combining these provides the result. We start with (i). The convergence  $\mathbb{P}_x(B_n^S|B_n^H) \to 1$  is in fact a direct consequence of Proposition 2.52, and the fact that given a branching Brownian motion survives until time  $y\alpha n$ , the probability of it surviving to time  $\varepsilon y\alpha n$  and having more than two particles alive at this time, is high. For the second part, it is therefore sufficient to show that  $\mathbb{P}_x(B_n^H)/\mathbb{P}_x(B_n^S) \to 1$  as  $n \to \infty$ . However, the convergence given by Proposition 2.47 allows us to compute an exact asymptotic for  $\mathbb{P}_x(B_n^S)$ , just as in for example [LG05, Section 1.4, p263]. It is easy to check that this is indeed equal to  $\frac{\varphi(z)}{y\sigma n} = \frac{\varphi(x)}{\alpha yn \int_D \varphi(z)^3 dz}$ .

For the proof of (ii) recall that we write  $\hat{S}^y$  for the process  $\hat{S}$  conditioned to reach y. This is where we will use Proposition 2.47. This, along with the Skorokhod representation theorem, tells us that there exists a sequence of processes  $(Z_t^{(n)})_{t\geq 0}$ , equal in distribution to  $(\hat{S}_{n^2t}^{y\sigma n}/n)_{t\geq 0}$ , such that

$$(Z_t^{(n)})_{t\geq 0} \to (\sigma e_t^y)_{t\geq 0} \tag{2.57}$$

uniformly almost surely as  $n \to \infty$ . Here we set the processes identically equal to zero after they reach zero, at times that we denote by  $(\tau_n)_{n\in\mathbb{N}}$  and  $\tau$ . Choose kpoints  $(z_i; 1 \le i \le k)$  uniformly from  $[0, \tau]$  and for each n set  $(z_i^n; 1 \le i \le k) = \frac{\tau_n}{\tau}(z_i; 1 \le i \le k)$ . Then for each n, the law of

$$\left( (Z_t^{(n)})_{t \ge 0}, (z_i^n; 1 \le i \le k) \right)$$

is that of  $(\hat{S}_{n^2t}^{y\sigma n}/n)_{t\geq 0}$  together with k points chosen uniformly from its length. Define the distance between  $z_i^n$  and  $z_j^n$  for  $1 \leq i < j \leq k$  by

$$d_n^Z(z_i^n, z_j^n) := Z_{z_i^n}^{(n)} + Z_{z_j^n}^{(n)} - 2 \inf_{s \in [z_i^n, z_j^n]} Z_s^{(n)}.$$
(2.58)

Setting

$$d_{e^{y}}(z_{i}, z_{j}) = e_{z^{i}}^{y} + e_{z^{j}}^{y} - 2 \inf_{s \in [z_{i}, z_{j}]} e_{s}^{y}$$

(corresponding to the normal metric  $d_{e^y}$  in  $\mathcal{T}_{e^y}$ ) it is immediate from (2.57) that

$$(z_i^n; 1 \le i \le k) \longrightarrow (z_i; 1 \le i \le k)$$

almost surely as  $n \to \infty$ , and so also that

$$\frac{1}{\sigma} \left( d_n^Z(z_i^n, z_j^n) \right)_{1 \le i < j \le k} \longrightarrow \left( d_{e^y}(z_i, z_j) \right)_{1 \le i < j \le k}$$
(2.59)

almost surely as  $n \to \infty$ . This is useful, because the law of the object on the right is the same as the law of the matrix of pairwise distances between k points chosen independently according to  $\mu_{e^y}$  from  $(\mathcal{T}_{e^y}, d_{e^y})$ . Note that although the metric space is actually a quotient of  $[0, \tau]$  in this case, the law above is not affected.

Moreover, the law of the object on the left is what we get if we choose k vertices uniformly from a branching Brownian motion tree conditioned on  $B_n^S$ , and take the matrix defined by  $d_n^S$  in Proposition 2.56. However, we know by Proposition 2.56 and Corollary 2.57 that if we condition instead on  $B_n^H$ , the difference between this matrix and the matrix of pairwise distances under  $\frac{1}{\alpha n} d_n^{\alpha y}$ , converges to 0 in probability as  $n \to \infty$ . By part (i), this is still true under conditioning on  $B_n^S$ . Since  $\Phi$  is continuous and bounded, the proof of part (ii) is complete.

We conclude by noticing that, since the above proof does not use anything specific about the branching diffusion (except the results that have been proven earlier in the paper and some constants), the more general statement of Theorem 2.1 holds.

**Remark 2.67.** The same result holds for a branching diffusion with generator L and offspring distribution A as in Theorem 2.1 where

$$\alpha = \sqrt{\frac{4(m-1)}{\lambda \langle 1, \varphi \rangle \mathbb{E}[A^2 - A] \int_D \varphi(y)^3 \, dy}}$$

so that

$$\frac{\sigma}{\alpha} = \frac{\lambda \mathbb{E}[A^2 - A]}{2(m-1)} \int_D \varphi(y)^3 \, dy$$

as in Remark 2.54.

# 3 Critical Gaussian chaos: convergence and uniqueness in the derivative normalisation

## 3.1 Introduction

The theory of Gaussian multiplicative chaos was developed by Kahane, [Kah85], in order to rigorously define measures of the form

$$\mu^{\gamma}(dx) := \mathrm{e}^{\gamma h(x) - \frac{\gamma^2}{2} \mathbb{E}[h(x)^2]} dx$$

where h is a rough centered Gaussian field, satisfying certain assumptions, and  $\gamma > 0$ is a real parameter. Since h is not defined pointwise, a regularisation procedure is required to define  $\mu^{\gamma}$ . In [Kah85], it is assumed that the covariance kernel K of h is  $\sigma$ -positive, meaning that K can be approximated by a series of smooth positive kernels  $K_n$ . It is then possible to associate to such an approximation the sequence of measures  $\mu_n(dx) := \exp{\{\gamma h_n(x) - (\gamma^2/2) \operatorname{var}(h_n(x))\}} dx$ . Kahane proved that these measures converge as  $n \to \infty$ , and that the limit is independent of the choice of approximation. We call this limit the  $\gamma$ -chaos measure associated to h.

However,  $\sigma$ -positivity can be hard to check pointwise, and in recent years this theory has been significantly generalised by several authors [RV10, Ber15a, JS17, Sha16]. When K is not  $\sigma$ -positive, a natural way to approximate h is to convolve it with a general mollifier function  $\theta$ . Writing  $h_{\varepsilon}$  for these regularisations, it has been shown that for log-correlated h, and under very general conditions on  $\theta$ , the approximate measures

$$\mu_{\varepsilon}^{\gamma}(dx) := \mathrm{e}^{\gamma h_{\varepsilon}(x)} \, \mathrm{e}^{-\frac{\gamma^2}{2} \operatorname{var}(h_{\varepsilon}(x))} \, dx \tag{3.1}$$

converge weakly in law [RV10] and in probability [Ber15a, Sha16] as  $\varepsilon \to 0$ . The limit is non-zero if and only if  $\gamma^2 < 2d$ . Moreover, it is universal in that it does not depend on the choice of regularisation [Ber15a, JS17, Sha16].

When  $\gamma^2 = 2d$ , an additional renormalisation is required in order to yield a nontrivial limiting measure. Motivated by the theory of multiplicative cascades and the branching random walk [BK04, AS14] one can hope to renormalise at criticality in one of two different ways. The first is called the Seneta-Heyde renormalisation, and involves premultiplying the sequence of measures (3.1) by the deterministic sequence  $\sqrt{\log(1/\varepsilon)}$ . The other is a random renormalisation, which is defined by taking a derivative of the measure (3.1) in  $\gamma$ . It has been shown in [DRSV14a, DRSV14b] that for a special class of fields h having so-called  $\star$ -scale invariant kernels, and for a specific sequence of approximations to h, both procedures yield the same nonzero limiting measure (up to a constant). However, the result in these papers relies heavily on the cut-off approximation used for the kernel of h, and does not generalise to arbitrary convolution approximations. These are somewhat more natural, local approximations to the field, and the goal of the paper will be to extend the theory to this set-up.

In this paper we will be particularly, but not exclusively, interested in the specific case where the underlying field h is a 2d Gaussian free field with zero-boundary conditions. In this case the measure  $\mu^{\gamma}$  (when it is defined and non-zero) is known as the Liouville measure with parameter  $\gamma$ . This has been an object of considerable recent interest due to its strong connection with 2d Liouville quantum gravity and the KPZ relations [DS11, RV11, Ber15b]. Recent works in the case  $\gamma < 2$  include [DS11, RV11, Ber15a], which among other things make an in-depth study of its moments, multifractal structure, and universality. Recently, in [APS17], it has also been shown that these measures can be approximated using so-called local sets of the Gaussian free field. This is a particularly natural construction because it is both local and conformally invariant.

The critical case  $\gamma = 2$  has also been considered for the Gaussian free field: [DRSV14b, HRV15, JS17, APS17]. In [DRSV14b], the authors generalised their construction for  $\star$ -scale invariant kernels to show convergence in the Seneta–Heyde and derivative renormalisations for a specific "white noise" approximation to the field. These both yield the same (up to a constant) non-trivial limiting measure  $\mu'$ , that we will call the critical Liouville measure. However, this proof again does not extend to convolution approximations.

The purpose of this article is to complete the picture for convolution approximations to critical chaos. We will focus specifically on the case of the 2d GFF, and fields with \*-scale invariant kernels. This builds on recent work of Junnila and Saksman [JS17] (and also [HRV15] in the case of the free field), who show that in either of the cases above, the critical measure can be constructed using convolution approximations in the Seneta–Heyde renormalisation.

To complete the story, therefore, it remains to show that the random "derivative" renormalisation procedure will also yield the same limit for general convolution approximations. This is the main result of the current paper. We remark that the derivative renormalisation is somewhat more natural, and in fact, it is usually easier to show convergence of this before convergence in the Seneta–Heyde renormalisation (which is then obtained by a comparison argument). Here we will reverse this procedure.

Suppose that h is a log-correlated field in  $D \subset \mathbb{R}^d$  with kernel K(x, y). By this

we mean that  $(h, \rho)_{\rho \in \mathcal{M}}$  is a centered Gaussian process, indexed by the set of signed measures  $\rho$  whose positive and negative parts  $\rho^{\pm}$  satisfy  $\int \int \rho^{\pm}(dx) |K(x,y)| \rho^{\pm}(dy) < \infty$ , with covariance structure

$$\operatorname{cov}((h,\rho)(h,\rho')) = \iint \rho(dx) K(x,y) \rho'(dy)$$

for  $\rho, \rho' \in \mathcal{M}$ . Also suppose that  $\theta \in \mathcal{M}$  is a positive measure of unit mass, supported in B(0, 1), and such that

$$\int \frac{1}{\sqrt{|u-v|}} \,\theta(du) = \mathcal{O}(1) \tag{3.2}$$

uniformly over  $v \in B(0,5)$ . Then we define a sequence of  $\theta$ -mollified approximations to h by setting for  $\varepsilon > 0$ ,

$$h_{\varepsilon} := h \star \theta_{\varepsilon}(x) = (h, \theta_{\varepsilon, x}), \tag{3.3}$$

where  $\theta_{\varepsilon}$  is the image of  $\theta$  under the map  $y \mapsto \varepsilon y$  and  $\theta_{\varepsilon,x}$  is the image of  $\theta_{\varepsilon}$ under the map  $y \mapsto y + x$ . We define the measures  $M_{\varepsilon}$  and  $D_{\varepsilon}$  associated with this approximation by setting for  $\mathcal{O} \subset D$ :

$$M_{\varepsilon}(\mathcal{O}) := \int_{\mathcal{O}} e^{2h_{\varepsilon}(x) - 2\operatorname{var}(h_{\varepsilon}(x))} dx;$$
$$D_{\varepsilon}(\mathcal{O}) := \int_{\mathcal{O}} (-h_{\varepsilon}(x) + 2\operatorname{var}(h_{\varepsilon}(x))) e^{2h_{\varepsilon}(x) - 2\operatorname{var}(h_{\varepsilon}(x))} dx$$

Note that  $M_{\varepsilon}$  is exactly the same as  $\mu_{\varepsilon}^{\gamma=2}$  (but we introduce the new notation to distinguish the special case  $\gamma = 2$  and avoid confusing notation.) Our aim will be to prove the following:

**Theorem 3.1.** Suppose that h is a 2d Gaussian free field and  $D_{\varepsilon}$  is defined as above, for a mollifier  $\theta$  satisfying (3.2). Then  $D_{\varepsilon}$  converges weakly in probability as  $\varepsilon \to 0$  to the critical Liouville measure  $\mu'$  constructed in [DRSV14b]. In particular  $\lim_{\varepsilon} D_{\varepsilon}$  does not depend on  $\theta$ .

**Theorem 3.2.** Suppose that h is a Gaussian field with  $\star$ -scale invariant kernel and  $D_{\varepsilon}$  is defined as above, for a mollifier  $\theta$  satisfying (3.2) and with Hölder continuous density. Then  $D_{\varepsilon}$  converges weakly in probability as  $\varepsilon \to 0$  to a limiting measure. This measure is independent of the choice of approximation, and agrees with the critical measure constructed in [DRSV14a, JS17] (see Theorems 3.8 and 3.9).

There is one further motivation for proving Theorem 3.1. In [APS17], the authors also construct a critical measure for the Gaussian free field, using a simple and natural approximation based on its local sets. This is closely related to the classical construction of multiplicative cascades [KP76], and we believe that this connection can be exploited to help us improve our understanding of the situation at criticality (in particular, to prove a conjecture given in [DRSV14a].) However, it is a priori hard to connect the measure of [APS17] to the measure  $\mu'$  of [DRSV14b]. It turns out that Theorem 3.1 is exactly what is needed to show that they are in fact equal (for details of this argument, see [APS17]). In conclusion, Theorem 3.1 gives us a universality statement for critical Liouville quantum gravity, that is now in line with the statement for the subcritical case [Ber15a, Sha16, APS17].

**Outline** We will begin in Section 3.2 by giving a brief introduction to log-correlated fields, and explaining how to approximate them using general mollifiers. We will also discuss here some of the existing literature concerning subcritical and critical Gaussian multiplicative chaos, and recall some basic facts about the 3-dimensional Bessel process. These occur naturally in critical Gaussian multiplicative chaos; roughly, as the value of the field locally about a typical point, and will be instrumental in the proof of Theorems 3.1 and 3.2. In Section 3.3 we concentrate on the case when h is a 2d Gaussian free field, and prove Theorem 3.1. We begin in Section 3.3.1 by showing that certain families of "cut-off" approximations to the derivative measures (that we shall call  $D_{\varepsilon}^{\beta}$ ) are uniformly integrable. In fact, this will not be used directly in the proof of Theorem 3.1, but is needed for the aforementioned application to [APS17], and introduces technical facts required for the rest of the proof. Section 3.3.2 contains the bulk of the proof. The main idea is to connect the derivative measures  $D_{\varepsilon}$  with the renormalised measures  $\sqrt{\log(1/\varepsilon)}M_{\varepsilon}$ , which we know converge by [JS17, HRV15]. To do this, we use a technique similar to that first applied in [AS14], and then in [DRSV14b], although the details of the proof are quite different. This is centred around the fact that for the circle average approximation to the free field, there is a natural "rooted measure" arising from the definition of  $D_{\varepsilon}$ , under which it becomes a 3d Bessel process. We can also show that for a general convolution approximation, under the corresponding rooted measure, the process is approximately a Bessel (unfortunately, this introduces many technicalities in the proof.) Properties of the Bessel process then allow us to conclude. Finally, in Section 3.4, we show how the proof can be adapted for the case of  $\star$ -scale invariant kernels, to give Theorem 3.2.

Acknowledgements I am especially grateful to Juhan Aru, Nathanaël Berestycki and Avelio Sepúlveda for many invaluable discussions concerning this paper, and Liouville measures in general. I would also particularly like to thank Nathanaël Berestycki for useful comments on a preliminary draft of the article. Further thanks
are due to Wendelin Werner for inviting me to visit ETH, where the idea for this project originated, and to Vincent Vargas, for useful advice on general chaos measures. This work was supported by the UK Engineering and Physical Sciences Research Council (EPSRC) grant EP/H023348/1 for the University of Cambridge Centre for Doctoral Training, the Cambridge Centre for Analysis.

#### 3.2 Preliminaries

## 3.2.1 Log-correlated fields, 2d Gaussian free field and $\star$ -scale invariant kernels.

Let us recap the definition of log-correlated fields from the introduction. Suppose we have a non-negative definite kernel K(x, y) on  $D \subset \mathbb{R}^d$  of the form

$$K(x,y) = \log(|x-y|^{-1}) + g(x,y)$$
(3.4)

where g is a  $C^1$  function on  $\overline{D} \times \overline{D}$ . As in the introduction, we let  $\mathcal{M}$  be the set of signed measures  $\rho := \rho^+ - \rho^-$  whose positive and negative parts satisfy  $\iint_{D \times D} |K(x,y)| \rho^{\pm}(dx) \rho^{\pm}(dy) < \infty$ . The centered Gaussian field h, with covariance K(x,y), is then defined as in [Ber15a] to be the unique centred Gaussian process  $(h, \rho)_{\rho \in \mathcal{M}}$  indexed by  $\mathcal{M}$ , such that

$$\operatorname{cov}((h,\rho),(h,\rho')) = \iint_{D \times D} K(x,y)\rho(dx)\rho'(dy)$$

for all  $\rho, \rho' \in \mathcal{M}$ .

We say that a kernel K is  $\star$ -scale invariant if it takes the form

$$K(x,y) = \int_{1}^{\infty} \frac{k(u|x-y|)}{u} \, du$$
 (3.5)

for  $k : [0, \infty) \to \mathbb{R}$  a compactly supported and positive-definite  $C^1$  function with k(0) = 1. One can easily check that such a K indeed has the form (3.4). Although this does not cover all kernels satisfying (3.4) it is still a natural family to consider, due to the nice scaling relations it possesses, [?]. Moreover, the sequence of "cut off" approximations to K given by

$$K_{\varepsilon}(x,y) = \int_{1}^{\frac{1}{\varepsilon}} \frac{k(u|x-y|)}{u} \, du$$

yields a family of approximating fields that exhibit a useful decorrelation property (see the proof of Theorem 3.2).

As mentioned in the introduction, we will also be interested in the special case when h is a 2-dimensional Gaussian free field. To define this, let  $D \subset \mathbb{C}$  be a simplyconnected domain. Then the zero boundary Gaussian free field h on D is defined as above, to be the log-correlated field whose kernel K is given by the Green function,  $G_D$ , for the Laplacian on D. This satisfies

$$G_D(x,y) = -\log|x-y| + g(x,y)$$
(3.6)

for g a smooth function on  $\overline{D} \times \overline{D}$ .

One feature that makes the Gaussian free field particularly nice to work with is that it satisfies the following spatial Markov property: if  $A \subset D$  is a closed subset, then we can write  $h = h^A + h_A$  where  $h^A$ ,  $h_A$  are independent,  $h^A$  is a zero-boundary GFF on  $D \setminus A$ , and  $h_A$  is harmonic when restricted to  $D \setminus A$ . We will see how this is useful to us in Section 3.3.

In the following we will always assume, for technical reasons and without loss of generality, that our domain  $D \subset \mathbb{R}^d$  contains the ball of radius 10 around the origin.

#### 3.2.2 Convolution with mollifiers

Suppose we have a field h with kernel K satisfying (3.4). As discussed in the introduction, since h is not defined pointwise, we need to use a regularisation procedure to define its chaos measures. A natural approach is convolve h with an approximation to the identity. Let  $\theta$  be a non-negative Radon measure on  $\mathbb{R}^d$ , satisfying the conditions described in the introduction, and define the convolution approximations  $(h_{\varepsilon}(x))_{\varepsilon>0}$  as in (3.3). The assumption (3.2) on  $\theta$  will be important to show various properties of the convolution approximations later on (cf. Lemma 3.3 and Corollary 3.6). We remark here that (3.2) is more restrictive than the condition given in [Ber15a], but includes most of the important examples. In particular, it includes the case when  $\theta$  is uniform measure on the unit circle, or when  $\theta$  has an  $L^p$  density with respect to Lebesgue measure for some p > 2.

We have the following estimate for the covariances of  $(h_{\varepsilon})_{\varepsilon}$ :

**Lemma 3.3** ([Ber15a]). Suppose  $\theta$  satisfies (3.2) and  $h_{\varepsilon}$  is defined as above. Then:

$$\operatorname{cov}(h_{\varepsilon}(x), h_{\varepsilon'}(y)) = \log(1/(|x-y| \lor \varepsilon \lor \varepsilon')) + \mathcal{O}(1).$$
(3.7)

where by O(1) we mean something that is uniformly bounded in  $\varepsilon, \varepsilon'$ , and x, y.

**Remark 3.4.** Similarly, whenever we use order notation in the sequel, we will mean the order in  $\varepsilon$ , uniformly in whatever spatial position(s) we are considering.

#### 3.2.3 Maxima of the mollified fields

It will also be important for us in this article to get a hold of how fast our approximations  $h_{\varepsilon}$  can blow up. For this we use the work of [Aco14]. Among other things, this gives us the following Lemma (in fact, we state here a slight modification of the result, that is proved in [HRV15]).

**Lemma 3.5.** Let  $(Y_{\varepsilon}^{x} : x \in [0,1]^{d})_{\varepsilon>0}$  be a family of Gaussian fields indexed by  $[0,1]^{d}$  for any integer d. Suppose that for some  $0 < C_{Y} < \infty$  and all  $x, y \in [0,1]^{d}$  we have

(1)  $|\operatorname{cov}(Y^x_{\varepsilon}, Y^y_{\varepsilon}) + \log(\max\{\varepsilon, ||x - y||\})| \le C_Y$  and

(2) 
$$\mathbb{E}\left[(Y_{\varepsilon}^{x}-Y_{\varepsilon}^{y})^{2}\right] \leq C_{Y}\sqrt{\|x-y\|}/\sqrt{\varepsilon} \text{ for all } \|x-y\| < \varepsilon.$$

Then, almost surely,

$$\inf_{\varepsilon} \inf_{x \in [0,1]^d} \{ -Y_{\varepsilon}^x + \sqrt{2d} \log(1/\varepsilon) \} > -\infty.$$

This Lemma, together with the assumption (3.2), allows us to deduce the following:

**Corollary 3.6.** Suppose that  $\theta$  satisfies our usual conditions, including (3.2), and that h has kernel K satisfying (3.23). Assume further that  $\mathcal{O} \subset \mathbb{R}^d$  is bounded. Then

$$\inf_{\varepsilon} \inf_{x \in \mathcal{O}} \{ -h_{\varepsilon}(x) + 2\log(1/\varepsilon) \} > -\infty$$

almost surely.

*Proof.* Condition (1) of Lemma 3.5 is easy to verify. To show condition (2) we write  $K(x,y) = (\log |x-y|^{-1}) + g(x,y)$  where g is  $C^1$  on  $\mathbb{R}^d \times \mathbb{R}^d$ . We need only prove that  $\mathbb{E}[h_{\varepsilon}(x)(h_{\varepsilon}(x) - h_{\varepsilon}(y))] \leq C\sqrt{||x-y||}/\sqrt{\varepsilon}$  for all  $||x-y|| < \varepsilon$  and an absolute constant C. The result then follows by symmetry.

To show this we write

$$\mathbb{E}[h_{\varepsilon}(x)(h_{\varepsilon}(x) - h_{\varepsilon}(y))] = \iint \log \left| \frac{x - y + \varepsilon(v - w)}{\varepsilon(v - w)} \right| \theta(dv)\theta(dw) + \iint g(x + \varepsilon v, y + \varepsilon w) - g(x + \varepsilon v, x + \varepsilon w)\theta(dv)\theta(dw)$$

Since g has continuous derivative and we are working on a bounded set, it is clear that the second term satisfies the required condition. To deal with the first we note that

$$\left|\log\left|\frac{x-y+\varepsilon(v-w)}{\varepsilon(v-w)}\right|\right| \le \max\left\{\log\left(1+\frac{|x-y|}{\varepsilon|v-w|}\right), \log\left(1+\frac{|x-y|}{|x-y+\varepsilon(v-w)|}\right)\right\},$$

where we have used that  $|x - y| < \varepsilon$  (and that  $|v - w| \le 1$ ) to remove the modulus inside the log on the left-hand side. Since  $\log(1 + |a|) \le \sqrt{|a|}$  for all a, we can conclude using assumption (3.2).

#### 3.2.4 Previous works on subcritical and critical Gaussian multiplicative chaos

As discussed in the introduction, Gaussian multiplicative chaos theory is a framework we can use to make sense of measures of the form " $e^{\gamma h(x)-\gamma^2/2 \operatorname{var}(h(x))} dx$ " for log-correlated Gaussian fields h. This stems from the classical martingale theory of the branching random walk [BK77, Kyp00] and multiplicative cascades [KP76], and was initiated by Kahane [Kah85] in the 1980's. In the special case where h is a 2d Gaussian free field, the Gaussian multiplicative chaos measure is often referred to as the Liouville measure [DS11]. Here we will state precisely some of the results mentioned in the introduction.

When  $\gamma < \sqrt{2d}$  (the subcritical regime) there are various approximation procedures that can be used to construct the chaos measure with parameter  $\gamma$ . One natural choice is to use the convolution approximations  $h_{\varepsilon}$  described in the previous section, and define approximate measures  $\mu_{\varepsilon}^{\gamma}$  by setting

$$\mu_{\varepsilon}^{\gamma}(dx) := \mathbb{E}[\mathrm{e}^{\gamma h_{\varepsilon}(x)}]^{-1} \,\mathrm{e}^{h_{\varepsilon}(x)} \,dx \tag{3.8}$$

for  $\varepsilon > 0$ . Note that the normalisation factor here is equal to  $\varepsilon^{\frac{\gamma^2}{2}}$  (up to a bounded constant that depends on x). We have the following result.

**Theorem 3.7** ([Ber15a]). For  $\gamma < \sqrt{2d}$  the measures  $\mu_{\varepsilon}^{\gamma}$  converge to a non-trivial measure  $\mu^{\gamma}$  weakly in probability. Moreover, for any fixed Borel set  $\mathcal{O}$  we have that  $\mu_{\varepsilon}^{\gamma}(\mathcal{O})$  converges in  $L^1$ .

We emphasise that this limit  $\mu^{\gamma}$  does not depend on the choice of mollifier  $\theta$ . In fact, one can approximate the field in other, completely different ways (for instance using a Karhunen–Loève expansion [Ber15a]) and find the same limit. For the case of the free field, this will even work for "non-Gaussian" approximations. Indeed, in [APS17] the authors construct (the same) Liouville measure for  $\gamma < 2$  using sequences of so-called "local sets" of the field.

For general h, the subcritical measures  $\mu^{\gamma}$  with  $\gamma < \sqrt{2d}$  are almost surely atomless, and assign positive mass to any open set. On the other hand, as discussed in the introduction, it is known that for  $\gamma \geq \sqrt{2d}$ , the measures  $\mu_{\varepsilon}^{\gamma}$  converge to zero [RV10]. To define the critical (and supercritical) measures we must therefore make an additional renormalisation. These cases turn out to be much more tricky to deal with than the subcritical case, in part because the limiting measure will not possess any moments of order greater than or equal to 1. Consequently a complete theory is still lacking, but some progress has been made (see [RV14] for a survey). Here and in the rest of this paper we will discuss the critical case  $\gamma = \sqrt{2d}$ .

#### 3.2.5 Critical measures

Motivated by the corresponding constructions for multiplicative cascades, [BK04, AS14], we expect to be able to obtain a non-trivial measure at criticality using either of two renormalisation procedures: one deterministic and one random. Let us outline how this should work. Suppose you have some approximations  $h_{\varepsilon}$  to a log-correlated field h, that are continuous fields for each  $\varepsilon$ . Then each of the following sequences should converge to the same (up to a constant) limiting measure.

- The sequence of measures  $\sqrt{\log(1/\varepsilon)}\mu_{\varepsilon}^{\gamma=2} := \sqrt{\log(1/\varepsilon)}M_{\varepsilon}$ , where  $\mu_{\varepsilon}^{\gamma}$  is defined by (3.8). This is known as the Seneta–Heyde renormalisation.
- The sequence of signed "derivative" measures, obtained by taking the derivative of  $\mu_{\varepsilon}^{\gamma}$  with respect to  $\gamma$  and evaluating at  $\gamma = \sqrt{2d}$ . That is, the sequence

$$D_{\varepsilon}(dz) := (-h_{\varepsilon}(z) + \gamma \mathbb{E}[h_{\varepsilon}(z)^{2}]) \exp\left(\gamma h_{\varepsilon}(z) - \frac{\gamma^{2}}{2} \mathbb{E}\left[h_{\varepsilon}(z)^{2}\right]\right) dz$$

(where we have also multiplied by -1 in order to yield a non-negative limit measure.)

This statement was verified for a specific set-up in [DRSV14a, DRSV14b].

**Theorem 3.8** ([DRSV14a, DRSV14b]). Suppose h has a  $\star$ -scale invariant kernel  $K(x, y) = \int_1^\infty k(u|x-y|)/u \, dx$  as in (3.5) and the approximate fields  $h_{\varepsilon}$  have kernels given by

$$K_{\varepsilon}(x,y) := \int_{1}^{1/\varepsilon} \frac{k(u|x-y|)}{u} \, du.$$

Then the two sequences of approximating measures described above converge weakly in probability to the same limiting measure, up to a constant  $\sqrt{2/\pi}$ . In particular, for any open set  $\mathcal{O} \subset \mathbb{R}^d$ ,  $\sqrt{\pi/2}\sqrt{\ln(1/\varepsilon)}M_{\varepsilon}(\mathcal{O})$  and  $D_{\varepsilon}(\mathcal{O})$  converge in probability and in  $L^p$  (any p < 1) to the same limit.

The authors in [DRSV14a, DRSV14b] were also able to generalise this approach to the case when h is a 2d Gaussian free field, using a white-noise decomposition for the field and another specific sequence of "cut-off" approximations for the kernel. However, both of these proofs rely strongly on a martingale property satisfied by the choice of approximating fields  $h_{\varepsilon}$ . In particular, they do not extend to general convolution approximations.

Convolution is clearly a natural way to approximate the field h, and so we would like to have a version of Theorem 3.8 for such approximations. Using comparison techniques, Junnila and Saksman were able to do this for the Seneta–Heyde renormalisation.

**Theorem 3.9** ([JS17]). Let h be a  $\star$ -scale invariant field, and assume that in addition to (3.2), the mollifier  $\theta$  has a Hölder continuous density. Then the measures  $\sqrt{\log(1/\varepsilon)}M_{\varepsilon}$  converge to a limiting measure weakly in probability as  $\varepsilon \to 0$ . This limit is equal to  $\sqrt{2/\pi}\mu'$  where

- $\mu'$  is the measure from Theorem 3.8,
- $\mathbb{E}\left[\mu'(\mathcal{O})\right] = \infty$  for any  $\mathcal{O} \subset \mathbb{R}^d$  and
- $\mu'(\mathcal{O})$  is positive almost surely for any  $\mathcal{O} \subset \mathbb{R}^d$ .

Again we have that for any open set  $\mathcal{O} \subset \mathbb{R}^d$ ,  $\sqrt{\pi/2}\sqrt{\ln(1/\varepsilon)}M_{\varepsilon}(\mathcal{O})$  and  $D_{\varepsilon}(\mathcal{O})$ converge in probability and in  $L^p$  (p < 1) to the same limit.

This has also been proven for the 2d-Gaussian free field.

**Theorem 3.10** ([HRV15, JS17]). Let h be a 2d-GFF, and take any mollifier  $\theta$  satisfying (3.2). Then the measures  $\sqrt{\log(1/\varepsilon)}M_{\varepsilon}$  converge to a limiting measure weakly in probability as  $\varepsilon \to 0$ . This limit is equal to  $\sqrt{2/\pi}\mu'$  where  $\mu'$  is the critical Liouville measure of [DRSV14b].

Note that Theorem 3.10 places a weaker constraint on the mollifier  $\theta$ . This is due to the proof given in [HRV15]. The aim of this paper will be to prove the analogues of Theorems 3.9 and 3.10 for the derivative renormalisation.

#### 3.2.6 Bessel processes

To conclude this introduction, we need to recall some basic properties of Brownian motion; in particular, of the 3-dimensional Bessel process. Let **P** denote the law of a standard Brownian motion  $B_t$  in  $\mathbb{R}$ , started from a possibly random position  $B_0$  such that  $\mathbf{P}(B_0 > 0) = 1$ . Then it is easy to check that for any  $\beta, \gamma > 0$ , the process

$$(-B_t + \gamma \operatorname{var}(B_t) + \beta) \mathbb{1}_{\{-B_u + \gamma \operatorname{var}(B_t) + \beta > 0 \,\forall u \in [0,t]\}} e^{\gamma B_t - \frac{\gamma^2}{2} \operatorname{var}(B_t)}$$
(3.9)

is a positive martingale. Let  $\mathcal{F}_t$  be the filtration generated by the Brownian motion and define a new measure  $\mathbf{Q}$  by letting its Radon–Nikodym derivative when restricted to  $\mathcal{F}_t$  be given by the martingale at time t. One can check that this yields a well-defined law  $\mathbf{Q}$ , under which the process  $(-B_t + \gamma \operatorname{var}(B_t) + \beta)_{t \geq 0}$  is a 3d Bessel process started from  $-B_0 + \gamma \operatorname{var}(B_0) + \beta$ . Note that this starting position will also be biased, and will be positive almost surely under  $\mathbf{Q}$ . The next lemma records some properties of the 3d Bessel process that we will use in our proofs.

**Lemma 3.11.** Let  $(X_t)_{t\geq 0}$  be a 3d Bessel process started from a random (positive) position  $X_0$  with finite variance, and law **Q**. Then

- (1)  $\mathbf{Q}[\frac{1}{X_t}] = \sqrt{\frac{2}{\pi t}} + o(t^{-1/2})$  where the error term is less than  $\frac{2}{\sqrt{t}}(\frac{\mathbf{Q}[X_0^2]}{t} + \frac{\mathbf{Q}[X_0]}{\sqrt{t}})$ . (2)  $\mathbf{Q}[\frac{1}{X_t^2}] \leq 2/t$ , uniformly in the starting position.
- (3)  $\mathbf{Q}[\frac{\sqrt{u}}{\log(2+u)^2} \le X_u \le (1+\sqrt{u\log(1+u)}) \ eventually] = 1$ (4)

$$\mathbf{Q}\left[\frac{\sqrt{u}}{R\log(2+u)^2} \le X_u \le R(1+\sqrt{u\log(1+u)}) \; \forall u \ge 0\right] \to 1$$

as  $R \to \infty$ , uniformly over  $X_0$  with  $\mathbf{Q}[X_0] \leq K$  for any K.

(5)  $\mathbf{Q}[\frac{1}{X_t}\mathbb{1}_{\{X_t \leq t^{1/4}\}}] \leq \frac{C}{2t}$ , uniformly in the starting position, where C is an absolute constant.

*Proof.* (1),(2) and (5) are straightforward to verify using direct calculation and scaling arguments. (3) is a classical result due to Motoo [Mot58] and then (4) follows by continuity and Markov's inequality.

#### 3.3 Proof of Theorem 3.1

In this section we will work to prove Theorem 3.1. Recall that this concerns the case when the underlying field h is a 2d Gaussian free field in a domain  $D \subset \mathbb{R}^2$ . For this choice of field, there is a particular convolution approximation, when  $\theta$  is uniform measure on the unit circle, that plays an important role. We call this the circle average process and distinguish it by writing  $\tilde{h}_{\varepsilon} := h \star \theta_{\varepsilon}$ . The Markov property of the field allows us to deduce the following:

**Lemma 3.12.** For each  $x \in D$  and  $\delta < d(x, \partial D)$ ,  $\{\tilde{h}_{e^{-u}}(x) : u \ge \log(1/\delta)\}$  is a Brownian motion started from  $\tilde{h}_{\delta}(x)$ .

We will also need to compare  $\tilde{h}_{\varepsilon}$  with a general convolution approximation  $h_{\varepsilon}$ .

**Lemma 3.13.** Let  $h_{\varepsilon}$  and  $h_{\varepsilon}$  be the mollified and circle averages of h at a point x with  $d(x, \partial D) > \varepsilon$ . Then we can write

$$h_{\varepsilon}(x) = \lambda_{\varepsilon}(x)\tilde{h}_{\varepsilon}(x) + Y_{\varepsilon}(x)$$
(3.10)

where  $\lambda_{\varepsilon}(x) = 1 + O(\log(1/\varepsilon)^{-1})$  (uniformly in x) and  $Y_{\varepsilon}(x)$  is independent of  $\tilde{h}_{\varepsilon}(x)$ , Gaussian, and has mean 0 and variance O(1).

*Proof.* For this, we observe (by an easy calculation using (3.6)) that

$$\operatorname{cov}(h_{\varepsilon}(x), h_{\varepsilon}(x)) = \log(1/\varepsilon) + O(1)$$

for any  $x \in D$  and  $\varepsilon < d(x, \partial D)$ . Let  $\lambda_{\varepsilon}(x) := \operatorname{cov}(h_{\varepsilon}(x), \tilde{h}_{\varepsilon}(x)) / \operatorname{cov}(\tilde{h}_{\varepsilon}(x), \tilde{h}_{\varepsilon}(x))$ , so that by direct calculation  $\operatorname{cov}(h_{\varepsilon} - \lambda_{\varepsilon}\tilde{h}_{\varepsilon}, \tilde{h}_{\varepsilon}) = 0$ . Then by Gaussianity,  $\tilde{h}_{\varepsilon}$  and  $Y_{\varepsilon} := h_{\varepsilon} - \tilde{h}_{\varepsilon}$  are independent. Using Lemmas 3.12 and 3.3, we see that the variance of  $Y_{\varepsilon}$  is O(1) and that  $\lambda_{\varepsilon} = 1 + O(\log(1/\varepsilon)^{-1})$ .

**Remark 3.14.** We will often drop the x from  $\lambda_{\varepsilon}(x)$  when it is clear from the context.

**Lemma 3.15.**  $Y_{\varepsilon}(x)$  also has bounded covariances with  $Y_{\varepsilon}(y)$  and  $h_{\varepsilon}(y)$  for any  $x, y \in D$ . Moreover, for  $\delta \geq \varepsilon$ , we have

$$-\rho_{\delta}^{\varepsilon}(x)/2 := \operatorname{cov}(Y_{\varepsilon}(x), h_{\delta}(x)) = \mathcal{O}(1),$$

uniformly in  $\varepsilon, \delta$  and x.

*Proof.* The first claims follow using direct calculation similar to the above. For the final claim note that  $\mathbb{E}[Y_{\varepsilon}(x)\tilde{h}_{\delta}(x)] = \mathbb{E}[h_{\varepsilon}(x)\tilde{h}_{\delta}(x)] - \lambda_{\varepsilon}(x)\mathbb{E}[\tilde{h}_{\varepsilon}(x)\tilde{h}_{\delta}(x)]$  where both expressions on the right-hand side are  $\log(1/\delta) + O(1)$ .

**Remark 3.16.** Lemma 3.13 implies that  $\rho_{\varepsilon}^{\varepsilon}(x) = 0$  for all  $\varepsilon, x$ .

Let us now move on to the proof of Theorem 3.1. By standard arguments, see [Ber15a], we need only prove that  $D_{\varepsilon}(\mathcal{O}) \to \mu'(\mathcal{O})$  in probability for each fixed  $\mathcal{O} \subset D$ . In fact, without loss of generality we may assume that  $\mathcal{O} := B(0,1)$  is the unit disc. From now on we will work with this assumption.

#### 3.3.1 A uniformly integrable family.

We know from [DRSV14b] that if  $\mu'$  is the critical Liouville measure,  $\mu'(\mathcal{O})$  has infinite expectation for any  $\mathcal{O} \subset D$ . Therefore, we cannot hope to have  $L^1$ -convergence or uniform integrability of  $D_{\varepsilon}(\mathcal{O})$ . Since we prefer to work with uniformly integrable families, we instead consider a sequence of cut-off approximations  $D_{\varepsilon}^{\beta}$  to  $D_{\varepsilon}$ . It will be very important to choose these cut-offs correctly, but for the right choice they *will* be uniformly integrable (for each  $\beta$ ) and moreover, will converge as  $\varepsilon \to 0$  (albeit in some slightly unusual sense, see Lemma 3.21). Obtaining the desired convergence in Theorem 3.1 then amounts to letting  $\beta \to \infty$  and using Lemma 3.5 to see that  $D_{\varepsilon}^{\beta}$  is actually very close to  $D_{\varepsilon}$  for large enough  $\beta$ . So, let us fix  $\varepsilon_0 > 0$ , such that  $B(x, \varepsilon) \subset D$  for every  $\varepsilon \leq \varepsilon_0$  and  $x \in \mathcal{O}$ . Then we define for  $\beta > 0$  and  $\varepsilon \in (0, \varepsilon_0]$ , the "cut-off" approximations

$$\begin{split} M^{\beta}_{\varepsilon}(\mathcal{O}) &:= \int_{\mathcal{O}} \mathrm{e}^{2h_{\varepsilon}(x) - 2\mathrm{var}(h_{\varepsilon}(x))} \, \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{-h_{\varepsilon}(x) + 2\mathrm{var}(h_{\varepsilon}(x)) + \beta > 1\}} \, dx; \text{ and} \\ D^{\beta}_{\varepsilon}(\mathcal{O}) &:= \int_{\mathcal{O}} (-h_{\varepsilon}(x) + 2\mathrm{var}(h_{\varepsilon}(x)) + \beta) e^{2h_{\varepsilon}(x) - \gamma^{2}/2\mathrm{var}(h_{\varepsilon}(x))} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{-h_{\varepsilon}(x) + 2\mathrm{var}(h_{\varepsilon}(x)) + \beta > 1\}} \, dx; \end{split}$$

where

$$L_{\varepsilon}(x) := \{-\tilde{h}_{\delta}(x) + 2\lambda_{\varepsilon}(x)\operatorname{var}(\tilde{h}_{\delta}(x)) + \beta - \rho_{\delta}^{\varepsilon}(x) > 0; \forall \delta \in [\varepsilon, \varepsilon_0]\}$$

Note that both  $M_{\varepsilon}^{\beta}(\mathcal{O})$  and  $D_{\varepsilon}^{\beta}(\mathcal{O})$  are positive by definition, and also that  $M_{\varepsilon}^{\beta}(\mathcal{O}) \leq D_{\varepsilon}^{\beta}(\mathcal{O})$ . For ease of notation we set

$$\begin{aligned} f_{\varepsilon,\gamma}^{\beta}(x) &= -h_{\varepsilon}(x) + \gamma \operatorname{var}(h_{\varepsilon}(x)) + \beta & g_{\varepsilon,\gamma}(x) = \gamma h_{\varepsilon}(x) - (\gamma^{2}/2) \operatorname{var}(h_{\varepsilon}(x)) \\ \tilde{f}_{\varepsilon,\gamma}^{\beta}(x) &= -\tilde{h}_{\varepsilon}(x) + \gamma \operatorname{var}(\tilde{h}_{\varepsilon}(x)) + \beta & \tilde{g}_{\varepsilon,\gamma}(x) = \gamma \tilde{h}_{\varepsilon}(x) - (\gamma^{2}/2) \operatorname{var}(\tilde{h}_{\varepsilon}(x)) \\ f_{\varepsilon,\gamma}^{\beta,Y}(x) &= -Y_{\varepsilon}(x) + \gamma \operatorname{var}(Y_{\varepsilon}(x)) + \beta & g_{\varepsilon,\gamma}^{Y}(x) = \gamma Y_{\varepsilon}(x) - (\gamma^{2}/2) \operatorname{var}(Y_{\varepsilon}(x)) \end{aligned}$$

recalling the definition of Y from Lemma 3.13. Then we have

$$M_{\varepsilon}^{\beta}(\mathcal{O}) := \int_{\mathcal{O}} \mathrm{e}^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}} \, dx \text{ and } D_{\varepsilon}^{\beta}(\mathcal{O}) := \int_{\mathcal{O}} f_{\varepsilon,2}^{\beta}(x) \mathrm{e}^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}>1\}} \, dx$$

and

$$L_{\varepsilon}(x) = \{ \tilde{f}^{\beta}_{\delta, 2\lambda_{\varepsilon}(x)}(x) - \rho^{\varepsilon}_{\delta} > 0 \,\forall \delta \in [\varepsilon, \varepsilon_0] \}$$

The decomposition

$$f_{\varepsilon,2}^{\beta}(\cdot) = \lambda_{\varepsilon}(\cdot)\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}(\cdot)}^{\beta}(\cdot) + f_{\varepsilon,2}^{0,Y}(\cdot) + (1 - \lambda_{\varepsilon}(\cdot))\beta \text{ and } g_{\varepsilon,2}(\cdot) = \tilde{g}_{\varepsilon,2\lambda_{\varepsilon}(\cdot)}(\cdot) + g_{\varepsilon,2}^{Y}(\cdot).$$

$$(3.11)$$

will also come in very useful it what follows.

**Proposition 3.17.** For fixed  $\beta > 0$ ,  $(D_{\varepsilon}^{\beta}(\mathcal{O}))_{\varepsilon \leq \varepsilon_0}$  is a uniformly integrable family.

*Proof.* The proof of this Lemma is inspired by that of Berestycki [Ber15a], who shows uniform integrability of  $\mu_{\varepsilon}^{\gamma}$  in the subcritical case. In analogy to his approach, for  $a \geq \varepsilon > 0$  we define the *good event* 

$$G^R_{\varepsilon,a}(x) :=$$

$$\left\{\frac{\sqrt{\log(1/u)}}{R\log(2+\log(1/u))^2} \le \tilde{f}_{u,2\lambda_{\varepsilon}(x)}^{\beta}(x) \le R(1+\sqrt{\log(1/u)\log(1+\log(1/u))}) \ \forall u \in [\varepsilon,a]\right\}$$

and write  $D_{\varepsilon}^{\beta}(\mathcal{O}) = J_{\varepsilon}^{\beta} + \hat{J}_{\varepsilon}^{\beta}$ , where  $J_{\varepsilon}^{\beta}$  is the integral over all "good" x, for which  $G_{\varepsilon,\varepsilon_0}^R(x)$  holds. <sup>5</sup> The rationale behind choosing G in this way is that it separates bad points of the field, which are "too thick" and make the second moment explode, from the good points.

To conclude, it is enough to prove the following two lemmas.

**Lemma 3.18.**  $\mathbb{E}[\hat{J}_{\varepsilon}^{\beta}] \leq p(R)$  for all  $\varepsilon \leq \varepsilon_0$  where  $p(R) \to 0$  as  $R \to \infty$ ;

**Lemma 3.19.** For fixed  $R, J_{\varepsilon}^{\beta}$  is uniformly bounded in  $L^2$ .

We first give a very rough idea of why these should hold:

•  $\mathbb{E}[\hat{J}_{\varepsilon}^{\beta}]$  corresponds to the probability of  $G_{\varepsilon,\varepsilon_0}^R(x)$  not holding under a weighted law: specifically, under the law with Radon–Nikodym derivative (with respect to  $\mathbb{P}$ ) proportional to

$$f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}}$$

Under this law we know that  $\tilde{f}_{u,2\lambda_{\varepsilon}(x)}^{\beta}(x)$  is (approximately) a Bessel process. Thus we know by Lemma 3.11 that this probability tends to 0 as  $R \to \infty$ .

• Now we move on to the  $L^2$  bound. Every time we write  $\approx$  it requires a lot of justification, usually because  $h_{\varepsilon}$  is not exactly a Brownian motion. First note that by the Markov property of the field and the fact that (3.9) is a martingale,

$$\mathbb{E}[f_{\varepsilon,2}^{\beta}(x)f_{\varepsilon,2}^{\beta}(y)\,\mathrm{e}^{g_{\varepsilon,2}(x)}\,\mathrm{e}^{g_{\varepsilon,2}(y)}\,\mathbb{1}_{L_{\varepsilon}(x)}\mathbb{1}_{L_{\varepsilon}(y)}] \approx \mathbb{E}[f_{\delta,2}^{\beta}(x)f_{\delta,2}^{\beta}(y)\,\mathrm{e}^{g_{\delta,2}(x)}\,\mathrm{e}^{g_{\delta,2}(y)}\,\mathbb{1}_{L_{\delta}(x)}\mathbb{1}_{L_{\delta}(y)}]$$

for  $x, y \in \mathcal{O}$ , where  $\delta = \delta(x, y) = (|x - y|/3) \vee \varepsilon$ . Now, on the event  $G^R_{\varepsilon, \varepsilon_0}(x) \cap G^R_{\varepsilon, \varepsilon_0}(y)$ ,

$$f_{\delta,2}^{\beta}(x) \approx \sqrt{\log(1/\delta)}, \ f_{\delta,2}^{\beta}(y) \approx \sqrt{\log(1/\delta)} \ \text{ and } g_{\delta,2}(y) \approx -2\sqrt{\log(1/\delta)} + 2\log(1/\delta).$$

We can use this to show that, roughly,

$$\begin{split} \mathbb{E}[f_{\delta,2}^{\beta}(x)f_{\delta,2}^{\beta}(y)\,\mathrm{e}^{g_{\delta,2}(x)}\,\mathrm{e}^{g_{\delta,2}(y)}\,\mathbbm{1}_{G^R_{\varepsilon,\varepsilon_0}(x)}\mathbbm{1}_{G^R_{\varepsilon,\varepsilon_0}(y)}] &\lesssim \delta^{-2}\log(1/\delta)\,\mathrm{e}^{-2\sqrt{\log(1/\delta)}}\,\mathbb{E}[\mathrm{e}^{g_{\delta,2}(x)}] \\ &= \delta^{-2}\log(1/\delta)\,\mathrm{e}^{-2\sqrt{\log(1/\delta)}}\,. \end{split}$$

We then only need to verify that this function of  $\delta(x, y)$  is integrable over  $\mathcal{O} \times \mathcal{O}$ .

We prove Lemmas 3.18 and 3.19 below. As already mentioned, there are several technical difficulties with making the above argument rigorous.

<sup>&</sup>lt;sup>5</sup>Note that we are setting  $a = \varepsilon_0$  here, but we define the more general notation  $G_{\varepsilon,a}^R$  for use later on.

Proof of Lemma 3.18. Consider for  $x \in \mathcal{O}$ 

$$\mathbb{E}\left[f_{\varepsilon,2}^{\beta}(x)e^{g_{\varepsilon,2}(x)}\mathbb{1}_{L_{\varepsilon}(x)}\mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}}\mathbb{1}_{G_{\varepsilon,\varepsilon_{0}}^{R}(x)^{c}}\right].$$
(3.12)

To prove the lemma, we need to show that this converges to 0 as  $R \to \infty$ , uniformly in  $\varepsilon$  and x. The strategy is to rewrite it as an expectation with respect to a different measure, under which we understand well the behaviour of  $\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}(x)$ . We set

$$\frac{d\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}}{d\mathbb{P}} = (\tilde{Z}_{\varepsilon}^{\beta}(x))^{-1} \tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)}; \quad \tilde{Z}_{\varepsilon}^{\beta}(x) = \mathbb{E}\left[\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)}\right].$$

This measure will be extremely important throughout the paper because, under  $\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}$ , the process

$$\{\tilde{f}_{u,2\lambda_{\varepsilon}}^{\beta}(x)-\rho_{u}^{\varepsilon}(x);\,u\in[\varepsilon,\varepsilon_{0}]\}$$

is a time changed 3d Bessel process. To see why this is true, we split the weighting that defines  $\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}$  into two steps. By decomposition (3.11) we have  $g_{\varepsilon,2}(x) = \tilde{g}_{\varepsilon,2\lambda_{\varepsilon}}(x) + g_{\varepsilon,2}^{Y}(x)$  and so we can first consider what happens if we only weight by  $\exp(g_{\varepsilon,2}^{Y}(x))$ . Let us call this intermediate law  $\hat{\mathbb{P}}$ . By the Cameron–Martin–Girsanov theorem, and the definition  $\rho_{\delta}^{\varepsilon}(x) := -2 \operatorname{cov}(\tilde{h}_{\delta}(x), Y_{\varepsilon}(x))$ , the process

$$-h_{\delta}(x) - \rho_{\delta}^{\varepsilon}(x)$$

is a time changed Brownian motion under  $\hat{\mathbb{P}}$ . For the second step in the weighting we use the definition of  $L_{\varepsilon}(x)$ , and the fact that  $\rho_{\varepsilon}^{\varepsilon}(x) = 0$ . This means that this second step is simply the Bessel process weighting described in Section 3.2.6, with  $\gamma = 2\lambda_{\varepsilon}(x)$ . The same argument also implies that  $\tilde{Z}_{\varepsilon}^{\beta}(x)$  does not depend on  $\varepsilon$  for each x, since (3.9) is a martingale.

To prove the lemma, and we will apply this technique over and over again, we rewrite (3.12) as

$$\tilde{Z}_{\varepsilon}^{\beta}(x)\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[\frac{f_{\varepsilon,2}^{\beta}(x)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)}\mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}>1\}}\mathbbm{1}_{G_{\varepsilon,\varepsilon_{0}}^{R}(x)^{c}}\right]$$

where by Lemma 3.11 part (4) we know that  $\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}(G_{\varepsilon,\varepsilon_0}^R(x)^c) \to 0$  as  $R \to \infty$ , uniformly in  $\varepsilon$  and x (using the uniform boundedness of  $(\rho_{\delta}^{\varepsilon}(x))_{\delta > \varepsilon}$ .)

Using the fact that

$$f_{\varepsilon,2}^{\beta}(x) = \lambda_{\varepsilon}(x)\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}(x)}^{\beta}(x) + \mathcal{O}(1) - Y_{\varepsilon}(x),$$

(cf. decomposition (3.11)), and Hölder's inequality, it is enough for us to show that

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \left( \frac{|Y_{\varepsilon}(x)| + \mathcal{O}(1)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} \right)^{3/2} \right] = \mathcal{O}(1).$$

However, this follows by Cauchy–Schwarz, since

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \left( \frac{|Y_{\varepsilon}(x)| + \mathcal{O}(1)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} \right)^{3/2} \right]^{2} \leq \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{(|Y_{\varepsilon}(x)| + \mathcal{O}(1))^{3}}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} \right] \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{1}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)^{2}} \right]$$

and

Q
<sup>β,ε</sup><sub>x</sub>[(f
<sup>β</sup><sub>ε,2λε</sub>(x))<sup>-2</sup>] is bounded by Lemma 3.11, part (2);
 Q
<sup>β,ε</sup><sub>x</sub>[|Y<sub>ε</sub>(x)|<sup>p</sup>/f
<sup>β</sup><sub>ε,2λε</sub>(x)] ≤ E[|Y<sub>ε</sub>(x)|<sup>p</sup> e<sup>g<sub>ε,2</sub>(x)</sup>] is bounded for p = 1, 2, 3.

Proof of Lemma 3.19. First, we make the simple bound

$$\mathbb{E}[(J_{\varepsilon}^{\beta})^{2}] \leq \iint_{\mathcal{O}^{2}} \mathbb{E}\left[|f_{\varepsilon,2}^{\beta}(x)||f_{\varepsilon,2}^{\beta}(y)| e^{g_{\varepsilon,2}(x)} e^{g_{\varepsilon,2}(y)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{L_{\varepsilon}(y)} \mathbb{1}_{G_{\varepsilon,\varepsilon_{0}}^{R}(x)} \mathbb{1}_{G_{\varepsilon,\varepsilon_{0}}^{R}(y)}\right] dy dx$$

$$(3.13)$$

and fix some  $x \in \mathcal{O}$ . For this fixed x, we will break the integral over y into two parts: those with  $|x - y| > 3\varepsilon$ , and those with  $|x - y| \le 3\varepsilon$ . Let us begin with the first case. For such a y, we set  $\delta = \delta(x, y) := |x - y|/3$ , so that the  $\delta$ -balls around x and y are disjoint. We are going to use the fact that the circle averages around x and ydecorrelate after this time. More precisely, if we let  $\mathcal{H}$  be the  $\sigma$ -algebra generated by  $h|_{D \setminus (B(x,\delta) \cap B(y,\delta))}$ , then we have the following observations, which we state as a lemma.

**Lemma 3.20.** (1) Conditionally on  $\mathcal{H}$ , the processes  $(\tilde{h}_{\delta e^{-t}}(x) - \tilde{h}_{\delta}(x))_{t \geq 0}$  and  $(\tilde{h}_{\delta e^{-t}}(y) - \tilde{h}_{\delta}(y))_{t \geq 0})$  are independent Brownian motions.

- (2) We can write  $Y_{\varepsilon}(x) = Y_{\varepsilon}^{1}(x) + Y_{\varepsilon}^{2}(x)$  and  $Y_{\varepsilon}(y) = Y_{\varepsilon}^{1}(y) + Y_{\varepsilon}^{2}(y)$  where:
  - $Y^1_{\varepsilon}(x)$  and  $Y^1_{\varepsilon}(y)$  are measurable with respect to  $\mathcal{H}$ ;
  - $Y_{\varepsilon}^2(x)$  is independent of  $\mathcal{H}$ ,  $Y_{\varepsilon}^2(y)$  and  $(\tilde{h}_{\eta}(y) \tilde{h}_{\delta}(y))_{\eta \leq \delta}$ ;
  - $Y^2_{\varepsilon}(y)$  is independent of  $\mathcal{H}$ ,  $Y^2_{\varepsilon}(x)$  and  $(\tilde{h}_{\eta}(x) \tilde{h}_{\delta}(x))_{\eta \leq \delta}$ ;
  - $Y^i_{\varepsilon}(x), Y^i_{\varepsilon}(y)$  for i = 1, 2 have bounded variance; and
  - $2 \operatorname{cov}(Y_{\varepsilon}^2(x), \tilde{h}_{\eta}(x) \tilde{h}_{\delta}(x)) = -\rho_{\eta}^{\varepsilon}(x) + \rho_{\delta}^{\varepsilon}(x)$  (similarly if x is replaced with y).

(3) We have

$$\mathbb{E}[\tilde{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mid \mathcal{H}] = (\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}} - \rho^{\varepsilon}_{\delta}(x)) \mathbb{1}_{\{\tilde{f}_{\eta,2\lambda_{\varepsilon}} - \rho^{\varepsilon}_{\eta}(x) > 0; \forall \eta \in [\delta,\varepsilon_0]\}} \times e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}}(x)} e^{2Y_1^{\varepsilon}(x) - 2\operatorname{var}(Y^1_{\varepsilon}(x))} e^{2\rho^{\varepsilon}_{\delta}(x)}.$$

- (4) We also have  $\mathbb{E}[|Y_{\varepsilon}^{2}(x)| e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} | \mathcal{H}] \leq C e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}}(x)} e^{2Y_{\varepsilon}^{1}(x)}$  where C is a universal constant.
- (5) Items (3) and (4) also hold when x is replaced by y.

Proof of Lemma 3.20. By the Markov property of the Gaussian free field, conditionally on  $\mathcal{H}$  we can write  $h = h^{\mathcal{H}} + h_{\mathcal{H}}$  where:

- $h_{\mathcal{H}}$  is measurable with respect to  $\mathcal{H}$  and harmonic when restricted to  $B(x, \delta) \cup B(y, \delta)$ ; and
- h<sup>H</sup>, independent of H, is a sum of two independent zero boundary GFFs: one in B(x, δ) and one in B(y, δ).

We use this to prove the points in turn.

(1) This follows from the fact that  $h_{\mathcal{H}}(x) = \tilde{h}_{\delta}(x)$  and  $h_{\mathcal{H}}(y) = \tilde{h}_{\delta}(y)$  (by harmonicity), and the fact that the circle average process of a Gaussian free field is a Brownian motion.

(2) We have  $Y_{\varepsilon}(x) = h_{\varepsilon}(x) - \lambda_{\varepsilon}(x)\tilde{h}_{\varepsilon}(x)$  by definition and so we can write  $Y_{\varepsilon}^{1}(x) = (h_{\mathcal{H}}, \theta_{\varepsilon,x}) - \lambda_{\varepsilon}(x)(h_{\mathcal{H}}, \tilde{\theta}_{\varepsilon,x})$  and  $Y_{\varepsilon}^{2} = (h^{\mathcal{H}}, \theta_{\varepsilon,x}) - \lambda_{\varepsilon}(x)(h^{\mathcal{H}}, \tilde{\theta}_{\varepsilon,x})$ , where  $\tilde{\theta}$  is uniform measure on the unit circle. The claimed properties of this decomposition are easy to see.

(3) We first take out the  $\mathcal{H}$ -measurable parts from the conditional expectation on the left-hand side. To this end we write for  $\eta < \delta$ 

$$\tilde{f}^{\beta}_{\eta,2\lambda_{\varepsilon}}(x) = \tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}}(x) - \rho^{\varepsilon}_{\delta} - (\tilde{h}_{\eta}(x) - \tilde{h}_{\delta}(x)) + 2\log(\delta/\eta) + \rho^{\varepsilon}_{\delta}$$
$$:= W - (\tilde{h}_{\eta}(x) - \tilde{h}_{\delta}(x)) + 2\log(\delta/\eta) + \rho^{\varepsilon}_{\delta}$$

where W is  $\mathcal{H}$  measurable. We can also write

$$\begin{split} L_{\varepsilon}(x) &= \{ \tilde{f}_{\eta,2\lambda_{\varepsilon}}^{\beta}(x) - \rho_{\eta}^{\varepsilon} > 0 \,\forall \, \eta \in [\delta,\varepsilon_{0}] \} \cap \{ \tilde{f}_{\eta,2\lambda_{\varepsilon}}^{\beta}(x) - (\rho_{\eta}^{\varepsilon} - \rho_{\delta}^{\varepsilon}) + W > 0 \,\forall \, \eta \in [\varepsilon,\delta] \} \\ &:= L_{\varepsilon}^{1} \cap L_{\varepsilon}^{2} \end{split}$$

where  $L^1_{\varepsilon}$  is also  $\mathcal{H}$ -measurable. Putting these together, and breaking up  $g_{\varepsilon,2}(x)$ 

using (3.11) and point (2), we see that

$$\begin{split} & \mathbb{E}[\tilde{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mid \mathcal{H}] = e^{2\rho^{\varepsilon}_{\delta}(x)} e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}}(x)} e^{2Y^{1}_{\varepsilon}(x) - 2\operatorname{var}(Y^{1}_{\varepsilon}(x))} \mathbb{1}_{L^{1}_{\varepsilon}(x)} \times \\ & (\mathbb{E}[(W - (\tilde{h}_{\eta}(x) - \tilde{h}_{\delta}(x)) + 2\log(\delta/\varepsilon) + \rho^{\varepsilon}_{\delta}) \\ & \times e^{2(\tilde{h}_{\eta}(x) - \tilde{h}_{\delta}(x)) - 2\log(\delta/\eta) - 2\rho^{\varepsilon}_{\delta}(x)} e^{2Y^{2}_{\varepsilon}(x) - 2\operatorname{var}(Y^{2}_{\varepsilon}(x))} \mathbb{1}_{L^{2}_{\varepsilon}(x)} \mid \mathcal{H}]) \end{split}$$

Now we can use Girsanov's theorem, as in the proof of Lemma 3.18, to get rid of the  $\exp\{2Y_{\varepsilon}^2(x) - 2\operatorname{var}(Y_{\varepsilon}^2(x))\}$  term. More precisely, changing measure by  $\exp\{2Y_{\varepsilon}^2(x) - 2\operatorname{var}(Y_{\varepsilon}^2(x))\}$  has the effect of shifting the law of  $(\tilde{h}_{\eta}(x) - \tilde{h}_{\delta}(x))_{\eta \in [\varepsilon, \delta]}$ by adding on the deterministic function  $\rho_{\delta}^{\varepsilon}(x) - \rho_{\eta}^{\varepsilon}(x)$ . We then see that the conditional expectation above is nothing but the expectation of the Brownian motion martingale (3.9), starting from W. The result follows.

(4) For this we bound the indicator above by 1, and take out the parts which are measurable with respect to  $\mathcal{H}$  as in part (3). Then we are left with the expectation of  $|Y_{\varepsilon}^{2}(x)|$  under a shifted law, where  $Y_{\varepsilon}^{2}(x)$  is still a Gaussian with O(1) mean and variance (since  $Y_{\varepsilon}^{2}(x)$  has bounded covariances with everything.) This proves the claim.

This lemma allows us to deduce that the integrand of (3.13), in the case  $|x-y| > 3\varepsilon$ , is less than or equal to some constant, depending on  $\beta$  only, times

$$\mathbb{E}\left[ (\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}(x)}(x) + 1 + |Y^{1}_{\varepsilon}(x)|) (\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}(y)}(y) + 1 + |Y^{1}_{\varepsilon}(y)|) \right. \\ \left. \times \mathrm{e}^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(x)}(x)} \mathrm{e}^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(y)}(y)} \mathrm{e}^{2Y^{1}_{\varepsilon}(x)} \mathrm{e}^{2Y^{1}_{\varepsilon}(y)} \, \mathbb{1}_{G^{R}_{\delta,\delta}(x)} \mathbb{1}_{G^{R}_{\delta,\delta}(y)} \right]$$

where  $\delta = \delta(x, y) = |x - y|/3$ . Here we used that  $\rho_{\delta}^{\varepsilon}(\cdot)$  and  $\operatorname{var}(Y_{\varepsilon}^{1}(\cdot))$  are uniformly bounded, and changed  $G_{\varepsilon,\varepsilon_{0}}^{R}(\cdot)$  to the larger  $\mathcal{H}$ -measurable event  $G_{\delta,\delta}^{R}(\cdot)$ , so that it would not interfere with the conditioning step.

Now we can use the definition of  $G^R_{\delta,\delta}$ . This, together with the fact that  $Y^1_{\varepsilon}(\cdot)$  has bounded variance and covariance with everything, tells us that the above is bounded by a constant times

$$\delta^{-2}F_R(\log(1/\delta))$$
;  $F_R(z) := R^2(1 + \sqrt{z\log(1+z)})^2 e^{-\frac{2\sqrt{z}}{R(\log(2+z))}}$ 

As in the sketch of this proof (given just after the statement of Lemmas 3.18 and 3.19) we have put deterministic bounds on  $\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}(x)}(x)$ ,  $\tilde{f}_{\delta,2\lambda_{\varepsilon}(y)}(y)$  and  $e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(y)}(y)}$ , and integrated over  $e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(x)}(x)}$ .

Hence we can bound the integral (3.13), restricted to the set  $x \in \mathcal{O}, y \in \mathcal{O} \setminus$ 

 $B(x, 3\varepsilon)$ , by a multiple of

$$\int_{x\in\mathcal{O}}\int_{y\notin B(x,3\varepsilon)}\frac{1}{|x-y|^2}F_R(-\log|x-y|)\,dy\,dx\leq C\int_{x\in\mathcal{O}}\int_0^{\log(1/\varepsilon)}F_R(u)du\,dx.$$

Since  $F_R$  is integrable we see that this is uniformly bounded in  $\varepsilon$ .

Finally, we must deal with the integral over the set  $x \in \mathcal{O}$ ,  $y \in B(x, 3\varepsilon)$ . By the same reasoning as above (although now we do not need to do any conditioning, since  $\delta(x, y) = \varepsilon$ ) we see that the integrand on this region is less than some constant times  $\varepsilon^{-2}F_R(\log(1/\varepsilon))$ . That the integral is uniformly bounded in  $\varepsilon$  then follows from that fact that  $F_R(\log(1/\varepsilon))$  is bounded, and that the area of  $B(x, 3\varepsilon)$  is  $O(\varepsilon^2)$ .

#### 3.3.2 Convergence

We now need to show that  $D_{\varepsilon}^{\beta}(\mathcal{O})$  converges (in some sense) as  $\varepsilon \to 0$ . To do this, we define the change of measure

$$\frac{d\mathbb{Q}^{\beta,\varepsilon}}{d\mathbb{P}} = \frac{D_{\varepsilon}^{\beta}(\mathcal{O})}{\mathbb{E}[D_{\varepsilon}^{\beta}(\mathcal{O})]}$$
(3.14)

for each  $\varepsilon > 0$ . Note that this is not a martingale change of measure, but it is well defined for each  $\varepsilon > 0$ . We will prove the following lemma (from now on we drop the dependence on  $\mathcal{O}$  from our notation for compactness.)

**Proposition 3.21.** For each fixed  $\beta$  and  $\varepsilon_0$ , and for any  $\delta > 0$ 

$$\mathbb{Q}^{\beta,\varepsilon}\left[\left|\frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}}\sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}}\right| > \delta\right] \to 0$$

as  $\varepsilon \to 0$ .

**Remark 3.22.** Since  $D_{\varepsilon}^{\beta}$  and  $M_{\varepsilon}^{\beta}$  are close to  $D_{\varepsilon}$  and  $M_{\varepsilon}$  for large  $\beta$  (Lemma 3.5) and  $\mathbb{Q}^{\beta,\varepsilon}$  is defined by a uniformly integrable change of measure (Proposition 3.17) this is almost exactly what we need (recall that by Theorem 3.10 we have  $M_{\varepsilon}\sqrt{\log(1/\varepsilon)} \rightarrow \sqrt{\pi/2}\mu'$  as  $\varepsilon \rightarrow 0$ .) Indeed, we will see that the proof of Theorem 3.1 follows in a straightforward manner once we have completed the proof of Proposition 3.21.

**Remark 3.23.** The proof of Proposition 3.21 follows the general outline of the main proof in [AS14]. However, the details of each step are somewhat different, and rely on the precise way we have constructed  $D_{\varepsilon}^{\beta}$ . One of the main difficulties is to make

exact statements about the behaviour of  $h_{\varepsilon}$  using what we know about the behaviour of  $\tilde{h}_{\varepsilon}$ .

Before starting the proof, we make a few remarks about the change of measure (3.14). Define

$$\hat{\mathbb{Q}}^{\beta,\varepsilon}(dx,dh) = \frac{f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{\{x \in \mathcal{O}\}} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}} dx \mathbb{P}[dh]}{\mathbb{E}[D_{\varepsilon}^{\beta}]}$$

to be the *rooted measure* on (h, x) where h is a field and x is a point in  $\mathcal{O}$ . Introducing this type of measure is a classical tool for dealing with branching processes, that also comes in very useful in the context of Gaussian multiplicative chaos. We have the following description of how the point x and the field h interact under  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$ :

- the marginal law of h under  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  is  $\mathbb{E}[D_{\varepsilon}^{\beta}]^{-1}D_{\varepsilon}^{\beta}d\mathbb{P}$  (i.e. the same law as under  $\mathbb{Q}^{\beta,\varepsilon}$ );
- the marginal law of x under  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$ , that we shall call  $dm^{\beta,\varepsilon}(x)$ , is proportional to

$$Z_{\varepsilon}^{\beta}(x) := \mathbb{1}_{\{x \in \mathcal{O}\}} \mathbb{E}[f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) > 1\}}].$$

• the conditional law of the field h given the point x is given by

$$\mathbf{Q}_x^{\beta,\varepsilon} := \hat{\mathbb{Q}}^{\beta,\varepsilon}(\cdot \mid x) = (Z_{\varepsilon}^{\beta}(x))^{-1} f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{\{x \in \mathcal{O}\}} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) > 1\}} d\mathbb{P}.$$

• the conditional law of the point x given the field h is proportional to

$$f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{\{x \in \mathcal{O}\}} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) > 1\}} dx.$$

Also note that

$$\frac{d\mathbf{Q}_x^{\beta,\varepsilon}}{d\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}} = \frac{\tilde{Z}_{\varepsilon}^{\beta}(x)}{Z_{\varepsilon}^{\beta}(x)} \frac{f_{\varepsilon,2}^{\beta}(x)}{\tilde{f}_{\varepsilon,2}^{\beta}(x)} \mathbbm{1}_{\{f_{\varepsilon,2\lambda_{\varepsilon}(x)}^{\beta}(x) \ge 1\}}$$

where we recall from the proof of Lemma 3.18 that under  $\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}$  the process

$$\{(\tilde{f}_{e^{-u},2\lambda_{\varepsilon}}^{\beta}(x)-\rho_{e^{-u}}^{\varepsilon}(x)\,;\,u\in[\log(1/\varepsilon_{0}),\log(1/\varepsilon)]\}$$

has the law of a 3d Bessel process, whose starting point is also biased (and a.s. positive.) In fact, one of the key ideas in the proof of Proposition 3.21 will be to say that  $f_{\varepsilon}^{\beta}(x)$  under  $\mathbf{Q}_{x}^{\beta,\varepsilon}$  also behaves essentially like a Bessel process. As a warm up, let us first prove the following:

#### Lemma 3.24.

$$\frac{Z_{\varepsilon}^{\beta}(x)}{\tilde{Z}_{\varepsilon}^{\beta}(x)} \to 1$$
(3.15)

uniformly in x as  $\varepsilon \to 0$ .

This justifies in some sense that the measures  $\mathbf{Q}_x^{\beta,\varepsilon}$  and  $\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}$  are similar for small  $\varepsilon$ , and is a result we will use many times.

*Proof.* We consider the ratio

$$Z_{\varepsilon}^{\beta}(x)/\tilde{Z}_{\varepsilon}^{\beta}(x) = (\tilde{Z}_{\varepsilon}^{\beta})^{-1}\mathbb{E}[f_{\varepsilon,2}^{\beta}(x) e^{g_{\varepsilon,2}(x)} \mathbb{1}_{L_{\varepsilon}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}}]$$
$$= \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ (f_{\varepsilon,2}^{\beta}(x)/\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)) \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}>1\}} \right].$$

To show this converges to 1 we write, using decomposition (3.11),

$$\frac{f_{\varepsilon,2}^{\beta}(x)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} = 1 + \mathrm{o}(1) + \frac{\mathrm{O}(1) - Y_{\varepsilon}(x)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)}$$

Then by exactly the same Cauchy–Schwarz argument as in the proof of Lemma 3.18, it is enough to show that

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}[f_{\varepsilon,2}^{\beta}(x) \le 1] \to 0$$
(3.16)

uniformly in x. Since  $\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)$  is close to  $\sqrt{\log(1/\varepsilon)}$  with high probability under  $\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}$ , and  $f_{\varepsilon,2}^{\beta}(x) = \lambda_{\varepsilon}(x)\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) + O(1) + Y_{\varepsilon}(x)$ , it is sufficient to control the tails of  $Y_{\varepsilon}(x)$  under  $\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}$ . For this we observe that, by Cauchy–Schwarz again,

$$\widetilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}[\mathbb{1}_{\{Y_{\varepsilon}(x)>a\}}]^{2} \leq \widetilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}[\widetilde{f}_{\varepsilon,2\lambda_{\varepsilon}}(x)] \mathbb{E}[\mathbb{1}_{\{Y_{\varepsilon}(x)>a\}} e^{\widetilde{g}_{\varepsilon,2\lambda_{\varepsilon}}(x)} e^{g_{\varepsilon,2}^{Y}(x)}] \qquad (3.17)$$

$$\leq \sqrt{\log(1/\varepsilon)} e^{-ka}$$

for some k, since  $Y_{\varepsilon}(x)$  is Gaussian with bounded variance under  $\mathbb{P}$ . This allows us to conclude.

*Proof of Proposition 3.21.* Our strategy to prove Proposition 3.21 is to show the following two things:

$$\mathbb{Q}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \right] = \sqrt{\frac{2}{\pi \log(1/\varepsilon)}} + o\left(\frac{1}{\sqrt{\log(1/\varepsilon)}}\right) \text{ as } \varepsilon \to 0; \text{ and}$$
(3.18)

$$\mathbb{Q}^{\beta,\varepsilon}\left[\left(\frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}}\right)^{2}\right] \leq \frac{2}{\pi \log(1/\varepsilon)} + o\left(\frac{1}{\log(1/\varepsilon)}\right) \text{ as } \varepsilon \to 0.$$
(3.19)

The result then follows using Jensen's and Markov's inequalities. (4.12) is relatively straightforward. Observe that, by the discussion preceeding this proof, we have

$$\frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} = \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mid h \right].$$
(3.20)

This means that

$$\mathbb{Q}^{\beta,\varepsilon}\left[M^{\beta}_{\varepsilon}/D^{\beta}_{\varepsilon}\right] = \hat{\mathbb{Q}}^{\beta,\varepsilon}\left[M^{\beta}_{\varepsilon}/D^{\beta}_{\varepsilon}\right] = \hat{\mathbb{Q}}^{\beta,\varepsilon}[f^{\beta}_{\varepsilon,2}(x)^{-1}] = \int_{\mathcal{O}} \mathbf{Q}^{\beta,\varepsilon}_{x}[f^{\beta}_{\varepsilon,2}(x)^{-1}] \, dm^{\beta,\varepsilon}(x),$$

which is a useful representation, because we can write

$$\mathbf{Q}_{x}^{\beta,\varepsilon}[f_{\varepsilon}^{\beta}(x)^{-1}] = \frac{\tilde{Z}_{\varepsilon}^{\beta}(x)}{Z_{\varepsilon}^{\beta}(x)} \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}[\tilde{f}_{\varepsilon,2}^{\beta}(x)^{-1}\mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}}]$$

for each  $x \in \mathcal{O}$ . The first moment estimate then follows by (3.16), Lemma 3.15 and Lemma 3.11, parts (1) and (2).

(3.19) is rather more difficult, and requires several steps.

**Step 1**: We show that restricting to an event of high probability under  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  does not affect our second moment too much. That is, we show that if we can find a sequence of events  $E_{\varepsilon} = E_{\varepsilon}(x)$  with  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}] \to 1$  and

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{E_{\varepsilon}} \right] \le \frac{2}{\pi \log(1/\varepsilon)} + o\left(\frac{1}{\log(1/\varepsilon)}\right), \tag{3.21}$$

then this will prove (3.19).

To see how this implies (3.19), take such an event  $E_{\varepsilon}$  and write

$$\begin{split} \mathbb{Q}^{\beta,\varepsilon} \left[ \left( \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \right)^{2} \right] &= \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \left( \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \right)^{2} \right] \\ &= \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \hat{\mathbb{Q}}^{\beta,\varepsilon} [f_{\varepsilon,2}^{\beta}(x)^{-1} \mathbb{1}_{E_{\varepsilon}} \mid h] \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \hat{\mathbb{Q}}^{\beta,\varepsilon} [f_{\varepsilon,2}^{\beta}(x)^{-1} \mathbb{1}_{E_{\varepsilon}^{c}} \mid h] \right] \\ &= \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{E_{\varepsilon}} \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \hat{\mathbb{Q}}^{\beta,\varepsilon} [f_{\varepsilon,2}^{\beta}(x)^{-1} \mathbb{1}_{E_{\varepsilon}^{c}} \mid h] \right]. \end{split}$$

We would like to show that the second term in the final expression is  $o(\log(1/\varepsilon)^{-1})$ . For this, it is enough by Cauchy–Schwarz, to show that

•  $\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ (M_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta})^2 \right] = \mathcal{O}(\log(1/\varepsilon)^{-1}), \text{ and}$ •  $\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \xi_{\varepsilon}^2 \right] = \mathcal{O}(\log(1/\varepsilon)^{-1}) \text{ where } \xi_{\varepsilon} := \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ f_{\varepsilon,2}^{\beta}(x)^{-1} \mathbb{1}_{E_{\varepsilon}^c} \mid h \right].$  We deal with each point in turn. For the first point, note that by conditional Jensen's inequality we have

$$\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[ (M_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta})^2 \right] \leq \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ f_{\varepsilon,2}^{\beta}(x)^{-2} \right] = \int_{\mathcal{O}} \mathbf{Q}_x^{\beta,\varepsilon} \left[ f_{\varepsilon,2}^{\beta}(x)^{-2} \right] \, dm^{\beta,\varepsilon}(x)$$

and then by changing measure and rearranging as usual, we can write

$$\mathbf{Q}_{x}^{\beta,\varepsilon}\left[f_{\varepsilon,2}^{\beta}(x)^{-2}\right] = \left(\tilde{Z}_{\varepsilon}^{\beta}(x)/Z_{\varepsilon}^{\beta}(x)\right)\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[\frac{1}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)f_{\varepsilon,2}^{\beta}(x)}\mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}}\right]$$

To show that this is  $O(\log(1/\varepsilon)^{-1})$  we need to be a little bit careful, although heuristically it is clear from the fact that  $\tilde{f}$  is a Bessel process and  $Y_{\varepsilon}(x)$  is small. The way to make this precise is to consider the expectation on the "good" event,

$$\{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}>\log(1/\varepsilon)^{1/4}\}\cap\{Y_{\varepsilon}(x)<(1/2)\log(1/\varepsilon)^{1/4}\}$$

and its complement separately. On the good event we have that  $f_{\varepsilon,2}^{\beta}(x) \geq c \tilde{f}_{\varepsilon,\lambda_{\varepsilon}}^{\beta}(x)$  for some constant c, and so the expectation is  $O(\log(1/\varepsilon)^{-1})$  by Lemma 3.11, part (2). On the bad event, we use (3.16) and Lemma 3.11, part (5), to see that the expectation is also  $O(\log(1/\varepsilon)^{-1})$ .

Now we treat the second point. By Jensen's inequality, and for any a > 0, we have

$$\begin{split} \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \xi_{\varepsilon}^{2} \right] &\leq \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ f_{\varepsilon}^{\beta}(x)^{-2} \mathbb{1}_{E_{\varepsilon}^{c}} \right] \\ &= \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{\mathbb{1}_{E_{\varepsilon}^{c}}}{f_{\varepsilon,2}^{\beta}(x)^{2}} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) \geq a\sqrt{\log(1/\varepsilon)}\}} \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{1}{f_{\varepsilon,2}^{\beta}(x)^{2}} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) < a\sqrt{\log(1/\varepsilon)}\}} \right] \end{split}$$

It is clear by definition that the first term is less than  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}^{c}]/(a^{2}\log(1/\varepsilon))$ , and for the second term, we write it as

$$\int_{\mathcal{O}} \frac{\tilde{Z}_{\varepsilon}^{\beta}(x)}{Z_{\varepsilon}^{\beta}(x)} \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{1}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) > 1\}} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) \geq a\sqrt{\log(1/\varepsilon)}\}} \right] dm^{\beta,\varepsilon}(x). \quad (3.22)$$

Similarly to before, we consider the expectation on the event

$$\{Y_{\varepsilon}(x) < a \log(1/\varepsilon)^{1/4}\} \cap \{\tilde{f}_{2,\lambda_{\varepsilon}}^{\beta}(x) > a \log(1/\varepsilon)^{1/4}\}$$

and its complement separately. This allows us to bound (3.22) by some constant times  $a \log(1/\varepsilon)^{-1} + \exp(-ak \log(1/\varepsilon)^{1/4})$ , where k > 0 is the constant from (3.17). Thus,  $\limsup_{\varepsilon} \log(1/\varepsilon) \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[\xi_{\varepsilon}^{2}\right] \leq Ca$  for any a > 0 and some fixed finite C. Taking  $a \to 0$ , this allows us to conclude step 1.

**Step 2**: We define the event  $E_{\varepsilon}$ , and set up the scales for the multiscale argument we will use.

To do this we let  $r_{\varepsilon} > \varepsilon$  be a sequence with

$$\frac{\log(1/r_{\varepsilon})}{\log(1/\varepsilon)^{1/3}} \to \infty \text{ and } \frac{\log(1/r_{\varepsilon})}{\log(1/\varepsilon)^{1/2}} \to 0$$
(3.23)

as  $\varepsilon \to 0$  (so  $r_{\varepsilon}$  is tending to 0 much slower than  $\varepsilon$ ). Given this, we break up D and M as

$$D_{\varepsilon}^{\beta} = D_{\varepsilon}^{\beta,in} + D_{\varepsilon}^{\beta,out} \quad \text{and} \quad M_{\varepsilon}^{\beta} = M_{\varepsilon}^{\beta,in} + M_{\varepsilon}^{\beta,out}$$

where the subscript *in* refers to the integral inside  $B(x, r_{\varepsilon})$  and the subscript *out* refers to the integral outside of it.

The basic idea is that  $D_{\varepsilon}^{\beta,in}$  and  $M_{\varepsilon}^{\beta,in}$  will be small with high probability (this will be part of the definition of  $E_{\varepsilon}$ ) and on this event,  $M_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta}$  will be close to  $M_{\varepsilon}^{\beta,out}/D_{\varepsilon}^{\beta,out}$ . Heuristically, this occurs with high probability because the limits of  $M_{\varepsilon}$  and  $D_{\varepsilon}$  should be atomless measures, and  $r_{\varepsilon}$  is tending to 0. Next, we claim that  $M_{\varepsilon}^{\beta,out}/D_{\varepsilon}^{\beta,out}$  is essentially independent of  $f_{\varepsilon,2}^{\beta}(x)$ . This is because  $r_{\varepsilon}$  is much larger than  $\varepsilon$  and f is approximately a (time changed) Bessel process, so its value at time  $\varepsilon$  is basically independent of its value at time  $r_{\varepsilon} - \varepsilon$ . From here (3.21) follows, since we already know that  $M_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta}$  and  $f_{\varepsilon,2}^{\beta}(x)^{-1}$  have (the same) expectation, of the right order.

We now choose our event  $E_{\varepsilon}$ , according to this plan. To do this, we first have to observe that, by the Markov property of the field,  $Y_{\varepsilon}(x) = h_{\varepsilon}(x) - \lambda_{\varepsilon}(x)\tilde{h}_{\varepsilon}(x)$  can be written as

$$Y_{\varepsilon}(x) := Y_{\varepsilon}^{1}(x) + Y_{\varepsilon}^{2}(x),$$

where  $Y_{\varepsilon}^2$  is independent of  $h|_{D\setminus B(x,r_{\varepsilon}-\varepsilon)}$  and  $Y_{\varepsilon}^1$  is measurable with respect to  $h|_{D\setminus B(x,r_{\varepsilon}-\varepsilon)}$  (see the proof of Lemma 3.20 for a more detailed explanation.) Given this definition, we set  $E_{\varepsilon} = E_{\varepsilon}^1 \cap E_{\varepsilon}^2$  where

$$E_{\varepsilon}^{1} = \{D_{\varepsilon}^{\beta,in} \leq \log(1/\varepsilon)^{-2}\}$$
$$E_{\varepsilon}^{2} = \{\tilde{f}_{u,2\lambda_{\varepsilon}}^{\beta}(x) \in [\sqrt[3]{\log(1/u)}, \log(1/u)] \ \forall u \in [r_{\varepsilon} - \varepsilon, r_{\varepsilon}]\} \cap \{Y_{\varepsilon}^{1}(x) < (\log(1/\varepsilon))^{1/4}\}$$

(we will prove that  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}] \to 1$  later on.) We remark that the event  $E_{\varepsilon}^2$  here is needed for the "independence" step.

**Step 3**: We split the left hand side of (3.21) into two parts: one concerning the measures restricted to  $\mathcal{O} \cap B(x, r_{\varepsilon})$ , and one concerning the measures restricted to

 $\mathcal{O} \cap (B(x,\varepsilon) \setminus B(x,r_{\varepsilon}))$ . We show that the first of these is negligible compared to  $\log(1/\varepsilon)$ .

More precisely, we write

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{E_{\varepsilon}} \right] \leq \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,in}}{D_{\varepsilon}^{\beta}} \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{E_{\varepsilon}} \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{E_{\varepsilon}^{2}} \right].$$
(3.24)

Then by definition we have that  $M_{\varepsilon}^{\beta,in} \leq D_{\varepsilon}^{\beta,in}$ , and so on the event  $E_{\varepsilon}$  it holds that  $M_{\varepsilon}^{\beta,in} \leq \log(1/\varepsilon)^{-2}$ . Moreover, we know that  $f_{\varepsilon,2}^{\beta}(x)$  is greater than 1 under  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  and also that  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[1/D_{\varepsilon}^{\beta}] = \mathbb{E}[D_{\varepsilon}^{\beta}]^{-1} = O(1)$ . Thus, the first term is  $o(\log(1/\varepsilon)^{-1})$  and we need only treat the second term.

**Step 4**: We condition on the field outside of  $B(x, r_{\varepsilon} - \varepsilon)$  in order to factorise the second term on the right-hand side of (3.24). We show that the conditional expectation of  $f_{\varepsilon,2}^{\beta}(x)^{-1}$  is of order  $\sqrt{2/\pi \log(1/\varepsilon)}(1 + o(1))$  uniformly on  $E_{\varepsilon}^{2}$ .

More precisely, we condition on  $\mathcal{F}_{r_{\varepsilon}-\varepsilon}$ , the  $\sigma$ -algebra generated by the point x and the field h restricted to  $D \setminus B(x, r_{\varepsilon} - \varepsilon)$ . Then  $M_{\varepsilon}^{\beta,out}$ ,  $D_{\varepsilon}^{\beta,out}$  and  $E_{\varepsilon}^{2}$  are measurable with respect to  $\mathcal{F}_{r_{\varepsilon}-\varepsilon}$ , meaning that the second term on the right-hand side of (3.24) is equal to

$$\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[\frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}}\mathbb{1}_{E_{\varepsilon}^{2}}\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[\frac{1}{f_{\varepsilon,2}^{\beta}(x)}\mid\mathcal{F}_{r_{\varepsilon}-\varepsilon}\right]\right].$$

Now we write

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \, \Big| \, \mathcal{F}_{r_{\varepsilon}-\varepsilon} \right] = \mathbf{Q}_{x}^{\beta,\varepsilon} \left[ \frac{1}{f_{\varepsilon,2}^{\beta}(x)} \, \Big| \, h|_{D \setminus B(x,r_{\varepsilon}-\varepsilon)} \right] \\ = \frac{\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{1}{\tilde{f}_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{\{f_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) \ge 1\}} \, \Big| \, h|_{D \setminus B(x,r_{\varepsilon}-\varepsilon)} \right]}{\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{f_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)}{\tilde{f}_{\varepsilon,2}^{\beta}(x)} \mathbb{1}_{\{f_{\varepsilon,2}^{\beta}(x) \ge 1\}} \, \Big| \, h|_{D \setminus B(x,r_{\varepsilon}-\varepsilon)} \right]}.$$

We will show that, on the event  $E_{\varepsilon}^2$ , the numerator in the final expression is less than or equal to  $\sqrt{2/(\pi \log(1/\varepsilon))} + o(\log(1/\varepsilon)^{-1/2})$  and the denominator is 1 + o(1). To do this we observe that, on  $E_{\varepsilon}^2$  and under the conditional law  $\tilde{\mathbf{Q}}_x^{\beta,\varepsilon}[\cdot \mid h_{D\setminus B(x,r_{\varepsilon}-\varepsilon)}]$ :

- $\tilde{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x)$  has the law of a Bessel process started from a position in  $[\sqrt[3]{\log(1/r_{\varepsilon}-\varepsilon)}, \log(1/r_{\varepsilon}-\varepsilon)]$  and evaluated at time  $\log(\varepsilon) \log(k_{\varepsilon}-\varepsilon);$
- by choice of  $r_{\varepsilon}$  and Lemma 3.11 part (1), this implies that the conditional expectation of  $(\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x))^{-1}$  is equal to  $\sqrt{2/(\pi \log(1/\varepsilon))}(1+o(1));$
- the conditional expectation of  $|Y_{\varepsilon}^2|/\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}(x)$  is o(1), by Cauchy–Schwarz; and

•  $|Y_{\varepsilon}^{1}(x)|$  times the conditional expectation of  $(\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x))^{-1}$  is also o(1), by the second point, and definition of  $E_{\varepsilon}^{2}$ .

Together these imply that, uniformly on  $E_{\varepsilon}^2$ ,

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ 1/f_{\varepsilon,2}^{\beta}(x) \mid \mathcal{F}_{r_{\varepsilon}-\varepsilon} \right] \leq \sqrt{2/(\pi \log(1/\varepsilon))} (1+\mathrm{o}(1)).$$

**Step 5**: We show that  $\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2}} \right]$  is also bounded above by  $\sqrt{2/(\pi \log(1/\varepsilon))}(1+o(1))$ . This completes the proof of (3.21) for our choice of  $E_{\varepsilon}$ .

This is the most delicate step. To do this, we define yet another event

$$E_{\varepsilon}^{3} = \{ \tilde{f}_{u,2\lambda_{\varepsilon}}^{\beta}(x) \ge \sqrt[6]{\log(1/r_{\varepsilon})} \, \forall u \in [\varepsilon, r_{\varepsilon}] \}.$$

Step 5(i): We will first show that

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2}} \mathbb{1}_{E_{\varepsilon}^{3}} \right] \ge \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2}} \right] (1+o(1))$$
(3.25)

so we can instead consider the term on the left-hand side (which turns out to be easier to deal with.) To see why (3.25) is true we again condition on  $\mathcal{F}_{r_{\varepsilon}-\varepsilon}$ . This gives us that

$$\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[\frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}}\mathbb{1}_{E_{\varepsilon}^{2}}\mathbb{1}_{E_{\varepsilon}^{3}}\right] \geq \hat{\mathbb{Q}}^{\beta,\varepsilon}\left[\frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}}\mathbb{1}_{E_{\varepsilon}^{2}}\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[E_{\varepsilon}^{3} \mid \mathcal{F}_{r_{\varepsilon}-\varepsilon}\right]\right]$$

where by changing measure as in step 4 we have

$$\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[ (E^3_{\varepsilon})^c \mid \mathcal{F}_{r_{\varepsilon}-\varepsilon} \right] = \frac{\tilde{\mathbf{Q}}_x^{\beta,\varepsilon} \left[ \frac{f^{\beta}_{\varepsilon,2}(x)}{\bar{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x)} \mathbb{1}_{\{f^{\beta}_{\varepsilon,2}(x) \ge 1\}} \mathbb{1}_{(E^3_{\varepsilon})^c} \mid h|_{D \setminus B(x,r_{\varepsilon}-\varepsilon)} \right]}{\tilde{\mathbf{Q}}_x^{\beta,\varepsilon} \left[ \frac{f^{\beta}_{\varepsilon,2}(x)}{\bar{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x)} \mathbb{1}_{\{f^{\beta}_{\varepsilon,2}(x) \ge 1\}} \mid h|_{D \setminus B(x,r_{\varepsilon}-\varepsilon)} \right]}$$

We already know that the denominator is 1 + o(1) uniformly on  $E_{\varepsilon}^2$  by our previous discussion. In fact, the numerator is also o(1) uniformly on  $E_{\varepsilon}^2$ . To show this, again using our observations from step 4, it is enough for us to prove that

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[E_{\varepsilon}^{3} \mid h|_{D \setminus (B(x,r_{\varepsilon}-\varepsilon))}\right] \to 1$$

uniformly on  $E_{\varepsilon}^2$ . For this we again use the fact that, under this conditional law, the process  $\tilde{f}_{u,2\lambda_{\varepsilon}}^{\beta}(x)$  for  $u \leq r_{\varepsilon} - \varepsilon$  is a time-changed Bessel process (plus a small deterministic fluctuation  $\rho_u^{\varepsilon}(x)$ ) starting from a position in  $[\sqrt[3]{\log(1/r_{\varepsilon} - \varepsilon)}, \log(1/r_{\varepsilon} - \varepsilon)]$ . Thus we need to calculate the probability that such a Bessel process, and we can

clearly forget about the fluctuations, remains greater than  $\log(1/r_{\varepsilon})$  up to time  $\log(1/\varepsilon) - \log(1/r_{\varepsilon} - \varepsilon)$ . It is clear that this probability is smallest if we take the starting point  $x_0$  to be  $\sqrt[3]{\log(1/r_{\varepsilon} - \varepsilon)}$ . In this case we have, writing  $\mathbf{Q}_y$  for the law of a Bessel process started at y, and by scaling, that

$$\mathbf{Q}_{x_0}(X_t \ge \sqrt[6]{\log(1/r_{\varepsilon})} \ \forall t \in [0, \log(1/\varepsilon) - \log(1/r_{\varepsilon} - \varepsilon)]) \ge \mathbf{Q}_1(X_t \ge \frac{\sqrt[6]{\log(1/r_{\varepsilon})}}{\sqrt[3]{\log(1/r_{\varepsilon} - \varepsilon)}} \ \forall t \in [0, \infty])$$

Taking  $\varepsilon \to 0$  we see that this converges to 1: the probability that a 3d Bessel process started at 1 never hits 0.

Step 5(ii): Having done this, we can now consider the left-hand side of (3.25) and instead try to show that this is bounded above by  $\sqrt{2/(\pi \log(1/\varepsilon))}(1+o(1))$ . Again this will require a few arguments. We write

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2} \cap E_{\varepsilon}^{3}} \right] \leq \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2} \cap E_{\varepsilon}^{3}} \mathbb{1}_{E_{\varepsilon}^{1}} \mathbb{1}_{\{D_{\varepsilon}^{\beta} > \log(1/\varepsilon)^{-1}\}} \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2} \cap E_{\varepsilon}^{3}} \mathbb{1}_{E_{\varepsilon}^{1}} \mathbb{1}_{\{D_{\varepsilon}^{\beta} \le \log(1/\varepsilon)^{-1}\}} \right] + \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta,out}}{D_{\varepsilon}^{\beta,out}} \mathbb{1}_{E_{\varepsilon}^{2} \cap E_{\varepsilon}^{3}} \mathbb{1}_{(E_{\varepsilon}^{1})^{c}} \right]$$

and will show that the first term is of the order we want, and the second and third are negligible. Indeed, it is clear that the first term is less than or equal to

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}} \right] (1 + \mathrm{o}(1)) \le \sqrt{\frac{2}{\pi \log(1/\varepsilon)}} (1 + \mathrm{o}(1))$$

by definition of the events (these imply that  $D_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta,out} = 1 + o(1)$ ) and our previous estimate (4.12) for the first moment. Moreover by Markov's inequality for  $(1/D_{\varepsilon}^{\beta})$ , and the fact that  $M_{\varepsilon}^{\beta}/D_{\varepsilon}^{\beta} \leq 1$ , the second is  $o(\sqrt{\log(1/\varepsilon)}^{-1})$ .

**Step 5(iii):** So we are left to deal with the third term, which we would also like to show is  $o(\sqrt{\log(1/\varepsilon)}^{-1})$ . To do this, we first observe that we can bound it above by

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ E_{\varepsilon}^2 \cap E_{\varepsilon}^3 \cap (E_{\varepsilon}^1)^c \right] \le \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ E_{\varepsilon}^3 \cap \left\{ D_{\varepsilon}^{\beta,in} > \frac{1}{\log(1/\varepsilon)^2} \right\} \right].$$
(3.26)

Our strategy here will be to use Markov's inequality for  $D_{\varepsilon}^{\beta,in}$ . For this we have to calculate the  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  expectation of  $D_{\varepsilon}^{\beta,in}$ , which by definition is the same as calculating the  $\mathbb{P}$  expectation of  $D_{\varepsilon}^{\beta} \times D_{\varepsilon}^{\beta,in}$ . Thus, we can use similar techniques to those in the proof of uniform integrability (Lemma 3.18), where we calculated the  $\mathbb{P}$  expectation of  $(D_{\varepsilon}^{\beta})^2$  on a "good" event.

As we did there, we will break up  $D_{\varepsilon}^{\beta,in}$  into two parts: the integral over  $B(x, 3\varepsilon)$ ,

and the rest. To deal with the integral over  $B(x, 3\varepsilon)$  we define a further event  $E_{\varepsilon}^4$ (which has high probability) and on which  $f_{\varepsilon,2}^{\beta}$  is close to  $\sqrt{\log(1/\varepsilon)}$ . Crude estimates on this event, using that  $|B(x, 3\varepsilon)| = O(\varepsilon^2)$ , then give the desired expectation. To deal with the integral over  $B(x, r_{\varepsilon}) \setminus B(x, 3\varepsilon)$  we need to be more careful. Here we use the definition of  $E_{\varepsilon}^3$ , which allows us to control the value of  $\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)$  at all times between  $r_{\varepsilon}$  and  $\varepsilon$ . Applying a decorrelation argument similar that in the proof of Lemma 3.18 then allows us to reach the desired conclusion.

Let us first define  $E_{\varepsilon}^4$ , using the following lemma:

**Lemma 3.25.** There exists  $p \in (0, 1/2)$  such that

$$E_{\varepsilon}^{4} = \{ f_{\varepsilon,2}^{\beta}(x) \in [\log(1/\varepsilon)^{1/2-p}, \log(1/\varepsilon)^{1/2+p}] \}$$

satisfies  $\hat{\mathbb{Q}}^{\beta,\varepsilon}\left[(E_{\varepsilon}^{4})^{c}\right] = o(\log(1/\varepsilon)^{-1/2}).$ 

Proof of Lemma 3.25. This is possible because for any p > 0

$$(E^4_{\varepsilon})^c \subset \{\tilde{f}^{\beta}_{\varepsilon,2\lambda_{\varepsilon}}(x) \notin [2\log(1/\varepsilon)^{1/2-p}, \frac{1}{2}\log(1/\varepsilon)^{1/2+p}]\} \cup \{|Y_{\varepsilon}(x)| > \log(1/\varepsilon)^{1/2-p}\}$$

and then

$$\begin{split} \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ (E_{\varepsilon}^{4})^{c} \right] &\leq \int_{\mathcal{O}} \mathbf{Q}_{x}^{\beta,\varepsilon} \left[ |Y_{\varepsilon}(x)| > \log(1/\varepsilon)^{1/2-p} \right] \\ &+ \mathbf{Q}_{x}^{\beta,\varepsilon} \left[ \tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x) \notin [2\log(1/\varepsilon)^{1/2-p}, \frac{1}{2}\log(1/\varepsilon)^{1/2+p}] \right] \, dm^{\beta,\varepsilon}(x). \end{split}$$

It is easy to see (using the definition of the measure  $\mathbf{Q}_x^{\beta,\varepsilon}$ ) that the first probability inside the integral decays exponentially in  $\varepsilon$ . For the second, we can write it as

$$\frac{Z_{\varepsilon}^{\beta}(x)}{\tilde{Z}_{\varepsilon}^{\beta}(x)}\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[\frac{f_{\varepsilon,2}^{\beta}(x)}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)}\mathbbm{1}_{\{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)\notin[2\log(1/\varepsilon)^{1/2-p},\frac{1}{2}\log(1/\varepsilon)^{1/2+p}]\}}\right].$$

which by Cauchy-Schwarz is less than or equal to

$$\begin{split} \lambda_{\varepsilon}(x) \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \mathbbm{1}_{\{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)\notin[2\log(1/\varepsilon)^{1/2-p},\frac{1}{2}\log(1/\varepsilon)^{1/2+p'}]\}} \right] + \\ \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{(Y_{\varepsilon}(x) + \mathcal{O}(1))^{2}}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} \right]^{1/2} \tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon} \left[ \frac{\mathbbm{1}_{\{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)\notin[2\log(1/\varepsilon)^{1/2-p},\frac{1}{2}\log(1/\varepsilon)^{1/2+p}]\}}{\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)} \right]^{1/2} \end{split}$$

A standard Bessel calculation, plus the fact that

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[(\mathcal{O}(1)+Y_{\varepsilon}(x))^{2}/\tilde{f}_{\varepsilon,2\lambda_{\varepsilon}}^{\beta}(x)\right]=\mathcal{O}(1)$$

(seen by changing back to the measure  $\mathbb{P}$ ), gives that this is  $o(\log(1/\varepsilon)^{-1/2})$  for some p < 1/2.

Using this new event  $E_{\varepsilon}^4$ , and Markov's inequality, we next bound the right-hand side of (3.26) above by

$$o(\log(1/\varepsilon)^{-1/2})$$

$$+2\log(1/\varepsilon)^{2} \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \mathbbm{1}_{E_{\varepsilon}^{4}} \int_{w \in B(x,3\varepsilon)} f_{\varepsilon,2}^{\beta}(w) \mathbbm{1}_{L_{\varepsilon}(w)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(w)>1\}} e^{g_{\varepsilon,2}(w)} dw \right]$$

$$+2\log(1/\varepsilon)^{2} \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \mathbbm{1}_{E_{\varepsilon}^{3}} \int_{w \in B(x,r_{\varepsilon}) \setminus B(x,3\varepsilon)} f_{\varepsilon,2}^{\beta}(w) \mathbbm{1}_{L_{\varepsilon}(w)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(w)>1\}} e^{g_{\varepsilon,2}(w)} dw \right]$$

$$(3.27)$$

where the first term comes from the  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  probability of  $E_{\varepsilon}^4$ . Recall that, to conclude, we need to show this whole expression is  $o(\log(1/\varepsilon)^{-1/2})$ . Let us look at the expectation in the second term. By definition of  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$  this is equal to

$$\begin{split} \mathbb{E}[D_{\varepsilon}^{\beta}]^{-1} & \int_{\mathcal{O}} \int_{w \in B(x,3\varepsilon)} \mathbb{E}\left[\mathbbm{1}_{E_{\varepsilon}^{4}} f_{\varepsilon,2}^{\beta}(w) \mathbbm{1}_{L_{\varepsilon}(w)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(w) > 1\}} e^{g_{\varepsilon,2}(w)} f_{\varepsilon,2}^{\beta}(x) \mathbbm{1}_{L_{\varepsilon}(x)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(x) > 1\}} e^{g_{\varepsilon,2}(x)}\right] dw dx \\ & \leq \mathbb{E}[D_{\varepsilon}^{\beta}]^{-1} \int_{\mathcal{O}} \int_{w \in B(x,3\varepsilon)} \varepsilon^{-2} \log(1/\varepsilon)^{1/2+p} e^{-2\log(1/\varepsilon)^{1/2-p}} \mathbb{E}\left[f_{\varepsilon,2}^{\beta}(w) \mathbbm{1}_{L_{\varepsilon}(w)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(w) > 1\}} e^{g_{\varepsilon,2}(w)}\right] dw dx \\ & \leq O(1) \times \log(1/\varepsilon)^{1/2+p} e^{-2\log(1/\varepsilon)^{1/2-p}} \end{split}$$

where the second line follows from the definition of  $E_{\varepsilon}^4$ . Hence, the second term of (3.27) is  $o(\log(1/\varepsilon)^{-1/2})$ .

We finish by dealing with the third term of (3.27). Writing  $A_{\varepsilon} = B(x, r_{\varepsilon}) \setminus B(x, 3\varepsilon)$ , we have that the expectation in this term is equal to

$$\mathbb{E}[D_{\varepsilon}^{\beta}]^{-1} \int_{\mathcal{O}} \int_{A_{\varepsilon}} \mathbb{E}\left[\mathbbm{1}_{E_{\varepsilon}^{3}} f_{\varepsilon,2}^{\beta}(w) \mathbbm{1}_{L_{\varepsilon}(w)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(w)>1\}} e^{g_{\varepsilon,2}(w)} f_{\varepsilon,2}^{\beta}(x) \mathbbm{1}_{L_{\varepsilon}(x)} \mathbbm{1}_{\{f_{\varepsilon,2}^{\beta}(x)>1\}} e^{g_{\varepsilon,2}(x)}\right] dw dx.$$

Then, by exactly the same reasoning used in the proof of Lemma 3.18, we can deduce that this is less than or equal to some constant times

$$\int_{\mathcal{O}} \int_{A_{\varepsilon}} \mathbb{E} \left[ e^{\tilde{g}_{\delta, 2\lambda_{\varepsilon}(x)}(x)} e^{\tilde{g}_{\delta, 2\lambda_{\varepsilon}(w)}(w)} \mathbb{1}_{\{\tilde{f}_{\delta, 2\lambda_{\varepsilon}}^{\beta}(x) \ge \log(1/r_{\varepsilon})^{1/6}\}} (\tilde{f}_{\delta, 2\lambda_{\varepsilon}(x)}^{\beta}(x) + 1) (\tilde{f}_{\delta, 2\lambda_{\varepsilon}(w)}^{\beta}(w) + 1) \right]$$
(3.28)

where  $\delta(x, y) = |x - y|/3$ . The final observation to make is that, by orthogonal projection, we have

$$\tilde{h}_{\delta}(w) = \alpha_{x,w}^{\delta} \tilde{h}_{\delta}(x) + Z_{x,w}^{\delta}$$

for  $\alpha_{x,w}^{\delta} = \operatorname{cov}(\tilde{h}_{\delta}(x), \tilde{h}_{\delta}(w))/\operatorname{var}(\tilde{h}_{\delta}(x))$  where  $Z_{x,w}^{\delta}$  is independent of  $\tilde{h}_{\delta}(x)$ ; distributed as a centered normal random variable with variance  $(1-(\alpha_{x,w}^{\delta})^2)\operatorname{var}(\tilde{h}_{\delta}(w))$ .

The proof of this is the same as the proof of Lemma 3.13. Moreover, by Lemma 3.3, we have  $\alpha_{x,w}^{\delta} = 1 + O(\log(1/\delta)^{-1})$  uniformly in x, w.

Thus, by conditioning on  $h_{\delta}(x)$ , we can calculate that the integrand in (3.28) is less than or equal to a constant times

$$\mathbb{E}\left[\mathbb{1}_{\{\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}}(x)\geq \log(1/r_{\varepsilon})^{1/6}\}}e^{\tilde{g}_{\delta,\lambda_{\varepsilon}(x)}(x)+\tilde{g}_{\delta,2\lambda_{\varepsilon}(w)\alpha_{x,w}^{\delta}}(x)}(\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}(x)}(x)+1)(\alpha_{x,w}^{\delta}\tilde{f}^{\beta}_{\delta,2\lambda_{\varepsilon}(w)\alpha_{x,w}^{\delta}}(x)+1)\right]$$

which by a simple calculation can be bounded again by

$$e^{-(\log(1/r_{\varepsilon}))^{1/6}} \,\delta^{-2} \mathbb{E}\left[e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(x)}(x)} (1+|\tilde{f}_{\delta,\lambda_{\varepsilon}(x)}^{\beta}(x)|^{2})\right]$$

times some constant. This last expectation can be estimated by observing that changing measure by  $e^{\tilde{g}_{\delta,2\lambda_{\varepsilon}(x)}(x)}$  turns  $\tilde{f}^{\beta}_{\delta,\lambda_{\varepsilon}(x)}(x)$  into a Gaussian random variable with mean  $\beta$  and variance  $\log(1/\delta) + O(1)$ . Hence the integrand of (3.28) is less than

$$C e^{-(\log(1/r_{\varepsilon}))^{1/6}} \delta^{-2} (1 + \log(1/\delta))$$

for some constant C. Integrating over  $A_{\varepsilon}$  gives that the third term of (3.27) is  $\leq e^{-\log(1/r_{\varepsilon})^{1/6}}\log(1/\varepsilon)^2$ , and so we conclude, using our assumption (3.23) on  $r_{\varepsilon}$ , that (3.27) is of the correct order.

**Step 6**: We show that  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}] \to 1 \text{ as } \varepsilon \to 0.$ 

Firstly, it is clear that  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}^2] \to 1$ . Then since we have already shown, see (3.26), that  $\hat{\mathbb{Q}}^{\beta,\varepsilon}[E_{\varepsilon}^2 \cap E_{\varepsilon}^3 \cap (E_{\varepsilon}^1)^c] \to 0$  and that

$$\tilde{\mathbf{Q}}_{x}^{\beta,\varepsilon}\left[E_{\varepsilon}^{3}\mid h|_{D\setminus(B(x,r_{\varepsilon}-\varepsilon))}\right]\to 1$$

uniformly on  $E_{\varepsilon}^2$ , the claim follows straight away. The proof is complete.

It is now relatively simple to show the convergence of  $D_{\varepsilon}(\mathcal{O})$ . To use Proposition 3.21, we must first compare  $M_{\varepsilon}^{\beta}(\mathcal{O})$  and  $D_{\varepsilon}^{\beta}(\mathcal{O})$  with  $M_{\varepsilon}(\mathcal{O})$  and  $D_{\varepsilon}(\mathcal{O})$ .

Lemma 3.26. We have

$$\mathbb{P}(C_{\beta}) := \mathbb{P}\left(\{\inf_{\varepsilon} \inf_{x \in D} (-\tilde{h}_{\varepsilon}(x) + 2\operatorname{var}(\tilde{h}_{\varepsilon}(x))) > -(\beta + 10)\}\right)$$
$$\cap \left\{\inf_{\varepsilon} \inf_{x \in D} (-h_{\varepsilon}(x) + 2\operatorname{var}(h_{\varepsilon}(x))) > -(\beta + 10)\}\right)$$

converges to 0 as  $\beta \rightarrow 0$ , uniformly in  $\varepsilon$ .

*Proof.* This is a consequence of Corollary 3.6.

134

**Remark 3.27.** Note that on the event  $C_{\beta}$  we have  $M_{\varepsilon}^{\beta}(\mathcal{O}) = M_{\varepsilon}(\mathcal{O})$  and also  $D_{\varepsilon}^{\beta}(\mathcal{O}) = D_{\varepsilon}(\mathcal{O}) + \beta M_{\varepsilon}(\mathcal{O})$  for all  $\mathcal{O} \subset D$ .

We are now ready to prove the main result.

Proof of Theorem 3.1. It is enough to show that for  $\mathcal{O} \subset D$  and  $\delta > 0$  fixed

$$\mathbb{P}\left[\left|\frac{M_{\varepsilon}(\mathcal{O})}{D_{\varepsilon}(\mathcal{O})}\sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}}\right| > \delta\right] \to 0$$
(3.29)

as  $\varepsilon \to 0$ . Then since  $\sqrt{\log(1/\varepsilon)}M_{\varepsilon}(\mathcal{O}) \to \sqrt{\frac{\pi}{2}}\mu'(\mathcal{O})$  in probability, by Theorem 3.10, we also have  $D_{\varepsilon}(\mathcal{O}) \to \mu'(\mathcal{O})$  in probability. Let us prove (3.29). By Proposition 3.21 we know that for any  $\beta > 0$ 

$$\hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \left| \frac{M_{\varepsilon}^{\beta}(\mathcal{O})}{D_{\varepsilon}^{\beta}(\mathcal{O})} \sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}} \right| > \delta \right] \to 0$$

as  $\varepsilon \to 0$ , and we also know that, on the event  $C_{\beta}$ , we can compare  $M_{\varepsilon}^{\beta}, D_{\varepsilon}^{\beta}$  with  $M_{\varepsilon}, D_{\varepsilon}$  by Remark 3.27. With this in mind, we bound (3.29) above by

$$\mathbb{P}\left[A_{\beta,\varepsilon}^{1}\right] + \mathbb{P}\left[A_{\beta,\varepsilon}^{2} \cap (A_{\beta,\varepsilon}^{1})^{c}\right]$$

where

$$A^{1}_{\beta,\varepsilon} = \left\{ \left| \frac{M_{\varepsilon}(\mathcal{O})}{D_{\varepsilon}(\mathcal{O}) + \beta M_{\varepsilon}(\mathcal{O})} \sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}} \right| > \frac{\delta}{2} \right\} \text{ and} \\ A^{2}_{\beta,\varepsilon} = \left\{ \sqrt{\log(1/\varepsilon)} \left| \frac{M_{\varepsilon}(\mathcal{O})}{D_{\varepsilon}(\mathcal{O}) + \beta M_{\varepsilon}(\mathcal{O})} - \frac{M_{\varepsilon}(\mathcal{O})}{D_{\varepsilon}(\mathcal{O})} \right| > \frac{\delta}{2} \right\}.$$

In fact, the event  $A^2_{\beta,\varepsilon} \cap (A^1_{\beta,\varepsilon})^c$  is deterministically non possible if  $\varepsilon$  is small enough. Thus, it is enough to show that  $\mathbb{P}(A^1_{\beta,\varepsilon})$  can be made arbitrarily small by choosing  $\beta$  large, and then  $\varepsilon$  small. To do this, we observe by Remark 3.27 that

$$\mathbb{P}(A_{\beta,\varepsilon}^1) \leq \mathbb{P}[C_{\beta}^c] + \mathbb{P}\left[\left\{ \left| \frac{M_{\varepsilon}^{\beta}(\mathcal{O})}{D_{\varepsilon}^{\beta}(\mathcal{O})} \sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}} \right| > \delta/2 \right\} \cap C_{\beta} \right].$$

Furthermore, by the definition of  $\hat{\mathbb{Q}}^{\beta,\varepsilon}$ , the left-hand side for any  $\eta > 0$ , is less than or equal to

$$\mathbb{P}[C_{\beta}^{c}] + \mathbb{P}[C_{\beta} \cap \{D_{\varepsilon}^{\beta} < \eta\}] + \frac{\mathbb{E}[D_{\varepsilon}^{\beta}]}{\eta} \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[ \left| \frac{M_{\varepsilon}^{\beta}(\mathcal{O})}{D_{\varepsilon}^{\beta}(\mathcal{O})} \sqrt{\log(1/\varepsilon)} - \sqrt{\frac{2}{\pi}} \right| > \delta \right].$$

Now note that by Markov's inequality

$$\mathbb{P}[C_{\beta} \cap \{D_{\varepsilon}^{\beta} < \eta\}] \leq \mathbb{P}[\sqrt{\log(1/\varepsilon)}M_{\varepsilon}(\mathcal{O}) < \eta^{1/4}] + \sqrt{\eta} \mathbb{E}[D_{\varepsilon}^{\beta}] \hat{\mathbb{Q}}^{\beta,\varepsilon} \left[\log(1/\varepsilon)\left(\frac{M_{\varepsilon}^{\beta}}{D_{\varepsilon}^{\beta}}\right)^{2}\right]$$

Hence using Proposition 3.21 and Lemma 3.26, together with the fact that  $\sqrt{\log(1/\varepsilon)}M_{\varepsilon}(\mathcal{O})$  converges to  $\sqrt{\pi/2}\mu'(\mathcal{O})$  (which is positive almost surely) and that  $\mathbb{E}[D_{\varepsilon}^{\beta}] = \int_{x} Z_{\varepsilon}^{\beta}(x)$  is bounded, we can conclude by letting  $\beta \to \infty$ , then  $\eta \to 0$ , and finally  $\varepsilon \to 0$ .

#### 3.4 \*-scale invariant kernels

In this section we prove Theorem 3.2 using a simple adaptation of our arguments from the previous section. Recalling the set-up, we have:

- $\theta : \mathbb{R}^d \to \mathbb{R}$  a mollifier, supported in B(0,1), with Hölder continuous density and satisfying (3.2);
- $k: [0, \infty) \to \mathbb{R}$ , a compactly supported and positive-definite  $C^1$  function with k(0) = 1; and
- $h \neq +$  scale invariant field on  $\mathbb{R}^d$  with covariance kernel

$$K(x,y) = \int_1^\infty \frac{k(u|x-y|)}{u} \, du.$$

We would like to prove that if  $h_{\varepsilon}(x) = h \star \theta_{\varepsilon}(x)$  is the  $\theta$ -convolution approximation to h, the signed measures

$$D_{\varepsilon}(dx) := \left(-h_{\varepsilon}(x) + \sqrt{2d}\log(1/\varepsilon)\right) e^{\sqrt{2d}h_{\varepsilon}(x)} \varepsilon^{d} dx$$

converge weakly in probability to a limiting measure. Moreover, we would like to show that this limiting measure is equal to the measure defined in [DRSV14a, JS17] (see Theorems 3.8 and 3.9).

*Proof.* First pick g such that  $g \star g(u) = k(u)$  (we can do this by our assumptions on k, [?]). Then we can define a field h with the correct covariance structure by setting

$$h(x) := \int_1^\infty \int_{\mathbb{R}^d} \frac{g(y - xu)}{\sqrt{u}} W(dy, du), \qquad (3.30)$$

where  $W(\cdot, \cdot)$  is a standard space-time white noise. It is then proved in [DRSV14a]

that if we let

$$\tilde{h}_{\varepsilon}(x) := \int_{1}^{\frac{1}{\varepsilon}} \int_{\mathbb{R}^d} \frac{g(y - xu)}{\sqrt{u}} W(dy, du),$$

the signed derivative measures  $\tilde{D}_{\varepsilon}(dx) := (-\tilde{h}_{\varepsilon}(x) + \sqrt{2d}\log(1/\varepsilon)) e^{\sqrt{2d}\tilde{h}_{\varepsilon}(x)} \varepsilon^{d} dx$ converge almost surely to a positive limiting measure  $\mu'$ . It is further shown in [DRSV14b] that

$$\tilde{M}_{\varepsilon}(dx) := \sqrt{\log(1/\varepsilon)} e^{\sqrt{2d}\tilde{h}_{\varepsilon}(x)} \varepsilon^d dx$$

converges to  $\sqrt{2/\pi}\mu'$  in probability and in [JS17] that

$$M_{\varepsilon} := \sqrt{\log(1/\varepsilon)} e^{\sqrt{2d}h_{\varepsilon}(x)} \varepsilon^d dx$$

also converges to  $\sqrt{2/\pi}\mu'$  in probability.

To prove the convergence of  $D_{\varepsilon}(dx)$  we use the same strategy as for the proof of Theorem 3.1, now letting  $\tilde{h}_{\varepsilon}$  play the role of the circle average. In particular we need only prove Proposition 3.21 (the result then following by Corollary 3.6 and Lemma 3.26 in exactly the same way.) We observe that:

- $\tilde{h}_{\varepsilon}(x)$  is a (time-changed) Brownian motion for each  $x \in \mathbb{R}^d$ ; and
- $\operatorname{cov}(h_{\varepsilon}(x), \tilde{h}_{\delta}(x)) = \log(1/(\varepsilon \wedge \delta)) + O(1)$ , so we can define  $\lambda_{\varepsilon}(x), Y_{\varepsilon}(x)$  and  $\rho_{\delta}^{\varepsilon}(x)$  as in Lemmas 3.13 and 3.15, and the statements of these lemmas will still hold.

This is enough to prove (4.12) and step 1 of (3.19). For step 2 we need to explain how we define a few things. We let  $r_{\varepsilon}$  be chosen as before, and without loss of generality we assume that  $\operatorname{supp}(k) \subset B(0,1)$ . It is then easy to check using the definition of h that  $h_{\varepsilon}(z)$  and  $(\tilde{h}_{\eta}(x) - \tilde{h}_{r_{\varepsilon}-\varepsilon}(x))_{\eta \leq r_{\varepsilon}-\varepsilon}$  are independent for all  $z \notin B(x, r_{\varepsilon})$ . We let  $\mathcal{F}_{\varepsilon} = \sigma(\{\tilde{h}_u(x) : u \geq r_{\varepsilon} - \varepsilon\}) \vee \sigma(\{h_{\varepsilon}(z) : z \in D \setminus B(x, r_{\varepsilon})\})$  so that  $(\tilde{h}_{\eta}(x) - \tilde{h}_{r_{\varepsilon}-\varepsilon}(x))_{\eta \leq r_{\varepsilon}-\varepsilon}(x)$  is independent of  $\mathcal{F}_{\varepsilon}$ , and  $M_{\varepsilon}^{\beta,out}/D_{\varepsilon}^{\beta,out}$  is measurable with respect to it. This is what we will use in place of  $\mathcal{F}_{r_{\varepsilon}-\varepsilon}$  from the original proof. Using standard properties of Gaussian processes we see that we can also write  $Y_{\varepsilon}(x) = Y_{\varepsilon}^{1}(x) + Y_{\varepsilon}^{2}(x)$  where  $Y_{\varepsilon}^{1}(x)$  is measurable with respect to  $\mathcal{F}_{\varepsilon}$  and  $Y_{\varepsilon}^{2}(x)$  is independent of it. From this point onwards we can define everything in the same way, and steps 3 and 4 follow, using only properties of the 3d Bessel process.

To conclude, we need only complete step 5, since step 6 is a straightforward consequence of this (as in the original proof.) For this step we note that by our assumption on  $\operatorname{supp}(k)$ ,  $(\tilde{h}_{\delta+\eta}(x) - \tilde{h}_{\delta}(x))_{\eta\geq 0}$  and  $(\tilde{h}_{\delta+\eta}(y) - \tilde{h}_{\delta}(y))_{\eta\geq 0}$  are independent as soon as  $|y - x| \geq \delta$ . Since this is the only extra property we used in this step, the proof of Proposition 3.21 goes through.

**Remark 3.28.** We remark here that the authors in [DRSV14a, DRSV14b] suggest that their constructions should hold for more general kernels than the  $\star$ -scale invariant ones. In particular, for any positive definite kernel of the form  $K(x,y) = -\log(|x - y|) + g(|x - y|)$  with g continuous, one has a white-noise decomposition for the corresponding field h, analogous to (3.30). This means that the theory in [DRSV14b, Appendix D] should go through, and as a consequence, the result of Theorem 3.2 should also hold. More generally we conjecture that Theorem 3.2 should hold for any K satisfying (3.4).

### 4 Approximating the Liouville measure using local sets of the Gaussian free field

#### 4.1 Introduction

Gaussian multiplicative chaos (GMC) theory, initiated by Kahane in the 80s [Kah85] as a generalization of multiplicative cascades, aims to give a meaning to " $\exp(\Gamma)$ " for rough Gaussian fields  $\Gamma$ . In a simpler setting it was already used in the 70s to model the exponential interaction of bosonic fields [HK71], and over the past ten years it has gained importance as a key component in constructing probabilistic models of so-called Liouville quantum gravity in 2D [DS11, DKRV16].

One of the important cases of GMC theory is when the underlying Gaussian field is equal to  $\gamma\Gamma$ , for  $\Gamma$  a 2D Gaussian free field (GFF) [DS11] and  $\gamma > 0$  a parameter. It is then possible to define random measures with area element " $\exp(\gamma\Gamma)dx \wedge dy$ ". These measures are sometimes also called Liouville measures [DS11] and we will do so for convenience in this article. Due to the recent work of many authors [RV10, DS11, Ber15a, Sha16] one can say that we have a rather thorough understanding of Liouville measures in the so-called subcritical regime ( $\gamma < 2$ ). The critical regime ( $\gamma = 2$ ) is trickier, but several constructions are also known [DRSV14a, DRSV14b, JS17, Pow17a].

Usually, in order to construct the GMC measure, one first approximates the underlying field using either a truncated series expansion or smooth mollifiers, then takes the exponential of the approximated Gaussian field, renormalizes it and shows that the limit exists in the space of measures. In [Aid15] the author proposed a different way to construct measures of multiplicative nature using nested conformally invariant loop ensembles, inspired by multiplicative cascades. He conjectured that in the subcritical and critical regime, and in the case where these loop ensembles correspond to certain same-height contour lines of the underlying GFF, the limiting measure should have the law of the Liouville measure. In this paper we confirm his conjecture. This is done by providing new constructions of the subcritical and critical Liouville measures using a certain family of so called local sets of the GFF [SS13, ASW15] and reinterpreting his construction as a special case of this general setting. Some of our local-set based constructions correspond to simple multiplicative cascades, and others in some sense to stopping lines constructions of the multiplicative cascade measures [Kyp00]. To our knowledge we provide a first "non-Gaussian" approximation of Liouville measures that is both local and conformally invariant. We also remark that our construction strongly uses the Markov property of the GFF and hence does not easily generalize to other log-correlated fields.

One simple, but important, consequence of our results is the simultaneous construction of a GFF in a simply connected domain and its associated Liouville measure using nested  $\text{CLE}_4$  and a collection of independent coin tosses. Start with a height function  $h_0 = 0$  on  $\mathbb{D}$  and sample a  $\text{CLE}_4$  in  $\mathbb{D}$ . Inside each connected component of its complement add either  $\pm \pi$  to  $h_0$  using independent fair coins. Call the resulting function  $h_1$ . Now repeat this procedure independently in each connected component: sample an independent  $\text{CLE}_4$ , toss coins and add  $\pm \pi$  to  $h_1$  to obtain  $h_2$ . Iterate. Then it is known [MS11, ASW15] that these piecewise constant fields  $h_n$  converge to a GFF  $\Gamma$ . It is also possible to show that the nested  $\text{CLE}_4$  used in this construction (we call the complement of the *n*th level loops  $A_n$ ) is a measurable function of  $\Gamma$ . Proposition 4.10 of the current article implies that one can construct the Liouville measures associated to  $\Gamma$  by just taking the limit of measures

$$M_n^{\gamma}(dz) = e^{\gamma h_n(z)} \operatorname{CR}(z; \mathbb{D} \setminus A^n)^{\frac{\gamma^2}{2}} dz.$$

Here  $\operatorname{CR}(z; \mathbb{D} \setminus A^n)$  is the conformal radius of the point z inside the n-th level loop.

Observe that the above approximation is different from taking naively the exponential of  $h_n$  and normalizing it pointwise by its expectation. In fact, it is not hard to see that in this setting the latter naive procedure that is used for mollifier and truncated series approximations would not give the Liouville measure.

In the critical case, and keeping to the above concrete approximation of the GFF, regularized Liouville measures can be given by the so-called derivative approximations:

$$D_n(dz) = \int_{\mathcal{O}} \left( -h_n(z) + 2\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A^n) \right) e^{2h_n(z)} \operatorname{CR}(z; \mathbb{D} \setminus A^n)^2 dz.$$

As the name suggests, they correspond to the derivative of the above measure  $M_n^{\gamma}$  w.r.t. to  $\gamma$ , taken at the critical parameter  $\gamma = 2$ . We show that these approximate signed measures converge to a positive measure that agrees (up to a constant factor 2) with the limiting measure of [Aid15] described in Section 4.3.3, and also to the critical Liouville measure constructed in [DRSV14b, Pow17a].

The connection between multiplicative cascades and the Liouville measure established by our construction makes it possible to directly adapt many techniques developed in the realm of branching random walks and multiplicative cascades to the study of the Liouville measure. For example, this allows us to prove a "Seneta– Heyde" rescaling result in the critical regime by following very closely the proof for the branching random walk in [AS14] and doing minimal extra work. Finally, our proofs are robust enough to study the Liouville measure in non-simply connected domains and also to study the boundary Liouville measure.

The rest of the article is structured as follows. We start with preliminaries on the GFF, its local sets and Liouville measure. Then, we treat the subcritical regime and discuss generalizations to non-simply connected domains and to the boundary Liouville measure. Finally, we handle the critical case: we first show that our construction agrees with both a construction by E. Aidekon (up to a constant factor 2) and a mollifier construction of the critical Liouville measure; then, we consider the case of Seneta-Heyde scaling.

# 4.2 Preliminaries on the Gaussian free field and its local sets

Let  $D \subseteq \mathbb{R}^2$  denote a bounded, open and simply connected planar domain. By conformal invariance, we can always assume that D is equal to  $\mathbb{D}$ , the unit disk. Recall that the Gaussian Free Field (GFF) in D can be viewed as a centered Gaussian process  $\Gamma$ , indexed by the set of continuous functions in D, with covariance given by

$$\mathbb{E}\left[(\Gamma, f)(\Gamma, g)\right] = \iint_{D \times D} f(x) G_D(x, y) g(y) dx dy.$$
(4.1)

Here  $G_D$  is the Dirichlet Green's function in D, normalized such that  $G_D(x, y) \sim \log(1/|x-y|)$  as  $x \to y$  for all  $y \in D$ .

Let us denote by  $\rho_z^{\varepsilon}$  the uniform measure on the circle of radius  $\varepsilon$  around z. Then for all  $z \in D$  and all  $\varepsilon > 0$ , one can define  $\Gamma_{\varepsilon} := (\Gamma, \rho_z^{\varepsilon})$ . We remark that this concrete choice of mollifying the free field is of no real importance, but is just a bit more convenient in the write-up of the critical case.

An explicit calculation then shows that:

$$\mathbb{E}\left[ \varepsilon^{\frac{\gamma^2}{2}} \exp\left(\gamma(\Gamma, \rho_z^{\varepsilon})\right) \right] \begin{cases} = \operatorname{CR}(z; D)^{\gamma^2/2} & \text{if } d(z, \partial D) \ge \varepsilon, \\ \le 1 & \text{if } d(z, \partial D) < \varepsilon, \end{cases}$$
(4.2)

where CR(z; D) is the conformal radius of z in the simply-connected domain D.

The Gaussian free field satisfies a spatial Markov property, and in fact it also satisfies a strong spatial Markov property. To formalise this, the concept of local sets was introduced in [SS13]. They can be thought as the generalisation of stopping times to a higher dimension.

**Definition 4.1** (Local sets). Consider a random triple  $(\Gamma, A, \Gamma_A)$ , where  $\Gamma$  is a GFF in D, A is a random, relatively closed subset of D and  $\Gamma_A$  a random distribution

that is equal to a harmonic function,  $h_A$ , when restricted to  $D \setminus A$ . We say that A is a local set for  $\Gamma$  if conditionally on A and  $\Gamma_A$ ,  $\Gamma^A := \Gamma - \Gamma_A$  is a GFF in  $D \setminus A$ .

Here, by a random closed set we mean a probability measure on the space of relatively closed subsets of D, endowed with the Hausdorff metric and its corresponding Borel  $\sigma$ -algebra. For simplicity, we will only work with local sets A that are measurable functions of  $\Gamma$  and such that  $A \cup \partial D$  is connected. In particular, this implies that all connected components of  $D \setminus A$  are simply-connected. We define  $\mathcal{F}_A = \sigma(A) \vee \sigma(\Gamma_A)$ .

Other than the Markov property apparent from the definition, we will use the following simple properties of local sets. See for instance [SS13, Wer] for further properties.

**Lemma 4.2.** Let  $(A^n)_{n \in \mathbb{N}}$  be an increasing sequence of local sets measurable w.r.t.  $\Gamma$ . Then

- 1.  $\mathcal{F}_{A^n} \subset \mathcal{F}_{A^{n+1}}$ ,
- 2.  $\overline{\bigcup A^n}$  is also a local set and  $\Gamma_{A_N} \to \Gamma_{\overline{\bigcup A^n}}$  in probability as distributions as  $N \to \infty$ ,
- 3. if  $\overline{\bigcup A^n} = \overline{D}$ , then the join of the  $\sigma$ -algebras  $\mathcal{F}_{A^n}$  is equal to  $\sigma(\Gamma)$ . Moreover,  $\Gamma_n := \Gamma_{A^n}$  then converges to  $\Gamma$  in probability in the space of distributions.

The property (1) follows from the fact that our local sets are measurable w.r.t.  $\Gamma$ and the characterization of local sets found in [SS13]. Properties (2) and (3) follow from the fact that when  $A^n \cup \partial D$  is connected we have that the Green's functions  $G_{D \setminus A^n} \to G_{D \setminus A}$ .

In other words, one can approximate the Gaussian free field by taking an increasing sequence of measurable local sets  $(A^n)_{n \in \mathbb{N}}$  and for each n defining  $\Gamma_n := \Gamma_{A_n}$ . In some sense these give very intrinsic approximations to the GFF. For example, one could intuitively think that  $A^n$  are the sets that discover the part of the surface described by the GFF that is linked to the boundary and has height between -nand n.

#### 4.2.1 Two useful families of local sets

One useful family of local sets are the so-called two-valued local sets introduced in [ASW15] and denoted by  $A_{-a,b}$ . For fixed a, b > 0,  $A_{-a,b}$  is a local set of the GFF such that: the value of  $h_{A_{-a,b}}$  inside each connected component of  $D \setminus A$  is constant with value either -a or b; and that is thin in the sense that for all f smooth we have  $(\Gamma_A, f) = \int_{D \setminus A} f(z)h_A(z) dz$ . Equivalently,  $\Gamma_A$  is equal to  $h_A$  as a distribution. The

prime example of such a set is CLE<sub>4</sub> coupled with the Gaussian free field as  $A_{-2\lambda,2\lambda}$ , where  $\lambda$  is an explicit constant equal to  $\lambda = \pi/2$  in our case [MS11, ASW15]. In analogy with stopping times, they correspond to exit times of Brownian motion from the interval [-a, b]. We recall the main properties of two-valued sets:

**Proposition 4.3.** Let us consider -a < 0 < b.

- 1. When  $a + b < 2\lambda$ , there are no local sets of  $\Gamma$  with the characteristics of  $A_{-a,b}$ .
- 2. When  $a + b \ge 2\lambda$ , it is possible to construct  $A_{-a,b}$  coupled with a GFF  $\Gamma$ . Moreover, the sets  $A_{-a,b}$  are
  - Unique in the sense that if A' is another local set coupled with the same
     Γ, such that for all z ∈ D, h<sub>A'</sub>(z) ∈ {-a, b} almost surely and A' is thin
     in the sense above, then A' = A<sub>-a,b</sub> almost surely.
  - Measurable functions of the GFF  $\Gamma$  that they are coupled with.
  - Monotonic in the following sense: if [a, b] ⊂ [a', b'] and -a < 0 < b with b + a ≥ 2λ, then almost surely, A<sub>-a,b</sub> ⊂ A<sub>-a',b'</sub>.
  - $A_{-a,b}$  has almost surely Lebesgue measure 0.
  - For any z, log CR(z; D\A<sub>−a,b</sub>) − log CR(z; D) has the distribution of the hitting time of {−a,b} by a standard Brownian motion.

Another nice class of local sets are those that only take one value in the complement of A. We call them first passage sets and denote them by  $A_a$  (if they only take the value a). These correspond to one-sided hitting times of the Brownian motion: hence the name. They are of interest in describing the geometry of the Gaussian free field and are treated in more detail in [ALS17]. Here, we only provide one working definition and refer to [ALS17] for a more intrinsic definition, uniqueness and other properties not needed in the current paper.

**Definition 4.4** (First passage set). Take  $a \ge 0$ . We say that  $A_a$  is the first passage set (FPS) of a GFF  $\Gamma$ , with height a, if it is given by  $\overline{\bigcup_n A_{-n,a}}$ .

We need a few properties of these sets. The first follows from the definition, the second and third from calculations in [ASW15] Section 6:

- We have that  $\Gamma_{A_a} = a \nu_a$ , where  $\nu_a$  is a positive measure supported on  $A_a$ ;
- $A_a$  has zero Lebesgue measure;
- For any  $a_n \to \infty$  we have that  $\overline{\bigcup A_{a_n}} = \overline{D}$ .

Note that because the circle-average of the GFF  $(\Gamma, \rho_z^{\varepsilon})$  is a.s. well-defined for all  $z \in D, \varepsilon > 0$  simultaneously, it also means that  $(\nu_a, \rho_z^{\varepsilon})$  is a.s. well-defined and positive for all  $z, \varepsilon$  as above.

In fact these three properties characterize  $A_a$  uniquely [ALS17]. However, in this paper we only need a weaker uniqueness statement that is a consequence of the following lemma:

**Lemma 4.5.** Denote  $A^1 = A_{-a,a}$  with  $a \ge \lambda$  and define iteratively  $A^n$  by exploring copies of  $A_{-a,a}$  in each connected component of the complement of  $A^{n-1}$ . Then, almost surely for a dense countable set  $z \in D$  the following holds: for  $k \in \mathbb{N}$ , let  $n_z$ be the first iteration when  $h_{A^{n_z}}(z) = ak$ , the connected component  $D \setminus A^{n_z}$  containing z is equal to the connected component of  $D \setminus A_{ak}$  containing z.

*Proof.* The proof follows from the uniqueness of two-valued sets  $A_{-a,b}$ . Indeed, construct sets  $B^n$  by taking  $B^1 = A^1$  and then repeating the construction of  $A^i$  only in the components where the value of  $h_{B^n}$  is not yet ak. Thus, by construction  $B^n \subset A^n$ . Moreover, for any z up to and including the first iteration where  $\Gamma_{B^k}(z) = ak$ , the connected component of the complement of  $A^n$  and  $B^n$  containing z coincide.

Now, note that for a fixed  $z \in D$ ,  $n_z$  is almost surely finite. Thus it suffices to prove that for all  $n \in \mathbb{N}$ , the set  $B^n$  is contained in  $A_{-\lceil an \rceil, ak}$  and that all connected components of  $D \setminus B^n$  where  $h_{B^n}$  takes the value ak are connected components of  $D \setminus A_{-\lceil an \rceil, ak}$  where  $h_{A_{-\lceil an \rceil, ak}}$  is equal to ak. To see this, first note that  $h_{B^n} \in$  $\{-an, -a(n-1), \ldots, ak\}$ . In particular, in each connected component where  $h_{B^n} =$  $c \notin \{-\lceil an \rceil, ak\}$  we can construct the two-valued sets  $A_{-\lceil an \rceil - c, ak - c}$ . This gives us a local set  $\tilde{B}$  s.t.  $h_{\tilde{B}}$  takes only values in  $\{-\lceil an \rceil, k\}$ . It is also possible to see that  $\tilde{B}$ is thin, by noting that inside each compact set its Minkowski dimension is smaller than 2 (e.g. see [Sep16, Proposition 4.3]). Then, by uniqueness of the two-valued sets, Lemma 4.3,  $\tilde{B}$  is equal to  $A_{\lceil an \rceil, k}$ . To finish, notice that we the connected components of  $D \setminus B^n$  where  $h_{B^n}$  took the value ak are also connected components of  $\tilde{B}$  with the same value.

In particular, from this lemma it follows that we can also construct  $A_a$  in a different way: denote  $A^1 = A_{-a,a}$  and define  $A^2$  by iterating independent copies of  $A_{-a,a}$  in each component of the complement of  $D \setminus A^1$  where  $h_{A_1} \neq a$ . Repeat this procedure again in all components of the complement for which the value still differs from a. This iteration gives an increasing sequence of local sets  $A^n$ , whose limit is equal to  $A_a$ . For a concrete example, one could take  $A_{-2\lambda,2\lambda}$  to be equal to  $\text{CLE}_4$  in its coupling with the GFF, and the above procedure would yield  $A_{2\lambda}$ . In fact the sets  $(A_{2\lambda n})_{n\in\mathbb{N}}$  are exactly the sets that the author [Aid15] proposes as a basis for the construction of the Liouville measure.
# 4.3 Overview of the Liouville measure and loop constructions of [Aid15]

There are many ways to define the Liouville measure in the subcritical case, the differences amounting to how one approximates the underlying GFF. We will first describe the approximations using circle averages in the subcritical case. Then we will discuss the critical regime, and finally present the nested-loop based constructions from [Aid15] that are conjectured to give the Liouville measure. From now on we will set  $D = \mathbb{D}$  for simplicity.

#### 4.3.1 Subcritical regime

Let us recall that we denote  $\Gamma_{\varepsilon}(z) = (\Gamma, \rho_z^{\varepsilon})$  the  $\varepsilon$ -circle average of the GFF around the point z as before. It is known that  $\Gamma_{\varepsilon}(z)$  is a continuous Gaussian processes that converge to  $\Gamma$  a.s. in the space of distributions as  $\varepsilon \to 0$ . Thus, one can define approximate Liouville measures

$$\mu_{\varepsilon}^{\gamma}(dz) := \varepsilon^{\frac{\gamma^2}{2}} \exp\left(\gamma \Gamma_{\varepsilon}(z)\right) dz.$$

In the subcritical regime we have the following result [DS11, Ber15a]:

**Theorem 4.6.** For  $\gamma < 2$  the measures  $\mu_{\varepsilon}^{\gamma}$  converge to a non-trivial measure  $\mu^{\gamma}$ weakly in probability. Moreover, for any fixed Borel set  $\mathcal{O} \subset \mathbb{D}$  we have that  $\mu_{\varepsilon}^{\gamma}(\mathcal{O})$ converges in  $L^1$  to  $\mu(\mathcal{O})$ .

In fact it is known that the measure is also unique, in the sense that the same limit can be obtained using any sufficiently nice mollifier instead of the circle average. We will show that the approximations using local sets give the same measure.

#### 4.3.2 Critical regime

It is known that for  $\gamma \geq 2$ , the measures  $\mu_{\gamma}^{\varepsilon}$  converge to zero [RV10]. To define the critical measures an additional renormalization is therefore required. One way to do it is to use the so-called derivative martingale, originating from studies on branching random walks. Define

$$\nu_{\varepsilon}(dz) := \left. \frac{\partial}{\partial \gamma} \right|_{\gamma=2} \mu_{\varepsilon}^{\gamma}(dz) = \left( -\Gamma_{\varepsilon}(z) + 2\log(1/\varepsilon) \right) \varepsilon^{2} \exp\left( 2\Gamma_{\varepsilon}(z) \right) dz$$

It has been recently shown in [Pow17a, Theorem 1.1] that  $\nu_{\varepsilon}$  converges weakly in probability to a non-trivial limiting measure  $\mu'_2$  as  $\varepsilon \to 0$ . Moreover,  $\mu'_2$  coincides

with the critical Liouville measure defined in [DRSV14a, DRSV14b]. We will again show that the approximations using local sets converge towards the same measure.

Another way to define the critical measure is to use the so-called Seneta-Heyde renormalization [AS14, DRSV14b]. In the case of the circle-average process the approximating measures would be defined as:

$$\bar{\nu_{\varepsilon}}(dz) := \sqrt{\log 1/\varepsilon} \mu_{\varepsilon}^2(dz)$$

It has been shown [HRV15, JS17] that  $\bar{\nu_{\varepsilon}}$  converges in probability to  $\sqrt{\frac{2}{\pi}}\mu'_2$  as  $\varepsilon \to 0$ . We will prove an analogous result in our setting.

#### 4.3.3 Measures constructed using nested loops

In [Aid15] the author proposes a construction of measures, analogous to the Liouville measure, using nested conformally-invariant loop ensembles. We will now describe it in a concrete context that is related to this paper.

Consider a  $\text{CLE}_4$ , and inside each loop toss an independent fair coin. Keep the loops with heads on top, and sample new  $\text{CLE}_4$  loops in the others. Also toss new independent coins inside these loops. Keep track of all the coin tosses for each loop and repeat the procedure inside each loop where the number of heads is not yet larger than the number of tails. Define the resulting set as  $\tilde{A}^1$ . Now define  $\tilde{A}^k$ iteratively by sampling an independent copy of  $\tilde{A}^1$  inside each connected component of  $\mathbb{D} \setminus \tilde{A}^{k-1}$ .

For any Borelian  $\mathcal{O} \subset \mathbb{D}$  we can now define

$$\tilde{M}_{k}^{\gamma}(\mathcal{O}) = \frac{1}{\mathbb{E}\left[\operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^{1})^{\gamma^{2}/2}\right]^{k}} \int_{\mathcal{O} \cap \mathbb{D} \setminus \tilde{A}^{k}} \operatorname{CR}(z, \mathbb{D} \setminus \tilde{A}^{k})^{\frac{\gamma^{2}}{2}} dz$$
(4.3)

It is shown in [Aid15] that for  $\gamma < 2$  the measures defined by  $\tilde{M}_k^{\gamma}$  converge weakly almost surely to a non-trivial measure  $\tilde{M}^{\gamma}$ . It is also conjectured there that the limiting measures coincide with the Liouville measures  $\mu^{\gamma}$ . We will prove this statement below.

It is further proved in [Aid15] that for  $\gamma \geq 2$ , these measures converge almost surely to zero. In the critical case, however, one can again define a derivative martingale  $\tilde{D}_k^{\gamma}$  by taking a derivative with respect to  $-\gamma$ . In other words one sets:

$$\tilde{D}_k^{\gamma}(\mathcal{O}) = -2\frac{\partial}{\partial\gamma}\tilde{M}_k^{\gamma}(\mathcal{O})$$

(we include the factor 2 here to be consistent with the definition in [Aid15]). It is shown in [Aid15] that the measures  $\tilde{D}_k := \tilde{D}_k^2$  converge to a non-trivial positive measure  $\tilde{D}_{\infty}$ . In this paper, we prove that  $\tilde{D}_{\infty} = 2\mu'_2$ .

# 4.4 Local set approximations of the subcritical Liouville measure

In this section we prove that one can approximate the Liouville measure of a GFF in a simply connected domain using increasing sequences of local sets  $(A^n)_{n \in \mathbb{N}}$  with  $\overline{\bigcup A^n} = \overline{\mathbb{D}}$ . In particular, the measure constructed in [Aid15] will fit in our framework and thus it agrees with the Liouville measure. In fact, for simplicity, we first present the proof of convergence in this specific case.

First, recall that we denote by  $h_A$  the harmonic function given by the restriction of  $\Gamma_A$  to  $\mathbb{D} \setminus A$ . For any local set A with Lebesgue measure 0 and bounded  $h_A$ , we define for any Borelian set  $\mathcal{O} \subseteq D$ :

$$M^{\gamma}(\mathcal{O}, A) := \int_{\mathcal{O}} e^{\gamma h_A} \operatorname{CR}(z; \mathbb{D} \setminus A)^{\gamma^2/2} dz.$$

Notice that as  $h_A$  is bounded, we can define it arbitrarily on the 0 Lebesgue measure set A.

**Proposition 4.7.** Fix  $\gamma \in [0,2)$ . For a > 0, let  $A_a$  be the a-FPS of  $\Gamma$  and  $\mu^{\gamma}$  be the Liouville measure defined by  $\Gamma$ . Then for each Borelian set  $\mathcal{O} \subset \mathbb{D}$ ,

$$M_a^{\gamma}(\mathcal{O}) := M^{\gamma}(\mathcal{O}, A_a) = e^{\gamma a} \int_{\mathcal{O}} CR(z; D \backslash A_a)^{\gamma^2/2} dz$$

is a martingale with respect to  $\mathcal{F}_{A_a}$  and converges a.s. to  $\mu^{\gamma}(\mathcal{O})$  as  $a \to \infty$ . Thus, a.s. the measures  $M_a^{\gamma}$  converge weakly to  $\mu^{\gamma}$ .

Before the proof, we make two remarks. First, we make the connection between our martingale and the martingales of [Aid15]:

**Remark 4.8.** As a consequence of Lemma 4.5, the fact that  $A_{-2\lambda,2\lambda}$  has the law of  $CLE_4$  and the fact that the value of its corresponding harmonic function is independent in each connected component of  $\mathbb{D}\setminus A_{-2\lambda,2\lambda}$  [MS11, ASW15], we see that  $\tilde{A}^1$  of Section 4.3.3 is equal in law to  $A_{2\lambda}$ . Furthermore, the sequence  $(\tilde{A}^k)_{k\in\mathbb{N}}$  has the same law has the sequence  $(A_{2\lambda k})_{k\in\mathbb{N}}$ .

Now, by the iterative construction and conformal invariance the random variables

$$\log \operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^i) - \log \operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^{i-1})$$

with  $A^0 = \emptyset$  are i.i.d. Thus,  $\mathbb{E}\left[\operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^1)^{\frac{\gamma^2}{2}}\right]^k = \mathbb{E}\left[\operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^k)^{\frac{\gamma^2}{2}}\right].$ 

Moreover, it is known from [SSW09, ASW15] that  $-\log \operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^k)$  corresponds precisely to the hitting time of  $k\pi$  by a standard Brownian motion started from zero. In our case, when  $2\lambda = \pi$ , we therefore see that

$$e^{\gamma 2\lambda k} = \mathbb{E}\left[\operatorname{CR}(0, \mathbb{D} \setminus \tilde{A}^1)^{\frac{\gamma^2}{2}}\right]^{-k}$$

Furthermore, since  $Leb(A_{2\lambda}) = 0$  implies that  $M_a^{\gamma}(\mathcal{O} \cap A_{2\lambda}) = 0$ , we have that  $M_{2\lambda k}^{\gamma}$  agrees with the measure  $\tilde{M}_k^{\gamma}$  defined in (4.3). Hence Proposition 4.7 confirms that the limit of  $\tilde{M}_k^{\gamma}$  corresponds to the Liouville measure.

**Remark 4.9.** Second, in order to avoid repetition, we recall here as a remark the standard argument showing that the almost sure weak convergence of measures is implied by the almost sure convergence of  $M_a^{\gamma}(\mathcal{O})$  over all boxes  $\mathcal{O}$  with dyadic coordinates. This follows from two observations: first, the subspace of Radon measures on  $\overline{\mathbb{D}}$  with bounded mass is compact and second, the boxes  $\mathcal{O}$  with dyadic coordinates generate the Borel  $\sigma$ -algebra. Notice that we do not show that we have strong convergence of measures, i.e. we do not know whether almost surely  $\mu^{\gamma}(\mathcal{O})$  is the limit of  $M_a^{\gamma}(\mathcal{O})$  for all Borelian  $\mathcal{O}$ .

Proof of Proposition 4.7. By Remark 4.9, it suffices to prove the convergence statement for  $M_a^{\gamma}(\mathcal{O})$ . When  $\gamma \in [0,2)$ , we know that  $\mu_{\varepsilon}^{\gamma}(\mathcal{O}) \to \mu^{\gamma}(\mathcal{O})$ , in  $\mathcal{L}^1$  as  $\varepsilon \to 0$ , where  $\mu_{\varepsilon}^{\gamma}$  is as in Theorem 4.6. Thus,

$$\mathbb{E}\left[\mu^{\gamma}(\mathcal{O}) \mid \mathcal{F}_{A_{a}}\right] = \lim_{\varepsilon \to 0} \mathbb{E}\left[\mu^{\gamma}_{\varepsilon}(\mathcal{O}) \mid \mathcal{F}_{A_{a}}\right].$$

The key is to argue that

$$\lim_{\varepsilon \to 0} \mathbb{E} \left[ \mu_{\varepsilon}^{\gamma}(\mathcal{O}) \mid \mathcal{F}_{A_a} \right] = M_a^{\gamma}(\mathcal{O}).$$
(4.4)

Then  $M_a^{\gamma}(\mathcal{O}) = \mathbb{E}\left[\mu^{\gamma}(\mathcal{O}) \mid \mathcal{F}_{A_a}\right]$  and we can conclude using the martingale convergence theorem and the fact that  $\overline{\bigcup A_a} = \overline{\mathbb{D}}$ .

To prove (4.4), define  $A_a^{\varepsilon}$  as the  $\varepsilon$ -enlargement of  $A_a$ . By writing  $\Gamma = \Gamma_{A_a} + \Gamma^{A_a}$ and using that  $(\Gamma_{A_a}, \rho_{\varepsilon}^z) = a$  for any  $z \in \mathbb{D} \setminus A_a^{\varepsilon}$ , we have

$$\mathbb{E}\left[\left.\int_{\mathcal{O}\setminus A_a^{\varepsilon}} \mathrm{e}^{\gamma(\Gamma,\rho_{\varepsilon}^z)} \,\varepsilon^{\gamma^2/2} \,dz\right| \mathcal{F}_{A_a}\right] = \int_{\mathcal{O}\setminus A_a^{\varepsilon}} \mathrm{e}^{\gamma a} \,\varepsilon^{\gamma^2/2} \mathbb{E}\left[\left.\mathrm{e}^{\left(\Gamma^{A_a},\rho_{\varepsilon}^z\right)}\right| \mathcal{F}_{A_a}\right] dz$$

Using (4.2) we recognize that the right hand side is just  $M_a^{\gamma}(\mathcal{O} \setminus A_a^{\varepsilon})$ .

But now for any fixed a, as  $\operatorname{CR}(z, \mathbb{D}) \leq 1$  and  $A_a$  has zero Lebesgue measure, we have that  $M_a^{\gamma}(\mathcal{O} \cap A_a^{\varepsilon}) = o_{\varepsilon}(1)$ . On the other hand, from the fact that  $(\Gamma_{A_a}, \rho_{\varepsilon}^z) \leq a$ 

for any z, and (4.2), it follows that

$$\mathbb{E}\left[\left.\int_{\mathcal{O}\cap A_a^{\varepsilon}} \mathrm{e}^{\gamma(\Gamma,\rho_{\varepsilon}^z)} \,\varepsilon^{\gamma^2/2} \,dz\right| \mathcal{F}_{A_a}\right] \leq Leb(A_a^{\varepsilon}) e^{\gamma a}.$$

Thus, we conclude (4.4) and the proof.

We now state a more general version of this result, which says that one can construct the Liouville measure using a variety of local set approximations. The proof is a simple adaptation of the proof above. We say that a generalized function T on  $\mathbb{D}$ , for which the circle-average process  $T_{\varepsilon}(z) := (T, \rho_z^{\varepsilon})$  can be defined, is bounded from above by K if for all  $z \in D$  and  $\varepsilon > 0$ , we have that  $T_{\varepsilon}(z) \leq K$ .

**Proposition 4.10.** Fix  $\gamma \in [0,2)$  and let  $(A^n)_{n \in \mathbb{N}}$  be an increasing sequence of local sets for a GFF  $\Gamma$  with  $\bigcup_{n \in \mathbb{N}} A^n = \overline{\mathbb{D}}$ . Let  $(K_n)_{n \in \mathbb{N}}$  be a sequence of positive integers. Suppose that almost surely for all  $n \in \mathbb{N}$ , we have that  $Leb(A^n) = 0$  and that  $\Gamma_{A^n}$ is bounded from above by  $K_n$ . Then for any Borel  $\mathcal{O} \subset \mathbb{D}$ ,  $M_n^{\gamma}(\mathcal{O})$  defined by

$$M_n^{\gamma}(\mathcal{O}) = \int_{\mathcal{O}} e^{\gamma h_{A^n}(z)} \operatorname{CR}(z; \mathbb{D} \setminus A^n)^{\gamma^2/2} dz$$

is a martingale with respect to  $\{\mathcal{F}_{A^n}\}_{n>0}$  and

$$\lim_{n \to \infty} M_n^{\gamma}(\mathcal{O}) = \mu^{\gamma}(\mathcal{O}) \ a.s.$$

where  $\mu^{\gamma}$  is the Liouville measure defined by  $\Gamma$ . Thus, almost surely the measures  $M_n^{\gamma}$  converge weakly to  $\mu^{\gamma}$ .

Let us mention two natural sequences of local sets for which this proposition applies. The first is when we take  $a_n, b_n \nearrow \infty$  and study the sequence  $(A_{-a_n,b_n})_{n \in \mathbb{N}}$ . The second is when we take the sequence  $(A_{-a,b}^n)_{n \in \mathbb{N}}$  for some a, b > 0, where  $A_{-a,b}^n$ is defined by iteration <sup>6</sup>. Note that in the case where  $a = b = 2\lambda$ , we recover the result described in the introduction for the iterated CLE<sub>4</sub>.

Observe that whereas our martingale agrees with the one given in [Aid15] for the case of first-passage sets, for any cases where  $h_{A^n}$  can take more than one value, the martingales are in fact different. Yet, we can still identify the limit of the martingale  $\tilde{M}_n^{\gamma}(\mathcal{O})$  of [Aid15], corresponding to an iterated CLE<sub>4</sub> (i.e.  $(\text{CLE}_4^n)_{n \in \mathbb{N}}$ .) In this case Aidekon's martingale converges in distribution to  $\eta^{\gamma}(\mathcal{O}) := \mathbb{E} [\mu^{\gamma}(\mathcal{O}) | \mathcal{F}_{\infty}]$ , where  $\mu^{\gamma}$  is the Liouville measure and  $\mathcal{F}_{\infty}$  is the  $\sigma$ -algebra containing only the geometric information from all iterations of the CLE<sub>4</sub>. This  $\sigma$ -algebra is strictly smaller than

<sup>&</sup>lt;sup>6</sup>We set  $A_{-a,b}^1 = A_{-a,b}$  and define  $A_{-a,b}^n$  by sampling the  $A_{-a,b}$  of  $\Gamma^{A_{-a,b}^{n-1}}$  inside each connected component of  $D \setminus A_{-a,b}^{n-1}$ 

 $\mathcal{F}_{A_{-2\lambda,2\lambda}^n}$ , which also contains information on the labels of CLE<sub>4</sub> in its coupling with the GFF. It is not hard to see that  $\eta^{\gamma}$  is not equal to  $\mu^{\gamma}$ .

## 4.5 Generalizations

In this section, we describe some other situations where an equivalent of Proposition 4.10 can be proven using the same techniques as the proof of Proposition 4.7. In the following we do not present any new methods, but focus instead on announcing the propositions in context, so that they may be used in other works. We also make explicit the places where the results are already, or may in the future, be used.

# 4.5.1 Non-simply connected domains and general boundary conditions.

Here we consider the case when  $\Gamma$  is a GFF in an n-connected domain  $D \subseteq \mathbb{D}$  (for more context see [ALS17]). First, let us note that in this set-up (4.2) becomes

$$\mathbb{E}\left[ \begin{array}{ll} \varepsilon^{\frac{\gamma^2}{2}} \exp\left(\gamma(\Gamma,\rho_z^{\varepsilon})\right) \right] \left\{ \begin{array}{ll} = e^{-\frac{\gamma^2}{2}\tilde{G}_D(z,z)} & \text{ if } d(z,\partial D) \geq \varepsilon, \\ \leq 1 & \text{ if } d(z,\partial D) < \varepsilon, \end{array} \right.$$

where we write  $G_D(z, w) = -\log |z-w| + \tilde{G}_D(z, w)$ , i.e. for any  $z \in D$ ,  $\tilde{G}_D(z, \cdot)$ , is the bounded harmonic function that has boundary conditions  $\log(|z-w|)$  for  $w \in \partial D$ . Additionally, if we work with local sets A such that all connected components of  $A \cup \partial D$  contain an element of  $\partial D$ , then Lemma 4.2 will hold. All local sets we refer to here are assumed to satisfy this condition. These facts and assumptions are enough to prove the following proposition:

**Proposition 4.11.** Fix  $\gamma \in [0,2)$  and let  $(A^n)_{n \in \mathbb{N}}$  be an increasing sequence of local sets for a GFF  $\Gamma$  with  $\overline{\bigcup_{n \in \mathbb{N}} A^n} = \mathbb{D}$ . Suppose that almost surely for all  $n \in \mathbb{N}$ , we have that  $Leb(A^n) = 0$  and that  $\Gamma_{A^n}$  is bounded from above by  $K_n$  for some sequence of finite  $K_n$ . Then for any Borel  $\mathcal{O} \subset D$ ,  $M_n^{\gamma}(\mathcal{O})$  defined by

$$M_n^{\gamma} := \int_{\mathcal{O}} e^{\gamma h_{A_n}(z) - \frac{\gamma^2}{2} \tilde{G}_{D \setminus A_n}(z, z)} \, dz, \quad \mathcal{O} \subset \mathbb{D}$$

is a martingale with respect to  $\{\mathcal{F}_{A^n}\}_{n>0}$  and

$$\lim_{n \to \infty} M_n^{\gamma}(\mathcal{O}) = \mu^{\gamma}(\mathcal{O}) \ a.s.$$

where  $\mu^{\gamma}$  is the Liouville measure defined by  $\Gamma$ . Thus, almost surely the measures  $M_n^{\gamma}$  converge weakly to  $\mu^{\gamma}$ .

The equivalent of the sets  $A_{-a,b}$  and  $A_a$  are defined in *n*-connected domains in [ALS17] and it is easy to see that their iterated versions satisfy the hypothesis of Proposition 4.11. In particular, the above construction allows the authors in [ALS17] to prove that the measure  $\Gamma_{A_a}$  is a measurable function of  $A_a$ .

#### 4.5.2 Dirichlet-Neumann GFF

In this section we take  $\Gamma$  to be a GFF with Dirichlet-Neumann boundary conditions in  $\mathbb{D}^+ = \mathbb{D} \cap \mathbb{H}$ . That is,  $\Gamma$  satisfies (4.1), with  $G_D$  replaced by  $G_{\mathbb{D}^+}$ : the Green's function in  $\mathbb{D}^+$  with Dirichlet boundary conditions on  $\partial \mathbb{D}$  and Neumann boundary conditions on [-1, 1]. To be more specific, we set  $G_{\mathbb{D}^+}(x, y) = G_{\mathbb{D}}(x, y) + G_{\mathbb{D}}(x, \bar{y})$ , with  $G_{\mathbb{D}}$  as in Section 4.2. Then  $G_{\mathbb{D}^+}(x, y) \sim \log(1/|x - y|)$  as  $x \to y$  in the interior of  $\mathbb{D}^+$  and  $G_{\mathbb{D}^+}(x, y) \sim 2\log(1/|x - y|)$  when  $y \in (0, 1)$ .

Let A be a closed subset of  $\overline{\mathbb{D}}^+$ . Suppose that  $\Gamma$  is a Dirichlet-Neumann GFF in  $\mathbb{D}^+ \setminus A$  with Neumann boundary conditions on  $[-1,1] \setminus A$  and Dirichlet boundary conditions on the rest of the boundary. Let  $z \in [-1,1]$  and define  $\varrho_z^{\varepsilon}$  to be the uniform measure on  $\partial B(z,\varepsilon) \cap \mathbb{D}^+$ . Then, in this set-up (4.2) becomes

$$\mathbb{E}\left[\varepsilon^{\gamma^{2}/4}\exp\left(\frac{\gamma}{2}(\Gamma,\varrho_{x}^{\varepsilon})\right)\right]\left\{\begin{array}{ll} = \operatorname{CR}(x;\mathbb{D}\backslash\check{A})^{\gamma^{2}/4} & \text{if } d(z,\partial(\mathbb{D}\backslash\check{A})) \ge \varepsilon, \\ \le 1 & \text{if } d(z,\partial(\mathbb{D}\backslash\check{A})) < \varepsilon. \end{array}\right.$$
(4.5)

Here we set  $\check{A} := A \cup \bar{A}$  for  $\bar{A} = \{z \in \mathbb{C} : \bar{z} \in A\}.$ 

There is also a notion of local sets for this Dirichlet-Neumann GFF. We say that  $(\Gamma, A, \Gamma_A)$  describes a local set coupling if, conditionally on  $(A, \Gamma_A)$ ,  $\Gamma^A := \Gamma - \Gamma_A$  is a GFF with Neumann boundary conditions on  $[-1, 1] \setminus A$  and Dirichlet on the rest. For connected local sets such  $\partial \mathbb{D}^+ \cup A$  is connected, Lemma 4.2 still holds (by the same proof given for the 0-boundary GFF).

We are interested in the boundary Liouville measure on [-1, 1]. Take  $\gamma < 2$ ,  $\varepsilon > 0$  and a Borel set  $\mathcal{O} \subseteq [-1, 1]$ . We define the approximate boundary Liouville measures as follows:

$$v_{\varepsilon}^{\gamma}(\mathcal{O}) := \varepsilon^{\gamma^2/4} \int_{\mathcal{O}} \exp\left(\frac{\gamma}{2}(\Gamma, \varrho_x^{\varepsilon})\right) dx$$

where here dx is the Lebesgue density on [-1, 1]. It is known (see [DS11, Ber15a]) that  $v_{\varepsilon}^{\gamma} \to v^{\gamma}$  in  $\mathcal{L}^1$  as  $\varepsilon \to 0$ . Moreover, it is also easy to see that  $v^{\gamma}$  is a measurable function of  $\mathcal{F}_{[-1,1]}$  - this just comes from the fact that the Dirichlet GFF contains no information on the boundary. Thus, we have all the necessary conditions to deduce the following Proposition using exactly the same proof as in Section 4.4.

**Proposition 4.12.** Fix  $\gamma \in [0,2)$  and let  $(A^n)_{n \in \mathbb{N}}$  be an increasing sequence of local sets for a GFF  $\Gamma$  with  $\bigcup_{n \in \mathbb{N}} A^n \supseteq [-1,1]$ . Suppose that almost surely for all  $n \in \mathbb{N}$ , we have that  $Leb_{[-1,1]}(A_n) = 0$  and that  $\Gamma_{A^n}$  restricted to  $A^n$  is bounded from above by  $K_n$  for some sequence of finite  $K_n$ . Then for any Borel  $\mathcal{O} \subset [-1,1]$ ,  $M_n^{\gamma}(\mathcal{O})$ defined by

$$M_n^{\gamma}(\mathcal{O}) := \int_{\mathcal{O}} e^{\frac{\gamma}{2}h_{A_n}(z)} \operatorname{CR}(z; \mathbb{D} \backslash \breve{A}_n)^{\frac{\gamma^2}{4}} dz$$

is a martingale with respect to  $\{\mathcal{F}_{A^n}\}_{n>0}$  and

$$\lim_{n \to \infty} M_n^{\gamma}(\mathcal{O}) = v^{\gamma}(\mathcal{O}) \ a.s.$$

where where  $\mu^{\gamma}$  is the boundary Liouville measure defined by  $\Gamma$ . Thus, almost surely the measures  $M_a^{\gamma}$  converge weakly to  $v^{\gamma}$ .

It has recently been proven in [QW17] that sets satisfying the above hypothesis do exist, and that they can be used to couple the Dirichlet GFF with the Neumann GFF. Let us describe some concrete examples of these sets. If  $\Gamma$  is a Dirichlet-Neumann GFF, then in [QW17] it is shown that there exists a (measurable) thin local set  $\tilde{A}(\Gamma)$  of the GFF such that:

- $\tilde{A}(\Gamma)$  has the law of the trace of an SLE<sub>4</sub>(0; -1) going from -1 to 1
- $h_{\tilde{A}(\Gamma)}$  is equal to 0 in the only connected component of  $\mathbb{D}^+ \setminus \tilde{A}(\Gamma)$  whose boundary intersects  $\partial \mathbb{D} \cap \mathbb{H}$
- in the other connected components,  $h_{\tilde{A}(\Gamma)}$  is equal to  $\pm 2\lambda$ , where conditionally on  $\tilde{A}(\Gamma)$  the sign is chosen independently in each component.

There are two interesting sequences of local sets we can construct using this basic building-block. The first one is the boundary equivalent of  $(A_{-2\lambda,2\lambda}^n)_{n\in\mathbb{N}}$ , and the second is the boundary equivalent of  $(A_{2\lambda n})_{n\in\mathbb{N}}$ . The first one is also described in [QW17, Section 3]. The construction goes as follows: choose  $A^1 = \tilde{A}(\Gamma)$  and construct  $A^n$  by induction. In the connected components O of  $\mathbb{D}\backslash A^n$  that contain an interval of  $\mathbb{R}$ , we have that  $\Gamma^{A_n}$  restricted to O is a Dirichlet-Neumann GFF (with Neumann boundary condition on  $\mathbb{R} \cap \partial O$ ). Thus, by conformal invariance we can explore the set  $\tilde{A}(\Gamma \mid_O)$  in each such component O. We define  $A^{n+1}$  to be the closed union of  $A^n$  with  $\tilde{A}(\Gamma \mid_O)$  over all explored components O. Note that  $h_{A^n} \in \{2\lambda k\}$ where k ranges between -n and n. It is also not hard to see that  $A^n$  is thin (it follows from the fact that  $h_A \in \mathcal{L}^1(\mathbb{D}\backslash A)$  and that for any compact set  $K \subseteq \mathbb{D}^+$  the Minkowski dimension of  $A^n \cap K$  is a.s. equal to 3/2, see e.g. [Sep16, Proposition 4.3]). Thus we deduce that  $\Gamma_{A^n} \leq 2\lambda n$ . Additionally, note that by adjusting [MS11, Lemma 6.4], we obtain from the construction of  $A^1$  that for any  $z \in (-1, 1)$  the law of  $2(\log(\operatorname{CR}^{-1}(z, \mathbb{D} \setminus \check{A}^1)) - \log(\operatorname{CR}^{-1}(z, \mathbb{D}))$  is equal to the first time that a BM exits  $[-2\lambda, 2\lambda]$ . It follows that for all  $n \in \mathbb{N}$ ,  $Leb_{\mathbb{R}}(A_n \cap [-1, 1]) = 0$  and also  $\overline{\bigcup_{n \in \mathbb{N}} A^n} \supseteq [-1, 1]$ . Hence we see that the sequence  $(A^n)_{n \in \mathbb{N}}$  satisfies the conditions of Proposition 4.12.

For the second sequence of local sets, take  $B^1 = \tilde{A}(\Gamma)$  and define  $B^{n+1}$  to be the closed union of  $B^n$  with all  $\tilde{A}(\Gamma |_O)$  such that O is a connected component of  $\mathbb{D}\backslash B^n$ ,  $h_{B^n} |_O \leq 2\lambda$  and  $\partial O$  contains an interval of  $\mathbb{R}$ . Denote  $A^1(\Gamma)$  the closed union of all the  $B^n$ . Due to the fact that  $B^n$  are BTLS with  $h_{B^n} \leq 2\lambda$  on [-1, 1], we have that  $\Gamma_{A^1}$  restricted to [-1, 1] is smaller than or equal to  $2\lambda$ . Additionally, note that  $2(\log(\operatorname{CR}^{-1}(z, \mathbb{D}\backslash \check{A}^1)) - \log(\operatorname{CR}^{-1}(z, \mathbb{D}))$  is distributed as the first time a BM hits  $2\lambda$ . Now, we iterate to define  $A^n(\Gamma)$  as the closed union of  $A^{n-1}(\Gamma)$  and  $A^1(\Gamma |_O)$ , where O ranges over all connected components of  $\mathbb{D}^+\backslash A^{(n-1)}$  containing an interval of  $\mathbb{R}$ . The sequence  $(A^n)_{n\in\mathbb{N}}$  satisfies the condition of Proposition 4.12. Note that in this case the martingale simplifies and contains only information on the geometry of the sets  $A^n$ :

$$M_n^{\gamma} := e^{\gamma 2\lambda n} \int_{\mathcal{O}} CR(z; \mathbb{D} \backslash \breve{A}^n)^{\gamma^2/4} dz.$$

The fact that this martingale is a measurable function of  $A^n$  allows us to use the same techniques as in [ALS17] to prove that the measure  $2\lambda n - \Gamma_{A^n}$  on  $\mathbb{R}$  is a measurable function of  $A^n$ .

It is also explained in [QW17] that the sets  $A^n$  we have just constructed, and the definition of the boundary Liouville measure using them, might help to reinterpret an SLE-type of conformal welding first studied in [She16].

#### 4.6 Critical and supercritical regimes

In this section it is technically simpler to restrict ourselves to the simply connected case and study a special family of sequences of local sets, though the results hold in a more general setting. Namely, we assume that our sets  $A^n$  are formed by an iterative procedure. That is,  $A^1 = A(\Gamma)$  is some measurable local set coupled with the GFF  $\Gamma$ , and  $A^{n+1}$  is formed from  $A^n$  by, in each component O of  $\mathbb{D} \setminus A^n$ , exploring  $A(\Gamma^{A^n})$ . Notice, for example, that the iterated  $\text{CLE}_4$  coupling described in the introduction is covered by this hypothesis, as are the couplings with  $A_{an}$  for any a > 0.

We first show that the martingales defined in Section 4.4 converge to zero for  $\gamma \geq 2$ . Then, in the critical case  $\gamma = 2$ , we define a derivative martingale and show it converges to the same measure as the critical measure  $\mu'_2$  from [DRSV14a,

DRSV14b, Pow17a], and 1/2 times the critical measure  $D_{\infty}$  from [Aid15]. Finally, we show that for  $A^n = A_{an}$  we can also construct the critical measure using the Seneta-Heyde rescaling (analogous to the main theorem of [AS14].) More precisely, for all Borelian  $\mathcal{O} \subset \mathbb{D}$ , we have that  $\sqrt{an}M^2(\mathcal{O}, A^n)$  converges in probability to  $\frac{4}{\sqrt{\pi}}\mu'_2(\mathcal{O})$  as  $n \to \infty$ .

#### 4.6.1 (Super)critical regime.

**Lemma 4.13.** Set  $\gamma \geq 2$  and assume that  $A^n$  is formed by iteration as above and that  $\overline{\bigcup_n A_n} = \overline{\mathbb{D}}$ . Assume further that A is such that  $h_A$  is constant in each connected component of  $\mathbb{D} \setminus A$  almost surely. Then  $M_n^{\gamma} \to 0$  almost surely.

**Remark 4.14.** Due to the iterative nature of the construction, the condition  $\overline{\bigcup_n A_n} = \mathbb{D}$ , i.e. that in the limit the iterated sets cover the whole domain, is implied for example by a simple requirement on the local set  $A = A_1$  - it suffices to have that there exists  $\varepsilon, \delta > 0$  such that for any point  $z \in \mathbb{D}$ , the probability that  $\operatorname{CR}(z; \mathbb{D} \setminus A_1) < (1 - \varepsilon) \operatorname{CR}(z; \mathbb{D})$  is bigger than  $\delta$ .

In [Aid15], Aïdekon also considers the critical and supercritical cases for his iterated loop measures. In particular, from his results one can read out that, with the notation of Proposition 4.7, for any a > 0 and  $\gamma \ge 2$ , we have  $M_{an}^{\gamma} \to 0$  almost surely as  $n \to \infty$ .

The proof follows from a classical technique stemming from the literature on branching random walks [Lyo97], but is based on the local set coupling with the GFF.

*Proof.* From (4.2) and a direct calculation we see that  $M_n^{\gamma}(\mathbb{D})/M_0^{\gamma}(\mathbb{D})$  is a mean one martingale, where  $M_0^{\gamma}(\mathbb{D}) = \int_{\mathbb{D}} \operatorname{CR}(z, \mathbb{D})^{\gamma^2/2} dz$ . Let us define a new probability measure  $\hat{\mathbb{P}}$  via the change of measure

$$\frac{d\hat{\mathbb{P}}}{d\mathbb{P}}\bigg|_{\mathcal{F}_{A^n}} = \frac{M_n^{\gamma}(\mathbb{D})}{M_0^{\gamma}(\mathbb{D})}.$$
(4.6)

It is well known, see for example [Dur], that in order to show that  $M_n^{\gamma}(\mathbb{D}) \to 0$ almost surely under  $\mathbb{P}$ , it suffices to prove that  $\limsup_n M_n^{\gamma}(\mathbb{D}) = +\infty$  a.s. under  $\hat{\mathbb{P}}$ .

To show this we actually consider a change of measure on an enlarged probability space. Define a measure  $\mathbb{P}^*$  on  $(\Gamma, (A^n)_n, Z)$  by sampling  $(\Gamma, (A^n)_n)$  from  $\mathbb{P}$  and then independently, sampling a random variable  $Z \in \mathbb{D}$  with law proportional to Lebesgue measure. Note that under  $\mathbb{P}^*$  the process

$$\xi_n = \mathrm{e}^{\gamma h_{A^n}(Z) + \gamma^2/2 \log \mathrm{CR}(Z, \mathbb{D} \setminus A^n)}$$

is a martingale with respect to the filtration  $\mathcal{F}_{A^n}^* = \mathcal{F}_{A^n} \vee \sigma(Z)$ . Thus we can define a new probability measure  $\hat{\mathbb{P}}^*$  by

$$\left. \frac{d\hat{\mathbb{P}}^*}{d\mathbb{P}^*} \right|_{\mathcal{F}^*_{A^n}} := \frac{\xi_n}{\mathbb{E}[\xi_0]} \tag{4.7}$$

Then if  $\hat{\mathbb{P}}$  is the restriction of  $\hat{\mathbb{P}}^*$  to  $\mathcal{F}_{A^n}$ ,  $\hat{\mathbb{P}}$  and  $\mathbb{P}$  satisfy (4.6). Therefore it suffices to prove that under  $\hat{\mathbb{P}}^*$  and conditionally on Z, we have  $\limsup_n M_n^{\gamma}(\mathbb{D}) = +\infty$ almost surely. By the Köebe-(1/4) Theorem and [Aid15, Lemma 2.4] we only need to prove that under this law almost surely

$$\limsup_{n} e^{\gamma h_{A^n}(Z) + (\gamma^2/2 + 2) \log \operatorname{CR}(Z, \mathbb{D} \setminus A^n)} = +\infty.$$

However we can calculate, using (4.2) that

$$\gamma h_{A^n}(Z) + (\gamma^2/2 + 2) \log \operatorname{CR}(Z, \mathbb{D} \setminus A^n)$$

is a random walk with non-negative mean (started from  $(\gamma^2/2 + 2) \log \operatorname{CR}(Z, \mathbb{D}))$ under this law. This allows us to conclude.

#### 4.6.2 The derivative martingale in the critical regime

We now show the convergence of the derivative martingale (when  $\gamma = 2$ , defined below) for the particular case of iterated  $A_{-a,a}$ ,  $a \geq \lambda$ . For any Borel set  $\mathcal{O} \subseteq \mathbb{D}$ and local set A, we define

$$D^{\gamma}(\mathcal{O}, A) := \int_{\mathcal{O}} \left( -h_A(z) + \gamma \log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A) \right) e^{\gamma h_A(z)} \operatorname{CR}(z; \mathbb{D} \setminus A)^{\gamma^2/2} dz.$$

The rest of this section is devoted to proving the following proposition.

**Proposition 4.15.** Assume that  $A^n$  is formed by iterating  $A_{-a,a}$  n times, for  $a \ge \lambda$ . Then for any Borel  $\mathcal{O} \subset \mathbb{D}$  we have that  $\hat{D}_n(\mathcal{O}) := D^2(\mathcal{O}, A^n)$  is a martingale and converges almost surely to a finite, positive limit  $\hat{D}_{\infty}(\mathcal{O})$  as  $n \to \infty$ . In particular the signed measures  $\hat{D}_n(\mathcal{O})$  converge weakly to a limiting measure that is independent of the choice of a > 0 and agrees with the critical measure  $\mu'_2$  defined in [DRSV14a, DRSV14b], and 1/2 times the critical measure  $\tilde{D}_{\infty}$  defined in Theorem

#### 1.3 of [Aid15].

Before the proof, let us first comment on the case where the set we are going to iterate is  $A_a$ ; so *n* iterations gives  $A_{an}$ . In the case  $a = 2\lambda$ , observe that twice the derivative martingale  $2D^2(\mathcal{O}, A_{2\lambda n})$  is equal to  $\tilde{D}_n$  defined in (1.3) of [Aid15] (see Remark 4.8). Thus, we know that when we iterate  $A_{2\lambda}$ , its associated sequence of measures converges to a limit  $\tilde{D}_{\infty}$ . In fact it follows from [Aid15], that for all dyadic  $a \ge 0$ ,  $D_n(\mathcal{O}) := 2D^2(\mathcal{O}, A_{an})$  converges to the same limit. Doob's maximal inequality then implies that there exists a modification of  $2D^2(\mathcal{O}, A_t)$  that also converges to  $\tilde{D}_{\infty}$  as  $t \to \infty$ .

These martingales are not uniformly integrable (U.I.) and thus, our previous techniques do not apply directly. However, we will discuss how to pass through certain U.I. martingales to get convergence in the case of  $D_n(\mathcal{O})$ . We will then use this case to show convergence for  $\hat{D}_n(\mathcal{O})$ . We remark that these U.I. martingales, given in the proof below, are similar but not exactly the same as the analogous U.I. martingales introduced in [DRSV14a, Aid15].

We make  $D_n(\mathcal{O})$  uniformly integrable via localization. To do this, let us introduce the following stopping times:

$$\tau_{\beta} := \inf \left\{ n \in \mathbb{N} : \inf_{z \in \mathbb{D} \setminus A_{an}} -h_{A_{an}}(z) + 2\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_{an}) \le -\beta \right\}.$$

Notice that  $A_{a(n\wedge\tau_{\beta})}$  is then also a local set of  $\Gamma$  and we can define  $D_n^{\beta}(\mathcal{O}) := 2D^2(\mathcal{O}, A_{a(n\wedge\tau_{\beta})})$ . As always, we include the factor 2 for comparison with martingales in [Aid15]. Then  $D_n^{\beta}(\mathcal{O})$  is a martingale, due to the fact that it is the derivative with respect to  $-\gamma$  of the martingale  $2M^{\gamma}(\mathcal{O}, A_a^{n\wedge\tau_{\beta}})$ .

 $D_n^{\beta}(\mathcal{O})$  is uniformly integrable. Let us first show the following claim:

**Claim 4.16.** The martingale  $D_n^{\beta}(\mathcal{O})$  is uniformly integrable for all  $\beta \geq 0$  and so converges almost surely and in  $\mathcal{L}^1$  to some limit  $L(\mathcal{O}, \beta)$  as  $n \to \infty$ .

*Proof.* Indeed, for  $\eta > 0$ , let  $E_{\eta}(n, z)$  be the event that

$$\{-am + 2\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_a^m) \ge -\eta \text{ for all } m \le n\}.$$

Then Proposition 3.2. of [Aid15] implies that for all  $\eta > 0$ ,

$$\bar{D}_{n}^{\eta}(\mathcal{O}) := \int_{\mathcal{O}} h_{1} \left( -2an + 4\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_{a}^{n}) + 2\eta \right) \mathbb{1}_{E_{\eta}(n,z)} e^{2an} \operatorname{CR}(z; \mathbb{D} \setminus A_{a}^{n})^{2} dz$$

$$(4.8)$$

is a U.I. martingale. Here  $h_1(u)$  is a so-called renewal function, that satisfies  $h_1(u) \ge Ru$  for some R > 0. We conclude that the stopped martingale  $\bar{D}^{\eta}_{n \wedge \tau_{\beta}}(\mathcal{O})$  is also U.I. Given that

$$-h_{A_a^{n\wedge\tau_{\beta}}}(z) + \gamma \log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_a^{n\wedge\tau_{\beta}}) \ge -a - \beta$$

we can bound

$$|D_n^{\beta}(\mathcal{O})| \le R^{-1} |\bar{D}_{n \wedge \tau_{\beta}}^{2\beta+2a}|.$$

The claim follows.

Comparison with Aïdekon's limit. We first show that our UI martingales converge to Aïdekon's limit, and then use this to treat the case where  $A^n$  is formed by iterating  $A_{-a,a}$ .

**Claim 4.17.** The martingales  $D_n^{\beta}(\mathcal{O})$  converge to  $\tilde{D}_{\infty}(\mathcal{O})$  as first  $n \to \infty$  and then  $\beta \to \infty$ .

*Proof.* From the definition,  $D_n^{\beta}(\mathcal{O})$  and  $2D^2(\mathcal{O}, A_{an})$  are equal on the event  $\{\tau_{\beta} = \infty\}$ . Additionally,  $\mathbb{P}(\tau_{\beta} = \infty)$  is equal to  $1 - o(\beta)$ , due to the fact that almost surely

$$\inf_{z\in\mathbb{D}}\inf_{n\in\mathbb{N}}\left(-2an+4\log\operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_{an})\right) > -\infty.$$
(4.9)

This is proved [Aid15] after the statement of Proposition 3.2 (see also Remark 4.8 of the current paper).

In particular, as  $2D^2(\mathcal{O}, A_{an})$  tends to  $\tilde{D}_{\infty}(\mathcal{O})$  by [Aid15] (together with the comments after Proposition 4.15) we see that  $\lim_{\beta \to \infty} L(\mathcal{O}, \beta)$  is also equal to  $\tilde{D}_{\infty}(\mathcal{O})$  in this case.

Now, let us come back to the case where  $A^n$  is formed by iterating  $A_{-a,a}$ . To do this, define  $\hat{D}_n^{\beta}(\mathcal{O}) := D^2(\mathcal{O}, A^{n \wedge \hat{\tau}_{\beta}})$ , for

$$\hat{\tau}_{\beta} := \inf_{n} \left\{ n \in \mathbb{N} : \inf_{z \in \mathbb{D} \setminus A^{n}} -h_{A^{n}}(z) + 2\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A^{n}) \le -\beta \right\}.$$

First, observe that when both  $A^n$ , the iterated  $A_{-a,a}$ , and  $A_{an}$  are coupled with the same GFF as local sets, Lemma 4.5 implies that a.s.  $\{\hat{\tau}_{\beta} = \infty\} = \{\tau_{\beta} = \infty\}$  and that for all  $n \leq m$ ,  $A^{n \wedge \hat{\tau}_{\beta}} \subseteq A_{a(m \wedge \tau_{\beta})}$ . Thus, as long as the limit on the lefthand side exists, we have

$$\lim_{\beta \to \infty} \lim_{n \to \infty} \hat{D}_n^{\beta}(\mathcal{O}) = \lim_{n \to \infty} \hat{D}_n(\mathcal{O}).$$

So, it suffices to argue that

$$2\hat{D}_n^{\beta}(\mathcal{O})\mathbf{1}_{\{\hat{\tau}_{\beta}=\infty\}} \to L(\mathcal{O},\beta)\mathbf{1}_{\{\hat{\tau}_{\beta}=\infty\}}, \quad \text{as } n \to \infty.$$

To see this we will use the strategy of the proof of Proposition 4.7.

Namely, consider a local set A with zero Lebesgue measure, and such that  $(\Gamma_A, f)$  is bounded from above by  $K \ge 0$  (in the sense explained before Proposition 4.10). Then from an explicit calculation similar to the key claim of Proposition 4.7 we have that for any  $\gamma > 0$ , a.s. and in  $\mathcal{L}^1(\Omega)$  as  $\varepsilon \to 0$ ,

$$\mathbb{E}\left[\int_{\mathcal{O}} (\gamma \log(1/\varepsilon) - \Gamma_{\varepsilon}(z)) e^{\gamma \Gamma_{\varepsilon}(z)} \varepsilon^{\frac{\gamma^2}{2}} dz \mid \mathcal{F}_A\right] \to D^{\gamma}(\mathcal{O}, A).$$

Now, for  $n \leq m$  let  $\hat{\mathcal{F}}_n$  and  $\mathcal{G}_m$  be the sigma-algebras corresponding to the local sets  $A^{n \wedge \hat{\tau}_\beta}$  and  $A_{a(m \wedge \tau_\beta)}$  respectively. Noting that a.s.  $A^{n \wedge \hat{\tau}_\beta} \subseteq A_{a(m \wedge \tau_\beta)}$ ,

$$\begin{split} \frac{1}{2} \mathbb{E} \left[ D_m^{\beta}(\mathcal{O}) \mid \hat{\mathcal{F}}_n \right] &= \mathbb{E} \left[ \lim_{\varepsilon \to 0} \mathbb{E} \left[ \int_{\mathcal{O}} (\gamma \log(1/\varepsilon) - \Gamma_{\varepsilon}(z)) e^{\gamma \Gamma_{\varepsilon}(z)} \varepsilon^{\frac{\gamma^2}{2}} dz \mid \mathcal{G}_m \right] \mid \hat{\mathcal{F}}_n \right] \\ &= \lim_{\varepsilon \to 0} \mathbb{E} \left[ \int_{\mathcal{O}} (\gamma \log(1/\varepsilon) - \Gamma_{\varepsilon}(z)) e^{\gamma \Gamma_{\varepsilon}(z)} \varepsilon^{\frac{\gamma^2}{2}} dz \mid \hat{\mathcal{F}}_n \right] = \hat{D}_n^{\beta}(\mathcal{O}). \end{split}$$

Due to the fact that  $D_m^{\beta}(\mathcal{O}) \to L(\mathcal{O},\beta)$  in  $L^1$  we have that  $\mathbb{E}\left[L(\mathcal{O},\beta)|\hat{\mathcal{F}}_n\right] = 2\hat{D}_n^{\beta}(\mathcal{O})$  and so  $2\hat{D}_n^{\beta}(\mathcal{O}) \to \mathbb{E}\left[L(\mathcal{O},\beta)|\hat{\mathcal{F}}_{\infty}\right]$ . However, the event  $\{\hat{\tau}_{\beta} = \infty\}$  is  $\hat{\mathcal{F}}_{\infty}$ -measurable and on this event the limit of  $A^{n\wedge\hat{\tau}_{\beta}}$  is  $\mathbb{D}$  (by Lemma 4.3 and Remark 4.14). Similarly to Lemma 4.2, it then follows that  $F(\Gamma)\mathbf{1}_{\{\hat{\tau}_{\beta}=\infty\}}$  is  $\hat{\mathcal{F}}_{\infty}$  measurable for any measurable function of  $\Gamma$ . Thus, we have that  $2\hat{D}_n^{\beta}(\mathcal{O})\mathbf{1}_{\{\tau_{\beta}=\infty\}} \to L(\mathcal{O},\beta)\mathbf{1}_{\{\tau_{\beta}=\infty\}}$ , as required.

Comparison with mollified measures. It remains to prove the latter claim of the proposition, i.e. to show that the limiting measure  $\hat{D}_{\infty} = \frac{1}{2}\tilde{D}_{\infty}$  is equal to the measure  $\mu'_2$  from [DRSV14b, Pow17a], described in Section 4.3.3. We again mollify our measures using the circle average, and choose a sequence  $\varepsilon_k \to 0$  such that  $\nu_{\varepsilon} \to \mu'_2$  a.s. Whenever we write  $\varepsilon \to 0$ , it means that we are converging to 0 via  $(\varepsilon_k)_{k\in\mathbb{N}}$ . We set, for fixed  $\mathcal{O} \subset \mathbb{D}$ ,

$$\nu_{\varepsilon}^{\beta}(\mathcal{O}) = \int_{\mathcal{O}} (-\Gamma_{\varepsilon}(z) + 2\log(1/\varepsilon)) \mathbf{1}_{\{T_{\beta}(z) \le \varepsilon\}} e^{2\Gamma_{\varepsilon}(z) - 2\log(1/\varepsilon)}$$

where  $T_{\beta}(z) = \sup\{\varepsilon \leq \varepsilon_0 : 2\Gamma_{\varepsilon}(z) - 2\log(1/\varepsilon) \leq -\beta\}$  and  $\varepsilon_0$  is some fixed starting point such that  $\operatorname{dist}(z, \partial \mathbb{D}) > \varepsilon_0$  for all  $z \in \mathcal{O}$ . It is shown in [Pow17a, Proposition 3.6] that  $\nu_{\varepsilon}^{\beta}(\mathcal{O})$  is uniformly integrable for fixed  $\beta \geq 0$ . Additionally, define

$$C_{\beta} := \{ -\Gamma_{\varepsilon}(z) + 2\log(1/\varepsilon) + \beta > 0 \text{ for all } z \in \mathbb{D}, 0 < \varepsilon \le d(z, \partial \mathbb{D}) \}$$

then  $\mathbb{P}(C_{\beta}) = 1 - o(1)$  thanks to [HRV15, Theorem 6.3].

The strategy is to prove, for  $A_n$  the *n*-FPS of the GFF, that

$$\lim_{\beta \to \infty} \lim_{n \to \infty} \lim_{\varepsilon \to 0} \mathbb{E} \left[ \nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_n} \right] \mathbf{1}_{\{\tau_{\beta} = \infty\}}$$
(4.10)

is equal to both  $\mu'_2(\mathcal{O})$  and  $\frac{1}{2}\tilde{D}_{\infty}(\mathcal{O})$  almost surely.

Let us first show that (4.10) is equal to  $\mu'_2(\mathcal{O})$ . Observe that since  $\nu_{\varepsilon}^{\beta}(\mathcal{O})$  is uniformly integrable, we have by Fatou's and reverse Fatou's lemma that, if the limit in  $\varepsilon$  exists (we will show that is does in the next step)

$$\mathbb{E}\left[\liminf_{\varepsilon \to 0} \nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_{n}}\right] \leq \lim_{\varepsilon \to 0} \mathbb{E}[\nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_{n}}] \leq \mathbb{E}\left[\limsup_{\varepsilon \to 0} \nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_{n}}\right].$$

Taking the limit as  $n, \beta \to \infty$  we obtain that

$$\lim_{\beta \to \infty} \liminf_{\varepsilon \to 0} \nu_{\varepsilon}^{\beta}(\mathcal{O}) \leq \lim_{\beta \to \infty} \lim_{n \to \infty} \lim_{\varepsilon \to 0} \mathbb{E}[\nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_{n}}] \leq \lim_{\beta \to \infty} \limsup_{\varepsilon \to 0} \nu_{\varepsilon}^{\beta}(\mathcal{O}).$$

However, since  $\nu_{\varepsilon}^{\beta}(\mathcal{O}) = \nu_{\varepsilon}(\mathcal{O})$  on the event  $C_{\beta}$ , and almost surely  $\mathbf{1}_{C_{\beta}} \uparrow 1$  as  $\beta \to \infty$ , the right and left hand sides of the above two expressions are equal to  $\mu'_{2}(\mathcal{O})$ . Since also almost surely  $\mathbf{1}_{\{\tau_{\beta}=\infty\}} \to 1$  as  $\beta \to \infty$ , we deduce that (4.10) is equal to  $\mu'_{2}(\mathcal{O})$ .

We now show that (4.10) is equal to  $\frac{1}{2}\tilde{D}_{\infty}(\mathcal{O})$ . Write  $\mathbb{E}[\nu_{\varepsilon}^{\beta}(\mathcal{O}) \mid \mathcal{F}_{A_n}] := E^1(n,\beta,\varepsilon) + E^2(n,\beta,\varepsilon)$  where

$$E^{1}(n,\beta,\varepsilon) := \int_{\mathcal{O} \setminus A_{n}^{\varepsilon}} \mathbb{E}^{A_{n}} \left[ (\Gamma_{\varepsilon}(z) + 2\log(1/\varepsilon)) \mathbf{1}_{\{T_{\beta}(z) \leq \varepsilon\}} e^{2\Gamma_{\varepsilon}(z) - 2\log(1/\varepsilon)} \right] dz;$$
  
$$E^{2}(n,\beta,\varepsilon) := \int_{\mathcal{O} \cap A_{n}^{\varepsilon}} \mathbb{E}^{A_{n}} \left[ (\Gamma_{\varepsilon}(z) + 2\log(1/\varepsilon)) \mathbf{1}_{\{T_{\beta}(z) \leq \varepsilon\}} e^{2\Gamma_{\varepsilon}(z) - 2\log(1/\varepsilon)} \right] dz;$$

and  $\mathbb{E}^{A_n}$  is the regular conditional expectation w.r.t.  $\mathcal{F}_{A_n}$ . Here we used that  $\Gamma_{\varepsilon}(z) = (\Gamma_{A_n}, \rho_z^{\varepsilon}) + \Gamma_{\varepsilon}^{A_n}(z)$ , where conditionally on  $\mathcal{F}_{A_n}$  (i.e. under  $\mathbb{P}^{A_n}$ )  $\Gamma^{A_n}$  is a GFF in  $\mathbb{D}\setminus A_n$ . This implies that  $\lim_{\varepsilon \to 0} E^2(n, \beta, \varepsilon) = 0$  almost surely. To see why, note that  $(\Gamma_{A_n}, \rho_z^{\varepsilon}) \leq n$  and that when  $\varepsilon \geq d(z, A_n)$ , the variance of  $\Gamma_{\varepsilon}^{A_n}(z)$  is uniformly bounded (independently of z and  $\varepsilon$ ). This implies that the integrand is of order  $\varepsilon^2 \log(1/\varepsilon)$  uniformly in z.

To deal with  $E^1$ , observe that if  $\varepsilon \leq d(A_n, z)$  then  $(\Gamma_{A_n}, \rho_z^{\varepsilon}) = n$ . Additionally, due to the Markov property of the GFF and an explicit computation, we have that conditionally on  $\mathcal{F}_{A_n}$ , i.e., under the probability  $\mathbb{P}^{A_n}$ ,

$$\left(-n - \Gamma_{\delta}^{A_n}(z) + 2\log(1/\delta)\right) \mathbf{1}_{\{T_{\beta}(z) \le \delta\}} e^{2n + 2\Gamma_{\delta}^{A_n}(z) - 2\log(1/\delta)}$$

is a (reverse) martingale for  $0 < \delta \leq \delta_n(z) := d(z, \partial \mathbb{D} \setminus A_n)$ . Thus, we have that

 $E^1(n,\beta,\varepsilon)$  is equal to

$$\int_{\mathcal{O}\setminus A_n^{\varepsilon}} \mathbb{E}^{A_n} \left[ (-n - \Gamma_{\delta_n(z)}^{A_n}(z) + 2\log(1/\delta_n(z))) \mathbf{1}_{\{T_\beta(z) \le \delta_n(z)\}} e^{2n + 2\Gamma_{\delta_n(z)}^{A_n}(z) - 2\log(1/\delta_n(z))} \right] dz.$$

Since the integrand does not depend on  $\varepsilon$ , taking the limit in  $\varepsilon$  simply yields the integral over the whole of  $\mathcal{O} \setminus A_n$ .

Now, we rewrite  $\lim_{\varepsilon \to 0} E^1(n, \beta, \varepsilon)$  as a difference between

$$\int_{\mathcal{O}\backslash A_n} \mathbb{E}^{A_n} \left[ \left( -n - \Gamma_{\delta_n(z)}^{A_n}(z) + 2\log(1/\delta_n(z)) \right) e^{2n + 2\Gamma_{\delta_n(z)}^{A_n}(z) - 2\log(1/\delta_n(z))} \right] dz$$

and

$$\int_{\mathcal{O}\backslash A_n} \mathbb{E}^{A_n} \left[ (-n - \Gamma_{\delta_n(z)}^{A_n}(z) + 2\log(1/\delta_n(z))) \mathbf{1}_{\{T_\beta(z) > \delta_n(z)\}} e^{2n + 2\Gamma_{\delta_n(z)}^{A_n}(z) - 2\log(1/\delta_n(z))} \right] dz$$

Notice that the first of these terms is equal to  $D_n(\mathcal{O})/2$ . Let us further rewrite the second term. First, we use Girsanov's theorem. Since  $\Gamma_{\delta_n(z)}^{A_n}$  is a normal random variable with mean 0 and variance  $\log(1/\delta_n(z)) - \log(1/\operatorname{CR}(z, \mathbb{D} \setminus A_n))$ , we see that this term is equal to

$$\int_{\mathcal{O}\setminus A_n} e^{2n-2\log \operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_n)} \tilde{\mathbb{E}}_z^{A_n} \left[ \left( -n - \Gamma_{\delta_n(z)}^{A_n} + 2\log\left(1/\delta_n(z)\right) \right) \mathbf{1}_{\{T_\beta(z) > \delta_n(z)\}} \right] dz$$

where  $\tilde{\mathbb{P}}_{z}^{A_{n}}$  is the measure under which the process  $(\Gamma_{\delta}^{A_{n}}(z))_{\delta}$  has the same covariance structure as under  $\mathbb{P}^{A_{n}}$  but with means shifted by  $0 \leq \operatorname{cov}(\Gamma_{\delta}^{A_{n}}(z), \Gamma_{\delta_{n}(z)}^{A_{n}}(z)) \leq 2$ .

Next, we further decompose this as a sum of  $E^3(n,\beta)$  and  $E^4(n,\beta)$  with:

$$E^{3}(n,\beta) := \int_{\mathcal{O}\backslash A_{n}} e^{2n-2\log\operatorname{CR}^{-1}(z,\mathbb{D}\backslash A_{n})} \tilde{\mathbb{E}}_{z}^{A_{n}} \left[ \left( -\Gamma_{\delta_{n}(z)}^{A_{n}} + 2\log\left(\frac{\operatorname{CR}(z,\mathbb{D}\backslash A_{n})}{\delta_{n}(z)}\right) \right) \mathbf{1}_{\{T_{\beta}(z) > \delta_{n}(z)\}} \right] dz;$$
$$E^{4}(n,\beta) := \int_{\mathcal{O}\backslash A_{n}} e^{2n-2\log\operatorname{CR}^{-1}(z,\mathbb{D}\backslash A_{n})} \left( -n + 2\log\operatorname{CR}^{-1}(z,\mathbb{D}\backslash A_{n}) \right) \tilde{\mathbb{P}}_{z}^{A_{n}} \left( T_{\beta}(z) > \delta_{n}(z) \right) dz.$$

To bound  $E^3$ , we notice that  $\Gamma^n_{\delta_n(z)}(z)$  has bounded variance under  $\tilde{\mathbb{E}}_z^{A_n}$ , and  $\frac{\operatorname{CR}(z,\mathbb{D}\setminus A_n)}{\delta_n(z)}$  is uniformly bounded by the Köebe 1/4 Theorem. This means that the whole expression is less than some constant times  $M_n(\mathcal{O})$ , which we know converges to 0 a.s. as  $n \to \infty$ .

For  $E^4$  first note that the law of  $\Gamma_{\delta}^{A_n}$  under  $\tilde{\mathbb{P}}_z^{A_n}$  is equal to its law under  $\mathbb{P}^{A_n}$  up to a shift that is uniformly bounded by 2. Thus we have that  $\tilde{\mathbb{P}}_z^{A_n}(T_{\beta}(z) > \delta_n(z)) \leq \mathbb{P}^{A_n}(C_{\beta-2}^c)$ . Additionally, on the event  $\{\tau_{\beta} = \infty\}$  we also have for all  $z \in \mathcal{O}$ 

$$-n + 2\log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_n) + \beta \ge 0.$$

This implies that

$$|E^4(n,\beta)|\mathbf{1}_{\{\tau_\beta=\infty\}} \le (|D_n(\mathcal{O})|/2 + \beta M_n(\mathcal{O}))\mathbb{E}[C^c_{\beta-2}|\mathcal{F}_{A_n}]\mathbf{1}_{\{\tau_\beta=\infty\}}.$$

But the limit of the RHS as  $n \to \infty$  is equal to  $\frac{1}{2}\tilde{D}_{\infty}(\mathcal{O})\mathbf{1}_{C_{\beta-2}^{c}}\mathbf{1}_{\{\tau_{\beta}=\infty\}}$ . As this tends to 0 as  $\beta \to \infty$  we conclude.

#### 4.6.3 Seneta-Heyde rescaling.

Finally, we show that one can also perform a so-called Seneta-Heyde rescaling for the construction of the critical Liouville measure using local sets. While this result itself is of interest, one of the other main objectives of this section is a proof to demonstrate how simple it is in this framework to transfer techniques and methods from multiplicative cascades and branching random walks to the study of the Liouville measure. We plan to make further use of this in a follow-up paper. The proof in this section follows very closely that of [AS14], so we only give an outline, point to concrete analogies, and highlight some minor differences. It might be helpful to have the article [AS14] on the side, although we aimed to make the section readable on its own too.

**Proposition 4.18** (Seneta-Heyde Rescaling). For all  $a \ge 0$ , and Borelian  $\mathcal{O} \subset \mathbb{D}$ , we have that  $\sqrt{an}M^2(\mathcal{O}, A_{an}) \to \frac{4}{\sqrt{\pi}}\mu'_2(O)$  in probability as  $n \to \infty$ . In particular the measures  $\sqrt{an}M_n^2$  converge weakly in probability to  $\frac{4}{\sqrt{\pi}}\mu'_2$ .

Again, by Remark 4.9, it suffices to prove the convergence statement for  $\sqrt{an}M^2(\mathcal{O}, A_{an})$ . For simplicity, we work in the case a = 1 and define  $M_n := M^2(O, A_n)$ . Before proving this proposition we need to define carefully a certain family  $\hat{\mathbb{Q}}^{\eta}$  of rooted measures. Recall that if  $(\Gamma, Z)$  has the law  $\hat{\mathbb{P}}^*(d\Gamma, dz)$  defined in (4.7), then the process

$$S_n := -2n + 4 \log \operatorname{CR}^{-1}(Z, \mathbb{D} \setminus A_n)$$

is a random walk with mean-zero increments under the conditional law  $\hat{\mathbb{P}}^*(d\Gamma|Z)$ . Recall also the definition of  $\bar{D}_n^{\eta}$  given in (4.8):

$$\bar{D}_n^{\eta}(\mathcal{O}) := \int_{\mathcal{O}} h_1 \left( -2n + 4 \log \operatorname{CR}^{-1}(z, \mathbb{D} \setminus A_n) + 2\eta \right) \mathbb{1}_{E_{\eta}(n, z)} e^{2n} \operatorname{CR}(z; \mathbb{D} \setminus A_n)^2 dz.$$

We already showed that  $\bar{D}_n^{\eta}$  is a positive martingale with respect to  $(\mathcal{F}_{A_n})_n$  and our initial probability measure  $\mathbb{P}$ . Hence we can define a new probability measure  $\mathbb{Q}^{\eta}$  by setting it, when restricted to  $\mathcal{F}_{A_n}$ , to have Radon-Nikodym derivative  $\bar{D}_n^{\eta}(\mathcal{O})/\bar{D}_0^{\eta}(\mathcal{O})$  with respect to  $\mathbb{P}$ .

Again we extend this to a *rooted* measure on the field  $\Gamma$  plus a distinguished point Z by setting  $\hat{\mathbb{Q}}^{\eta}(d\Gamma, dz)$  restricted to  $\mathcal{F}_{A_n}^* = \mathcal{F}_{A_n} \vee \sigma(Z)$  to be

$$h_1(-2n+4\log \operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_n)+2\eta)\,\mathrm{e}^{2n-2\log \operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_n)}\,\mathbf{1}_{E_\eta(z,n)}\frac{\mathbf{1}_{\mathcal{O}}(z)}{\bar{D}_0^\eta}\,dz\,\mathbb{P}[d\Gamma].$$

We make the following observations:

- 1. The marginal law of Z under  $\hat{\mathbb{Q}}^{\eta}$  is proportional to  $h_1(4 \log \operatorname{CR}^{-1}(z, \mathbb{D}) + 2\eta) \operatorname{CR}(z, \mathbb{D})^2 \mathbf{1}_{\mathcal{O}}(z) dz$ .
- 2. The marginal law of the field  $\Gamma$  under  $\hat{\mathbb{Q}}^{\eta}$  is given by  $\mathbb{Q}^{\eta}$ .
- 3. Write  $\hat{\mathbb{Q}}_z^{\eta} = \hat{\mathbb{Q}}^{\eta}[\cdot | Z = z]$  for the law of  $\Gamma$  given the point Z = z. The law of the sequence  $(A_n)_n$  under this measure can be described as follows. First sample  $A_1$  with law weighted by

$$\frac{h_1(-2+4\log \operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_1)+2\eta)}{h_1(\eta+4\log \operatorname{CR}^{-1}(z,\mathbb{D}))}\mathbf{1}_{E_\eta(z,1)}\,\mathrm{e}^{2-2\log \operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_1)}\,.$$

Then given  $A_k$  for any  $k \ge 1$ , construct an independent copy of  $(A_n)_n$  inside each component of  $\mathbb{D} \setminus A_k$  that does *not* contain the point z. Inside the component containing z, let us call this  $\mathcal{B}_k$ , construct the components of  $A_{k+1} \cap \mathcal{B}_k$ by weighting their laws by

$$\frac{h_1(-2(k+1)-4\log\operatorname{CR}(z,\mathbb{D}\setminus A_{k+1})+2\eta)}{h_1(-2k-4\log\operatorname{CR}(z,\mathbb{D}\setminus A_k)+2\eta)}\mathbf{1}_{E_\eta(z,k+1)}e^{2+2\log\operatorname{CR}(z,\mathbb{D}\setminus A_{k+1})-2\log\operatorname{CR}(z,\mathbb{D}\setminus A_k)}$$

This defines the law of the sets  $A_n$ , and hence also by iteration the law of  $\Gamma$ .

It follows directly from the above construction that the law of  $S_n = -2n + 4 \log \operatorname{CR}(z, \mathbb{D} \setminus A_n)$  under  $\hat{\mathbb{Q}}_z^{\eta}$  has the same as its law under  $\hat{\mathbb{P}}^*[\cdot | Z = z]$ , but conditioned to stay above  $-2\eta$ .

Now, we note some useful properties of the renewal function see for example [AS14, Section 2]):.

- First, recall from Claim 4.16 that  $h_1(u) \ge Ru$  for all  $u \ge 0$  for some positive R.
- By the renewal theorem,  $c_0 := \lim_{u \to \infty} \frac{h_1(u)}{u}$  exists and lies in  $(0, \infty)$ .
- Let  $\theta = 2/(\sqrt{\pi}c_0)$  (in the case of  $A_{an}$ ,  $\theta = 2/(\sqrt{\pi}ac_0)$ ). Then

$$\hat{\mathbb{P}}^* \left[ \min_{1 \le i \le n} S_i \ge -u \mid z \right] \sim \frac{\theta h_1(S_0 + u)}{\sqrt{n}}$$
(4.11)

as  $n \to \infty$ , for any  $u \ge 0$ . Moreover, the above holds uniformly in  $u \in [0, b_n]$ 

for any sequence  $b_n \in \mathbb{R}_+$  such that  $\lim_{n\to\infty} b_n/\sqrt{n} = 0$ .

Now set

$$\bar{M}_n^{\eta}(\mathcal{O}) := \int_{\mathcal{O}} e^{2n} \operatorname{CR}(x; \mathbb{D} \setminus A_n)^2 \mathbf{1}_{E_{\eta}(x,n)} \, dx.$$

Using the fact that  $\bar{D}_n^{\eta}$  converges a.s. to a positive measure and that  $\bar{D}_n^{\eta} = \bar{D}_n$ for  $\eta$  large enough, one can show, following exactly [AS14, Proof of Theorem 1.1], that in order to prove the Proposition 4.18 it suffices to establish the next claim:

**Claim 4.19.** For any  $\eta > 0$ ,  $\sqrt{n} \frac{\overline{M}_n^{\eta}}{\overline{D}_n^{\eta}} \to \theta$  in  $\mathbb{Q}^{\eta}$  probability as  $n \to \infty$ .

*Proof.* The overall strategy follows very closely [AS14] and is to control the first and second moments of  $\frac{\bar{M}_n^{\eta}}{\bar{D}_n^{\eta}}$  as  $n \to \infty$ :

$$\hat{\mathbb{Q}}^{\eta} \left[ \frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}} \right] = \frac{\theta}{\sqrt{n}} + o\left(\frac{1}{\sqrt{n}}\right) \text{ and } \hat{\mathbb{Q}}^{\eta} \left[ \left(\frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}}\right)^{2} \right] \le \frac{\theta^{2}}{n} + o\left(\frac{1}{n}\right)$$
(4.12)

These estimates prove the result by Jensen and Chebyshev's inequalities. The key observation lies in rewriting the moments using the rooted measure. Indeed, we can write for  $(\Gamma, Z)$  distributed under  $\hat{\mathbb{Q}}^{\eta}$ 

$$\frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}} = \hat{\mathbb{Q}}^{\eta} \left[ \frac{1}{h_{1}(-2n+4\log\operatorname{CR}(Z,\mathbb{D}\setminus A_{n})+2\eta)} \mid \mathcal{F}_{A_{n}} \right]$$
(4.13)

and thus we have  $\hat{\mathbb{Q}}^{\eta}[\bar{M}_{n}^{\eta}/\bar{D}_{n}^{\eta}] = \int_{z} \hat{\mathbb{Q}}_{z}^{\eta} [1/h_{1}(-2n+4\log \operatorname{CR}(z,\mathbb{D}\setminus A_{n}))] d\hat{\mathbb{Q}}^{\eta}[dz].$ The first moment estimate (4.12) then follows easily using estimates on the renewal function, as in [AS14, proof of Proposition 4.1, Equation (4.1)].

We now move on to the second moment claim. Using random walk estimates and Jensen inequality, exactly as in [AS14, Lemmas 4.3-4.4], and (4.13), one can see that

$$\hat{\mathbb{Q}}_{z}^{\eta}\left[\left(\frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}}\right)^{2}\right] = \hat{\mathbb{Q}}_{z}^{\eta}\left[\frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}}\frac{1}{h_{1}(-2n+4\log\operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_{n})+2\eta)}\right] = O\left(\frac{1}{n}\right).$$
(4.14)

Thus, to prove the second moment bound, it suffices to find a sequence of events  $E_n$  with  $\hat{\mathbb{Q}}^{\eta}(E_n) \to 1$  as  $n \to \infty$  so that

$$\hat{\mathbb{Q}}_{z}^{\eta} \left[ \frac{\bar{M}_{n}^{\eta}}{\bar{D}_{n}^{\eta}} \frac{\mathbf{1}_{E_{n}}}{h_{1}(-2n+4\log\operatorname{CR}^{-1}(z,\mathbb{D}\setminus A_{n})+2\eta)} \right] \leq \frac{\theta^{2}}{n} + o\left(\frac{1}{n}\right)$$
(4.15)

holds uniformly in z. To do this, we pick a sequence  $k_n \to \infty$  such that  $k_n/\sqrt{n} \to 0$ and  $k_n/(\log n)^6 \to \infty$  as  $n \to \infty$  (the reason for this choice will become clear later). We then decompose  $\bar{M}_n^{\eta}$  and  $\bar{D}_n^{\eta}$  by writing

$$\bar{M}_n^{\eta} = \bar{M}_n^{\eta,[0,k_n]} + \bar{M}_n^{\eta,[k_n,n]}; \quad \bar{D}_n^{\eta} = \bar{D}_n^{\eta,[k_n,n]} + \bar{D}_n^{\eta,[0,k_n]}$$

where the superscript  $[0, k_n]$  refers to the integral over  $\mathcal{B}_{k_n}$  and the superscript  $[k_n, n]$ refers to the integral over  $\mathcal{O} \setminus \mathcal{B}_{k_n}$ , where  $\mathcal{B}_{k_n}$  is the connected component of  $D \setminus A_n$ containing z.

We now define our sequence of events  $E_n$  by setting  $E_n = E_n^1 \cap E_n^2$ , where

$$E_n^1 := \{ \bar{D}_n^{\eta, [k_n, n]} \le 1/n^2 \}; \ E_n^2 = \{ S_{k_n} \in [k_n^{1/3}, k_n] \}.$$

Since under  $\hat{\mathbb{Q}}_z^{\eta}$ ,  $S_n$  is a centered random walk conditioned to stay above  $-2\eta$ , it is clear at least that  $\hat{\mathbb{Q}}^{\eta} [E_n^2] \to 1$  as  $n \to \infty$ . Putting aside the issue of whether or not  $\hat{\mathbb{Q}}^{\eta} [E_n^1] \to 1$  for the moment, the next step is to bound (4.15) above by

$$\hat{\mathbb{Q}}_{z}^{\eta} \left[ \frac{\bar{M}_{n}^{\eta,[k_{n},n]}}{\bar{D}_{n}^{\eta}} \frac{\mathbf{1}_{E_{n}^{1}}}{h_{1}(-2n+4\log\operatorname{CR}^{-1}(z,\mathcal{B}_{n})+2\eta)} \right] + \hat{\mathbb{Q}}_{z}^{\eta} \left[ \frac{\bar{M}_{n}^{\eta,[0,k_{n}]}}{\bar{D}_{n}^{\eta,[0,k_{n}]}} \frac{\mathbf{1}_{E_{n}^{2}}}{h_{1}(-2n+4\log\operatorname{CR}^{-1}(z,\mathcal{B}_{n})+2\eta)} \right]$$

Then, using that  $R\bar{M}_n^{\eta,[k_n,n]} \leq \bar{D}_n^{\eta,[k_n,n]} \leq 1/n^2$ , as in [AS14, Proof of Lemma 4.5], it can be deduced that the first term is o(1/n). For the second term, we use that the two products in the expectation are conditionally independent given  $\mathcal{F}_{A_{k_n}}^*$ . We then have, by (4.11) and the assumption that  $k_n/\sqrt{n} \to 0$ , that

$$\mathbf{1}_{E_{n}^{2}}\hat{\mathbb{Q}}_{z}^{\eta}\left[h_{1}^{-1}(-2n+4\log\operatorname{CR}^{-1}(z,\mathcal{B}_{n})+2\eta) \mid \mathcal{F}_{A_{k_{n}}}^{*}\right] = \theta/\sqrt{n} + o(1/\sqrt{n})$$

uniformly in  $\omega$  and z. Since  $E_n^2$  is  $\mathcal{F}_{A_{k_n}}^*$  measurable, it therefore remains to prove that

$$\hat{\mathbb{Q}}_{z}^{\eta} \left[ (\bar{M}_{n}^{\eta, [0, k_{n}]} / \bar{D}_{n}^{\eta, [0, k_{n}]}) \mathbf{1}_{E_{n}^{2}} \right] \leq \theta / \sqrt{n} + o(1/\sqrt{n}).$$
(4.16)

This is a consequence of our first moment estimate and the fact that  $\bar{M}_n^{\eta,[0,k_n]}/\bar{D}_n^{\eta,[0,k_n]}$  is comparable to  $\bar{M}_n^{\eta}/\bar{D}_n^{\eta}$  on the event  $E_n^1$ . The details of this claim are exactly as in [AS14, Lemma 4.5].

Thus, to finish the prove the proposition, it remains to establish that  $\hat{\mathbb{Q}}^{\eta} [E_n^1] \rightarrow 1$  as  $n \rightarrow \infty$ . In fact we need to prove a stronger slightly stronger statement to also deduce (4.16) as described above:

**Lemma 4.20.** Suppose that  $k_n/\sqrt{n} \to 0$  and  $k_n/(\log n)^6 \to \infty$  as  $n \to \infty$ . Then there exists a deterministic sequence  $p_n \nearrow 1$  such that  $\mathbf{1}_{E_n^2} \hat{\mathbb{Q}}_z^{\eta} \left[ E_n^1 \mid \mathcal{F}_{A_{k_n}}^* \right] \ge p_n$ .

It is only in the proof of this lemma that we need to do a bit of extra work. The extra work comes from the fact that, unlike in the case of multiplicative cascades, in our setting the sets at the n-th level have different shapes and sizes.

Proof of Lemma 4.20. Define further events  $E_n^3$  and  $E_n^4$  by setting

$$E_n^3 = \bigcap_{k_n \le j \le n} \{ S_j \ge k_n^{1/6} \}; \ E_n^4 = \bigcap_{k_n \le j \le n} \{ \sup_{w \in \mathcal{B}_j} |z - w| \le j^c \operatorname{CR}(z, \mathcal{B}_j) \}$$

where c is some fixed constant to be chosen just below (as in [Aid15, Lemma 3.5]). We argue that:

- (i)  $\mathbf{1}_{E_n^2} \hat{\mathbb{Q}}_z^{\eta} \left[ E_n^3 \mid \mathcal{F}_{A_{k_n}}^* \right] \ge p_n$ , where  $p_n \to 1$  is deterministic;
- (ii)  $\mathbf{1}_{E_n^2} \hat{\mathbb{Q}}_z^{\eta} \left[ E_n^4 \mid \mathcal{F}_{A_{k_n}}^* \right] \ge q_n$ , where  $q_n \to 1$  is deterministic; and finally
- (iii)  $\hat{\mathbb{Q}}_z^{\eta}[\bar{D}_n^{\eta,[k_n,n]}\mathbf{1}_{E_n^3\cap E_n^4} \mid \mathcal{F}_{A_{k_n}}^*] \leq r_n$  where  $r_n = o(1/n^2)$  is deterministic.

This proves the lemma by conditional Markov's inequality. For (i), one uses the fact that under the given conditional law,  $(S_j - S_{k_n}; j \ge S_{k_n})$  is a centered random walk conditioned to stay above  $-S_{k_n} + 2\eta$ . The details rely on estimates for the renewal function, and are as in [AS14, Proof of Lemma 4.7]. Claim (ii) follows from the proof of [Aid15, Lemma 3.5]. This proof shows that, uniformly in z,

$$\hat{\mathbb{Q}}_{z}^{\eta}\left[\sup_{w\in\mathcal{B}_{j}}|z-w|>j^{c}\operatorname{CR}(z,\mathcal{B}_{j})\mid\mathcal{F}_{A_{k_{n}}}^{*}\right]\leq c'\sqrt{j-k_{n}}j^{-c'c},$$

for some positive constant c' that does not depend on c (note the right-hand side is deterministic.) Choosing c large enough gives (ii). For (iii), we condition further on all the *brother loops* of the point z (that is, for each  $k_n \leq j \leq n-1$ , the components of  $D \setminus A_j$  contained in  $\mathcal{B}_j$  but not  $\mathcal{B}_{j+1}$ ). After applying this conditioning, and using the description of  $(A_n)_n$  given after the statement of Proposition 4.18, we see that  $\hat{\mathbb{Q}}_z^{\eta} \left[ \bar{D}_n^{\eta,[k_n,n]} \mathbf{1}_{E_n^3 \cap E_n^4} \mid \mathcal{F}_{A_{k_n}}^* \right]$  is equal to

$$\sum_{j=k_n}^{n-1} s_j + \hat{\mathbb{Q}}_z^{\eta} \left[ \int_{\mathcal{B}_n} h_1(-2n + 4\log \operatorname{CR}^{-1}(w, \mathcal{B}_n) + 2\eta) \, \mathrm{e}^{2n-2\log \operatorname{CR}^{-1}(w, \mathcal{B}_n)} \, \mathbf{1}_{E_n^3 \cap E_n^4} \, dw \mid \mathcal{F}_{A_{k_n}}^* \right]$$

where  $s_i$  is

$$\hat{\mathbb{Q}}_{z}^{\eta} \left[ \mathbf{1}_{E_{n}^{3} \cap E_{n}^{4}} \int_{\mathcal{B}_{j} \setminus \mathcal{B}_{j+1}} h_{1}(4 \log \operatorname{CR}^{-1}(w, \mathbb{D} \setminus A_{j+1}) + 2\eta - 2(j+1)) \operatorname{e}^{2(j+1) - 2 \log \operatorname{CR}^{-1}(w, \mathbb{D} \setminus A_{j+1})} dw \mid \mathcal{F}_{A_{k_{n}}}^{*} \right]$$

$$\leq C \hat{\mathbb{Q}}_{z}^{\eta} \left[ \mathbf{1}_{E_{n}^{3} \cap E_{n}^{4}} \int_{\mathcal{B}_{j} \setminus \mathcal{B}_{j+1}} h_{1}(4 \log \operatorname{CR}^{-1}(w, \mathcal{B}_{j}) + 2\eta - 2j) \operatorname{e}^{2j - 2 \log \operatorname{CR}^{-1}(w, \mathcal{B}_{j})} dw \mid \mathcal{F}_{A_{k_{n}}}^{*} \right],$$

for some constant  $C \in \mathbb{R}$ . The inequality here follows because  $\operatorname{CR}(w, \mathbb{D} \setminus A_j)$  is decreasing in j and  $h_1$  is bounded on either side by a linear function. Now, note that on the event  $E_n^3 \cap E_n^4$ , thanks to Koebe's theorem,  $2j - 2\log CR^{-1}(w, \mathcal{B}_j)$  is smaller than  $S_j + 2c \log(j) + 2 \log \operatorname{CR}^{-1}(z, \mathcal{B}_j)$ , and the area of each  $B_j$  is also  $O(\operatorname{CR}(z, \mathcal{B}_j)^2)$ . This means that each  $s_j$  is  $O(\exp(-\sqrt[6]{k_n}/2)n^{4c+1})$ , and the assumption that  $k_n/(\log n)^6 \to \infty$  therefore implies (iii).  $\Box$ 

#### Acknowledgements.

We would like to thank R. Rhodes and V. Vargas for elucidating the existing literature in the critical case, N. Berestycki for advice on connecting our measure with the existing Liouville measure in this case, and W. Werner for advice on the construction of the Liouville boundary measure. We are grateful to W. Werner for inviting E. Powell to visit ETH on two occasions, where a large part of this work was carried out, and for presenting us the Neumann-Dirichlet set-up. J. Aru and A. Sepúlveda are supported by the SNF grant #155922, and are happy to be part of the NCCR Swissmap. E. Powell is supported by a Cambridge Centre for Analysis EPSRC grant EP/H023348/1.

# 5 Level Lines of the Gaussian free field with general boundary data

## 5.1 Introduction

The relationship between Schramm-Loewner Evolution (SLE) and the twodimensional Gaussian free field (GFF) is at the heart of recent breakthroughs in Liouville quantum gravity, imaginary geometry and more generally, random conformal geometry. Starting with the seminal papers of [Dub09], [SS13], [SS09], one key idea is to make sense of SLE-type curves as a level lines of an underlying Gaussian free field h in a domain, which we take to be the upper half plane  $\mathbb{H}$  without loss of generality in the rest of the paper. When the field h is given the boundary values  $\lambda := \pi/2$  on  $\mathbb{R}_+$  and  $-\lambda$  on  $\mathbb{R}_-$ , the corresponding level line is a chordal SLE<sub>4</sub> curve. A considerable extension of that theory is described in [MS16a], which introduced the notion of flow lines and counter flow lines of the GFF. In this case it turns out that the curves are given by SLE<sub> $\kappa$ </sub> processes with  $\kappa \in (0, 4)$  and  $\kappa \in (4, \infty)$ respectively.

It is also natural to wonder for which sort of boundary data the notion of level line makes sense. In [MS16a] and [WW16], the hypothesis on the boundary data is extended from the above to any arbitrary piecewise constant function on the real line. The goal of this paper will be to relax this assumption. Assuming solely that the boundary data F is a *regulated* function, i.e., the left and right limits

$$F(t^{+}) = \lim_{h \to 0^{+}} F(t+h); \quad F(t^{-}) = \lim_{h \to 0^{-}} F(t+h)$$
(5.1)

exist and are finite for all  $t \in \overline{\mathbb{R}}$ , and that for some c > 0

$$F(x) \le \lambda - c, \quad x < 0; \quad F(x) \ge -\lambda + c, \quad x \ge 0$$
(5.2)

which roughly corresponds to the non existence of a continuation threshold, we can show that the corresponding level line is well defined almost surely as a continuous transient curve. Moreover, it is almost surely determined by the field.

This also allows us, for a zero boundary GFF h, to consider the set of level lines of different heights. By this we mean the level lines of h+F, where F ranges over (the bounded harmonic extensions of) all regulated functions on  $\mathbb{R}$ . Strengthening the results of [MS16a], [WW16], we are able to prove a general monotonicity principle for the level lines, which is both a key tool in our existence proof, and an interesting result in its own right. This is deeply intertwined with the reversibility property of the level lines, which we are also able to prove in general; see Theorems 5.4 and 5.5.

A further point of interest is that we obtain some continuity in the level lines as a consequence of our proof. That is, if we take a sequence of piecewise constant functions  $F_n$  converging monotonically uniformly to some F, then the level lines of height  $F_n$  for a zero boundary GFF converge almost surely to the level line of height F. This convergence is with respect to Hausdorff distance, after conformally mapping everything to the unit disc.

We remark that our hypothesis on the boundary data is satisfied by a wide range of functions, including the special class of *functions of bounded variation*. Any such function can be described almost everywhere as the integral of a finite Radon measure  $\rho$ , and this connection allows us to deduce that the marginal law of a level line with such boundary data is given by what we call an SLE<sub>4</sub>( $\rho$ ) process. This is the natural analogue of an SLE<sub>4</sub>( $\rho$ ) process, where the vector  $\rho$  is replaced by a measure. Our results therefore demonstrate the existence of such processes, as well as establishing some further properties.

We first recall the definition of what it means for a curve, and more generally a Loewner chain, to be a level line. If we have a Loewner chain  $(K_t, t \ge 0)$  in  $\mathbb{H}$ , with associated sequence of conformal maps  $g_t : \mathbb{H} \setminus K_t \to \mathbb{H}$ , we will often want to describe the *image under*  $g_t$  of a point x on the real line. To do this, for any  $x \le 0$ we define a process  $V_t^L(x)$  by setting it equal to  $g_t(x)$  if  $x \notin K_t$  and if  $x \in K_t$ , taking it to be the image of the leftmost point of  $\mathbb{R} \cap K_t$  under  $g_t$ . We define a process  $V_t^R(x)$  for  $x \ge 0$  analogously. The process  $V_t^L(x)$  for  $x \in \mathbb{R}_-$ , or  $V_t^R(x)$  for  $x \in \mathbb{R}_+$ , is what we define to be the *image of* x under  $g_t$ .

**Definition 5.1** ([MS16a, WW16]). Suppose that F is  $L^1$  with respect to harmonic measure on  $\mathbb{R}$  viewed from some point in  $\mathbb{H}$  and that h is a zero boundary GFF in  $\mathbb{H}$ . If  $(K_t, t \ge 0)$  is a Loewner chain and  $(g_t, t \ge 0)$  is the corresponding sequence of conformal maps, set  $f_t = g_t - W_t$ , and let  $V_t^R(x)$  (resp.  $V_t^L(x)$ ) be the image of  $x \ge 0$  (resp.  $x \le 0$ ) under  $g_t$ . Let  $\eta_t^0$  be the bounded harmonic function on  $\mathbb{H}$  with boundary values (see Figure 23)

$$\begin{cases} F(f_t^{-1}(x)), & \text{if } x \ge V_t^R(0^+) - W_t, \\ \lambda, & \text{if } 0 \le x < V_t^R(0^+) - W_t, \\ -\lambda, & \text{if } V_t^L(0^-) - W_t \le x < 0, \\ F(f_t^{-1}(x)), & \text{if } x < V_t^L(0^-) - W_t. \end{cases}$$

Define, for  $z \in \mathbb{H} \setminus K_t$ ,

$$\eta_t(z) = \eta_t^0(f_t(z)).$$

We say that K is a level line of h + F if there exists a coupling (h, K) such that the following domain Markov property holds: for any finite K-stopping time  $\tau$ , given  $K_{\tau}$ , the conditional law of  $(h + F)|_{\mathbb{H} \setminus K_{\tau}}$  is equal to the law of  $h \circ f_{\tau} + \eta_{\tau}$ .



Figure 23: The left hand side shows the boundary values of the harmonic function  $\eta_{\tau}$  in  $\mathbb{H} \setminus K_{\tau}$ . This is the image under  $f_{\tau}^{-1}$  of the harmonic function  $\eta_{\tau}^{0}$  in  $\mathbb{H}$ , whose boundary values are shown on the right hand side.

Note that this definition is the same for any two functions  $F_1$  and  $F_2$  which are equal almost everywhere, since the harmonic extensions of such functions are necessarily equal. From Definition 5.1, we can see that the so-called level lines of the GFF have an intriguing property that distinguishes them from level lines of an ordinary smooth function. Namely, once one conditions on a level line, the conditional expectation of the field on one side of the curve differs by  $2\lambda$  from the value on the other side. In a sense, a level line is more like a "level cliff" where there is a prescribed jump between the two sides of the curve.

More generally, we say that a Loewner chain  $(K_t, t \ge 0)$  is a level line of a GFF h in a domain D from  $a \in \partial D$  to  $b \in \partial D$  if  $(\varphi(K_t), t \ge 0)$  is a level line of  $\varphi(h)$  as in Definition 5.1, where  $\varphi$  is a conformal map from  $D \to \mathbb{H}$  sending a to 0 and b to  $\infty$ .

**Theorem 5.2.** [Coupling] Assume the same notations as in Definition 5.1. Suppose that the function F is regulated and satisfies (5.2) for some c > 0. Then there exists a coupling satisfying the conditions in Definition 5.1. Moreover, in this coupling, the Loewner chain K is almost surely generated by a continuous and transient curve  $\gamma$ with almost surely continuous driving function.

The inequality on F in Theorem 5.2 guarantees that the corresponding level line will reach its target point  $\infty$  before "dying" at some continuation threshold. Indeed, the level line of a GFF with piecewise constant boundary data is only defined until the first time that it hits a section of  $\mathbb{R}_+$  where the boundary data is less than  $-\lambda$  or a section of  $\mathbb{R}_-$  where it is greater than  $\lambda$ . In our case, if we allowed F to approach  $-\lambda$  (resp.  $\lambda$ ) at some point in  $\mathbb{R}_+$  (resp. in  $\mathbb{R}_-$ ), then our current framework would not control the behaviour of the level line around this point (see discussion below.) Thus, we do not treat this situation here.

**Theorem 5.3.** [Determination] If  $(h, \gamma)$  are coupled as in Theorem 5.2, then  $\gamma$  is almost surely determined by h. Moreover, the curve  $\gamma$  is almost surely simple. We call  $\gamma$  the level line of h + F.

With this in hand, we can consider the collection of level lines determined by a given field. The following two theorems describe the interactions between the curves; corresponding to what one might expect from the level lines of a smooth function.

**Theorem 5.4.** [Monotonicity] Suppose that F, G are functions satisfying the conditions in Theorem 5.2, and that  $F(x) \ge G(x)$  for  $x \in \mathbb{R}$ . Suppose that h is a zero boundary GFF on  $\mathbb{H}$  and  $\gamma_F$  (resp.  $\gamma_G$ ) is the level line of h + F (resp. h + G). Then  $\gamma_F$  lies to the left of  $\gamma_G$  almost surely.

**Theorem 5.5.** [Reversibility] Suppose that h is a GFF on  $\mathbb{H}$  whose boundary value satisfies the conditions in Theorem 5.2. Let  $\gamma$  be the level line of h from 0 to  $\infty$  and  $\gamma'$  be the level line of -h from  $\infty$  to 0. Then the two paths  $\gamma$  and  $\gamma'$  (viewed as sets) are equal almost surely.

Now we will explain the relevance of Conditions (5.1) and (5.2), which we need for our approach to work. Although one can make sense of what it means to be a level line of h + F for any F in  $L^1$  (as in Definition 5.1), before this work the existence of the coupling was only known for piecewise constant boundary data. The assumption that the boundary data F is regulated corresponds precisely to the fact that F can be uniformly approximated by piecewise constant functions. Indeed, our argument will use an approximation of F by such functions, and a limit of the corresponding level lines. Thus with our current approach we are unable to say anything about functions which are not regulated. However, since Definition 5.1 still makes sense for a wider class of functions, it is an interesting question to determine the most general restrictions under which a coupling exists. For example, if one takes a GFF with boundary data which is  $-\lambda$  in a neighbourhood to the left of 0 and  $\lambda$  in a neighbourhood to the right of 0 then one can allow much rougher boundary data away from these neighbourhoods (for example, even Neumann boundary conditions, see [KI13]), and construct a weaker form of "local coupling" with an SLE variant. Whether these types of coupling can be extended to a strong coupling as in Definition 5.1, where the curve is also determined by the field, or whether the condition near 0 can be relaxed is currently unknown.

Concerning Condition (5.2); the key to the proof of Theorem 5.2 is the continuity and transience of the approximating level lines (with piecewise constant boundary data). This allows us to use the results of [KS16] (see details in Section 5.2.2) to obtain a continuous limiting curve. If Condition (5.2) failed, the approximating level lines would only be defined up to a continuation threshold, and we would not be able to obtain such a limit. The continuity of the limiting curve is absolutely crucial to the proofs of Theorems 5.3 to 5.5. In fact, if the existence and the continuity of level lines were obtained for other boundary data, one could use similar proofs to get the corresponding theorems. However, whether continuity still holds in this set up is also a difficult open problem. Although it is natural to conjecture that for general regulated boundary data the level line will exist as a continuous curve until hitting a point on the boundary where Condition (5.2) fails, a "continuation threshold" as in [MS16a],[WW16], it is unclear whether or not the continuity will break down around this point.

Finally, we identify the law of the level lines. It is proved in [MS16a, WW16] that the level lines of GFF with piecewise constant boundary data are  $\text{SLE}_4(\underline{\rho})$  processes where  $\underline{\rho}$  is a vector. In our context, when the boundary data is of bounded variation, the level lines turn out to be  $\text{SLE}_4(\rho)$  processes, where  $\rho$  is now a Radon measure. With the help of the GFF, we are able to obtain the existence, the continuity, and the reversibility of such processes, properties which are far from clear by the definition of the process through Loewner evolution.

**Theorem 5.6.** Assume the same notations as in Theorem 5.2. Suppose further that F is of bounded variation. Then in the coupling  $(h, \gamma)$  given by Theorem 5.2, the marginal law of  $\gamma$  is that of an  $\text{SLE}_4(\rho^L; \rho^R)$  process (see Section 5.2.5) where  $\rho^R$  (resp.  $\rho^L$ ) is a finite Radon measure on  $\mathbb{R}_+$  (resp. on  $\mathbb{R}_-$ ) and

$$\begin{cases} F(x) = \lambda(1 + \rho^{R}([0, x])), & x \ge 0; \\ F(x) = -\lambda(1 + \rho^{L}((x, 0])), & x < 0 \end{cases} \end{cases}$$

almost everywhere. In particular, we have the following properties of the  $SLE_4(\rho^L; \rho^R)$  process. Suppose that there exists c > 0 such that

$$\rho^{L}((x,0]) \ge -2 + c, \quad x < 0, \quad \rho^{R}([0,x]) \ge -2 + c, \quad x > 0.$$

Then

- (1) There exists a law on continuous curves from 0 to  $\infty$  in  $\overline{\mathbb{H}}$  with almost surely continuous driving functions, for which the associated Loewner chain is an  $\operatorname{arSLE}_4(\rho^L; \rho^R)$  process.
- (2) The above continuous curve is almost surely simple and transient.

(3) The time reversal of the above  $SLE_4(\rho^L; \rho^R)$  process has the same law as  $SLE_4(\tilde{\rho}^L; \tilde{\rho}^R)$ , where

$$\tilde{\rho}^R([x,\infty]) = \rho^L((x,0]), \quad x < 0; \quad \tilde{\rho}^L((x,\infty]) = \rho^R([0,x]), \quad x > 0.$$

**Remark 5.7.** Although Theorem 5.6 gives us existence of  $\text{SLE}_4(\rho^L; \rho^R)$  processes, we do not derive uniqueness in law. That is, we have not excluded the possibility that there exists another law on Loewner chains satisfying the definition of an  $\text{SLE}_4(\rho^L; \rho^R)$  process.

**Remark 5.8.** Item (3) is the so-called reversibility of SLE. The reversibility was derived previously for  $SLE_{\kappa}$  in [Zha08b], for  $SLE_{\kappa}(\rho)$  where  $\rho$  is a vector in [Zha08a, MS16b, MS16c, WW13]. In Theorem 5.6, we derive the reversibility of  $SLE_4(\rho)$  where  $\rho$  is a Radon measure.

**Outline.** The structure of the paper is as follows. In Section 5.2, we discuss briefly the necessary background theory, and collect some results that will be important to us. We also define the class of  $SLE_{\kappa}(\rho)$  process and generalize some of the theory from [MS16a, WW16] which will help us in the sequel. In Sections 5.3 and 5.4, we set up a general framework for the level lines of a GFF, under the assumption that they exist and are given by continuous transient curves. In particular, we show that they are monotonic in the boundary data, and describe where they can and cannot hit the boundary. Sections 5.5 and 5.6 address the existence of continuous transient curves which can be coupled as level lines of a GFF, provided the boundary data satisfies the conditions of Theorem 5.2. The proof of this is via an approximation argument; using a general theory for the weak convergence of curves, as set out in [KS16]. The key point in the proof is the monotonicity obtained in Section 5.4. In Section 5.7 we prove Theorems 5.3 to 5.5 using the ideas from Sections 5.3 and 5.4. Finally, we complete the proof of Theorem 5.6 in Section 5.8.

Acknowledgments. We thank Nathanaël Berestycki, Jason Miller, Steffen Rohde, and Scott Sheffield for helpful discussions. We thank Avelio Seplveda and Juhan Aru for precious comments on the previous version of this paper. The main part of this work was done while H. Wu was at MIT and H. Wu's work is funded by NSF DMS-1406411. E. Powell's is funded by a Cambridge Centre for Analysis EPSRC studentship.

## 5.2 Preliminaries

#### 5.2.1 Regulated functions and functions of bounded variation

We say that a function F on  $\mathbb{R}$  is *regulated* if it admits finite left and right limits

$$F(t^+) = \lim_{h \to 0^+} F(t+h); \quad F(t^-) = \lim_{h \to 0^-} F(t+h)$$

at every point  $t \in \mathbb{R}$ , including  $\infty$ . Equivalently, see [Die69, Section 7.6], F is regulated if it can be uniformly approximated on  $\mathbb{R}$  by piecewise constant functions which change value only finitely many times. It is this formulation of the definition that will be useful to us in the sequel.

Another type of function which is of particular interest in the current paper is the class of functions of *bounded variation*. Let us consider the connection (5.5) between pairs of Radon measures ( $\rho^L; \rho^R$ ) and functions F on the real line. We saw above that piecewise constant functions correspond to purely atomic measures. In general, finite Radon measures are in one-to-one correspondence with right-continuous functions of bounded variation.

The space of functions of bounded variation are those F which satisfy

$$\sup_{a < b} \left( \sup \left\{ \sum_{i} |F(x_i) - F(x_{i-1})| : \{x_i\} \text{ a finite partition of } [a, b] \right\} \right) < \infty.$$

For a proof of this equivalence, see [Fol99, Theorem 3.29]. Note that these functions are clearly regulated. So, provided they satisfy the correct bounds on  $\mathbb{R}_{-}$  and  $\mathbb{R}_{+}$ , functions of bounded variation meet the conditions of Theorem 5.2.

Furthermore, if a bounded variation function is also absolutely continuous, then the corresponding measures  $(\rho^L; \rho^R)$  are absolutely continuous with respect to Lebesgue measure, and writing

$$\rho^L(dx) = f^L(x) \, dx, \quad \rho^R(dx) = f^R(x) \, dx$$

we have that the function is differentiable almost everywhere with derivative equal to  $f^{L}(x)$  on  $\mathbb{R}_{-}$  and  $f^{R}(x)$  on  $\mathbb{R}_{+}$ .

#### 5.2.2 A result on the convergence of curves

To show existence of the level line of a GFF with general boundary data as given in Theorem 5.2, we will attempt to approximate it by level lines of the field with piecewise constant boundary data. For this, a result from [KS16] on the weak convergence of curves, satisfying certain conditions on crossing probabilities, will be crucial.

In order to state the result, we need to define what we mean by *crossings of* topological quadrilaterals.

**Definition 5.9.** A topological quadrilateral  $Q = (V; S_k, k = 0, 1, 2, 3)$  consists of a domain V, along with four boundary arcs  $S_0, S_1, S_2, S_3$ , which can be mapped homeomorphically to a square in such a way that the boundary arcs are in counterclockwise order and correspond to the edges of the square. For any topological quadrilateral, there exists a unique positive L and a conformal map from Q onto the rectangle  $[0, L] \times [0, 1]$ , such that the boundary arcs are mapped to the edges of the quadrilateral and, in particular,  $S_0$  is mapped to  $\{0\} \times [0, 1]$ . We call this unique L the modulus of Q, denoted by m(Q).

**Definition 5.10.** We will often consider topological quadrilaterals in  $\mathbb{H}$  which lie on the boundary in the sense that  $S_1, S_3 \subset \mathbb{R}$  and  $S_0, S_2 \subset \mathbb{H}$ . If we have such a quadrilateral, then we say that a curve  $\gamma : [T_0, T_1] \to \mathbb{C}$  crosses Q if there is a subinterval  $[t_0, t_1] \subset [T_0, T_1]$ , such that  $\gamma(t_0, t_1) \subset V$  but  $\gamma[t_0, t_1]$  intersects both  $S_0$ and  $S_2$ .

Essentially, the condition that will be required for weak convergence will be the following:

**Condition 5.11.** For any simple curve  $\gamma$  on  $\mathbb{H}$  we say that Q is a topological quadrilateral in  $H_{\tau} := \mathbb{H} \setminus \gamma[0,\tau]$  if it is the image of the square  $(0,1)^2$  under a homeomorphism  $\psi$ . We define the sides of  $Q: S_0, S_1, S_2, S_3$ , to be the images of

$$\{0\} \times (0,1), \quad (0,1) \times \{0\}, \quad \{1\} \times (0,1), \quad (0,1) \times \{1\}$$

under  $\psi$ . We consider Q such that the opposite sides  $S_1, S_3$  are contained in  $\partial H_t$ and define a crossing of Q to be a curve in  $H_t$  which connects the two opposite sides  $S_0$  and  $S_2$ . Finally, we say that Q is avoidable if it doesn't disconnect  $\gamma(\tau)$  and  $\infty$ inside  $H_t$ .

A family  $\Sigma$  of probability measures on simple curves from 0 to  $\infty$  in  $\mathbb{H}$  is said to satisfy a conformal bound on an unforced crossing if there exists a constant M > 0such that for any  $\mathbb{P} \in \Sigma$ , for any stopping time  $\tau$ , and any avoidable quadrilateral Q of  $H_{\tau}$  whose modulus m(Q) is greater than M,

$$\mathbb{P}\left(\gamma[\tau, \infty) \operatorname{crosses} Q \mid \gamma[0, \tau]\right) \leq 1/2.$$

Now we may state the result.



Figure 24: In the left panel, the grey part is an avoidable quadrilateral of  $H_{\tau} = \mathbb{H} \setminus \gamma[0, \tau]$ ; in the right panel, the grey part is an unavoidable quadrilateral.

**Proposition 5.12.** Suppose that  $(W^{(n)})_{n\in\mathbb{N}}$  is a sequence of driving processes of random Loewner chains that are generated by continuous simple random cuves  $(\gamma^{(n)})_{n\in\mathbb{N}}$  in  $\mathbb{H}$ , satisfying Condition 5.11. Suppose that the  $(\gamma^{(n)})_{n\in\mathbb{N}}$  are parameterized by half plane capacity. Then

- (W<sup>(n)</sup>)<sub>n∈ℕ</sub> is tight in the metrisable space of continuous functions on [0,∞) with the topology of uniform convergence on compact subsets of [0,∞).
- $(\gamma^{(n)})_{n \in \mathbb{N}}$  is tight in the metrisable space of continuous functions on  $[0, \infty)$ with the topology of uniform convergence on the compact subsets of  $[0, \infty)$ .

Moreover, if the sequence converges weakly in either of the topologies above, then it also converges weakly in the other and the limits agree in the sense that the law of the limiting random curve is the same as the that of the random curve generated under the law of the limiting driving process. In particular, any subsequential limit of the sequence of curves a.s. generates a Loewner chain with continuous driving function.

*Proof.* This may be found in [KS16] cf. Theorem 1.5 and Corollary 1.7.  $\Box$ 

In fact, we will need to apply this theorem when the curves  $(\gamma^{(n)})_{n \in \mathbb{N}}$  correspond to certain  $\mathrm{SLE}_4(\underline{\rho}^L; \underline{\rho}^R)$  processes. In this case they may hit the real line, and so are not necessarily contained in  $\mathbb{H}$ , as required by the Proposition. However, as discussed before the proof of Theorem 1.10 in [KS16], the result extends to curves such as ours, and so we may apply it without concern.

#### 5.2.3 The zero boundary Gaussian free field

In this section we will describe the zero boundary Gaussian free field (GFF) in an arbitrary domain  $D \subsetneq \mathbb{C}$ . We will always assume that the domain has harmonically non-trivial boundary, meaning that a Brownian motion started from a point in the interior will hit the boundary almost surely.

We start with the *Green's function*  $G_D$  in D, which is the unique function in D such that

- $\Delta G_D(z, \cdot) = 2\pi \delta_z(\cdot)$  for each  $z \in D$ , and
- $G_D(z, w) = 0$  if z or w is in  $\partial D$ .

Explicitly,

$$G_D(z,w) = -\log|z-w| - \tilde{G}_z(w)$$

where  $\tilde{G}_z(w)$  is the harmonic extension of  $w \mapsto -\log |z - w|$  from  $\partial D$  to D. The Green's function is conformally invariant in the sense that for any conformal map  $\phi$  on D, and  $z, w \in D$ , we have

$$G_D(z,w) = G_{\phi(D)}(\phi(z),\phi(w)).$$

Roughly speaking, the GFF will be the random Gaussian "function" on D with  $\operatorname{cov}(h(z), h(w)) = G_D(z, w)$ . However, it can only be made sense of rigorously as a random distribution on D. For  $H_s(D)$  the space of smooth compactly supported functions on D, we let  $(\cdot, \cdot)$  denote the normal  $L^2$  inner product on  $H_s(D)$ . We may also endow  $H_s(D)$  with the Dirichlet inner product defined by

$$(f,g)_{\nabla} = \frac{1}{2\pi} \int_D \nabla f(z) \cdot \nabla g(z) d^2 z$$

and we denote its Hilbert space completion under Dirichlet inner product by H(D).

For  $\{\phi_n\}_{n\geq 0}$  an orthonormal basis of H(D), we define the zero boundary GFF h to be the random sum  $h := \sum_n \alpha_n \phi_n$ , where the  $\alpha_n$ 's are i.i.d. Gaussians with mean 0 and variance 1. This almost surely diverges in H(D), but makes sense as a distribution. That is, the limit  $\sum_n \alpha_n(\phi_n, p) := (h, p)$  almost surely exists for each  $p \in H_s(D)$ , and  $p \mapsto (h, p)$  is almost surely a continuous linear functional on  $H_s(D)$ . Note that for any  $f \in H_s(D)$  we have that  $-\Delta f = p$  is also in  $H_s(D)$  and so can define

$$(h,f)_{\nabla} := \frac{1}{2\pi}(h,p).$$

Then  $(h, f)_{\nabla}$  is a Gaussian with mean 0 and variance

$$\frac{1}{4\pi^2} \sum_{n} (\phi_n, p)^2 = \sum_{n} (\phi_n, f)_{\nabla}^2 = (f, f)_{\nabla}.$$

In fact, this characterizes the Gaussian free field. Furthermore, noticing that for  $p \in H_s(D)$ 

$$\Delta^{-1}p := \frac{1}{2\pi} \int_D G_D(\cdot, w) p(w) \, dw$$

is a smooth function in D whose Laplacian is p and vanishes on  $\partial D$ , we see that for any  $f, g, p, q \in H_s(D)$ 

$$\operatorname{cov}((h,f)_{\nabla},(h,g)_{\nabla}) = (f,g)_{\nabla}, \quad \operatorname{cov}((h,p),(h,q)) = \int \int_{D \times D} p(z) G_D(z,w) q(w) d^2 z d^2 w$$

**Proposition 5.13.** [The Markov Property] Let  $W \subset D$  be open and h be a zero boundary GFF on D. Then we can write

$$h = h_1 + h_2$$

where  $h_1$  and  $h_2$  are independent,  $h_1$  is harmonic in W, and  $h_2$  is a zero boundary GFF in W.

This tells us that, given  $h|_{D\setminus W}$ , the conditional law of  $h|_W$  is that of a zero boundary GFF in W, plus the harmonic extension of  $h|_{D\setminus W}$  to W.

Suppose that F is  $L^1$  with respect to harmonic measure on  $\mathbb{R}$  viewed from some point (hence every point) in  $\mathbb{H}$ ; we also denote its bounded harmonic extension to  $\mathbb{H}$  by F. Then the GFF with mean F is defined to be the sum, h + F, of a zero boundary GFF and F.

**Proposition 5.14.** Suppose that  $D_1$  and  $D_2$  are two simply connected domains with non empty intersection, and  $h_i$  is a zero boundary GFF on  $D_i$  for i = 1, 2. Let  $F_i$ be harmonic on  $D_i$ , i = 1, 2 and  $U \subset D_1 \cap D_2$  be a simply connected open domain. Then

(1) If  $dist(U, \partial D_i) > 0$  for i = 1, 2, then the laws of

$$(h_1 + F_1)|_U$$
 and  $(h_2 + F_2)|_U$ 

are mutually absolutely continuous.

(2) Suppose there is a neighbourhood  $\overline{U} \subset U'$  such that  $D_1 \cap U' = D_2 \cap U'$  and that  $F_1 - F_2$  tends to 0 along sequences approaching points in  $\partial D_i \cap U'$ . Then the laws of

$$(h_1 + F_1)|_U$$
 and  $(h_2 + F_2)|_U$ 

are mutually absolutely continuous.

Proof. [MS16a, Proposition 3.2].

#### 5.2.4 Local sets for the GFF

The theory of *local sets* for the Gaussian free field was first introduced by Schramm and Sheffield in [SS13], and we quote several of their results here. For D a simply connected domain and A a random closed subset of  $\overline{D}$ , we let

$$A_{\delta} := \{ z \in D : d(z, A) \le \delta \}$$

and  $\mathcal{A}_{\delta}$  be the smallest  $\sigma$ -algebra for which A and the restriction of h to the interior of  $A_{\delta}$  are measurable. Setting  $\mathcal{A} = \bigcap_{\delta \in \mathbb{Q}_+} \mathcal{A}_{\delta}$  we obtain a  $\sigma$ -algebra which is intuitively the smallest such making A, and h restricted to some infinitesimal neighbourhood of A, measurable. With this in mind, we will often refer to  $\mathcal{A}$  as  $(A, h|_A)$ .

**Definition 5.15.** Suppose that (h, A) is a coupling of a GFF in D and a random closed subset  $A \subset \overline{D}$ . Then we say that A is a local set for h if either of the following equivalent statements hold:

- (1) For any deterministic open subset U ⊂ D we have that, given the orthogonal projection of h onto h<sup>⊥</sup>(U), the event {A∩U = Ø} is independent of the orthogonal projection of h onto H(U). This means that the conditional probability of {A∩U = Ø} given h is a measurable function of the orthogonal projection of h onto H<sup>⊥</sup>(U).
- (2) Given A, the conditional law of h is that of h<sub>1</sub> + h<sub>2</sub>, for h a zero boundary GFF on D \ A and h<sub>1</sub> an A-measurable random distribution which is almost surely harmonic on D \ A.

In this case, we let  $C_A$  be the conditional expectation of h given  $(A, h|_A)$ , corresponding to  $h_1$  in Item (2).

The interactions between local sets display some nice properties, which we will describe in the following propositions.

**Proposition 5.16.** Suppose that  $A_1$ ,  $A_2$  are local sets for a GFF h, which are conditionally independent given h. Then  $A = A_1 \cup A_2$  is also local for h and moreover,

given  $(A_1, A_2, A, h|_A)$ , the conditional law of h is given by  $\mathcal{C}_A$  plus an instance of the zero boundary GFF in  $D \setminus A$ .

*Proof.* [SS13, Lemma 3.10].

**Proposition 5.17.** Let  $A_1, A_2$  be connected local sets which are conditionally independent and  $A = A_1 \cup A_2$ . Then  $C_A - C_{A_2}$  is almost surely a harmonic function in  $D \setminus A$  which tends to zero along any sequence converging to a limit in

- a connected component of  $A_2 \setminus A_1$  which is larger than a singleton, or
- a connected component of A<sub>1</sub> ∩ A<sub>2</sub> which is larger than a singleton, if the limit is at a positive distance from either A<sub>2</sub> \ A<sub>1</sub>, or A<sub>1</sub> \ A<sub>2</sub>.

*Proof.* [SS13, Lemma 3.11] and [MS16a, Proposition 3.6].

**Proposition 5.18.** Let  $A_1, A_2$  be connected local sets which are conditionally independent and  $A = A_1 \cup A_2$ . Suppose that C is a  $\sigma(A_1)$ -measurable connected component of  $D \setminus A_1$  such that  $\{C \cap A_2 = \emptyset\}$  almost surely. Then  $C_A|_C = C_{A_1}|_C$  almost surely, given  $A_1$ .

Proof. [MS16a, Proposition 3.7].

**Proposition 5.19.** Let h be a GFF and  $(Z(t), t \ge 0)$  a family of closed sets such that  $Z(\tau)$  is local for every Z-stopping time  $\tau$ . Suppose further that for a fixed  $z \in D$ ,  $CR(z, D \setminus Z(t))$  is almost surly continuous and monotonic in t. Then, if we reparameterise time by

$$\log \operatorname{CR}(z, D \setminus Z(0)) - \log \operatorname{CR}(z, D \setminus Z(t)),$$

the process  $C_{Z(t)}(z) - C_{Z(0)}(z)$  has a modification which is Brownian motion until the first time that Z(t) accumulates at z. In particular,  $C_{Z(t)}(z)$  has a modification which is almost surely continuous in t.

*Proof.* This is proved in [MS16a, Proposition 6.5]. Since we need the argument in the proof later, we briefly recall the proof here. For  $s \ge 0$ , set

$$\tau(s) := \inf\{t \ge 0 : \log \operatorname{CR}(z, D \setminus Z(0)) - \log \operatorname{CR}(z, D \setminus Z(t)) = s\}.$$

We need only show that the increments of the process  $C_{Z(\tau(t))}(z)$  are independent, and stationary with Gaussian distribution. By [MS16a, Lemma 6.4], we know that for any s < t, the conditional law of

$$\mathcal{C}_{Z(\tau(t))}(z) - \mathcal{C}_{Z(\tau(s))}(z),$$

given  $(Z(\tau(s)), h|_{Z(\tau(s))})$ , is a Gaussian with mean 0 and variance

$$\log \operatorname{CR}(z, D \setminus Z(\tau(s))) - \log \operatorname{CR}(z, D \setminus Z(\tau(t))) = t - s.$$

This means it must also be independent of  $(Z(\tau(s)), h|_{Z(\tau(s))})$ , and so of  $\mathcal{C}_{Z(\tau(s))}(z)$ . This completes the proof.

#### **5.2.5** SLE<sub> $\kappa$ </sub>( $\rho$ ) processes

We call a compact set  $K \subset \mathbb{H}$  an  $\mathbb{H}$ -hull if  $H := \mathbb{H} \setminus K$  is simply connected. For any such hull one can show that there exists a unique conformal map  $\phi$  from  $H \to \mathbb{H}$ which is normalized at  $\infty$  in the sense that

$$\phi(z) = z + \frac{2a}{z} + o(\frac{1}{z}), \quad \text{as } z \to \infty,$$

for some constant a which we call the half-plane capacity of K. For a continuous real-valued function  $(W_t, t \ge 0)$  with  $W_0 = 0$  we can define the solution  $g_t(z)$  to the chordal Loewner equation

$$\partial_t g_t(z) = rac{2}{g_t(z) - W_t}, \quad g_0(z) = z.$$

This is well defined for each  $z \in \mathbb{H}$  until the first time,  $\tau(z)$ , that  $g_t(z) - W_t$  hits 0. Setting  $K_t = \{z \in \overline{\mathbb{H}} : \tau(z) \leq t\}$  and  $H_t = \mathbb{H} \setminus K_t$  we find that  $g_t$  is the conformal map from  $H_t$  to  $\mathbb{H}$  normalized at  $\infty$ , and the half-plane capacity of  $K_t$  is equal to 2t. We call the family  $(K_t, t \geq 0)$  the Loewner chain driven by  $(W_t, t \geq 0)$ . One class of Loewner chains that we will be particularly interested in are those generated by continuous curves; that is, those for which there exists a continuous curve  $\gamma$  such that  $K_t$  is the hull generated by  $\gamma[0, t]$  for all t.

Chordal SLE<sub> $\kappa$ </sub> is the Loewner chain driven by  $W_t = \sqrt{\kappa}B_t$ , where  $B_t$  is a standard one-dimensional Brownian motion. It is characterised by the special properties of conformal invariance and the domain Markov property. Specifically,  $(\mu^{-1}K_{\mu^2 t}, t \ge 0)$ has the same law as  $(K_t, t \ge 0)$  for any  $\mu > 0$ , and for any stopping time  $\tau$ , the law of  $(f_{\tau}(K_{t+\tau}), t \ge 0)$  is the same as that of K. Here  $f_{\tau} := g_{\tau} - W_{\tau}$ .

It is known that  $SLE_{\kappa}$  is almost surely generated by a continuous curve for all  $\kappa$ . In the special case  $\kappa \in [0, 4]$ , it has also been shown that the curve is almost surely simple. Moreover we know that  $\lim_{t\to\infty} \gamma(t) = \infty$  almost surely; a property we refer to as *transience*. These facts were all proved in [RS05].

**Definition 5.20.** Let  $\rho^L$  and  $\rho^R$  be finite Radon measures on  $\mathbb{R}_- = (-\infty, 0]$  and  $\mathbb{R}_+ = [0, \infty)$  respectively, and  $(B_t, t \ge 0)$  be a standard one-dimensional Brownian
motion. We say that

$$\left(W_t, (V_t^L(x))_{x \in \mathbb{R}_-}, (V_t^R(x))_{x \in \mathbb{R}_+}\right)_{t \ge 0}$$

describe an  $SLE_{\kappa}(\rho^L; \rho^R)$  process, if they are adapted to the filtration of B and the following hold:

(1) The processes  $W_t$ ,  $B_t$ ,  $(V_t^L(x))_{x \in \mathbb{R}_-}$  and  $(V_t^R(x))_{x \in \mathbb{R}^+}$  satisfy the following SDE on time intervals where  $W_t$  does not collide with any of the  $V_t^{L,R}(x)$ :

$$dW_t = \sqrt{\kappa} dB_t + \left(\int_{\mathbb{R}_-} \frac{\rho^L(dx)}{W_t - V_t^L(x)}\right) dt + \left(\int_{\mathbb{R}_+} \frac{\rho^R(dx)}{W_t - V_t^R(x)}\right) dt \qquad (5.3)$$

and

$$dV_t^L(x) = \frac{2dt}{V_t^L(x) - W_t}, \quad x \in \mathbb{R}_-; \quad dV_t^R(x) = \frac{2dt}{V_t^R(x) - W_t}, \quad x \in \mathbb{R}_+.$$
(5.4)

(2) We have instantaneous reflection of  $W_t$  off the  $V_t^{L,R}(x)$ , i.e. it is almost surely the case that for Lebesgue almost all times t we have that  $W_t \neq V_t^{L,R}(x)$  for each  $x \in \mathbb{R}$ .

The  $SLE_{\kappa}(\rho^L; \rho^R)$  process is then defined to be the Loewner chain driven by W.

**Remark 5.21.** Note that it is not immediate from the definition that such a process exists. Indeed, we will only show the existence for  $\kappa = 4$  and a specific subset of  $(\rho^L; \rho^R)$ .

We define the *continuation threshold* of the process to be the be the infinum of values of t for which

either 
$$\rho^L \left( \{ x \in \mathbb{R}_- : V_t^L(x) = W_t \} \right) \le -2$$
 or  $\rho^R \left( \{ x \in \mathbb{R}^+ : V_t^R(x) = W_t \} \right) \le -2$ 

Observe that the case  $\rho^L \equiv 0, \rho^R \equiv 0$  corresponds simply to  $\text{SLE}_{\kappa}$ . Another special case is when the Radon measures are purely atomic. If this occurs we instead consider  $(\rho^L; \rho^R)$  to be a pair of vectors

$$\underline{\rho}^L = (\rho_l^L, \cdots, \rho_1^L), \quad \underline{\rho}^R = (\rho_1^R, \cdots, \rho_r^R)$$

with associated force points

$$\underline{x}^L = (x_l^L < \dots < x_1^L \le 0), \quad \underline{x}^R = (0 \le x_1^R < \dots x_r^R)$$

in the obvious way. In this case, it is proved in [MS16a, Theorem 2.2] that a slightly

stronger version of Definition 5.20 determines a unique law on  $\text{SLE}_{\kappa}(\underline{\rho}^{L};\underline{\rho}^{R})$  processes, defined for all time up until the continuation threshold. The additional condition they impose is that  $W_t$ ,  $B_t$ ,  $(V_t^L(x))_{x\in\mathbb{R}_-}$  and  $(V_t^R(x))_{x\in\mathbb{R}^+}$  in fact must satisfy (5.3) and (5.4) at all times. This ensures the uniqueness in law of these processes.

Through their connection with the GFF, which we will discuss in the next section, it was shown in [MS16a] that  $SLE_{\kappa}(\underline{\rho}^{L};\underline{\rho}^{R})$  processes are almost surely generated by continuous curves up to and including the continuation threshold. Moreover, on the event that the continuation threshold is not hit before the curves reach  $\infty$ , the curves are almost surely transient. One can also show that the curves are absolutely continuous with respect to  $SLE_{\kappa}$  as long as they are away from the boundary.

#### 5.2.6 Level lines of the GFF with piecewise constant boundary data

As discussed in the introduction, the theory of level lines and flow lines of a GFF with piecewise constant boundary data has been studied previously in a number of works, including [Dub09], [MS16a],[SS13] and [WW16]. We collect in this section some results that will be useful in our article.

Suppose that F is a bounded harmonic function in  $\mathbb{H}$  whose boundary value is piecewise constant on  $\mathbb{R}$  and changes only finitely many times. Then F can be described almost everywhere in terms of a pair of purely atomic finite Radon measures  $(\rho^L; \rho^R)$ , corresponding to vectors  $(\rho^L; \rho^R)$ , via the relation

$$F(x) = \lambda(1 + \rho^R([0, x])), \quad x \ge 0; \quad F(x) = -\lambda(1 + \rho^L((x, 0])), \quad x < 0.$$
(5.5)

When  $\kappa = 4$ , which corresponds to level lines of the GFF, the following results are known for any  $(\rho^L; \rho^R)$ : (see [WW16, Theorems 1.1.1 and 1.1.2])

- There exists a coupling (K, h) where K is an  $SLE_4(\underline{\rho}^L; \underline{\rho}^R)$  process and h is a zero boundary GFF, such that K is a level line of h + F.
- If h is a zero boundary GFF and K an  $SLE_4(\underline{\rho}^L; \underline{\rho}^R)$  process, coupled such that K is a level line of h + F, then K is almost surely determined by h.

This allows us, for any such F and an instance of the zero boundary GFF h in  $\mathbb{H}$ , to define *the* level line,  $\gamma$ , of h + F. It has been shown in [WW16, Theorem 1.1.3] that  $\gamma$  is in fact almost surely continuous up to and including the continuation threshold, and it is transient when the continuation threshold is not hit.

More generally, for any simply connected domain D and x, y in  $\partial D$ , we say that  $\gamma$  is the level line of a GFF h in D started at x and targeted at y, if  $\phi(\gamma)$  is the level line of  $h \circ \phi^{-1}$ , where  $\phi$  is any conformal map from D to  $\mathbb{H}$  which sends x to 0 and y to  $\infty$ .

One nice property of the level lines is what we call *monotonicity*. Suppose that h is a GFF with piecewise constant boundary values, changing only finitely many times. For  $u \in \mathbb{R}$ , we define the level line of h with height u to be the level line of h + u, and denote it by  $\gamma_u$ . Then, for any  $u_1 \ge u_2$ , the level line  $\gamma_{u_1}$  lies to the left of  $\gamma_{u_2}$  almost surely, see [WW16, Theorem 1.1.4].

Another property of the level lines is their *reversibility*. Suppose that h is a GFF with piecewise constant boundary values changing only finitely many times. Let  $\gamma$ be the level line of h from 0 to  $\infty$  and  $\gamma'$  be the level line of -h from  $\infty$  to 0. Then, on the event that neither hit their continuation thresholds before reaching their target points, we have  $\gamma = \gamma'$  almost surely as sets. This implies the *reversibility* of the SLE<sub>4</sub>( $\underline{\rho}^R; \underline{\rho}^R$ ) process: conditioned on the event that the continuation threshold is not hit, the time reversal of the process is another SLE<sub>4</sub>( $\underline{\rho}^L; \underline{\rho}^R$ ) process, now from  $\infty$  to 0 in  $\mathbb{H}$  with appropriate weights and force points, conditioned not to hit its continuation threshold. See [WW16, Theorem 1.1.6]. Finally, we include a list of results from [WW16] that will be useful for the later proofs.

**Lemma 5.22.** Suppose that h is a zero-boundary GFF and F is the bounded harmonic extension of the piecewise constant boundary data which changes finitely many times. Let  $\gamma$  be the level line of h + F. We already know that  $\gamma$  is almost surely continuous up to and including the continuation threshold.

- (1) [WW16, Theorem 1.1.3] The curve  $\gamma$  is almost surely simple and is continuous up to and including the continuation threshold.
- (2) [WW16, Remark 2.5.15] For any open interval I of  $(-\infty, 0) \cup (0, \infty)$ , assume that

either  $F(x) \ge \lambda$ ,  $\forall x \in I$ , or  $F(x) \le -\lambda$ ,  $\forall x \in I$ .

Then almost surely  $\gamma \cap I = \emptyset$ .

(3) [WW16, Proposition 2.5.11] For any point  $x_0 \in (0, \infty)$ , assume that there exists c > 0 such that  $F \ge -\lambda + c$  in a neighborhood of  $\{x_0\}$ , then almost surely  $\gamma$  does not hit  $\{x_0\}$ . Symmetrically, for  $x_0 \in (-\infty, 0)$ , assume that there exists c > 0 such that  $F \le \lambda - c$  in a neighborhood of  $\{x_0\}$ , then almost surely  $\gamma$  does not hit  $\{x_0\}$ .

#### 5.2.7 First generalizations to the GFF with general boundary data

In this section, we generalize some results concerning level lines with piecewise constant boundary data to general boundary data. In fact, the ideas in the proof for Lemma 5.24 when the boundary condition is piecewise constant [SS13, Lemmas 2.4-2.6] work for general boundary data with proper adjustment. In order to be self-contained, we still give a complete proof here.

**Lemma 5.23.** Suppose that  $(K_t, t \ge 0)$  is a Loewner chain driven by a continuous process  $(W_t, t \ge 0)$ . Denote by  $(g_t, t \ge 0)$  the corresponding sequence of conformal maps and  $f_t = g_t - W_t$  the centered conformal maps. For any fixed  $z \in \mathbb{H}$ , define

$$C_t(z) = \log \operatorname{CR}(z, \mathbb{H}) - \log \operatorname{CR}(z, \mathbb{H} \setminus K_t).$$

Then, we have that

$$dC_t(z) = \frac{4\Im(f_t(z))^2}{|f_t(z)|^4} dt.$$

Proof. The conformal radius  $\operatorname{CR}(z, \mathbb{H} \setminus K_t)$  is equal to  $2/|\phi'_t(z)|$  for  $\phi_t$  any conformal map from  $\mathbb{H} \setminus K_t$  to  $\mathbb{H}$  which sends z to i, an example of which is given by  $m_t \circ f_t$ , where  $m_t : \mathbb{H} \to \mathbb{H}$  is the Möbius transformation defined by

$$m_t(w) = \frac{\Im(f_t(z))w}{\Re(f_t(z))^2 + \Im(f_t(z))^2 - \Re(f_t(z))w}$$

This gives us that

$$C_t(z) - C_0(z) = -\log 2 + \Re(\log m'_t(f_t(z))) + \Re(\log g'_t(z)).$$

However, in this case we can calculate  $m'_t(f_t(z))$  explicitly, and find that  $-\Re(\log m'_t(f_t(z))) = \log \Im f_t(z)$ . Since we also know that

$$dg'_t(z) = \frac{-2g'_t(z)}{f_t(z)^2} dt, \quad d\Im(f_t(z)) = \frac{-2\Im(f_t(z))}{|f_t(z)|^4} dt,$$

we can compute

$$dC_t(z) = \frac{4\Im(f_t(z))^2}{|f_t(z)|^4} dt$$

and this implies the result.

**Lemma 5.24.** Assume the same notations as in Definition 5.1. Suppose that the Loewner chain K is almost surely generated by a random continuous curve  $\gamma$  on  $\overline{\mathbb{H}}$  from 0 to  $\infty$  whose driving function W is almost surely continuous. For  $z \in \mathbb{H}$  and  $t \geq 0$ , set

$$\tau(t) = \inf\{s : \log \operatorname{CR}(z, \mathbb{H}) - \log \operatorname{CR}(z, \mathbb{H} \setminus K_s) = t\}.$$

Then the pair (h, K) can be coupled as in Definition 5.1 if and only if  $(\eta_{\tau(t)}(z), t \ge 0)$ is a Brownian motion with respect to the filtration generated by  $(W_{\tau(t)}, t \ge 0)$  for any  $z \in \mathbb{H}$ .

Proof. If (h, K) is coupled as in Definition 5.1, by Proposition 5.19, we know that  $(\eta_{\tau(t)}(z), t \ge 0)$  is a Brownian motion. Moreover, by the proof of Proposition 5.19, we see that, for s < t, the variable  $\eta_{\tau(t)}(z) - \eta_{\tau(s)}(z)$  is independent of  $K_{\tau(s)}$  and has the law of Gaussian with mean zero and variance t-s. This implies that  $(\eta_{\tau(t)}(z), t \ge 0)$  is a Brownian motion with respect to the filtration generated by  $(W_{\tau(t)}, t \ge 0)$ .

For the converse, assume that, for each  $z \in \mathbb{H}$ , the process  $(\eta_{\tau(t)}(z), t \geq 0)$  is a Brownian motion with respect the filtration generated by  $(W_{\tau(t)}, t \geq 0)$ . We will begin by showing that there exists a Brownian motion  $(B_t, t \geq 0)$  (with respect to the filtration of  $(W_t, t \geq 0)$ ) such that, for all z, we have

$$d\eta_t(z) = \Im \frac{2}{f_t(z)} dB_t.$$
(5.6)

Define

$$U_t(z) = \eta_t(z) + \arg(f_t(z)).$$
 (5.7)

We have the following observations.

- By the definition of η<sub>t</sub>(·), we know that U<sub>t</sub>(·) is the bounded harmonic function on ℍ \ K<sub>t</sub> with the boundary values given by F + 2λ on ℝ<sub>-</sub> \ K<sub>t</sub>, λ along the boundary of K<sub>t</sub>, and F on ℝ<sub>+</sub> \ K<sub>t</sub>. Therefore, for fixed z, process (U<sub>t</sub>(z), t ≥ 0) is of bounded variation and is measurable with respect to the filtration generated by (W<sub>t</sub>, t ≥ 0).
- By the assumption, for fixed z, the process (η<sub>t</sub>(z), t ≥ 0) is a Brownian motion when parameterized by

$$C_t(z) = \log \operatorname{CR}(z, \mathbb{H}) - \log \operatorname{CR}(z, \mathbb{H} \setminus K_t).$$

Thus, by Lemma 5.23, we see that

$$d\langle \eta_t(z) \rangle = dC_t(z) = 4\left(\Im \frac{1}{f_t(z)}\right)^2 dt$$

Moreover, since  $(\eta_{\tau(t)}(z), t \ge 0)$  is a Brownian motion with respect to the filtration of  $(W_{\tau(t)}, t \ge 0)$ , we know that, for any s < t, the variable  $\eta_t(z) - \eta_s(z)$  is independent of  $(W_u, u \le s)$  (we implicitly use the fact that K is a Loewner chain generated by a continuous curve with continuous driving function), thus the process  $(\eta_t(z), t \ge 0)$  is a local martingale with respect to the filtration of  $(W_t, t \ge 0)$ .

Combining these two facts, we know that  $\arg(f_t(z)) = \arg(g_t(z) - W_t)$  is a semimartingale, and hence  $W_t$  is a semimartingale at least up to the first time that z is swallowed by  $K_t$ . Note that

$$d\arg(f_t(z)) = \Im \frac{-1}{f_t(z)} dW_t + \Im \left( \frac{2dt}{(f_t(z))^2} - \frac{d\langle W_t \rangle}{2(f_t(z))^2} \right),$$

and therefore, we have  $d\langle W_t \rangle = 4dt$  for all such t. Note however that the process W does not depend on z, and since we can always choose z far away as we want, we can argue that  $(W_u, 0 \le u \le t)$  is a semimartingale up to time t for any t > 0, with  $d\langle W_t \rangle = 4dt$ . Thus, there exists a Brownian motion  $(B_t, t \ge 0)$  and a process of bounded variation  $(V_t, t \ge 0)$  such that  $W_t = 2B_t - V_t$ . We emphasize that the processes B and V do not depend on z. Plugging in Equation (5.7), we have that

$$d\eta_t(z) = \Im \frac{2}{f_t(z)} dB_t, \quad dU_t(z) = \Im \frac{1}{f_t(z)} dV_t, \tag{5.8}$$

as desired.

With (5.6) in hand, we know that for  $z, w \in \mathbb{H}$ , we have

$$d\langle \eta_t(z), \eta_t(w) \rangle = \Im\left(\frac{1}{f_t(z)}\right)\Im\left(\frac{1}{f_t(w)}\right)dt.$$

We know that, for each z,  $\eta_t(z)$  is a continuous martingale. We can also extend the definition of  $\eta_t(z)$  by setting it equal to its limit as  $s \uparrow \tau(z)$  at all times after  $\tau(z)$ . We further define for  $z, w \in \mathbb{H}$  and  $t \leq \tau(z) \land \tau(w)$ ,

$$G_t(z,w) := G_{\mathbb{H}}(f_t(z), f_t(w))$$

where we again extend this to all times after  $\tau(z) \wedge \tau(w)$ , by setting it constant and equal to its limit as  $t \uparrow \tau(z) \wedge \tau(w)$ . Observe that, in each connected component of  $\mathbb{H} \setminus \gamma[0, t]$ , the function  $\eta_t$  is the bounded harmonic function with boundary values shown in Figure 25. We also know that  $G_t(z, w)$  is non-decreasing in t for any fixed



Figure 25: The function  $\eta_t(\cdot)$  is a harmonic function in each connected component of  $\mathbb{H} \setminus \gamma[0,t]$  with boundary values as above.

z, w.

Putting all of the above together, we can deduce by stochastic calculus that for

any  $p \in H_s(\mathbb{H})$ ,  $(\eta_t, p)$  is a continuous martingale with

$$d\langle (\eta_t, p) \rangle = -dE_t(p), \quad \text{where} \quad E_t(p) := \int \int p(z)p(w)G_t(z, w) \, d^2z d^2w.$$

Now we are ready to show that the pair (h, K) is coupled as in Definition 5.1. Since for each  $z, w \in \mathbb{H}$  and non-negative  $p \in H_s(\mathbb{H})$  we have that  $\eta_t(z)$  is a martingale and  $G_t(z, w)$ ,  $E_t(p)$  are non-decreasing, it must be that all the limits  $\eta_{\infty}(z)$ ,  $G_{\infty}(z, w)$  and  $E_{\infty}(p)$  exist. We let  $\tilde{h}$  be equal to  $\eta_{\infty} - \eta_0$  plus a sum of independent zero boundary GFF's; one in each connected component of  $\mathbb{H} \setminus \gamma$ . To show that  $(K, \tilde{h})$  are coupled in the correct way we must verify that the marginal law of  $\tilde{h}$ is that of a zero boundary GFF in  $\mathbb{H}$ , and that  $(K, \tilde{h})$  satisfies the correct domain Markov property. This amounts to showing that for each non-negative  $p \in H_s(\mathbb{H})$ :

- $(\tilde{h}, p)$  is a Gaussian with mean 0 and variance  $E_0(p)$ .
- For any K-stopping time  $\tau$ , the conditional law of  $\left((\tilde{h} + \eta_0)|_{\mathbb{H}\setminus K_{\tau}}, p\right)$  given  $K_{\tau}$  is a Gaussian with mean  $(\eta_{\tau}, p)$  and variance  $E_{\tau}(p)$ .

To see the first point, for any  $\mu > 0$  we calculate

$$\begin{split} \mathbb{E}[\exp(-\mu(\tilde{h},p))] &= \mathbb{E}[\mathbb{E}[\exp(-\mu(\tilde{h},p))|K]] \\ &= \mathbb{E}\left[\exp\left(-\mu(\eta_{\infty}-\eta_{0},p) - \frac{\mu^{2}}{2}E_{\infty}(p)\right)\right] \\ &= \mathbb{E}\left[\exp\left(-\mu(\eta_{\infty}-\eta_{0},p) + \frac{\mu^{2}}{2}(E_{0}(p) - E_{\infty}(p))\right)\right]\exp\left(-\frac{\mu^{2}}{2}E_{0}(p)\right) \\ &= \exp\left(-\frac{\mu^{2}}{2}E_{0}(p)\right), \end{split}$$

where the last line follows from the fact that  $(\eta_t, p)$  is a continuous bounded martingale with mean  $\eta_0(p)$  and quadratic variation  $E_0(p) - E_{\infty}(p)$ . The second point follows similarly, replacing the initial expectation with a conditional one.  $\Box$ 

#### 5.3 Non-boundary intersecting regime

All the conclusions in Sections 5.3 and 5.4 are proved in [MS16a, WW16] for level lines with piecewise constant boundary data and constant height difference. Although many of the ideas from these papers are fundamental to our proofs, there are several places where they fail for general boundary data. Therefore, we treat the general case here and give complete proofs in the next two sections.

**Lemma 5.25.** Suppose that  $\gamma$  is a random continuous curve from 0 to some  $\gamma$ -stopping time T with almost surely continuous driving function. Assume that  $\gamma$  is

coupled with a zero boundary GFF h as a level line of h + F up to time T where

$$F(x) \ge -\lambda, \quad \forall x < 0; \quad F(x) \ge \lambda, \quad \forall x \ge 0.$$

Then almost surely  $\gamma[0,T] \cap (0,\infty) = \emptyset$ .

Proof. Assume the same notations as in Definition 5.1. First, for any  $z \in \mathbb{H}$ , define  $U_t(z)$  in the same way in Equation (5.7), and we will explain that the process  $(U_t(z), 0 \leq t \leq T)$  is non-increasing. By the definition of  $\eta_t(z)$ , we know that  $U_t(\cdot) - \lambda$  is the harmonic function on  $\mathbb{H} \setminus K_t$  with the boundary values given by  $F + \lambda \geq 0$  on  $\mathbb{R}_- \setminus K_t$ , zero along the boundary of  $K_t$ , and  $F - \lambda \geq 0$  on  $\mathbb{R}_+ \setminus K_t$ . This harmonic function is non-increasing in t, and thus  $U_t(z)$  is non-increasing.

Next, we will show that the process

$$Z_t := V_t^R(0^+) - W_t$$

cannot hit zero. By the proof of Lemma 5.24, we know that there exist a Brownian motion B and a process of bounded variation V such that W = 2B - V and Equation (5.8) holds. Since  $U_t(z)$  in non-increasing in t, the process  $V_t$  is non-decreasing in t up to the time that z is swallowed. However, the process V does not depend on z, and since we can always choose z far away, we have that  $(V_u, 0 \le u \le t)$  is non-decreasing in u for any t > 0. Then we have, for all t,

$$dZ_t \ge -2dB_t + \frac{2dt}{Z_t}.$$

We can compare  $Z_t/2$  with a Bessel process of dimension 2, and so may conclude that  $Z_t$  cannot hit 0. This implies that the curve cannot hit  $(0, \infty)$ .

Remark 5.26. The proof of Lemma 5.25 also applies to the case when

$$F(x) \le -\lambda, \quad x < 0; \quad F(x) \le \lambda, \quad x \ge 0$$

by symmetry. In this case we see that for  $\gamma$  satisfying the same conditions as in Lemma 5.25, we have  $\gamma[0,T] \cap (-\infty,0) = \emptyset$  almost surely.

**Lemma 5.27.** Suppose that  $\gamma$  is a random continuous curve from 0 to some  $\gamma$ stopping time T with almost surely continuous driving function. Assume that  $\gamma$  is coupled with a zero boundary GFF h as a level line of h + F up to time T where

$$F(x) \le -\lambda, \quad \forall x < 0; \quad F(x) \ge \lambda, \quad \forall x \ge 0.$$



(a) The boundary value of -h - G given (b)  $\gamma'_G$  cannot hit the left side of  $\gamma[0, T_{\varepsilon}]$  or  $\gamma[0, T_{\varepsilon}]$ .  $\mathbb{R}_{-}$ .

Figure 26: Explanation of the boundary values in the proof of Lemma 5.27.

Then, almost surely, the curve  $(\gamma(t), 0 \le t \le T)$  does not hit the boundary except the two end points.

*Proof.* It is sufficient to show that, for any  $0 < a < b < \infty$ , the curve  $\gamma$  does not hit the interval I = (a, b). We prove by contradiction.

Suppose that  $\gamma$  does hit I with positive probability, and on this event, define  $T_{\varepsilon}$  to be the first time that  $\gamma$  gets within  $\varepsilon$  of I. Since F is bounded, suppose that  $F \geq -C$  for some finite  $C \geq \lambda$ . Let G be the bounded harmonic extension of the function which is equal to -C on  $\mathbb{R}_{-}$  and is equal to  $\lambda$  on  $\mathbb{R}_{+}$ . Note that  $F \geq G$ . Let  $\gamma'_{G}$  be the level line of -h - G from  $\infty$  to 0. By Lemma 5.22(1) and (2), we know that  $\gamma'_{G}$  is almost surely continuous and transient; and that  $\gamma'_{G}$  almost surely does not hit I.

Let  $\tilde{h}$  be h restricted to the unbounded connected component of  $\mathbb{H} \setminus \gamma[0, T_{\varepsilon}]$ , then conditionally on  $\gamma[0, T_{\varepsilon}]$ , the field  $-\tilde{h} - G$  is a GFF with boundary data as shown in Figure 26(a). Moreover, given  $\gamma[0, T_{\varepsilon}]$ , the curve  $\gamma'_G$  is coupled with  $\tilde{h}$  so that it is a level line of  $-\tilde{h} - G$  up until the first time that  $\gamma'_G$  hits  $\gamma[0, T_{\varepsilon}]$  (by Propositions 5.16 to 5.18). Since F - G is positive on  $\mathbb{H}$ , we see from Lemma 5.25 that  $\gamma'_G$  cannot hit the left side of  $\gamma[0, T_{\varepsilon}]$  or  $(-\infty, 0]$  before hitting the right side of  $\gamma[0, T_{\varepsilon}]$  or the tip  $\gamma(T_{\varepsilon})$ , see Figure 26(b). In any case, this implies that  $\gamma'_G$  has to get within  $\varepsilon$  of I. Since this holds for any  $\varepsilon > 0$  on the event that  $\gamma$  hits I and  $\gamma'_G$  is continuous, we can conclude that  $\gamma'_G$  hits I with positive probability, contradiction.  $\Box$ 

**Lemma 5.28.** Assume the same notations as in Lemma 5.27. Then  $\gamma$  is almost surely simple.

*Proof.* First, we argue that, for any  $\gamma$ -stopping time  $\tau$ , we have  $\gamma[0,\tau) \cap \gamma(\tau,T) = \emptyset$ 

almost surely. Given  $\gamma[0, \tau]$ , denote by  $\tilde{h}$  the restriction of h + F to  $\mathbb{H} \setminus \gamma[0, \tau]$  (since  $\gamma$  does not hit the boundary, this set only has one connected component). By the domain Markov property in Definition 5.1, we know that, given  $\gamma[0, \tau]$ , the curve  $\gamma|_{[\tau,T)}$  is coupled with  $\tilde{h}$  as its level line. Note that the boundary value of  $\tilde{h} + F$  is  $F \leq -\lambda$  on  $\mathbb{R}_-$ , is  $-\lambda$  along the left side of  $\gamma[0, \tau]$ , is  $\lambda$  along the right side of  $\gamma[0, \tau]$ , and is  $F \geq \lambda$  along  $\mathbb{R}_+$ . By Lemma 5.27, we know that  $\gamma(\tau, T)$  cannot hit  $\gamma(0, \tau)$ .

Next, we show that  $\gamma$  is almost surely simple. For any q > 0, define  $A_q$  to be the event that  $\gamma(0,q) \cap \gamma(q,T) \neq \emptyset$ . If  $\gamma$  has double point, then  $A_q$  happens for some positive rational q, since  $\gamma$  is continuous. However, by the above argument, we know that  $\cup_{q \in \mathbb{Q}_+} A_q$  has zero probability. Therefore,  $\gamma$  is almost surely simple.  $\Box$ 

**Proposition 5.29.** Suppose that h is a zero boundary GFF and that F is bounded and satisfies

$$F(x) \le -\lambda, \quad \forall x < 0; \quad F(x) \ge \lambda, \quad \forall x \ge 0.$$

Suppose that  $\gamma$  (resp.  $\gamma'$ ) is a random continuous transient curve from 0 to  $\infty$  (resp. from  $\infty$  to 0) with almost surely continuous driving function.

Assume that  $\gamma$  is coupled with h as a level line of h + F, that  $\gamma'$  is coupled with h as a level line of -h - F, and that the triple  $(h, \gamma, \gamma')$  are coupled so that  $\gamma$  and  $\gamma'$  are conditionally independent given h. Then almost surely  $\gamma'$  equals  $\gamma$ . In particular, this implies that  $\gamma$  is almost surely determined by h.

Proof. First, we argue that, for any  $\gamma'$ -stopping time  $\tau'$ , given  $\gamma'[0, \tau']$ , the curve  $\gamma$  almost surely first exits  $\mathbb{H} \setminus \gamma'[0, \tau']$  at  $\gamma'(\tau')$ . Denote by  $\tilde{h}$  the restriction of h to  $\mathbb{H} \setminus \gamma'[0, \tau']$ . Given  $\gamma'[0, \tau']$ , the curve  $\gamma$  is coupled with h as a level line of  $\tilde{h} + F$ . The boundary value of  $\tilde{h} + F$  is  $F \leq -\lambda$  on  $\mathbb{R}_-$ , is  $-\lambda$  along the left side of  $\gamma'[0, \tau']$ , is  $\lambda$  along the right side of  $\gamma'[0, \tau']$ , and is  $F \geq \lambda$  on  $\mathbb{R}_+$ . Thus, by Lemma 5.27, we know that  $\gamma$  must exit  $\mathbb{H} \setminus \gamma'[0, \tau']$  at  $\gamma'(\tau')$ .

Next, we show that  $\gamma$  and  $\gamma'$  are equal. Since  $\gamma$  hits  $\gamma'[0, \tau']$  for the first time at  $\gamma'(\tau')$  for any  $\gamma'$ -stopping time  $\tau'$ , we know that  $\gamma$  hits a dense countable set of points along  $\gamma'$  in reverse chronological order. By symmetry,  $\gamma'$  hits a dense countable set of points along  $\gamma$ . Since both  $\gamma$  and  $\gamma'$  are continuous simple curves, the two curves (viewed as sets) are equal.

#### 5.4 Monotonicity

**Lemma 5.30.** Suppose that h is a zero boundary GFF and that F is bounded. Suppose that  $\gamma$  is a random continuous curve from 0 to some  $\gamma$ -stopping time T with almost surely continuous driving function. Assume that  $\gamma$  is coupled with h as a level line of h + F up to time T. (1) Then the curve  $(\gamma(t), 0 \le t \le T)$  almost surely does not intersect any open interval I of  $(0, \infty)$  such that

$$F(x) \ge \lambda \quad \forall x \in I.$$

Symmetrically, it does not intersect any open interval of  $(-\infty, 0)$  where  $F(x) \leq -\lambda$ .

(2) In addition, if (γ(t), 0 ≤ t ≤ T) is almost surely simple, then it does not hit any open interval I of (-∞, 0) where F(x) ≥ λ. Symmetrically, it does not intersect any open interval of (0,∞) where F(x) ≤ -λ.

Proof of Lemma 5.30, Item (1). We first show the conclusion when I = (a, b) for 0 < a < b and  $F(x) \geq \lambda, \forall x \in I$ . Pick  $\tilde{a}, \tilde{b}$  such that  $a < \tilde{a} < \tilde{b} < b$ . It is sufficient to show that, for any such  $\tilde{a}, \tilde{b}$ , the curve  $(\gamma(t), 0 \leq t \leq T)$  does not hit the interval  $\tilde{I} = [\tilde{a}, \tilde{b}]$ . We prove by contradiction. Suppose that the curve  $(\gamma(t), 0 \leq t \leq T)$  hits  $\tilde{I}$  with positive probability. Since F is bounded, we have that  $F \geq -C$  for some  $C \geq \lambda$ . Let G be the bounded harmonic extension of the function which is equal to -C on  $\mathbb{R}_- \cup (0, a) \cup (b, \infty)$  and  $\lambda$  on (a, b). Note that  $F \geq G$ . Let  $\gamma'_G$  be the level line of -h - G from b to a. Note that since G is piecewise constant we know by Lemma 5.22(1) that the curve  $\gamma'_G$  is continuous from b to a, and the boundary data also means, by Lemma 5.22(2), that it does not hit  $\tilde{I}$ . This means we can repeat the same argument as in the proof of Lemma 5.27 to show that  $\gamma'_G$  hits  $\tilde{I}$  with positive probability and obtain a contradiction.

Proof of Lemma 5.30, Item (2). Now, let I = (a, b) for a < b < 0, and suppose that  $F(x) \ge \lambda, \forall x \in I$  and  $(\gamma(t), 0 \le t \le T)$  is almost surely simple. It will be sufficient for us to prove that  $\gamma$  does not hit  $\tilde{I} = [\tilde{a}, \tilde{b}]$  for any  $a < \tilde{a} < \tilde{b} < b$ . First note that if  $\gamma$  hits  $[-\infty, a]$  before hitting I, since  $\gamma$  grows towards  $\infty$ , it can never hit  $\tilde{I}$  thereafter and we are done. If not, let  $\varphi$  be the Möbius transform of  $\mathbb{H}$  that sends the triplet  $(b, 0, \infty)$  to  $(\infty, 0, 1)$ . Then  $(\varphi(\gamma(t)), 0 \le t \le T)$  is a continuous curve, coupled with a zero-boundary GFF  $\tilde{h}$  as a level line of  $\tilde{h} + F \circ \varphi^{-1}$  until the first time it hits  $[1, \infty]$ . By Item (1), we know that  $(\varphi(\gamma(t)), 0 \le t \le T)$  cannot hit  $\tilde{I}$  without first hitting the point b. Let  $\tau$  be the time at which  $\gamma$  hits  $\tilde{I}$ , again setting  $\tau' = T$  if this never happens, and  $\tau'$  be the first time at which  $\gamma$  hits  $\tilde{I}$ , again setting  $\tau' = T$  if necessary. By the previous reasoning, if we do not have  $\{\tau < \tau' < T\}$  then we are done, so assume this occurs with positive probability. On this event, since  $\gamma$  is a continuous curve with continuous Loewner driving function, we see that  $\{\tau \le t \le \tau' : \gamma(t) \in \mathbb{R}\}$  has Lebesgue measure 0, and so there exists a time  $\tau < \sigma < \tau'$  with  $\gamma(\sigma) \notin \mathbb{R}$ . Let

w be the left-most point in  $(\gamma(t), 0 \le t \le \sigma) \cap (-\infty, 0)$ . Then applying the same argument as above, now to  $(\gamma(t), \sigma \le t \le T)$  in the domain  $\mathbb{H} \setminus (\gamma(t), 0 \le t \le \sigma)$ with (a, b) replaced by (a, w), we see that  $(\gamma(t), \sigma \le t \le T)$  must first hit w before it can hit  $\tilde{I}$ . This is a contradiction to the simplicity of  $\gamma$ .

**Lemma 5.31.** Suppose that h is a zero boundary GFF and that F is bounded. Suppose that  $\gamma$  is a random continuous curve from 0 to some  $\gamma$ -stopping time T with almost surely continuous driving function. Assume that  $\gamma$  is coupled with h as a level line of h + F up to time T.

- (1) For any fixed point x<sub>0</sub> ∈ (0,∞), if there exists c > 0 such that F ≥ -λ + c in a neighborhood of x<sub>0</sub>, then the curve (γ(t), 0 ≤ t ≤ T) almost surely does not hit {x<sub>0</sub>}. Symmetrically, for any fixed point x<sub>0</sub> ∈ (-∞,0), if there exists c > 0 such that F ≤ λ - c in a neighborhood of x<sub>0</sub>, then the curve (γ(t), 0 ≤ t ≤ T) almost surely does not hit {x<sub>0</sub>}.
- (2) If there exists  $X \in (0, \infty)$  and c > 0 such that

$$F(x) \ge \lambda, \quad \forall x \in (-X, \infty); \quad F(x) \ge -\lambda + c, \quad \forall x \in (X, \infty)$$

and in addition the curve  $(\gamma(t), 0 \leq t \leq T)$  is almost surely simple, then  $(\gamma(t), 0 \leq t \leq T)$  almost surely does not hit  $\infty$ . Symmetrically, if there exists  $X \in (0, \infty)$  and c > 0 such that

$$F(x) \le -\lambda, \quad \forall x \in (X,\infty); \quad F(x) \le \lambda - c, \quad \forall x \in (-X,\infty)$$

and the curve  $(\gamma(t), 0 \le t \le T)$  is almost surely simple, then  $(\gamma(t), 0 \le t \le T)$ almost surely does not hit  $\infty$ 

We point out that Item (2) is not a consequence of Item (1) in Lemma 5.31. In fact, if F is piecewise constant and  $F(x) \in (-\lambda + c, \lambda - c)$  on  $(-\infty, -X) \cup (X, \infty)$ for  $c \in (0, \lambda)$ , then the level line of h + F is transient, and hence hits  $\infty$  almost surely.

Proof of Lemma 5.31, Item (1). We may assume that  $F \ge -\lambda + c$  on (a, b) where  $0 < a < x_0 < b$  and again prove by contradiction. Suppose that the curve  $(\gamma(t), 0 \le t \le T)$  does hit  $\{x_0\}$  with some positive probability. Since F is bounded, suppose that  $F \ge -C$  for some  $C \ge \lambda$ . Let G be the function which is equal to -C on  $\mathbb{R}_- \cup (0, a) \cup (b, \infty)$ , and is  $(-\lambda + c) \land \lambda$  on (a, b). Note that  $F \ge G$ . Let  $\gamma'_G$  be the level line of -h - G from b to a. Note that  $\gamma'_G$  is continuous and does not the point  $\{x_0\}$  by Lemma 5.22(3). Thus we can repeat the same argument as in the

proof of Lemma 5.27 and show that  $\gamma'_G$  hits  $\{x_0\}$  with positive probability, which is a contradiction.





(a) Suppose that  $\gamma_F$  hits  $\infty$  with positive probability. Let  $T_{\varepsilon}$  be the first time that it enters  $\{z : |z| > 1/\varepsilon\}$ . Since  $F \ge \lambda$  on  $(-\infty, -X)$ , the curve  $\gamma_F$  can never hit  $(-\infty, -X)$ . Given  $\gamma_F[0, T_{\varepsilon}]$ , the boundary data of -h - G is shown in this figure.

(b) By the choice of G, we see that  $\gamma'_G$  cannot hit the union of (-X, 0) and the left side of  $\gamma_F[0, T_{\varepsilon}]$  before hitting  $\gamma_F[0, T_{\varepsilon}]$ . Therefore,  $\gamma'_G$ has to enter  $\{z : |z| > 1/\varepsilon\}$ . This holds for all  $\varepsilon > 0$ , thus  $\gamma'_G$  hits  $\infty$  with positive chance, contradiction.

Figure 27: Explanation of the boundary values in the proof of Lemma 5.31, Item (2).

Proof of Lemma 5.31, Item (2). We prove by contradiction. Suppose that  $\gamma$  does hit  $\infty$  with positive probability. Since F is bounded, suppose that  $F \geq -C$  for some  $C \geq \lambda$ . Let G be the function which is equal to -C on  $(-X,0) \cup (0,X)$  and is  $(-\lambda + c) \wedge \lambda$  on  $(X, \infty) \cup (-\infty, -X)$ . Note that  $F \geq G$ . Let  $\gamma'_G$  be the level line of -h - G from -X to X. Since G is piecewise constant, we know the curve  $\gamma'_G$ is continuous and does not hit the point  $\infty$ . By Lemma 5.30(2), since  $\gamma$  is almost surely simple, we know that  $\gamma$  cannot hit  $(-X, \infty)$  before it hits  $\infty$ . Thus, we can repeat the same argument as in the proof of Lemma 5.27 and show that  $\gamma'_G$  hits  $\infty$ with positive probability, contradiction. See more details in Figure 27.

**Lemma 5.32.** Suppose that h is a zero boundary GFF and that F is bounded and satisfies Condition (5.2). Suppose that  $\gamma$  is a random continuous transient curve from 0 to  $\infty$  with almost surely continuous driving function. Assume that  $\gamma$  is coupled with h as a level line of h + F. Then  $\gamma$  is almost surely simple.

*Proof.* We can repeat the same argument as in the proof of Lemma 5.28 replacing Lemma 5.27 by Lemmas 5.30(1) and 5.31(1).

**Lemma 5.33.** Suppose that F and G are bounded, F satisfies Condition (5.2), and that

$$F(x) \ge G(x), \quad \forall x \in \mathbb{R}.$$

Suppose that  $\gamma_F$  (resp.  $\gamma'_G$ ) is a random continuous transient curve from 0 to  $\infty$  (resp. from  $\infty$  to 0) with almost surely continuous driving function.

Assume that  $\gamma_F$  is coupled with a zero boundary GFF h as a level line of h + Ffrom 0 to  $\infty$  and that  $\gamma'_G$  is coupled with h as a level line of -h - G from  $\infty$  to 0, and that the triple  $(h, \gamma_F, \gamma'_G)$  is coupled so that  $\gamma_F$  and  $\gamma'_G$  are conditionally independent given h. Then almost surely  $\gamma_F$  stays to the left of  $\gamma'_G$ .

*Proof.* Note that, by Lemma 5.32,  $\gamma_F$  is almost surely simple. It is sufficient to show that, for any  $\gamma'_G$ -stopping time  $\tau'$ , the point  $\gamma'_G(\tau')$  is to the right of  $\gamma_F$ .

Let  $\tilde{h}$  be h restricted to  $\mathbb{H} \setminus \gamma'_G[0, \tau']$ . Then we know that, given  $\gamma'_G[0, \tau']$ , the conditional law of  $\tilde{h} + F$  is that of a GFF with boundary data as shown in Figure 28(a). Moreover,  $\gamma_F$  is coupled as a level line of  $\tilde{h} + F$  up until the first time it hits  $\gamma'_G[0, \tau']$ . Given  $\gamma'_G[0, \tau']$ , let  $\tau$  be the first time that  $\gamma_F$  hits  $\gamma'_G[0, \tau']$ .

Consider the set  $\gamma'_G[0, \tau']$ , there are two possibilities for the intersection  $\gamma'_G[0, \tau'] \cap (0, \infty)$ : Case (a), the intersection is nonempty, and in this case we denote by  $x_G$  the last point in the intersection; Case (b), the intersection is empty and in this case we set  $x_G = +\infty$  ie. to the right of  $\gamma'_G[0, \tau']$ .



Figure 28: The curve  $\gamma_F$  cannot hit the blue section in the figure.

Since the boundary data on the right hand side of  $\gamma'_G[0, \tau']$  is greater than  $\lambda$  and the boundary data is bounded away from  $-\lambda$  in a neighborhood of  $x_G$ , by Lemma 5.30 we know that  $\gamma_F$  cannot hit the right hand side of  $\gamma'_G[0, \tau']$  before hitting the left side of  $\gamma'_G[0, \tau']$  or exiting  $\mathbb{H}$  at  $\infty$ , approaching from the left. We also know that  $\gamma_F$  cannot hit  $\{x_G\}$  before this time, by Lemma 5.31(1) in Case (a) and by Lemma 5.31(2) in Case (b). Therefore  $\gamma_F$  cannot hit the union of the right hand side of  $\gamma'_{C}[0,\tau']$  and  $\{x_{C}\}$  (i.e. the blue section in Figure 28(a)) of the boundary before hitting the left hand side of  $\gamma'_G[0, \tau']$  or exiting  $\mathbb{H}$  at  $\infty$ , approaching from the left. In the latter case, we are done. In the former case,  $\gamma_F$  first hits  $\gamma'_G[0, \tau']$  from its left hand side at time  $\tau$ . If  $\gamma'_G(\tau')$  is strictly to the left of  $\gamma_F$ , then it must be the case that after time  $\tau$ ,  $\gamma_F$  wraps around  $\gamma'_G[0, \tau']$  and then hits the right hand side of  $\gamma'_G[0, \tau']$ or exits at  $x_G$ . Let  $\tau_{\delta}$  be the first time after  $\tau$  that  $\gamma_F$  is in the right connected component of  $\mathbb{H} \setminus (\gamma_F[0,\tau] \cup \gamma'_G[0,\tau'])$  and  $\operatorname{dist}(\gamma_F(t),\gamma'_G[0,\tau]) \geq \delta$ , setting  $\tau_{\delta} = \infty$ if this never happens. If  $\gamma'_G(\tau')$  is strictly to the left of  $\gamma_F$  (so in particular not on the curve  $\gamma_F$ ) with positive probability then we know that  $\{\tau_{\delta} < \infty\}$  occurs with strictly positive probability. However, given  $\gamma'_G[0,\tau'] \cup \gamma_F[0,\tau_{\delta}]$ , the conditional law of h+Fis that of a GFF with boundary values as shown in Figure 28(b), and  $(\gamma_F(t), t \geq \tau_{\delta})$ is a level line of this field (by Propositions 5.16 to 5.18.) Therefore, by Lemmas 5.30 and 5.31 again, we know that it cannot hit the right hand side of  $\gamma'_G[0,\tau']$  or exit at  $x_G$ , and hence cannot reach  $\infty$ . Thus we obtain a contradiction. 

**Lemma 5.34.** Suppose that F is bounded and satisfies Condition (5.2). Suppose that  $\gamma_F$  (resp.  $\gamma'_F$ ) is a random continuous transient curve from 0 to  $\infty$  (resp. from  $\infty$  to 0) with almost surely continuous driving functions.

Assume that  $\gamma_F$  is coupled with a zero boundary GFF h as a level line of h + Ffrom 0 to  $\infty$ , that  $\gamma'_F$  is coupled with h as a level line of -h - F from  $\infty$  to 0, and that the triple  $(h, \gamma_F, \gamma'_F)$  is coupled so that  $\gamma_F$  and  $\gamma'_F$  are conditionally independent given h. Then almost surely  $\gamma_F = \gamma'_F$ . In particular,  $\gamma_F$  is almost surely determined by h.

*Proof.* By Lemma 5.33, we know that  $\gamma_F$  almost surely stays to the left of  $\gamma'_F$  and (by the same arguments) that  $\gamma_F$  almost surely stays to the right of  $\gamma'_F$ . Combining with the fact that  $\gamma_F, \gamma'_F$  are simple by Lemma 5.32, we know that almost surely  $\gamma_F = \gamma'_F$ . Since  $\gamma_F$  and  $\gamma'_F$  are coupled with h so that they are conditionally independent given  $h, \gamma_F = \gamma'_F$  implies that  $\gamma_F$  must be almost surely determined by h.

**Lemma 5.35.** Suppose that F and G are bounded, and satisfy Condition (5.2). Suppose further that

$$F(x) \ge G(x), \quad x \in \mathbb{R}.$$

Suppose that  $\gamma_F, \gamma_G$  (resp.  $\gamma'_G$ ) are random continuous transient curves from 0 to  $\infty$  (resp. from  $\infty$  to 0) with almost surely continuous driving functions.

Assume that  $\gamma_F$  (resp.  $\gamma_G$ ) is coupled with a zero boundary GFF h as a level line of h + F (resp. h + G), that  $\gamma'_G$  is coupled with h as a level line of -h - G from  $\infty$  to 0, and that  $(h, \gamma_F, \gamma_G, \gamma'_G)$  is coupled so that  $\gamma_F$ ,  $\gamma_G$  and  $\gamma'_G$  are conditionally independent given h. Then almost surely  $\gamma_F$  stays to the left of  $\gamma_G$ .

*Proof.* We have the following observations.

- By Lemma 5.33, we know that  $\gamma_F$  stays to the left of  $\gamma'_G$ .
- By Lemma 5.34, we know that  $\gamma_G = \gamma'_G$ .

Combining these two facts, we see that  $\gamma_F$  stays to the left of  $\gamma_G$ .

**Corollary 5.36.** Suppose that F and G are piecewise constant functions changing value only finitely many times and that they satisfy Condition (5.2). Suppose further that

$$F(x) \ge G(x), \quad x \in \mathbb{R}.$$

Let  $\gamma_F$  (resp.  $\gamma_G$ ) be the level line of h + F (resp. h + G) for h a zero boundary GFF as in Section 5.2.6. Then almost surely  $\gamma_F$  stays to the left of  $\gamma_G$ .

*Proof.* From the results in Section 5.2.6, we have the existence, the continuity and transience of  $\gamma_F$  and  $\gamma_G$ , and also  $\gamma'_G$  which is the level line of -h - G from  $\infty$  to 0. Moreover, we know that each of  $\gamma_F$ ,  $\gamma_G$  and  $\gamma'_G$  is almost surely determined by h. By Lemma 5.35, we know that  $\gamma_F$  stays to the left of  $\gamma_G$  almost surely.

## 5.5 Estimates on crossing probabilities

In this section, we will consider  $SLE_4(\underline{\rho}^L; \underline{\rho}^R)$  processes for vectors

$$\underline{\rho}^L = (\rho_l^L, \cdots, \rho_1^L), \qquad \underline{\rho}^R = (\rho_1^R, \cdots, \rho_r^R),$$

with associated force points

$$\underline{x}^L = (x_l^L < \dots < x_1^L \le 0), \qquad \underline{x}^R = (0 \le x_1^R < \dots x_r^R),$$

such that for some  $c > 0, C < \infty$ ,

$$-2 + \frac{c}{\lambda} \le \sum_{i=1}^{j} \rho_i^L \le -1 + \frac{C}{\lambda}, \quad 1 \le j \le l, \quad -2 + \frac{c}{\lambda} \le \sum_{i=1}^{k} \rho_i^R \le -1 + \frac{C}{\lambda}, \quad 1 \le k \le r.$$

$$(5.9)$$

We will show that if  $(\gamma^{(n)})_{n \in \mathbb{N}}$  are a family  $\text{SLE}_4(\underline{\rho}^L; \underline{\rho}^R)$  processes as above (with the same c, C), then they satisfy Condition 5.11. Here, we know that the processes are generated by continuous curves, due to the results of [MS16a, WW16].

Note that these processes correspond to level lines of  $(h + F_n)_{n \in \mathbb{N}}$  for h a zero boundary GFF, where Condition (5.9) means that the  $F_n$ 's are uniformly bounded (lying in (-C, C)) and satisfy, for all  $n \ge 0$ ,

$$F_n(x) \le \lambda - c, \quad x < 0; \quad F_n(x) \ge -\lambda + c, \quad x \ge 0.$$

These are the same conditions we require on F in Theorem 5.2. Therefore, the tactic will be to approximate such an F by piecewise constant functions  $F_n$  on  $\mathbb{R}$ , and show that the laws of the corresponding  $\text{SLE}_4(\underline{\rho}^{L,n}; \underline{\rho}^{R,n})$  processes converge weakly using Proposition 5.12. This limiting law will be our candidate for the level line of h + F.

**Lemma 5.37.** Suppose that  $(\gamma^{(n)})_{n \in \mathbb{N}}$  are a family of  $\text{SLE}_4(\underline{\rho}^L, \underline{\rho}^R)$  processes satisfying Condition (5.9) for some  $c > 0, C < \infty$  and all n. Then they satisfy Condition 5.11.

Proof. Recall, we would like to show that our family satisfies a conformal bound on an unforced crossing. That is, that there exists a constant M > 0, such that for any of our processes  $\gamma^{(n)}$ , any stopping time  $\tau$  and any avoidable quadrilateral of  $H_{\tau} = \mathbb{H} \setminus K_{\tau}^{(n)}$  whose modulus m(Q) is greater than M,

$$\mathbb{P}\left(\gamma^{(n)}[\tau,\infty)\operatorname{crosses} Q \mid \gamma^{(n)}[0,\tau]\right) \le 1/2.$$

Here  $(K_t^{(n)}, t \ge 0)$  denotes the sequence of hulls generated by  $\gamma^{(n)}$ .

For  $n \geq 0$ , the law of  $\gamma^{(n)}$  is that of an  $\text{SLE}_4(\underline{\rho}^{L,n}; \underline{\rho}^{R,n})$  process with force points located at  $(\underline{x}^{L,n}; \underline{x}^{R,n})$ . Denote its driving function by  $W^{(n)}$ , its sequence of conformal mappings by  $g^{(n)}$ , and set  $f^{(n)} = g^{(n)} - W^{(n)}$ . By the results of [WW16], we know that  $\gamma^{(n)}$  can be coupled with h a zero boundary GFF in  $\mathbb{H}$ , as the level line of  $h + F_n$ , for

$$F_{n}(x) = \begin{cases} -\lambda \left( 1 + \sum_{\{i:x_{i}^{L,n} \ge x\}} \rho_{i}^{L,n} \right), & x < 0\\ \lambda \left( 1 + \sum_{\{i:x_{i}^{R,n} \le x\}} \rho_{i}^{R,n} \right), & x \ge 0. \end{cases}$$

Moreover, for any stopping time  $\tau$ , we know by the domain Markov property that, conditionally on  $\gamma^{(n)}[0,\tau]$ , the curve evolves from time  $\tau$  onwards as a level line of a GFF with boundary conditions  $\eta_{\tau}^{(n)}$  in the remaining domain. Here  $\eta^{(n)}$  is defined corresponding to  $F_n$  as in Definition 5.1. The important thing to notice is that, as a result of the condition (5.9), we have

$$\eta_{\tau}^{(n)} \circ (f_{\tau}^{(n)})^{-1} \ge -C, \quad \text{on} \ (-\infty, 0); \quad \eta_{\tau}^{(n)} \circ (f_{\tau}^{(n)})^{-1} \ge -\lambda + c \quad \text{on} \ [0, \infty)$$

for any  $\tau$  and n. Therefore, if we set

$$G_1(x) := \begin{cases} -C, & x < 0\\ -\lambda + c, & x \ge 0 \end{cases}$$

we have that

$$G_1 \le \eta_\tau^{(n)} \circ (f_\tau^{(n)})^{-1}, \quad \text{on } \mathbb{R}.$$

Similarly if we set

$$G_2(x) := \begin{cases} \lambda - c, & x < 0 \\ C, & x \ge 0 \end{cases}$$

then

$$G_2 \ge \eta_\tau^{(n)} \circ (f_\tau^{(n)})^{-1}, \quad \text{on } \mathbb{R}.$$

Now, consider an avoidable topological quadrilateral Q of  $H_{\tau}$ . The avoidability assumption means that, when we map it to  $\mathbb{H}$  via  $f_{\tau}^{(n)}$ , its image Q' is a topological quadrilateral in  $\mathbb{H}$  as in Definition 5.9 with  $S'_1$ ,  $S'_3$  (the arcs touching the boundary) either both lying in  $[0, \infty)$ , or both in  $(-\infty, 0]$ .

Suppose we are in the first case. We would like to bound above the probability of  $\gamma^{(n)}[\tau,\infty)$  crossing Q, where Q has modulus greater than M for some positive M. Equivalently, we must bound the probability of  $f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty))$  crossing Q', noting by conformal invariance that Q' also has modulus greater than M. If Q' = $(V', (S'_k)_{1 \leq k \leq 4})$ , we let  $Q'' = (V'', S''_0, S''_2)$  be the doubly connected domain where V''is the interior of the closure of  $V' \cup V' *$  (V' \* the reflection of V' in the real line) and  $S''_0, S''_2$  are it's inner and outer boundary. Following the arguments in the proof of [KS16, Theorem 1.10], we let  $x = \min(\mathbb{R} \cap S''_0) > 0$  and  $r = \max\{|z-x| : z \in S''_0\} > 0$ . We see that Q'' is a doubly connected domain separating x and a point on  $\partial B(x, r)$ from 0 and  $\infty$  (see Figure 29.) However, [Ahl73, Thoerem 4.7] tells us that among all such domains, the one with the largest modulus (here defined as the extremal length of the curve family connecting  $S''_0$  and  $S''_2$  in V'', which satisfies m(Q'') = m(Q')/2) is the domain formed by removing  $(-\infty, 0] \cup [x, x + r]$  from the complex plane. This modulus is also calculated explicitly in [Ahl73] and so we may deduce that

$$\exp(2\pi m(Q'')) \le 16\left(\frac{x}{r}+1\right).$$

Since

$$m(Q'') = \frac{m(Q)}{2} \ge \frac{M}{2},$$

this means that  $r \leq \nu x$  for

$$\nu = \left(\frac{1}{16}\exp(\pi M) - 1\right)^{-1}.$$
(5.10)

Note that  $\nu$  can be made as small as we like by choosing M large.



Figure 29: Since Q'' separates x and a point on  $\partial B(x, r)$  from  $\infty$ , we obtain a lower bound on m(Q'') = m(Q)/2. Since  $m(Q) \ge M$  this gives us an upper bound on r/x. Moreover, we know that for a curve to cross Q it must necessarily intersect  $\overline{B(x, r)}$ .

It is also clear that for  $f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty))$  to cross Q', it must necessarily intersect  $\overline{B(x,r)}$ . However, the law of  $f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty))$  is that of the level line of  $\tilde{h} + \eta_{\tau}^{(n)} \circ (f_{\tau}^{(n)})^{-1}$ , for  $\tilde{h}$  a zero boundary GFF in  $\mathbb{H}$ . By the monotonicity result Corollary 5.36, we see that this level line lies to the left of the level line of  $\tilde{h} + G_1$  almost surely (see Figure 30.) Thus, the probability of  $f_{\tau}^{(n)}(\gamma^{(n)}[0,\tau])$  intersecting  $\overline{B(x,r)}$  is less than the probability of an  $\mathrm{SLE}_4(\rho^L;\rho^R)$  process with

$$\rho^L = -1 + \frac{C}{\lambda}; \quad \rho^R = -2 + \frac{c}{\lambda} \tag{5.11}$$

(left and right force points at the origin), intersecting it.

Therefore, we have

$$\mathbb{P}\left(\gamma^{(n)}[\tau,\infty) \operatorname{crosses} Q \mid \gamma^{(n)}[0,\tau]\right) \\
\leq \mathbb{P}\left(f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty)) \operatorname{crosses} Q' \mid \gamma^{(n)}[0,\tau]\right) \\
\leq \mathbb{P}\left(f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty)) \operatorname{intersects} \overline{B(x,r)} \mid \gamma^{(n)}[0,\tau]\right) \\
\leq \mathbb{P}\left(\operatorname{SLE}_{4}(\rho^{L};\rho^{R}) \operatorname{intersects} \overline{B(x,r)}\right) \quad (\rho^{L},\rho^{R} \text{ are defined in Equation (5.11)}) \\
= \mathbb{P}\left(\operatorname{SLE}_{4}(\rho^{L};\rho^{R}) \operatorname{intersects} \overline{B(1,r/x)}\right) \quad (\nu \text{ is defined in Equation (5.10)}) \\
\leq \mathbb{P}\left(\operatorname{SLE}_{4}(\rho^{L};\rho^{R}) \operatorname{intersects} \overline{B(1,\nu)}\right). \quad (\nu \text{ is defined in Equation (5.10)})$$



Figure 30: The boundary values of  $h+F_n$  given  $\gamma^{(n)}[0,\tau]$  are marked in the left panel. Thus  $f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty])$  is the level line of a zero boundary GFF  $\tilde{h}$  + boundary data as depicted in the right panel. By monotonicity, it must therefore lie to the left of the level line of  $\tilde{h}+G_1$  (marked in red.) Consequently, the probability that  $f_{\tau}^{(n)}(\gamma^{(n)}[\tau,\infty])$  intersects  $\overline{B(x,r)}$  is less than the probability that the red level line does.

Since we know that the  $\text{SLE}_4(\rho^L; \rho^R)$  process with left and right force points at 0 almost surely does not hit the point 1 (in fact, there is exact estimate on this event, see for instance [MW16, Theorem 1.8]), we see that by choosing M large enough, and so  $\nu$  small enough, we can make the right hand side less than 1/2. Thus there exists an M such that the left hand side is bounded above uniformly by 1/2 whenever  $m(Q) \ge M$ .

For the second case, when the boundary arcs  $S'_1, S'_3$  of Q' both lie on the negative real line, we may use symmetrical arguments, replacing  $G_1$  by  $G_2$ .

**Corollary 5.38.** Suppose that  $(\gamma^{(n)})_{n\in\mathbb{N}}$  are a family of  $\operatorname{SLE}_4(\underline{\rho}^L;\underline{\rho}^R)$  processes satisfying Condition (5.9) for all n. Suppose further than they are all parameterised by half plane capacity and that  $(W^{(n)})_{n\in\mathbb{N}}$  are the corresponding family of driving functions. Then

- (W<sup>(n)</sup>)<sub>n∈ℕ</sub> is tight in the metrisable space of continuous functions on [0,∞) with the topology of uniform convergence on compact subsets of [0,∞).
- (γ<sup>(n)</sup>)<sub>n∈ℕ</sub> is tight in the metrisable space of continuous functions on [0,∞) with the topology of uniform convergence on the compact subsets of [0,∞).

Moreover, if the sequence converges weakly in either of the topologies above, then it also converges weakly in the other and the limits agree in the sense that the law of the limiting random curve is the same as the that of the random curve generated under the law of the limiting driving process.

*Proof.* This is a direct consequence of Proposition 5.12 and the remarks in Theorem 1.10 of [KS16].  $\Box$ 

#### 5.6 Existence of the coupling—proof of Theorem 5.2

In this section we will show existence of the coupling described by Theorem 5.2. Recall we would like to prove that for F on  $\mathbb{R}$  which is regulated, so can be approximated uniformly by piecewise constant functions changing value only finitely many times, and which satisfies Condition (5.2), there exists a coupling of a Loewner chain K with a zero boundary GFF h, such that K is a level line of h + F. Moreover, we will show that K is almost surely generated by a continuous and transient curve  $\gamma$ .

To do this we will take a sequence of piecewise constant functions  $F_n$  (changing value only finitely many times), which uniformly approximate F, and consider the level lines, denoted by  $\gamma^{(n)}$ , of  $h + F_n$  for a zero boundary GFF h. Observe that we can choose the  $F_n$  so that the level lines are a family of  $\text{SLE}_4(\underline{\rho}^L; \underline{\rho}^R)$  processes satisfying the conditions of Corollary 5.38. Thus, the tightness given by the corollary will allow us to extract a subsequential limit.

**Proposition 5.39.** Let F satisfy the conditions of Theorem 5.2. Suppose that  $(F_n)_{n \in \mathbb{N}}$  are piecewise constant functions on  $\mathbb{R}$ , changing value only finitely many times. Let h be a zero boundary GFF and  $\gamma^{(n)}$  be the level line of  $h + F_n$  for each n. Suppose further that they are all parameterized by half plane capacity and that  $(W^{(n)})_{n \in \mathbb{N}}$  are the corresponding family of driving functions.

Then, if the  $(F_n)$  converge uniformly to F on  $\mathbb{R}$ , we have that:

- There exists a subsequence of the γ<sup>(n)</sup> which converges weakly in the space of continuous functions on [0,∞) with the topology of uniform convergence on compact subsets of [0,∞).
- (2) The limiting law describes a continuous curve from 0 to  $\infty$  in  $\mathbb{H}$  which generates a Loewner chain with a.s. continuous driving function.
- (3) The limiting curve can be coupled with a zero boundary GFF h, as a level line of h + F.

**Remark 5.40.** We will later see that this limiting law does not depend on the choice of approximation, as any continuous curve which can be coupled with a zero boundary GFF as a level line of h + F must have a unique law: see Remark 5.44. In particular, this tells us that we actually have convergence of the whole sequence in distribution.

Proof of Theorem 5.2. Theorem 5.2 is a direct consequence of Proposition 5.39.  $\Box$ 

Proof of Proposition 5.39, Items (1), (2). Note that the weak convergence directly follows from Corollary 5.38, as does the fact that the limiting law corresponds to a

continuous curve generating a Loewner chain with almost surely continuous driving function.  $\hfill \Box$ 

**Definition 5.41.** Suppose that F is  $L^1$  with respect to harmonic measure on  $\mathbb{R}$  and that  $\gamma$  is a continuous curve with continuous Loewner driving function. We set  $\eta_t^0$  in the same way as in Definition 5.1. Then for any  $z \in \mathbb{H}$  we can define, for t less than the first time that  $\gamma$  swallows z,

$$\eta_t(F,\gamma,z) = \eta_t^0(f_t(z))$$

as in Definition 5.1, emphasising the dependence on F and  $\gamma$ . Let

$$C_t(\gamma, z) = \log \operatorname{CR}(z, \mathbb{H}) - \log \operatorname{CR}(z, \mathbb{H} \setminus K_t),$$

$$\tilde{\eta}_t(F,\gamma,z) = \eta_{\tau(t)}(F,\gamma,z), \quad \text{where } \tau(t) := \inf\{s \ge 0 : C_s(\gamma,z) = t\}.$$

Finally, define  $(\tilde{W}_t(\gamma, z))_{t\geq 0}$  for  $\gamma$  to be the driving function of  $\gamma$  reparameterised by  $C_t(\gamma, z)$ .

To prove Proposition 5.39, Item (3), i.e. to see that the limiting curve can be coupled as a level line in the way we want, we will use Lemma 5.24. This tells us that if we define  $\tilde{\eta}_t(F, \gamma, z)$  as above for our limiting curve  $\gamma$ , we need only show that for each  $z \in \mathbb{H}$ , the process  $(\tilde{\eta}_t(F, \gamma, z), t \ge 0)$  is a Brownian motion with respect to the filtration generated by  $(\tilde{W}_t(\gamma, z), t \ge 0)$ .

**Lemma 5.42.** Let  $(\gamma^{(n_k)})$  be a subsequence of the random curves in Proposition 5.39, parameterised by half plane capacity, which converge weakly to some  $\gamma$  in the space of continuous functions on  $[0, \infty)$  with the topology of uniform convergence on compacts. Then for every  $z \in \mathbb{H}$ ,

$$\left(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z)\right) \xrightarrow{d} \left(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z)\right)$$

in  $C([0,\infty);\mathbb{R}) \times C([0,\infty);\mathbb{R})$  with respect to the product topology of uniform convergence on compacts.

We postpone the proof of Lemma 5.42 and first tell the readers how we obtain Proposition 5.39 from Lemma 5.42.

**Lemma 5.43.** Let  $(\gamma^{(n_k)})$  be a subsequence of the random curves in Proposition 5.39, parameterised by half plane capacity, which converge weakly to some  $\gamma$  in the space of continuous functions on  $[0, \infty)$  with the topology of uniform convergence on

compacts. Then for every  $z \in \mathbb{H}$ ,

$$\left(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F_{n_k}, \gamma^{(n_k)}, z)\right) \xrightarrow{d} \left(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z)\right)$$

in  $C([0,\infty);\mathbb{R}) \times C([0,\infty);\mathbb{R})$  with respect to the product topology of uniform convergence on compacts.

*Proof.* By Lemma 5.42, we have that

$$\left(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z)\right) \xrightarrow{d} \left(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z)\right)$$
(5.12)

with respect to the product topology of uniform convergence on compacts. It is also clear that, for all t, z and any curve  $\gamma'$ 

$$\left|\tilde{\eta}_t(F_{n_k},\gamma',z) - \tilde{\eta}_t(F,\gamma',z)\right| \le \sup_{x \in \mathbb{R}} |F(x) - F_{n_k}(x)|.$$

Indeed,  $\tilde{\eta}_t(F_{n_k}, \gamma', \cdot)$  and  $\tilde{\eta}_t(F, \gamma', \cdot)$  are by definition harmonic extensions of functions whose boundary values differ by at most the right hand side. Since  $\sup_{x \in \mathbb{R}} |F(x) - F_{n_k}(x)| \to 0$  by assumption, we may conclude that, for any T > 0, almost surely as  $k \to \infty$ ,

$$\sup_{t \in [0,T]} \left| \tilde{\eta}_t(F_{n_k}, \gamma^{(n_k)}, z) - \tilde{\eta}_t(F, \gamma^{(n_k)}, z) \right| \to 0.$$
 (5.13)

Combining Equations (5.12) and (5.13), we obtain the conclusion.

Proof of Proposition 5.39, Item (3). Fix  $z \in \mathbb{H}$ . Since  $\gamma^{(n_k)}$  is coupled as a level line of  $h + F_{n_k}$  we know by Lemma 5.24 that  $(\tilde{\eta}_t(F_{n_k}, \gamma^{(n_k)}, z), t \ge 0)$  is a Brownian motion for each k, with respect to the filtration of  $(\tilde{W}_t(\gamma^{(n_k)}, z), t \ge 0)$ . Therefore, by the weak convergence in Lemma 5.43, we have that if  $\gamma$  is the limiting law of the  $\gamma^{(n_k)}$ 's, the process  $\tilde{\eta}_t(F, \gamma, z)$  must also have the law of Brownian motion, with respect to the filtration of  $(\tilde{W}_t(\gamma, z), t \ge 0)$ . Applying Lemma 5.24 again proves the proposition.

Proof of Lemma 5.42. Fix  $z \in \mathbb{H}$ . We will show that the laws of  $(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z))$  converge weakly in k to the law of  $(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z))$ . To do this, we begin by showing that this family of laws is tight in  $C([0,\infty);\mathbb{R}) \times C([0,\infty);\mathbb{R})$  with respect to the product topology of uniform convergence on compacts. This allows us to extract a further subsequence along which the  $(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z))$ 's converge. We then argue that the limit of this subsequence must be equal to that of  $(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z))$ , so in fact our whole original subsequence converged, and the limit is  $(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z))$ . Note that the proof of this lemma would be trivial if  $(\tilde{W}(\cdot, z), \tilde{\eta}(F, \cdot, z))$  was a continuous function on the set of curves, however, this is not quite the case. It is essentially a continuous function when restricted to a set in which the  $\gamma^{(n_k)}$ 's lie with high probability.

By the proof of [KS16, Theorem 1.5], we know that for every M > 0 we can find a subset E of the space of continuous curves in  $\mathbb{H}$  such that

$$\inf_{k} \mathbb{P}(\gamma^{(n_k)} \in E) \ge 1 - \frac{1}{M}$$
(5.14)

when the  $(\gamma^{(n_k)})$  are parameterised by half plane capacity, and

- E is relatively compact with respect to the topology of uniform convergence on compacts,
- curves in *E* correspond to Loewner chains with continuous driving functions parameterised by half plane capacity, and
- if a sequence of curves in *E* converges with respect to uniform convergence on compacts, then their driving functions also converge uniformly on compacts along a further subsequence, and the limits agree.

For the construction of such an E, see Section 3.5 of [KS16], in particular the definition (60) and the discussion in the closing paragraphs. See also the opening paragraph of Section 3.6.

We argue that the set  $\{(\tilde{W}(\gamma', z), \tilde{\eta}(F, \gamma', z)) : \gamma' \in E\}$  is a relatively compact subset of  $C([0, \infty); \mathbb{R}) \times C([0, \infty); \mathbb{R})$  with respect to the product topology of uniform convergence on compacts. Thus by (5.14) the laws of the

$$\left(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z)\right)$$

are tight in this topology. It is sufficient to verify the following claim: if  $\gamma'_n \to \gamma'$  is any convergent sequence of curves in E, whose driving functions also converge uniformly on compacts, then for any T > 0, as  $n \to \infty$ ,

$$\sup_{t\in[0,T]} \left| \tilde{\eta}_t(F,\gamma'_n,z) - \tilde{\eta}_t(F,\gamma',z) \right| \to 0$$
(5.15)

and

$$\sup_{t\in[0,T]} \left| \tilde{W}_t(\gamma'_n, z) - \tilde{W}_t(\gamma', z) \right| \to 0.$$
(5.16)

Relative compactness then follows because the choice of E means that any sequence of curves in E has a convergent subsequence along which the driving functions also converge.

We will prove the above claim now. We let  $K_t$  (resp.  $K_t^n$ ) be the hull generated by  $\gamma'$  (resp.  $\gamma'_n$ ) in the capacity parameterisation and  $W_t, g_t$  (resp.  $W_t^n, g_t^n$ ) be the corresponding driving functions, and functions  $\mathbb{H} \setminus K_t$  (resp.  $\mathbb{H} \setminus K_t^n$ ) to  $\mathbb{H}$ , normalised at  $\infty$ . We define  $f_t = g_t - W_t$  and  $f_t^n = g_t^n - W_t^n$  as usual, and consider these to be extended to the boundary, also writing  $f_t(0^+)$  for  $V_t^R(0^+) - W_t$ . Write

$$C_t = C_t(\gamma', z), \quad C_t^n = C_t(\gamma'_n, z);$$
  
$$\tau(t) := \inf\{s \ge 0 : C_s = t\}, \quad \tau^n(t) := \inf\{s \ge 0 : C_s^n = t\}.$$

First, we will show that for any T > 0 before the first time that  $\gamma'$  swallows z, as  $n \to \infty$ ,

$$\sup_{t \in [0,T]} |C_t - C_t^n| \to 0.$$
(5.17)

We have the following observations.

• By Lemma 5.23, and since  $C_0 = C_0^n = 0$ , we have

$$C_t = \int_0^t \frac{-4\Im(f_s(z))^2}{|f_s(z)|^4} ds; \quad C_t^n = \int_0^t \frac{-4\Im(f_s^n(z))^2}{|f_s^n(z)|^4} ds.$$

- $W_t^n \to W_t$  uniformly on [0, T].
- $g_t^n \to g_t$  uniformly on  $\{(t, z) \in [0, T] \times \overline{\mathbb{H}} : d(z, K_t) > \delta\}$  for any  $\delta > 0$ . (See for instance [KS16, Lemmas A.3 and A.4])

Combining these three facts, we obtain Equation (5.17).

Second, we show that, for any T > 0 before  $\gamma'$  swallows z, as  $n \to \infty$ ,

$$\sup_{t \in [0,T]} \left| \eta_{\tau^n(t)}(F,\gamma',z) - \eta_{\tau(t)}(F,\gamma',z) \right| \to 0.$$
(5.18)

By Equation (5.17), we have that  $[0, \tau^n(T) \vee \tau(T)] \subset [0, \tau(S)]$  for *n* large enough, where S > T, and  $\tau(S)$  is a time before *z* is swallowed by  $\gamma'$ . By (5.17) again, we therefore have that, as  $n \to \infty$ 

$$c_n := \sup_{t \in [0,T]} |C_{\tau^n(t)} - t| \le \sup_{t \in [0,\tau^n(T) \lor \tau(T)]} |C_t - C_t^n| \to 0.$$

Since

$$\sup_{t \in [0,T]} \left| \eta_{\tau^{n}(t)}(F,\gamma',z) - \eta_{\tau(t)}(F,\gamma',z) \right| \le \sup_{s,t \in [0,S], |s-t| \le c_{n}} \left| \tilde{\eta}_{t}(F,\gamma',z) - \tilde{\eta}_{s}(F,\gamma',z) \right|,$$

and  $\tilde{\eta}_t(F, \gamma', z)$  is uniformly continuous on [0, S], we see that it must converge to 0.

Third, we show that, for any T > 0 before  $\gamma'$  swallows z, as  $n \to \infty$ ,

$$\sup_{t \in [0,T]} \left| \eta_{\tau^n(t)}(F,\gamma'_n,z) - \eta_{\tau^n(t)}(F,\gamma',z) \right| \to 0.$$
(5.19)

We need only show that, on any time interval [0, S] such that S is strictly less than the time  $\gamma'$  swallows z, the quantity  $|\eta_t(F, \gamma'_n, z) - \eta_t(F, \gamma', z)|$  converges uniformly to 0. We have the following observations.

- By Definition 5.41, we know that  $\eta_t(F, \gamma', \cdot)$  (resp.  $\eta_t(F, \gamma'_n, \cdot)$ ) is the bounded harmonic function with boundary values equal F on  $\mathbb{R} \setminus K_t$  (resp. on  $\mathbb{R} \setminus K_t^n$ ),  $-\lambda$  on the left side of  $K_t$  (resp.  $K_t^n$ ), and  $\lambda$  on the right side of  $K_t$  (resp.  $K_t^n$ ).
- $W_t^n \to W_t$  uniformly on [0, S].
- $g_t^n \to g_t$  uniformly on  $\{(t, z) \in [0, S] \times \overline{\mathbb{H}} : d(z, K_t) > \delta\}$  for any  $\delta > 0$ . Same reason as above.

Combining these three facts, we have that the quantity  $|\eta_t(F, \gamma'_n, z) - \eta_t(F, \gamma', z)|$ converges uniformly to 0 on  $t \in [0, S]$ , implying Equation (5.19).

Combining Equations (5.17), (5.18) and (5.19), we obtain Equation (5.15) by noting that

$$\sup_{t \in [0,T]} \left| \eta_{\tau^{n}(t)}(F,\gamma'_{n},z) - \eta_{\tau(t)}(F,\gamma',z) \right| \leq \sup_{t \in [0,T]} \left| \eta_{\tau^{n}(t)}(F,\gamma'_{n},z) - \eta_{\tau^{n}(t)}(F,\gamma',z) \right| + \sup_{t \in [0,T]} \left| \eta_{\tau^{n}(t)}(F,\gamma',z) - \eta_{\tau(t)}(F,\gamma',z) \right|.$$
(5.20)

We obtain Equation (5.16) by the same method as above, which is much simpler in this case, and so we omit the details.

Finally, we show that if  $(\gamma^{(n_k)})_{k\in\mathbb{N}}$  converges weakly, and there exists a further subsequence along which  $(\tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z))$  converges, then the limit must be  $(\tilde{W}(\gamma, z), \tilde{\eta}(F, \gamma, z))$ . To do this, for any  $M \in \mathbb{N}$  take E relatively compact such that (5.14) holds, and note that by the above claim we have that

$$A_E := \left\{ (\gamma', \tilde{W}(\gamma', z), \tilde{\eta}(F, \gamma', z)) : \gamma' \in E \right\}$$

is relatively compact in  $C([0,\infty);\mathbb{C}) \times C([0,\infty);\mathbb{R}) \times C([0,\infty);\mathbb{R})$ , and its closure is equal to

$$\left\{(\gamma',\tilde{W}(\gamma',z),\tilde{\eta}(F,\gamma',z)):\gamma'\in\overline{E}\right\}.$$

This means that the joint laws of  $(\gamma^{(n_k)}, \tilde{W}(\gamma^{(n_k)}, z), \tilde{\eta}(F, \gamma^{(n_k)}, z))$  are also tight,

and thus we can extract an even further subsequence along which we have joint convergence. If  $\mathbb{P}^*$  is the law of this joint limit then,

$$\mathbb{P}^*\left(\overline{A_E}\right) \ge \inf_k \mathbb{P}\left(\gamma^{(n_k)} \in \overline{E}\right) \ge 1 - \frac{1}{M}$$

and so we see that the probability of our marginal laws agreeing in the sense we want must be greater than  $1 - \frac{1}{M}$ . Since this holds for every M, agreement must hold almost surely, and as these marginal laws are equal to the limiting laws of the individually convergent sequences, the result follows.

# 5.7 Proof of Theorems 5.3 to 5.5

Proof of Theorem 5.4. Suppose that  $\gamma_F$  and  $\gamma_G$  are continuous transient curves from 0 to  $\infty$  in  $\mathbb{H}$ , coupled with a zero-boundary GFF h as level lines of h + F and h + G respectively. Suppose further that  $\gamma'_G$  is a continuous transient curve from  $\infty$  to 0 and is coupled with h as a level line of -h - G from  $\infty$  to 0, such that the four objects  $h, \gamma_F, \gamma_G, \gamma'_G$  are coupled with  $\gamma_F, \gamma_G, \gamma'_G$  are conditionally independent given h. From Theorem 5.2, we have the existence of  $\gamma_F, \gamma_G$  and  $\gamma'_G$ . By Lemma 5.35, we know that  $\gamma_F$  stays to the left of  $\gamma_G$  almost surely.

Proof of Theorems 5.3 and 5.5. Suppose that  $\gamma_F$  is a continuous transient curve which is coupled with h as a level line of h + F from 0 to  $\infty$ , as in Theorem 5.2. Let  $\gamma'_F$  be a continuous curve coupled with h as a level line of -h - F from  $\infty$  to 0, such that  $\gamma_F$  and  $\gamma'_F$  are conditionally independent given h. The existence of  $\gamma'_F$ is given by Theorem 5.2. Lemma 5.34 then tells us that  $\gamma_F = \gamma'_F$  almost surely. In particular,  $\gamma_F$  is almost surely determined by h.

**Remark 5.44.** By applying Theorem 5.3, we see that if  $\gamma$  is the weak limit of any sequence of level lines as in Proposition 5.39, then  $\gamma$  can be coupled as the level line of a GFF and is moreover determined by the GFF in this coupling. Thus, the law of  $\gamma$  is uniquely determined. In particular, it does not depend on the sequence of approximating level lines.

**Lemma 5.45.** Let F be as in Theorem 5.2. Suppose that  $F_n \downarrow F$  approximate F uniformly on the real line, where the  $F_n$  are decreasing, and are piecewise constant with value changing only finitely many times.

Let h be a zero boundary GFF in  $\mathbb{H}$ ,  $\gamma_n$  be the level line of  $h + F_n$  for each n, and  $\gamma$  be the level line of h + F. Denote by  $H_n$  the open sets corresponding to the strict right hand sides of  $\gamma_n$ . By monotonicity these are almost surely decreasing. Define

$$H = \cap_n \overline{H_n}.$$

Then  $\partial H$  coincides with  $\gamma$  almost surely. In other words, the sequence of curves  $\gamma_n$  converges to  $\gamma$  almost surely.

*Proof.* First, we show that  $\partial H$  has the same law as  $\gamma_F$ . We use a conformal mapping to take everything to the unit disc, as it will be more convenient to work in a space where our sets are compact. We endow  $\mathbb{H}$  with the metric it inherits from the unit disc  $\mathbb{U}$  via the map  $\varphi(z) = (z - i)/(z + i)$ . Namely, let  $d_*(\cdot, \cdot)$  denote the metric on  $\mathbb{H}$  given by

$$d_*(z,w) = |\varphi(z) - \varphi(w)|.$$

We write  $\overline{\mathbb{H}}$  for the completion of  $\mathbb{H}$  with respect to  $d_*$ . For compact sets  $A, B \subset \overline{\mathbb{H}}$ , we have the  $d_*$ -induced Hausdorff distance

$$d^{H}_{*}(A,B) = \inf\{\varepsilon > 0 : A \subset B^{(\varepsilon)}, B \subset A^{(\varepsilon)}\},\$$

where  $A^{(\varepsilon)}$  denotes the open  $\varepsilon$ -neighborhood of A with respect to the metric  $d_*$ . Note that  $d_*^H$  makes the set of all non-empty compact subsets of  $\overline{\mathbb{H}}$  (with metric  $d_*$ ) into a compact metric space. We have the following observations.

- The sets  $\overline{H_n}$  form an almost surely decreasing sequence of compact subsets of  $\overline{\mathbb{H}}$ , which therefore converge to H with respect to  $d_*^H$ . This implies that  $\gamma_n = \partial H_n$  almost surely converges to  $\partial H$  with respect to  $d_*^H$ .
- By the assumptions on F<sub>n</sub>, we know that the laws of γ<sub>n</sub> fall in to the framework of Proposition 5.39. This means that we can extract a subsequence which converges weakly in the space of continuous functions on [0,∞) with respect to uniform convergence on compacts. Moreover, the limiting curve can be coupled with a zero boundary GFF h̃ as the level line of h̃ + F. Furthermore, the subsequence converges weakly, to the same limit, in the space of curves from [0, 1] → H̃ with respect to the topology of uniform convergence modulo reparameterisation, where the metric on H̃ is given by d<sub>\*</sub>. This requires a slight extension of Proposition 5.12, which was stated here, but is nonetheless still true, by the extended version given in [KS16, Corollary 1.7]. By continuity, we therefore have that along this subsequence the curves also converge weakly to the same limit with respect to d<sup>H</sup><sub>\*</sub>. Thus ∂H has the law of a continuous curve which can be coupled with a zero boundary GFF h̃ as the level line of h̃ + F.
- By Theorem 5.3, we know that the law on continuous curves which can be coupled with a GFF  $\tilde{h}$  as a level line of  $\tilde{h} + F$ , is unique.

Combining these three facts, we may conclude that  $\partial H$  has the same law as  $\gamma_F$ .

Next, we show that  $\partial H$  coincides with  $\gamma$  almost surely. We have the following observations.

- By the above analysis, we know that  $\partial H$  has the same law as  $\gamma$ .
- By Theorem 5.4, we know that  $\partial H$  lies to the left of  $\gamma$  almost surely.

Combining these two facts, we obtain that  $\partial H$  coincides with  $\gamma$  almost surely.  $\Box$ 

## 5.8 Proof of Theorem 5.6 and concluding remarks

In this section, we prove Theorem 5.6: the key ingredient being the proof of Lemma 5.46. This lemma is proved in [MS16a, WW16] for  $SLE_4(\rho)$  process when  $\rho$  is a vector. The proof given in these papers will work with minor modifications for the case when  $\rho$  is a Radon measure but, to be self-contained, we still give a complete proof here.

**Lemma 5.46.** Suppose we are given a random continuous curve in  $\overline{\mathbb{H}}$  from 0 to  $\infty$  whose Loewner driving function W is almost surely continuous. If  $(\rho^L; \rho^R)$  are a pair of finite Radon measures on  $\mathbb{R}_-, \mathbb{R}_+$  and F is the corresponding function of bounded variation, define  $(\eta_t, t \ge 0)$  as in Definition 5.1. For  $z \in \mathbb{H}$  and  $t \ge 0$ , define

$$\tau(t) = \inf\{s : \log \operatorname{CR}(z, \mathbb{H}) - \log \operatorname{CR}(z, \mathbb{H} \setminus K_s) = t\}.$$

Then  $(W, (V^L(x))_{x \in \mathbb{R}_-}, (V^R(x))_{x \in \mathbb{R}_+})$  can be coupled with a standard Brownian motion to describe an  $\mathrm{SLE}_4(\rho^L; \rho^R)$  process if  $(\eta_{\tau(t)}(z), t \ge 0)$  evolves as a Brownian motion with respect to the filtration generated by  $(W_{\tau(t)}, t \ge 0)$  for any  $z \in \mathbb{H}$ .

Proof. Suppose that  $(\eta_{\tau(t)}(z), t \ge 0)$  is a Brownian motion with respect to the filtration generated by  $(W_{\tau(t)}, t \ge 0)$  for each  $z \in \mathbb{H}$ . This implies that  $(\eta_t(z), t \ge 0)$  is a local martingale with respect to the filtration generated by  $(W_t, t \ge 0)$ . Our first step will be to show that  $W_t$  is a continuous semi-martingale. By the definition of  $\eta_t(\cdot)$ , we know that, for each  $z \in \mathbb{H}$ ,

$$2\eta_t(z) = -\int_{\mathbb{R}_-} \arg(g_t(z) - V_t^L(x)) \,\rho^L(dx) - \arg(g_t(z) - W_t) \\ + (\pi - \arg(g_t(z) - W_t)) + \int_{\mathbb{R}_+} \left(\pi - \arg(g_t(z) - V_t^R(x))\right) \,\rho^R(dx).$$
(5.21)

This follows from the integration by parts formula for functions of bounded variation, and the integral expression for the harmonic extension of a bounded function on the real line. Note here that the integrals are well defined, since for each fixed t, z the integrands are continuous, bounded functions in x, and  $\rho^L, \rho^R$  are assumed to be finite measures. Indeed,  $g_t(z)$  and  $(V_t^{L,R}(x))_{x\in\mathbb{R}}$  are adapted and differentiable, and we may also differentiate under the integral in (5.21) by finiteness of  $\rho^L, \rho^R$ . Therefore, we can deduce that all the terms in (5.21) apart from the only one,  $\arg(g_t(z) - W_t)$ , involving  $W_t$ , are semi-martingales. Since  $\eta_t(z)$  is itself a local martingale, this means that  $\arg(g_t(z) - W_t)$  must also be a semi-martingale. Now, note that by Schwartz's formula, we can write  $\log(g_t(z) - W_t)$ , up to a constant, as a linear functional (an integral against a test function) of  $\arg(g_t(z) - W_t)$ . So  $\log(g_t(z) - W_t)$  is also a semi-martingale, and thus it's exponential, and consequently  $W_t$  itself, must be a semi-martingale also. Hence we can write  $W_t := M_t - V_t$ for M a local martingale and V of bounded variation.

Substituting this into the expression (5.21) we see that, on intervals where  $W_t$  does not collide with the  $V_t^{L,R}$ , the drift of  $2\eta_t$  is equal to the imaginary part of

$$\frac{\int_{\mathbb{R}_{-}} \frac{2\rho^{L}(dx)}{V_{t}^{L}(x) - W_{t}}}{g_{t}(z) - W_{t}} dt + \frac{-2dV_{t}}{(g_{t}(z) - W_{t})} + \frac{d\langle W_{t} \rangle - 4dt}{(g_{t}(z) - W_{t})^{2}} + \frac{\int_{\mathbb{R}_{+}} \frac{2\rho^{R}(dx)}{V_{t}^{R}(x) - W_{t}}}{g_{t}(z) - W_{t}} dt,$$

which of course must vanish. Therefore, multiplying by  $(g_t(z) - W_t)^2$  and evaluating at z such that  $g_t(z) - W_t$  is arbitrarily close to 0, we can deduce that  $d\langle W_t \rangle = 4dt$ . On subsequently removing the third term, we also find an expression for  $dV_t$ , and can conclude that  $W_t$  satisfies (5.3) in Definition 5.20 on intervals where  $W_t$  does not collide with the  $V_t^{L,R}$ .

All that remains is to show that we have instantaneous reflection of  $W_t$  off the  $V_t^{L,R}(x)$ . It suffices to show that the number of times the curve  $\gamma$  hits the real line has Lebesgue measure 0. However, this is always the case for a continuous curve with continuous driving function, which we know for example by [MS16a, Lemma 2.5].

**Remark 5.47.** We believe that Lemma 5.46 could be made into an if and only if statement if we strengthened Definition 5.20 of an  $SLE_{\kappa}(\rho^{L};\rho^{R})$  process to also require that, almost surely,

$$V_t^L(x) = x + \int_0^t \frac{2ds}{V_s^L(x) - W_s}, \quad x \in \mathbb{R}_-; \quad V_t^R(x) = x + \int_0^t \frac{2ds}{V_s^R(x) - W_s}, \quad x \in \mathbb{R}_+$$
(5.22)

and

$$W_t = \sqrt{\kappa}B_t + \int_0^t ds \int_{\mathbb{R}_-} \frac{\rho^L(dx)}{W_s - V_s^L(x)} + \int_0^t ds \int_{\mathbb{R}_+} \frac{\rho^R(dx)}{W_s - V_s^R(x)},$$
(5.23)

as in [WW16] and [MS16a].

That is, using the stronger definition, we could show that any such process can always be coupled with the Gaussian Free Field as generalized level line. This would also give us uniqueness in law for the  $SLE_4(\rho^L; \rho^R)$  process among continuous curves. However, it seems that (5.22) and (5.23) are hard to verify assuming only that  $(\eta_{\tau(t)}(z), t \ge 0)$  evolves as a Brownian motion.

Proof of Theorem 5.6. Combining Theorem 5.2 with Lemmas 5.24 and 5.46 in the case that F is of bounded variation, we know that in the coupling  $(h, \gamma)$  given by Theorem 5.2, the marginal law of  $\gamma$  is that of an  $\text{SLE}_4(\rho^L; \rho^R)$  process. This gives us existence of the process. Moreover, we know the curve  $\gamma$  is almost surely continuous and transient and also satisfies the reversibility property (3) of Theorem 5.6, by Theorem 5.5.

**Remark 5.48.** We can also generalize the construction of flow lines and counterflow lines to GFF with general boundary data. A similar approximating idea works for flow lines and counterflow lines. Since the flow lines and counterflow lines have a duality property, instead of reversibility as for level line case, some extra work is needed for the proof of monotonicity as in Section 5.4. The details are left to interested readers.

**Remark 5.49.** As explained in the introduction, we restrict to boundary values satisfying Condition (5.2) throughout the paper. This condition guarantees that there is no continuation threshold. The continuity of the level lines when there does exist a continuation threshold is still open.

# References

<ul> <li>[ADH13] R. Abraham, J.F. Delmas, and P. Hoscheit. A note on the Gromov-Hausdorff–Prokhorov distance between (locally) compact metric measure spaces. <i>Electronic Journal of Probability</i>, 18, 2013.</li> <li>[AH83] S. Asmussen and H. Hering. <i>Branching processes</i>. Birkhäuser, Boston MA, 1983.</li> <li>[Ah173] L. V. Ahlfors. <i>Conformal invariants: topics in geometric function the ory</i>. McGraw-Hill Book Co., New York, 1973.</li> <li>[Aid15] E. Aidekon. The extremal process in nested conformal loops. <i>Preprint</i> 2015.</li> </ul>	[Aco14]	J. Acosta. Tightness of the recentered maximum of log-correlated Gaussian fields. <i>Elec. Journal Probab.</i> , 19:1–25, 2014.
<ul> <li>[AH83] S. Asmussen and H. Hering. Branching processes. Birkhäuser, Boston MA, 1983.</li> <li>[Ahl73] L. V. Ahlfors. Conformal invariants: topics in geometric function the ory. McGraw-Hill Book Co., New York, 1973.</li> <li>[Aid15] E. Aidekon. The extremal process in nested conformal loops. Preprint 2015.</li> </ul>	[ADH13]	R. Abraham, J.F. Delmas, and P. Hoscheit. A note on the Gromov– Hausdorff–Prokhorov distance between (locally) compact metric mea- sure spaces. <i>Electronic Journal of Probability</i> , 18, 2013.
<ul> <li>[Ahl73] L. V. Ahlfors. Conformal invariants: topics in geometric function the ory. McGraw-Hill Book Co., New York, 1973.</li> <li>[Aid15] E. Aidekon. The extremal process in nested conformal loops. Preprint 2015.</li> </ul>	[AH83]	S. Asmussen and H. Hering. <i>Branching processes</i> . Birkhäuser, Boston, MA, 1983.
[Aid15] E. Aidekon. The extremal process in nested conformal loops. <i>Preprint</i> 2015.	[Ahl73]	L. V. Ahlfors. Conformal invariants: topics in geometric function the- ory. McGraw-Hill Book Co., New York, 1973.
	[Aid15]	E. Aidekon. The extremal process in nested conformal loops. <i>Preprint</i> , 2015.

- [Ald91] D. Aldous. The Continuum Random Tree I. Annals of Probability, 19:1–28, 1991.
- [Ald93] D. Aldous. The Continuum Random Tree III. Annals of Probability, 21:248–289, 1993.
- [ALS17] J. Aru, T. Lupu, and A. Sepúlveda. First passage sets of the 2d continuum GFF. *To appear*, 2017.
- [APS17] J. Aru, E. Powell, and A. Sepúlveda. Approximating the Liouville measure using local sets of the Gaussian free field. *Preprint*, arxiv::1701.05872, 2017.
- [AS14] E. Aïdekon and Z. Shi. Seneta–Heyde rescaling for the branching random walk. *Annals of Probability*, 42, no.3:959–993, 2014.
- [ASW15] J. Aru, A. Sepúlveda, and W. Werner. On bounded type thin local sets (BTLS) of the two-dimensional Gaussian Free Field. *Preprint*, 2015.
- [Bañ99] R. Bañuelos. Intrinsic ultracontractivity and eigenfunction estimates for Schrödinger operators. Journal of Functional Analysis, 100:181– 205, 1999.
- [BBI01] D. Burago, Y. Burago, and S. Ivanov. A course in metric geometry. AMS, Boston MA, 2001.
- [BBS11] J. Berestycki, N. Berestycki, and J. Schweinsberg. Survival of nearcritical branching Brownian motion with absorption. *Journal of Statistical Physics*, 143:833–854, 2011.
- [BBS14] J. Berestycki, N. Berestycki, and J. Schweinsberg. Critical branching Brownian motion with absorption: survival probability. *Probability Theory and Related Fields*, 160:489–520, 2014.
- [Bef08] V. Beffara. The dimension of the SLE curves. Ann. Probab., 36(4):1421–1452, 2008.
- [Ber15a] N. Berestycki. An elementary approach to Gaussian multiplicative chaos. *Preprint*, 2015.
- [Ber15b] N. Berestycki. Introduction to the Gaussian free field and Liouville quantum gravity. Lecture Notes, available on the webpage of the author, 2015.
- [Bil95] P. Billingsley. Probability and Measure. John Wiley and Sons, New York, 3rd edition, 1995.
- [BK77] J.D. Biggins and A.E. Kyprianou. Martingale convergence in the branching random walk. J. Appl. Probability, 14(1):25–37, 1977.

- [BK04] J.D. Biggins and A.E. Kyprianou. Measure changes in multitype branching. *Adv. in Appl. Probab.*, 36:544–581, 2004.
- [BN] N. Berestycki and J. Norris. Lectures on Schramm-Loewner evolutions. Lecture notes, available on the webpages of the authors.
- [CDCH<sup>+</sup>14] D. Chelkak, H. Duminil-Copin, C. Hongler, A. Kempainnen, and S. Smirnov. Convergence of Ising interfaces to Schramm's SLE curves. *C.R. Math. Acad. Sci. Paris*, 352(2):157–161, 2014.
- [CN07] F. Camia and C.M. Newman. Critical percolation exploration path and SLE<sub>6</sub>: a proof of convergence. Probability Theory and Related Fields, 139(3-4):473–519, 2007.
- [CR88] B. Chauvin and A. Rouault. KPP equation and supercritical branching Brownian motion in the subcritical speed area. Application to spatial trees. *Probability Theory and Related Fields*, 80:299–314, 1988.
- [CS12] D. Chelkak and S. Smirnov. Universality in the 2D Ising model and conformal invariance of fermionic observables. *Invent. Math.*, 189(3):515– 580, 2012.
- [Die69] J. Dieudonné. Foundations of modern analysis. Academic Press, New York, 1969.
- [DKRV16] F. David, A. Kupiainen, R. Rhodes, and V. Vargas. Liouville quantum gravity on the Riemann sphere. Communications in Mathematical Physics, 342(3):869–907, 2016.
- [DMS12] B. Duplantier, J. Miller, and S. Sheffield. Liouville quantum gravity as a mating of trees. *Preprint*, 2012.
- [DN80] D. J. Daley and P. Narayan. Series Expansions of Probability Generating Functions and Bounds for the Extinction Probability of a Branching Process, volume 17, no.4. 1980.
- [Doo57] J.L. Doob. Conditioned Brownian motion and the boundary limits of harmonic functions. Bulletin of the Mathematical Society of France, 85:431–458, 1957.
- [DRSV14a] B. Duplantier, R. Rhodes, S. Sheffield, and V. Vargas. Critical Gaussian multiplicative chaos: convergence of the derivative martingale. Ann. Probab., 42(5):1769–1808, 2014.
- [DRSV14b] B. Duplantier, R. Rhodes, S. Sheffield, and V. Vargas. Renormalization of critical Gaussian multiplicative chaos and KPZ relation. Comm. Math. Phys., 330(1):283–330, 2014.

[DS84]	E.B. Davies and B. Simon. Ultracontractivity and heat kernels for Schrödinger operators and Dirichlet Laplacians. <i>Journal of Functional</i> <i>Analysis</i> , 59:335–395, 1984.
[DS11]	B. Duplantier and S. Sheffield. Liouville quantum gravity and KPZ formula. <i>Invent. Math.</i> , 185(2):333–393, 2011.
[Dub09]	J. Dubédat. SLE and the free field: partition functions and couplings.J. Amer. Math. Soc., 22(4):995–1054, 2009.
[Dur]	R. Durrett. Probability: Theory and Examples. Wadsworth, Pacific Grove, California.
[EK04]	J. Engländer and A.E. Kyprianou. Local extinction vs. local exponential growth for spatial branching processes. <i>Annals of Probability</i> , 32:78–99, 2004.
[Eva98]	L.C. Evans. <i>Partial Differential Equations</i> . Graduate Studies in Mathematics, Volume 19. AMS, 1998.
[Fol99]	Gerald B. Folland. Real analysis: modern techniques and their appli- cations. John Wiley and Sons inc., 2nd edition, 1999.
[GPW09]	A. Greven, P. Pfaffelhuber, and A. Winter. Convergence in distribu- tion of random metric measure spaces ( $\lambda$ -coalescent measure trees). <i>Probability Theory and Related Fields</i> , 145(1-2):285–322, 2009.
[GT83]	D. Gilbarg and N.S. Trudinger. <i>Elliptic partial differential equations of second order</i> . Springer-Verlag, Berlin, 2nd edition, 1983.
[HH07]	J.W. Harris and S.C. Harris. Survival probabilities for branching Brownian motion with absorption. <i>Electronic Communications in Probability</i> , 12:81–92, 2007.
[HH09]	<ul> <li>R. Hardy and S.C. Harris. A spine approach to branching diffusions with applications to Lp-convergence of martingales, pages 280–330.</li> <li>Séminaire de Probabilité XLII, volume 1979, Lecture Notes in Math.</li> <li>Springer, Berlin, 2009.</li> </ul>
[HHK12]	S.C. Harris, M. Hesse, and A.E. Kyprianou. Branching Brownian mo- tion in a strip: Survival near criticality. <i>Annals of Probability</i> , 44 (1):235–275, 2012.
[HK71]	R. Høegh-Krohn. A general class of quantum fields without cut-offs in two space-time dimensions. <i>Comm. Math. Phys.</i> , 21(3):244–255, 1971.

[HR14]	S.C. Harris and M. Roberts. A strong law of large numbers for branch- ing processes: almost sure spine events. <i>Electronic Communications in</i> <i>Probability</i> , 19,28, 2014.
[HR15]	S.C. Harris and M. Roberts. The many-to-few lemma and multiple spines. Annales de l'Institut Henri Poincaré (B) Probability and Statistics, 2015.
[HRV15]	Y. Huang, R. Rhodes, and V. Vargas. Liouville Quantum Gravity on the unit disk. <i>Preprint</i> , 2015.
[IM65]	K. Itô and H.P. McKean. <i>Diffusion processes and their sample paths</i> . Springer, 1965.
[JS87]	J. Jacod and A.N. Shiryaev. <i>Limit Theorems for Stochastic Processes</i> . Springer-Verlag Berlin Heidelberg, 1987.
[JS17]	J. Junnila and E. Saksman. Uniqueness of critical Gaussian chaos. Electronic Journal of Probability, 22:1–31, 2017.
[Kah85]	J.P. Kahane. Sur le chaos multiplicatif. Ann. Sci. Math. Québec, 9(2):105–150, 1985.
[Kes78]	H. Kesten. Branching Brownian motion with absorption. <i>Stochastic Processes and Applications</i> , 7:9–47, 1978.
[KI13]	K. Kytölä and K. Izyurov. Hadamard's formula and couplings of SLEs with free field. <i>Probability Theory and Related Fields</i> , 155(1-2):35–69, 2013.
[Kol38]	<ul> <li>A.N. Kolmogorov. Zur Löslung einer biologischen Aufgabe. Izv.</li> <li>NaučnIssled. Inst. Mat. i Meh. Tomsk. Gosudarstv. Univ., 2:1–6, 1938.</li> </ul>
[KP76]	J.P. Kahane and J. Peyrière. Sur certaines martingales de Benoit Man- delbrot. <i>Advances in Math.</i> , 22(2):131–145, 1976.
[KS16]	A. Kemppainen and S. Smirnov. Random curves, scaling limits and Loewner evolutions. <i>Annals of Probability</i> , page to appear, 2016.
[Kyp00]	A.E. Kyprianou. Martingale convergence and the stopped branching random walk. <i>Probab. Theory Related Fields</i> , 116(3):405–419, 2000.
[Law05]	G. Lawler. <i>Conformally invariant processes in the plane</i> . Mathemat- ical Surveys and Monographs, 114. American Mathematical Society, Providence, RI, 2005.
[LG05]	J.F. Le Gall. Random trees and applications. <i>Probability Surveys</i> , 2, 2005.

- [LG06] J.F. Le Gall. Random real trees. Ann. Fac. Sci. Toulouse Math., 15(1):35–62, 2006.
- [LG13] J.F. Le Gall. Uniqueness and universality of the Brownian map. Ann. Probab., 41(4):2880–2960, 2013.
- [LGD02] J.F. Le Gall and T. Duquesne. Random trees, Lévy processes and spatial branching processes. *Astérisque*, 281, 2002.
- [LSW03] G. Lawler, O. Schramm, and W. Werner. Conformal restriction : the chordal case. J. Amer. Math. Soc., 16(4):917–955, 2003.
- [LSW04] G. Lawler, O. Schramm, and W. Werner. Conformal invariance of planar loop-erased random walks and uniform spanning trees. Ann. Probab., 32(IB):939–995, 2004.
- [Lyo97] R. Lyons. A simple path to Biggins' martingale convergence for branching random walk, pages 217–221. Number 84 in IMA Volumes in Mathematics and its Applications. Springer, New York, 1997.
- [McK75] H.P. McKean. Application of Brownian motion to the equation of Kolmogorov-Petrovski-Piscounov. Communications in Pure and Applied Mathematics, 28:323–331, 1975.
- [Mie08] G. Miermont. Invariance principles for spatial multitype Galton-Watson trees. Annales de l'institut Henri Poincaré (B) Probabilités et Statistiques, 44(6):1128–1161, 2008.
- [Mie09] G. Miermont. Tesselations of random maps of arbitrary genus. Ann. Sci. Éc. Norm. Supér, 42(5:725–781, 2009.
- [Mie13] G. Miermont. The Brownian map is the scaling limit of uniform random plane quadrangulations. *Acta Math.*, 210(2):319–401, 2013.
- [MNV09] M. Meerschaert, E. Nane, and P. Vellaisamy. Fractional cauchy problems on bounded domains. *Annals of Probability*, 37:997–1007, 2009.
- [Mot58] M. Motoo. Proof of the law of iterated logarithm through diffusion equation. Ann. Inst. Statist. Math., 10:21–28, 1958.
- [MP10] P. Mörters and Y. Peres. *Brownian Motion*. Cambridge University Press, 2010.
- [MS11] J. Miller and S. Sheffield. The GFF and CLE(4). Slides of 2011 talks and private communications, 2011.
- [MS15a] J. Miller and S. Sheffield. Liouville quantum gravity and the Brownian map I: the QLE(8/3,0) metric. *Preprint*, 2015.
| [MS15b]  | J. Miller and S. Sheffield. Liouville quantum gravity and the Brownian map II: geodesics and continuity of the embedding. <i>Preprint</i> , 2015.                                                                    |
|----------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| [MS16a]  | J. Miller and S. Sheffield. Imaginary geometry I: interacting SLEs. <i>Probab. Theory Related Fields</i> , 164(3-4):553–705, 2016.                                                                                   |
| [MS16b]  | J. Miller and S. Sheffield. Imaginary geometry II: Reversibility of $SLE_{\kappa}(\rho_1; \rho_2)$ for $\kappa \in (0, 4)$ . Ann. Probab., 44(3):1647–1722, 2016.                                                    |
| [MS16c]  | J. Miller and S. Sheffield. Imaginary geometry III: reversibility of SLE $_{\kappa}$ for $\kappa$ in (4, 8). Ann. of Math., 184(2):455–486, 2016.                                                                    |
| [MS16d]  | J. Miller and S. Sheffield. Imaginary geometry IV: interior flow lines, whole-plane reversibility and space filling paths. <i>Preprint</i> , 2016.                                                                   |
| [MS16e]  | J. Miller and S. Sheffield. Liouville quantum gravity and the Brownian map III: the conformal structure is determined. <i>Preprint</i> , 2016.                                                                       |
| [MSW16]  | J. Miller, S. Sheffield, and W. Werner. Cle percolations. <i>Preprint</i> , 2016.                                                                                                                                    |
| [MW16]   | J. Miller and H. Wu. Intersections of SLE paths: the double and cut point dimension of SLE. <i>Probability Theory and Related Fields</i> , page to appear, 2016.                                                     |
| [Pin85]  | R.G. Pinsky. On the convergence of diffusion processes conditioned to remain in a bounded region for large time to limiting positive recurrent diffusion processes. <i>Annals of Probability</i> , 13:363–378, 1985. |
| [Pol81a] | A.M. Polyakov. Quantum geometry of bosonic strings. <i>Phys. Lett. B.</i> , 103(3):207–210, 1981.                                                                                                                    |
| [Pol81b] | A.M. Polyakov. Quantum geometry of fermionic strings. <i>Phys. Lett.</i> B., 103(3):211–213, 1981.                                                                                                                   |
| [Pow17a] | E. Powell. Critical Gaussian chaos: convergence and uniqueness in the derivative renormalisation. <i>Preprint; available on the webpage of the author</i> , 2017.                                                    |
| [Pow17b] | E. Powell. Invariance principles for branching diffusions in bounded domains. <i>Preprint, arxiv::1512.00031</i> , 2017.                                                                                             |
| [PW17]   | E. Powell and H. Wu. Level lines of the Gaussian Free Field with general boundary data. Annales de l'Institute Henri Poincaré (B), To appear, 2017.                                                                  |
| [QW17]   | W. Qian and W. Werner. Coupling the Gaussian free fields with free and with zero boundary conditions via common level lines. <i>arXiv</i> preprint arXiv:1703.04350, 2017.                                           |

[Rob10]	<ul> <li>M.I. Roberts. Spine changes of measure and branching diffusions. <i>PhD</i> thesis, University of Bath, http://people.bath.ac.uk/mir20/thesis.pdf, 2010.</li> </ul>
[RS05]	S. Rohde and O. Schramm. Basic properties of SLE. Ann. of Math. (2), 161(2):883–924, 2005.
[Rud76]	W. Rudin. <i>Principles of Mathematical Analysis</i> . McGraw-Hill Book Co., New York, 3rd edition, 1976.
[RV10]	R. Robert and V. Vargas. Gaussian multiplicative chaos revisited. Ann. Probab., 38(2):605–631, 2010.
[RV11]	R. Rhodes and V. Vargas. KPZ formula for log-infinitely divisible multifractal random measures. <i>ESAIM, P and S</i> , 15:358–371, 2011.
[RV14]	R. Rhodes and V. Vargas. Gaussian multiplicative chaos and applica- tions: a review. <i>Probab. Surv.</i> , 11:315–392, 2014.
[RY91]	D. Revuz and M. Yor. <i>Continuous Martingales and Brownian Motion</i> . Springer, 1991.
[Sch00]	O. Schramm. Scaling limits of loop-erased random walks and uniform spanning trees. <i>Israel J. Math.</i> , 118:221–288, 2000.
[Sep 16]	A. Sepúlveda. On thin local sets of the GFF. Preprint, 2016.
[Sev58]	B.A. Sevst'yanov. Branching stochastic processes for particles diffusing in a bounded domain with absorbing boundaries. <i>Theory of Probability</i> <i>and its Applications</i> , 1,3:111–126, 1958.
[Sha16]	A. Shamov. On Gaussian multiplicative chaos. J. Funct. Anal., 270(9):3224–3261, 2016.
[She07]	S. Sheffield. Gaussian Free Fields for mathematicians. <i>Probab. Theory Related Fields</i> , 139(3-4):521–541, 2007.
[She09]	S. Sheffield. Exploration trees and conformal loop ensembles. <i>Duke Math. J.</i> , 147(1):79–129, 2009.
[She16]	S. Sheffield. Conformal weldings of random surfaces: SLE and the quantum gravity zipper. Ann. Probab., 44(5):3474–3545, 2016.
[Smi01]	S. Smirnov. Critical percolation in the plane: conformal invariance, Cardy's formula, scaling limits. C.R. Acad. Sci. Paris. Ser. I Math., 333(3):239–244, 2001.
[SS09]	O. Schramm and S. Sheffield. Contour lines of the two-dimensional discrete Gaussian Free Field. <i>Acta Math.</i> , 202(1):21–137, 2009.

[SS13]	O. Schramm and S. Sheffield. A contour line of the continuum Gaussian free field. <i>Probab. Theory Related Fields</i> , 157(1-2):47–80, 2013.
[SSW09]	O. Schramm, S. Sheffield, and D.B. Wilson. Conformal radii for con- formal loop ensembles. <i>Communications in Mathematical Physics</i> , 288:43–53, 2009.
[SW12]	S. Sheffield and W. Werner. Conformal loop ensembles: the Markovian characterization and the loop-soup construction. <i>Ann. of Math.</i> , 176(3):1827–1917, 2012.
[Tho06]	V. Thomee. Galerkin Finite Element Methods for Parabolic Problems. Springer, Berlin, 2006.
[Wat65]	S. Watanabe. On the branching process for Brownian particles with an absorbing boundary. <i>Jour. Math. Kyoto Univ.</i> , 4:385–398, 1965.
[Wat67]	S. Watanabe. <i>Limit theorem for a class of branching processes.</i> , pages 205–232. J. Chover ed., Markov Processes and Potential Theory. Wiley, New York, 1967.
[Wer]	W. Werner. Topics on the two-dimensional Gaussian Free Field. Lecture notes, available on the webpage of the author.
[WW13]	W. Werner and H. Wu. From $\text{CLE}(\kappa)$ to $\text{SLE}(\kappa, \rho)$ s. Electron. J. Probab, 18(36):1–20, 2013.
[WW16]	M. Wang and H. Wu. Level Lines of Gaussian Free Field I: Zero- boundary GFF. <i>Stochastic Processes and their Applications</i> , 2016.
[Zha08a]	D. Zhan. Duality of chordal SLE. Inventiones mathematicae, 174(2):309–353, 2008.
[Zha08b]	D. Zhan. Reversibility of chordal SLE. <i>The Annals of Probability</i> , 36(4):1472–1494, 2008.
Contact: $\epsilon$	egpowell 12@gmail.com

219